Totally positive spaces: topology and applications

by

Pavel Galashin

Submitted to the Department of Mathematics in partial fulfillment of the requirements for the degree of

Doctor of Philosophy

at the

MASSACHUSETTS INSTITUTE OF TECHNOLOGY

June 2019

© Massachusetts Institute of Technology 2019. All rights reserved.

Author

Department of Mathematics May 3, 2019

Certified by..... Alexander Postnikov Professor of Applied Mathematics Thesis Supervisor

Chairman, Department Committee on Graduate Theses

Totally positive spaces: topology and applications

by

Pavel Galashin

Submitted to the Department of Mathematics on May 3, 2019, in partial fulfillment of the requirements for the degree of Doctor of Philosophy

Abstract

This thesis studies topological spaces arising in total positivity. Examples include the totally nonnegative Grassmannian $\operatorname{Gr}_{\geq 0}(k, n)$, Lusztig's totally nonnegative part $(G/P)_{\geq 0}$ of a partial flag variety, Lam's compactification of the space of electrical networks, and the space of (boundary correlation matrices of) planar Ising networks. We show that all these spaces are homeomorphic to closed balls. In addition, we confirm conjectures of Postnikov and Williams that the CW complexes $\operatorname{Gr}_{\geq 0}(k, n)$ and $(G/P)_{\geq 0}$ are regular. This implies that the closure of each positroid cell inside $\operatorname{Gr}_{\geq 0}(k, n)$ is homeomorphic to a closed ball. We discuss the close relationship between the above spaces and the physics of scattering amplitudes, which has served as a motivation for most of our results.

In the second part of the thesis, we investigate the space of planar Ising networks. We give a simple stratification-preserving homeomorphism between this space and the totally nonnegative orthogonal Grassmannian, describing boundary correlation matrices of the planar Ising model by inequalities. Under our correspondence, Kramers–Wannier's high/low temperature duality transforms into the cyclic symmetry of $\text{Gr}_{>0}(k, n)$.

Thesis Supervisor: Alexander Postnikov Title: Professor of Applied Mathematics

Acknowledgments

I would like to thank my advisor Alex Postnikov for explaining lots of beautiful mathematics to me, providing me with support and advice for the past 5 years, and for teaching me how to be more "chill". He is mostly responsible for my decision to enter the field of Algebraic Combinatorics. I am also indebted to my coauthors Thomas Lam and Pasha Pylyavskyy. I learned a great deal from each of them, and had a lot of fun collaborating with them. Next, I am grateful to Sergey Fomin for sharing his infinite wisdom with me, and for the fact that the majority of my mathematical work so far has been centered around objects and questions introduced by him and his coauthors. I thank Igor Pak, Lauren Williams, Nicolai Reshetikhin, Alexei Borodin, and Pavel Etingof for their advice and help with my job search. In addition to the above people, the works of Nima Arkani-Hamed, Allen Knutson, George Lusztig, Robert Marsh, Konni Rietsch, Jeanne Scott, David Speyer, Richard Stanley, Andrei Zelevinsky, and many others have served as a constant source of inspiration for my research.

This thesis is based on joint papers with Steven Karp, Thomas Lam, and Pasha Pylyavskyy. I thank them for letting me reproduce our work here.

Throughout these years, I have benefited from conversations and collaborations with many of my colleagues, including Pavle Blagojević, Jake Bourjaily, Cesar "Misha" Cuenca, Philippe Di Francesco, Miriam Farber, Darij Grinberg, Patricia Hersh, Sam Hopkins, Rinat Kedem, Alisa Knizel, Gaku Liu, Thomas McConville, Gleb Nenashev, Nevena Palić, Vic Reiner, Tom Roby, Hugh Thomas, Nathan Williams, and Günter Ziegler. Before coming to MIT, I had a wonderful time learning from my mentors Gaiane Panina, Mikhail Lifshits, Sergei Ivanov, and Anatoly Vershik, as well as my peers Vladimir Zolotov and Andrei Alpeev — the regular participants of the all night long geometry seminar held in my car.

Finally, I would like to acknowledge my family and friends, most importantly, my wife and parents. Special thanks go to my wife Valentina for keeping me in check throughout my PhD journey, listening patiently to my theorems, and being supportive of my slightly irregular working schedule. I am also particularly grateful to my mother Irina for the enormous effort that she has put into my education.

Contents

1	Intr	oduction	11
	1.1	The totally nonnegative Grassmannian	11
	1.2	The totally nonnegative part of a partial flag variety	12
	1.3	The totally nonnegative part of the unipotent radical	13
	1.4	The cyclically symmetric amplituhedron	14
	1.5	The space of planar electrical networks	14
	1.6	The space of planar Ising networks	15
	1.7	Connections	17
	1.8	Outline	18
2	The	cyclic shift vector field	19
	2.1	Contractive flows	19
	2.2	The totally nonnegative Grassmannian	23
	2.3	The cyclically symmetric amplituhedron	31
	2.4	The space of planar electrical networks	34
3	Reg	ularity theorem for totally nonnegative flag varieties	39
	3.1	Stars, links, and the Fomin–Shapiro atlas	39
	3.2	Overview of the proof	42
	3.3	Topological results	45
	3.4	G/P: preliminaries	53
	3.5	Subtraction-free parametrizations	68
	3.6	Bruhat projections and total positivity	79

	3.7	Affine Bruhat atlas for the projected Richardson stratification $\ldots \ldots \ldots$	90	
	3.8	From Bruhat atlas to Fomin–Shapiro atlas	98	
	3.9	The case $G = SL_n \dots \dots$	101	
	3.10	Further directions	120	
	3.A	Appendix: Kac–Moody flag varieties	122	
4	т.		105	
4	Ising model			
	4.1	Main results	128	
	4.2	Consequences of the main construction	132	
	4.3	Background on the combinatorics of the totally nonnegative Grassmannian .	147	
	4.4	The totally nonnegative orthogonal Grassmannian	153	
	4.5	From the Ising model to the orthogonal Grassmannian	165	
	4.6	Cyclic symmetry and a homeomorphism with a ball	178	
	4.7	Generalized Griffiths' inequalities	182	
	4.8	Open problems and future directions	190	

List of Figures

1-1	Connections between various totally positive spaces.	17
3-1	The map $\bar{\nu}_g$ for the case $G = SL_3$ and $P = B$ from Example 3.1.1.	40
3-2	The map from Le diagrams to bounded affine permutations. \ldots . \ldots .	107
4-1	Doubling map: an example	130
4-2	Transforming a graph G into a bipartite graph G^\square and a medial graph $G^\times.$.	136
4-3	Some almost perfect matchings of G^{\Box}	138
4-4	A planar graph G, its dual G^* , and the corresponding plabic graph G^{\Box}	139
4-5	A planar graph G , the medial graph G^{\times} and its medial strands	141
4-6	Uncrossing the pairs $\{i, j\}$ and $\{i', j'\}$.	158
4-7	The Hasse diagram of the poset P_3	160
4-8	The map θ between almost perfect matchings and minimal spinned flows	176

1

Introduction

Постепенно человек утрачивает свою форму и становится шаром. И став шаром, человек утрачивает все свои желания. – Даниил Хармс

We investigate various topological spaces arising in the theory of total positivity. Each space comes equipped with a natural decomposition into cells, and is conjectured to be a regular CW complex homeomorphic to a closed ball. A *regular CW complex* is a topological space subdivided into cells, such that the closure of each cell is homeomorphic to a ball, and the boundary of each cell is homeomorphic to a sphere. A prototypical example of a regular CW complex is a convex polytope.

We show that each of the totally nonnegative spaces introduced in Sections 1.1–1.6 is homeomorphic to a closed ball. Additionally, we show that the spaces from Sections 1.1–1.3 are regular CW complexes. As we discuss in Section 1.7, the spaces that we consider are surprisingly closely related to each other and to the physics of scattering amplitudes.

This thesis is based on papers [GKL17, GKL19, GP18].

1.1 The totally nonnegative Grassmannian

Let $\operatorname{Gr}(k, n)$ denote the Grassmannian of k-planes in \mathbb{R}^n . Postnikov [Pos07] defined its totally nonnegative part $\operatorname{Gr}_{\geq 0}(k, n)$ as the set of $X \in \operatorname{Gr}(k, n)$ whose Plücker coordinates are all nonnegative. Postnikov conjectured that $\operatorname{Gr}_{\geq 0}(k, n)$ is a regular CW complex homeomorphic to a closed ball. He gave a decomposition of $\operatorname{Gr}_{\geq 0}(k, n)$ into *positroid cells*, where each cell is specified by requiring some subset of the Plücker coordinates to be strictly positive, and requiring the rest to equal zero.

Over the past decade, much work has been done towards Postnikov's conjecture. The face poset of the positroid cell decomposition (described in [Rie98, Rie06, Pos07]) was shown to be thin and shellable by Williams [Wil07]. Combined with Björner's results [Bjö84], this implies that there exists *some* regular CW complex with the same face poset. Postnikov, Speyer, and Williams [PSW09] showed that the cell decomposition is a CW complex, and Rietsch and Williams [RW10] showed that the closure of each cell is contractible. We start by giving a simple proof of the following result.

Theorem 1.1.1. $\operatorname{Gr}_{\geq 0}(k,n)$ is homeomorphic to a k(n-k)-dimensional closed ball.

This proves a special case of Postnikov's conjecture: there is one top-dimensional positroid cell in $\operatorname{Gr}_{\geq 0}(k, n)$ whose closure is the whole $\operatorname{Gr}_{\geq 0}(k, n)$, and thus Theorem 1.1.1 shows that the closure of this cell is homeomorphic to a ball. Our proof of Theorem 1.1.1 employs a certain cyclic shift vector field τ on $\operatorname{Gr}_{\geq 0}(k, n)$. The flow defined by τ contracts all of $\operatorname{Gr}_{\geq 0}(k, n)$ to the unique cyclically symmetric point $X_0 \in \operatorname{Gr}_{\geq 0}(k, n)$. We construct a homeomorphism from $\operatorname{Gr}_{\geq 0}(k, n)$ to a closed ball $B \subset \operatorname{Gr}_{\geq 0}(k, n)$ centered at X_0 , by mapping each trajectory in $\operatorname{Gr}_{\geq 0}(k, n)$ to its intersection with B. A feature of our construction is that we do not rely on any cell decomposition of $\operatorname{Gr}_{\geq 0}(k, n)$.

We also prove Postnikov's conjecture in full generality, although in this case the proof is much more involved, see Sections 3.1 and 3.2.

Theorem 1.1.2. $\operatorname{Gr}_{\geq 0}(k,n)$ is a regular CW complex homeomorphic to a closed ball.

1.2 The totally nonnegative part of a partial flag variety

Let G be a simple and simply connected algebraic group, split over \mathbb{R} , and let $P \subset G$ be a parabolic subgroup. Lusztig [Lus94] introduced the totally nonnegative part of the partial flag variety G/P, denoted $(G/P)_{\geq 0}$. He called $(G/P)_{\geq 0}$ a "remarkable polyhedral subspace", and conjectured that $(G/P)_{\geq 0}$ has a decomposition into cells $\Pi_g^{>0} \cong \mathbb{R}^{\dim(g)}$, which was proved by Rietsch [Rie99]. The following result was conjectured by Williams [Wil07].

Theorem 1.2.1. $(G/P)_{\geq 0}$ is a regular CW complex homeomorphic to a closed ball.

For a specific choice of G and P, $(G/P)_{\geq 0}$ becomes the totally nonnegative Grassmannian $\operatorname{Gr}_{\geq 0}(k, n)$, thus Theorems 1.1.1 and 1.1.2 are special cases of Theorem 1.2.1. Similarly to the case of $\operatorname{Gr}_{\geq 0}(k, n)$, Williams [Wil07] proved that the face poset of $(G/P)_{\geq 0}$ is thin and shellable, and then Rietsch–Williams [RW08, RW10] showed that $(G/P)_{\geq 0}$ is a CW complex such that the closure of every cell is contractible.

1.3 The totally nonnegative part of the unipotent radical

The theory of total positivity originated in the 1930's, and concerns real matrices whose minors are all nonnegative [Sch30, GK50, Whi52]. Later, Lusztig [Lus94] was motivated by a question of Kostant to consider connections between totally nonnegative matrices and his theory of canonical bases for quantum groups [Lus90]. This led him to introduce the totally nonnegative part $G_{\geq 0}$ of a split semisimple G.

Fomin and Shapiro [FS00] realized that Lusztig's work may be used to address a longstanding problem in poset topology. Namely, the Bruhat order of the Weyl group W of Ghad been shown to be shellable by Björner and Wachs [BW82], and by general results of Björner [Bjö84] it follows that there exists a "synthetic" regular CW complex whose face poset coincides with (W, \leq) . The motivation of [FS00] was to answer a natural question due to Bernstein and Björner of whether such a regular CW complex exists "in nature". Let $U \subset G$ be the unipotent radical of the standard Borel subgroup, and let $U_{\geq 0} := U \cap G_{\geq 0}$ be its totally nonnegative part. For $G = SL_n$, $U_{\geq 0}$ is the semigroup of upper-triangular unipotent matrices with all minors nonnegative. The work of Lusztig [Lus94] implies that $U_{\geq 0}$ has a cell decomposition whose face poset is (W, \leq) . The space $U_{\geq 0}$ is not compact, but Fomin and Shapiro [FS00] conjectured that taking the link of the identity element in $U_{\geq 0}$, which also has (W, \leq) as its face poset, gives the desired regular CW complex. Their conjecture was confirmed by Hersh [Her14]. Alternatively, it follows as a corollary to our proof of Theorem 1.2.1, see Remark 3.3.12. Thus we provide an independent proof of the Fomin–Shapiro conjecture.

Corollary 1.3.1 ([Her14]). The link of the identity in $U_{\geq 0}$ is a regular CW complex.

1.4 The cyclically symmetric amplituhedron

A robust connection between the totally nonnegative Grassmannian and the physics of scattering amplitudes was developed in [AHBC⁺16], which led Arkani-Hamed and Trnka [AHT14] to define topological spaces called *amplituhedra*.

Let k, m, n be nonnegative integers with $k+m \leq n$, and Z be a $(k+m) \times n$ matrix whose $(k+m) \times (k+m)$ minors are all positive. We regard Z as a linear map $\mathbb{R}^n \to \mathbb{R}^{k+m}$, which induces a map Z_{Gr} on Gr(k, n) taking the subspace X to the subspace $\{Z(v) : v \in X\}$. The *(tree) amplituhedron* $\mathcal{A}_{n,k,m}(Z)$ is the image of $\text{Gr}_{\geq 0}(k, n)$ in Gr(k, k+m) under the map Z_{Gr} [AHT14, Section 4]. When k = 1, the totally nonnegative Grassmannian $\text{Gr}_{\geq 0}(1, n)$ is a simplex in \mathbb{P}^{n-1} , and the amplituhedron $\mathcal{A}_{n,1,m}(Z)$ is a *cyclic polytope* in \mathbb{P}^m [Stu88].

We now take m to be even, and $Z = Z_0$ such that the rows of Z_0 span the unique element of $\operatorname{Gr}_{\geq 0}(k+m,n)$ invariant under $\mathbb{Z}/n\mathbb{Z}$ -cyclic action (cf. [Kar18]). We call $\mathcal{A}_{n,k,m}(Z_0)$ the cyclically symmetric amplituhedron. When k = 1 and m = 2, $\mathcal{A}_{n,1,2}(Z_0)$ is a regular n-gon in the plane. More generally, $\mathcal{A}_{n,1,m}(Z_0)$ is a polytope whose vertices are n regularly spaced points on the trigonometric moment curve in \mathbb{P}^m .

Theorem 1.4.1. The cyclically symmetric amplituhedron $\mathcal{A}_{n,k,m}(Z_0)$ is homeomorphic to a km-dimensional closed ball.

It is expected that every amplituhedron is homeomorphic to a closed ball. In some special cases, this is indeed true, see [KW19, BGPZ19].

1.5 The space of planar electrical networks

Let Γ be an electrical network consisting only of resistors, modeled as an undirected graph whose edge weights (conductances) are positive real numbers. The electrical properties of Γ are encoded by the response matrix $\Lambda(\Gamma) : \mathbb{R}^n \to \mathbb{R}^n$, sending a vector of voltages at n distinguished boundary vertices to the vector of currents induced at the same vertices. The response matrix can be computed using (only) Kirchhoff's law and Ohm's law. Following Curtis, Ingerman, and Morrow [CIM98] and Colin de Verdière, Gitler, and Vertigan [CdVGV96], we consider the space Ω_n of response matrices of planar electrical networks: those Γ embedded into a disk, with boundary vertices on the boundary of the disk. This space is not compact; a compactification E_n was defined by Lam [Lam18]. It comes equipped with a natural embedding $\iota : E_n \hookrightarrow \operatorname{Gr}_{\geq 0}(n-1, 2n)$. We exploit this embedding to establish the following result.

Theorem 1.5.1. The space E_n is homeomorphic to an $\binom{n}{2}$ -dimensional closed ball.

A cell decomposition of E_n was defined in [Lam18], extending earlier work in [CIM98, CdVGV96]. The face poset of this cell decomposition had been defined and studied by Kenyon [Ken12, Section 4.5.2]. Theorem 1.5.1 says that the closure of the unique cell of top dimension in E_n is homeomorphic to a closed ball. In [Lam15], Lam showed that the face poset of the cell decomposition of E_n is Eulerian, and conjectured that it is shellable. Hersh and Kenyon recently proved this conjecture [HK18]. Björner's results [Bjö84] therefore imply that this poset is the face poset of some regular CW complex homeomorphic to a ball. We expect that E_n forms such a CW complex, so that the closure of every cell of E_n is homeomorphic to a closed ball. Proving this remains an open problem.

1.6 The space of planar Ising networks

The Ising model, introduced by Lenz in 1920 as a model for ferromagnetism and solved by Ising [Isi25] in dimension 1, plays a central role in statistical mechanics and conformal field theory. One of the main features of this model is that it undergoes a phase transition in dimensions larger than 1, and in particular its critical temperature $\frac{1}{2}\log(\sqrt{2}+1)$ in dimension 2 has been computed by Kramers and Wannier [KW41], who found a duality transformation exchanging subcritical and supercritical temperatures. The free energy of the model was computed by Onsager [Ons44] and Yang [Yan52], and since then it became a subject of active mathematical and physical research. Conformal invariance of the scaling limit was conjectured in [BPZ84b, BPZ84a] in relation to conformal field theory, and proven more recently as a part of a series of groundbreaking results by Smirnov, Chelkak, Hongler, Izyurov, and others [Smi10, CHI15, CS12, HS13, CDCH⁺14].

Among the most important quantities associated with the Ising model are two-point correlation functions. It was shown in [Gri67] and later generalized in [KS68] that these correlation functions satisfy natural inequalities, and in particular, the question of characterizing correlation functions coming from the Ising model was raised in the appendix of [KS68].

A starting point for our results was recent insightful work of Lis [Lis17], where he discovered a deep connection between the planar Ising model and total positivity, and used it to prove new inequalities on boundary two-point correlation functions in the planar case.

Despite the enormous amount of research on the planar Ising model, some basic questions seem to have remained unanswered. Let us denote by $\mathcal{X}_n \subset \operatorname{Mat}_n(\mathbb{R})$ the space of all boundary correlation matrices of planar Ising networks with n boundary nodes embedded in a disk. This is a subspace of the space $\operatorname{Mat}_n(\mathbb{R})$ of $n \times n$ matrices with real entries. Every matrix in \mathcal{X}_n is symmetric and has diagonal entries equal to 1, but \mathcal{X}_n is neither a closed nor an open subset of the space of such matrices. Let $\overline{\mathcal{X}}_n$ denote the closure of \mathcal{X}_n inside $\operatorname{Mat}_n(\mathbb{R})$, i.e., $\overline{\mathcal{X}}_n$ is the space of boundary correlation matrices of a slightly more general class of planar Ising networks, as discussed in Section 4.5.

Two fundamental questions about $\overline{\mathcal{X}}_n$ that we answer (see Theorem 4.1.3) are:

- Describe $\overline{\mathcal{X}}_n$ by equalities and inequalities inside $\operatorname{Mat}_n(\mathbb{R})$.
- Describe the topology of $\overline{\mathcal{X}}_n$.

Using a construction similar to the one in [Lis17], we give a simple embedding ϕ of the space $\overline{\mathcal{X}}_n$ as a subset of $\operatorname{Gr}_{\geq 0}(n, 2n)$ which turns out to be precisely the *totally nonnegative or*thogonal Grassmannian $\operatorname{OG}_{\geq 0}(n, 2n)$, introduced in [HW13, HWX14] in the study of ABJM scattering amplitudes. This gives a solution to the first question. Next, we show that $\overline{\mathcal{X}}_n$ is homeomorphic to an $\binom{n}{2}$ -dimensional closed ball using the same cyclic shift vector field τ that was used in the proof of Theorem 1.1.1. Surprisingly, we find (Theorem 4.2.4) that the



Figure 1-1: Connections between various totally positive spaces, see Section 1.7.

celebrated Kramers–Wannier's duality [KW41] is translated by our map ϕ into the cyclic shift on $\operatorname{Gr}_{\geq 0}(k, n)$.

1.7 Connections

Totally positive spaces have attracted a lot of interest due to their appearances in other contexts such as cluster algebras [FZ02] and the physics of scattering amplitudes [AHBC⁺16]. In particular, our original motivation came from studying the amplituhedron of [AHT14] and the more general Grassmann polytopes of [Lam16]. The faces of a Grassmann polytope are linear projections of closures of positroid cells, which is why it is essential to understand the topology of these closures in order to develop a theory of Grassmann polytopes.

The relationship between the following objects is shown schematically in Figure 1-1:

- 1. the planar $\mathcal{N} = 4$ supersymmetric Yang–Mills theory;
- 2. the three-dimensional $\mathcal{N} = 6$ supersymmetric Chern–Simons matter theory (also known as *ABJM theory*);
- 3. the amplituhedron $\mathcal{A}_{n,k,m}(Z)$ from Section 1.4;
- 4. the totally nonnegative Grassmannian $\operatorname{Gr}_{\geq 0}(k, n)$ from Section 1.1;
- 5. the totally nonnegative part $(G/P)_{\geq 0}$ of a partial flag variety from Section 1.2;
- 6. the compactification E_n of the space of electrical networks from Section 1.5;
- 7. the space $\overline{\mathcal{X}}_n$ of planar Ising networks from Section 1.6;

8. the totally nonnegative orthogonal Grassmannian $OG_{\geq 0}(n, 2n)$ from Section 1.6.

The scattering amplitudes in the planar $\mathcal{N} = 4$ supersymmetric Yang–Mills theory can be computed by formally integrating a certain differential form over the amplituhedron $\mathcal{A}_{n,k,m}(Z)$, cf. [AHBC⁺16, AHT14]. By definition, $\mathcal{A}_{n,k,m}(Z)$ is a linear projection of $\operatorname{Gr}_{\geq 0}(k,n)$, see Section 1.4. As we explained in Section 1.2, Postnikov's $\operatorname{Gr}_{\geq 0}(k,n)$ is a special case of Lusztig's $(G/P)_{\geq 0}$. Next, Lam [Lam18] constructed an embedding of his compactification E_n of the space of electrical networks into $\operatorname{Gr}_{\geq 0}(n-1,2n)$, see (2.4.1). Similarly, the space $\overline{\mathcal{X}}_n$ of planar Ising networks from Section 1.6 is identified via Theorem 4.1.3 with the totally nonnegative orthogonal Grassmannian $OG_{\geq 0}(n, 2n)$, which is a subset of $Gr_{\geq 0}(n, 2n)$ by definition. The spaces E_n and $\overline{\mathcal{X}}_n$ share a lot of common properties (cf. Section 4.8), for example, their face posets are isomorphic. However, the precise relationship between them remains completely mysterious to us, see Question 4.8.2. By Remark 4.4.2, each of these spaces can be realized as a subset of G/P for a suitable choice of G and P, but this subset does not coincide with the totally nonnegative part $(G/P)_{\geq 0}$. Finally, the totally nonnegative orthogonal Grassmannian $OG_{\geq 0}(n, 2n)$ (which is homeomorphic to $\overline{\mathcal{X}}_n$) was first introduced in [HW13, HWX14] in connection with the $\mathcal{N} = 6$ ABJM theory, analogously to the relationship [AHBC⁺16] between $\operatorname{Gr}_{\geq 0}(k, n)$ and the $\mathcal{N} = 4$ SYM theory.

1.8 Outline

Chapters 2, 3, 4 correspond to papers [GKL17, GKL19, GP18]. Theorems 1.1.1, 1.4.1, and 1.5.1 are proved in Chapter 2 using the cyclic shift vector field τ . Theorem 1.2.1 and its special case Theorem 1.1.2 are proved in Chapter 3. Chapter 4 develops the theory of planar Ising networks, see Section 4.1 for the precise statement of the main results. $\mathbf{2}$

The cyclic shift vector field

In this chapter, give a simple unified approach to show that the spaces $\operatorname{Gr}_{\geq 0}(k, n)$, $\mathcal{A}_{n,k,m}(Z_0)$, and E_n are homeomorphic to closed balls, thus proving Theorems 1.1.1, 1.4.1, and 1.5.1. We note that the same approach can be used to show that the spaces $U_{\geq 0}$ (when $G = \operatorname{SL}_n$) and $(G/P)_{\geq 0}$ (when G and P are arbitrary) are homeomorphic to closed balls as well, see [GKL17, §4] and [GKL18].

In Section 2.1, we introduce *contractive flows*, and then we use them in Sections 2.2, 2.3, and 2.4 to prove Theorems 1.1.1, 1.4.1, and 1.5.1, respectively.

2.1 Contractive flows

In this section we prove Lemma 2.1.3, which we will repeatedly use in establishing our theorems. Consider a real normed vector space $(\mathbb{R}^N, \|\cdot\|)$. Thus for each r > 0, the closed ball $B_r^N := \{p \in \mathbb{R}^N : \|p\| \le r\}$ of radius r is a compact convex body in \mathbb{R}^N whose interior contains the origin. We denote its boundary by ∂B_r^N , which is the sphere of radius r.

Definition 2.1.1. We say that a map $f : \mathbb{R} \times \mathbb{R}^N \to \mathbb{R}^N$ is a *contractive flow* if the following conditions are satisfied:

- (1) the map f is continuous;
- (2) for all $p \in \mathbb{R}^N$ and $t_1, t_2 \in \mathbb{R}$, we have f(0, p) = p and $f(t_1 + t_2, p) = f(t_1, f(t_2, p))$; and

(3) for all $p \neq 0$ and t > 0, we have ||f(t, p)|| < ||p||.

The condition (2) says that f induces a group action of $(\mathbb{R}, +)$ on \mathbb{R}^N . In particular, f(t, p) = q is equivalent to f(-t, q) = p, so (3) implies that if $t \neq 0$ and f(t, p) = p, then p = 0. The converse to this statement is given below in Lemma 2.1.2(i).

For $K \subset \mathbb{R}^N$ and $t \in \mathbb{R}$, we let f(t, K) denote $\{f(t, p) : p \in K\}$.

Lemma 2.1.2. Let $f : \mathbb{R} \times \mathbb{R}^N \to \mathbb{R}^N$ be a contractive flow.

- (i) We have f(t, 0) = 0 for all $t \in \mathbb{R}$.
- (ii) Let $p \neq 0$. Then the function $t \mapsto ||f(t,p)||$ is strictly decreasing on $(-\infty, \infty)$.
- (iii) Let $p \neq 0$. Then $\lim_{t \to \infty} ||f(t,p)|| = 0$ and $\lim_{t \to -\infty} ||f(t,p)|| = \infty$.

Proof. (i) By (1), the function $s \mapsto ||f(s,0)||$ is continuous on \mathbb{R} , and it equals 0 when s = 0. If $f(t,0) \neq 0$ for some t > 0, then 0 < ||f(s,0)|| < ||f(t,0)|| for some $s \in (0,t)$, which contradicts (3) applied to p = f(s,0) and t - s. Therefore f(t,0) = 0 for all $t \ge 0$. By (2), for $t \ge 0$ we have 0 = f(0,0) = f(-t,f(t,0)) = f(-t,0), and so f(-t,0) = 0 as well.

(ii) This follows from (3) and the fact that f induces a group action of \mathbb{R} on \mathbb{R}^N , once we know that f(t, p) is never 0. But if f(t, p) = 0 then f(-t, 0) = p, which contradicts part (i).

(iii) Let $r_1(p)$ and $r_2(p)$ denote the respective limits. By part (ii), both limits exist, where $r_1(p) \in [0, \infty)$ and $r_2(p) \in (0, \infty]$. For any $r \in (0, \infty)$, consider the compact set $K_r := \bigcap_{s \ge 0} f(s, B_r^N)$. By (2), we have $K_r \subset f(t, K_r) = \bigcap_{s \ge t} f(s, B_r^N)$ for any $t \ge 0$. On the other hand, if $q \in K_r$ is a point with maximum norm, and $q \ne 0$, then (3) implies that $q \notin f(t, K_r)$ for any t > 0. Thus $K_r = \{0\}$. Taking r = ||p|| implies that $r_1(p) = 0$. Suppose now that $r_2(p) \ne \infty$. Then for any $t \ge 0$, we have $f(-t, p) \in B_{r_2(p)}^N$, i.e. $p \in f(t, B_{r_2(p)}^N)$. Thus $p \in K_{r_2(p)} = \{0\}$, a contradiction.

Lemma 2.1.3. Let $Q \subset \mathbb{R}^N$ be a smooth embedded submanifold of dimension $d \leq N$, and $f : \mathbb{R} \times \mathbb{R}^N \to \mathbb{R}^N$ a contractive flow. Suppose that Q is bounded and satisfies the condition

$$f(t,\overline{Q}) \subset Q \quad \text{for } t > 0.$$
 (2.1.1)

Then the closure \overline{Q} is homeomorphic to a closed ball of dimension d, and $\overline{Q} \setminus Q$ is homeomorphic to a sphere of dimension d-1.

Proof. Since Q is bounded, its closure \overline{Q} is compact. By Lemma 2.1.2(iii) and (2.1.1) we have $0 \in \overline{Q}$, and therefore $0 \in Q$. Because Q is smoothly embedded, we can take r > 0 sufficiently small so that $B := B_r^N \cap Q$ is homeomorphic to a closed ball of dimension d. We let ∂B denote $(\partial B_r^N) \cap Q$, which is a (d-1)-dimensional sphere.

For any $p \in \mathbb{R}^N \setminus \{0\}$, consider the curve $t \mapsto f(t,p)$ starting at p and defined for all $t \in \mathbb{R}$. By Lemma 2.1.2(ii), this curve intersects the sphere ∂B_r^N for a unique $t \in \mathbb{R}$, which we denote by $t_r(p)$. Also, for $p \in \overline{Q} \setminus \{0\}$, define $t_{\partial}(p) \in (-\infty, 0]$ as follows. Let $T(p) := \{t \in \mathbb{R} : f(t,p) \in \overline{Q}\}$. We have $0 \in T(p)$, and T(p) is bounded from below by Lemma 2.1.2(iii) because \overline{Q} is bounded. By (2.1.1), if $t \in T(p)$ then $[t,\infty) \subset T(p)$. Also, T(p) is closed since it is the preimage of \overline{Q} under the continuous map $t \mapsto f(t,p)$. It follows that $T(p) = [t_{\partial}(p), \infty)$ for some $t_{\partial}(p) \in (-\infty, 0]$.

Claim. The functions t_r and t_{∂} are continuous on $\overline{Q} \setminus \{0\}$.

Proof. First we prove that t_r is continuous on $\mathbb{R}^N \setminus \{0\}$. It suffices to show that the preimage of any open interval $I \subset \mathbb{R}$ is open. To this end, let $q \in t_r^{-1}(I)$. Take $t_1, t_2 \in I$ with $t_1 < t_r(q) < t_2$. By Lemma 2.1.2(ii), we have $||f(t_1,q)|| > r > ||f(t_2,q)||$. Note that the map $\gamma_1 : \mathbb{R}^N \to \mathbb{R}^N, p \mapsto f(t_1,p)$ is continuous and $\mathbb{R}^N \setminus B_r^N$ is open, so $\gamma_1^{-1}(\mathbb{R}^N \setminus B_r^N)$ is an open neighborhood of q. Similarly, defining $\gamma_2 : \mathbb{R}^N \to \mathbb{R}^N, p \mapsto f(t_2,p)$, we have that $\gamma_2^{-1}(\operatorname{int}(B_r^N))$ is an open neighborhood of q. Therefore $\gamma_1^{-1}(\mathbb{R}^N \setminus B_r^N) \cap \gamma_2^{-1}(\operatorname{int}(B_r^N))$ is an open neighborhood of q, whose image under t_r is contained in $(t_1, t_2) \subset I$. This shows that t_r is continuous on $\mathbb{R}^N \setminus \{0\}$.

Next, let us define

$$R := \{ f(t,p) : (t,p) \in \mathbb{R} \times \overline{Q} \}.$$

$$(2.1.2)$$

The map $b : \mathbb{R} \times \partial B \to R \setminus \{0\}$ defined by $(t, p) \mapsto f(t, p)$ is a continuous bijection. (Recall that $\partial B = (\partial B_r^N) \cap Q$.) Its inverse $p \mapsto (-t_r(p), f(t_r(p), p))$ is continuous as well. Therefore b is a homeomorphism. We claim that Q is relatively open in R. Indeed, since $Q \setminus \{0\}$ is a submanifold of $R \setminus \{0\}$ of the same dimension d, we deduce that $Q \setminus \{0\}$ is an open subset of $R \setminus \{0\}$. Also, Q contains the neighborhood $int(B_r^N) \cap R$ of 0 in R. Thus Q is an open subset of R.

We now prove that the map $t_{\partial}: \overline{Q} \setminus \{0\} \to \mathbb{R}$ is continuous, by a very similar argument. Let $I \subset \mathbb{R}$ be an open interval and consider a point $q \in t_{\partial}^{-1}(I)$. Take $t_1, t_2 \in I$ with $t_1 < t_{\partial}(q) < t_2$. By the definition of t_{∂} , we have $f(t_1, q) \in R \setminus \overline{Q}$. By (2.1.1), we have $f(t_2, q) \in Q$. Note that the map $\gamma_1 : R \to R, p \mapsto f(t_1, p)$ is continuous and $R \setminus \overline{Q}$ is open in R, so $\gamma_1^{-1}(R \setminus \overline{Q})$ is an open neighborhood of q in R. Similarly, defining $\gamma_2 : R \to R, p \mapsto f(t_2, p)$, we have that $\gamma_2^{-1}(Q)$ is an open neighborhood of q in R. Therefore $\gamma_1^{-1}(R \setminus \overline{Q}) \cap \gamma_2^{-1}(Q) \cap \overline{Q}$ is an open neighborhood of q in \overline{Q} , whose image under t_{∂} is contained inside $(t_1, t_2) \subset I$. This finishes the proof of the claim.

Define the maps $\alpha: \overline{Q} \to B$ and $\beta: B \to \overline{Q}$ by

$$\alpha(p) := f(t_r(p) - t_{\partial}(p), p), \quad \beta(p) := f(t_{\partial}(p) - t_r(p), p)$$

for $p \neq 0$, and $\alpha(0) := 0$, $\beta(0) := 0$. Let us verify that α sends \overline{Q} inside B and β sends B inside \overline{Q} . If $p \in \overline{Q} \setminus \{0\}$, then $f(t_r(p), p) \in B$ and $t_{\partial}(p) \leq 0$, whence the contractive property (3) implies $\alpha(p) = f(-t_{\partial}(p), f(t_r(p), p)) \in B$. Similarly, if $p \in B \setminus \{0\}$, then $f(t_{\partial}(p), p) \in \overline{Q}$ and $t_r(p) \leq 0$, whence (2.1.1) implies $\beta(p) = f(-t_r(p), f(t_{\partial}(p), p)) \in \overline{Q}$.

Now we check that α and β are inverse maps. For any $p \in \overline{Q}$ and $\Delta t \in \mathbb{R}$ such that $f(\Delta t, p) \in \overline{Q}$, we have

$$t_r(f(\Delta t, p)) = t_r(p) - \Delta t, \quad t_\partial(f(\Delta t, p)) = t_\partial(p) - \Delta t.$$

Taking $\Delta t := t_{\partial}(p) - t_r(p)$, we find

$$\alpha(\beta(p)) = \alpha(f(\Delta t, p)) = f(t_r(f(\Delta t, p)) - t_\partial(f(\Delta t, p)), f(\Delta t, p)) = f(-\Delta t, f(\Delta t, p)) = p.$$

We can similarly verify that $\beta(\alpha(p)) = p$, by instead taking $\Delta t := t_r(p) - t_{\partial}(p)$.

By the claim, t_r and t_{∂} are continuous on $\overline{Q} \setminus \{0\}$, so α is continuous everywhere except possibly at 0. Also, $t_r(p) > t_{\partial}(p)$ for all $p \in \overline{Q} \setminus \{0\}$, so α is continuous at 0 by Lemma 2.1.2(ii).

Thus α is a continuous bijection from a compact space to a Hausdorff space, so it is a homeomorphism. This shows that \overline{Q} is homeomorphic to a closed *d*-dimensional ball.

It remains to prove that $\overline{Q} \setminus Q$ is homeomorphic to a (d-1)-dimensional sphere. We claim that α restricts to a homeomorphism from $\overline{Q} \setminus Q$ to ∂B . We need to check that α sends $\overline{Q} \setminus Q$ inside ∂B , and β sends ∂B inside $\overline{Q} \setminus Q$. To this end, let $p \in \overline{Q} \setminus Q$. By condition (2), we have $p = f(-t_{\partial}(p), f(t_{\partial}(p), p))$. Hence if $t_{\partial}(p) < 0$, then (2.1.1) implies $p \in Q$, a contradiction. Therefore $t_{\partial}(p) = 0$, and $\alpha(p) = f(t_r(p), p) \in \partial B$. Now let $q \in \partial B$. We have $t_r(q) = 0$, so $\beta(q) = f(t_{\partial}(q), q)$. If $\beta(q) \in Q$, then $f(t_{\partial}(q) - t, q) \in Q$ for t > 0 sufficiently small (as Q is open in R from (2.1.2)), contradicting the definition of $t_{\partial}(q)$. Thus $\beta(q) \in \overline{Q} \setminus Q$.

2.2 The totally nonnegative Grassmannian

Let $\operatorname{Gr}(k,n)$ denote the *real Grassmannian*, the space of all k-dimensional subspaces of \mathbb{R}^n . We set $[n] := \{1, \ldots, n\}$, and let $\binom{[n]}{k}$ denote the set of k-element subsets of [n]. For $X \in \operatorname{Gr}(k,n)$, we denote by $(\Delta_I(X))_{I \in \binom{[n]}{k}} \in \mathbb{P}^{\binom{n}{k}-1}$ the *Plücker coordinates* of $X: \Delta_I(X)$ is the $k \times k$ minor of X (viewed as a $k \times n$ matrix modulo row operations) with column set I.

Recall that $\operatorname{Gr}_{\geq 0}(k, n)$ is the subset of $\operatorname{Gr}(k, n)$ where all Plücker coordinates are nonnegative (up to a common scalar). We also define the *totally positive Grassmannian* $\operatorname{Gr}_{>0}(k, n)$ as the subset of $\operatorname{Gr}_{\geq 0}(k, n)$ where all Plücker coordinates are positive.

2.2.1 Global coordinates for $Gr_{>0}(k, n)$

For each k and n, we introduce several distinguished linear operators on \mathbb{R}^n . Define the left cyclic shift $S \in \mathfrak{gl}_n(\mathbb{R}) = \operatorname{End}(\mathbb{R}^n)$ by $S(v_1, \ldots, v_n) := (v_2, \ldots, v_n, (-1)^{k-1}v_1)$. The sign $(-1)^{k-1}$ can be explained as follows: if we pretend that S is an element of $\operatorname{GL}_n(\mathbb{R})$, then the action of S on $\operatorname{Gr}(k, n)$ preserves $\operatorname{Gr}_{\geq 0}(k, n)$ (it acts on Plücker coordinates by rotating the index set [n]).

Note that the transpose S^T of S is the *right cyclic shift* given by $S^T(v_1, \ldots, v_n) = ((-1)^{k-1}v_n, v_1, \ldots, v_{n-1})$. Let $\tau := S + S^T \in \text{End}(\mathbb{R}^n)$. We endow \mathbb{R}^n with the standard inner product, so that τ (being symmetric) has an orthogonal basis of eigenvectors

 $u_1, \ldots, u_n \in \mathbb{R}^n$ corresponding to real eigenvalues $\lambda_1 \geq \cdots \geq \lambda_n$. Let $X_0 \in Gr(k, n)$ be the linear span of u_1, \ldots, u_k . The following lemma implies that X_0 is totally positive and does not depend on the choice of eigenvectors u_1, \ldots, u_n .

Lemma 2.2.1.

- (i) The eigenvalues $\lambda_1 \geq \cdots \geq \lambda_n$ are given as follows, depending on the parity of k:
 - if k is even, $\lambda_1 = \lambda_2 = 2\cos(\frac{\pi}{n}), \ \lambda_3 = \lambda_4 = 2\cos(\frac{3\pi}{n}), \ \lambda_5 = \lambda_6 = 2\cos(\frac{5\pi}{n}), \ldots;$
 - if k is odd, $\lambda_1 = 1$, $\lambda_2 = \lambda_3 = 2\cos(\frac{2\pi}{n})$, $\lambda_4 = \lambda_5 = 2\cos(\frac{4\pi}{n})$,

In either case, we have

$$\lambda_k = 2\cos\left(\frac{k-1}{n}\pi\right) > 2\cos\left(\frac{k+1}{n}\pi\right) = \lambda_{k+1}$$

(ii) [Sco79] The Plücker coordinates of X_0 are given by

$$\Delta_I(X_0) = \prod_{i,j \in I, \ i < j} \sin\left(\frac{j-i}{n}\pi\right) > 0 \quad \text{for all } I \in \binom{[n]}{k}.$$
(2.2.1)

For an example in the case of Gr(2, 4), see Section 2.2.5. (We remark that in the example, the Plücker coordinates of X_0 are scaled by a factor of 2 compared to the formula above.)

Proof. In this proof, we work over \mathbb{C} . Let $\zeta \in \mathbb{C}$ be an *n*th root of $(-1)^{k-1}$. There are *n* such values of ζ , each of the form $\zeta = e^{i\pi m/n}$ for some integer *m* congruent to k-1 modulo 2. Let $z_m := (1, \zeta, \zeta^2, \ldots, \zeta^{n-1}) \in \mathbb{C}^n$. We have $S(z_m) = \zeta z_m$ and $S^T(z_m) = \zeta^{-1} z_m$, so

$$\tau(z_m) = (\zeta + \zeta^{-1}) z_m = 2\cos(\frac{\pi m}{n}) z_m.$$
(2.2.2)

The *n* distinct z_m 's are linearly independent (they form an $n \times n$ Vandermonde matrix with nonzero determinant), so they give a basis of \mathbb{C}^n of eigenvectors of τ .

We deduce part (i) from (2.2.2). For part (ii), we apply Vandermonde's determinantal identity, following an argument outlined by Scott [Sco79]. That is, by (2.2.2), the C-linear

span of u_1, \ldots, u_k is the same as the span of $z_{-k+1}, z_{-k+3}, z_{-k+5}, \ldots, z_{k-1}$. Let M be the matrix whose rows are $z_{-k+1}, z_{-k+3}, z_{-k+5}, \ldots, z_{k-1}$, i.e.

$$M_{r,j} = e^{i\pi(-k-1+2r)(j-1)/n}$$
 for $1 \le r \le k$ and $1 \le j \le n$.

Then the Plücker coordinates of X_0 are the $k \times k$ minors of M (up to a common nonzero complex scalar), which can be computed explicitly by Vandermonde's identity after appropriately rescaling the columns. We refer the reader to [Kar18] for details.

Denote by Mat(k, n - k) the vector space of real $k \times (n - k)$ matrices. Define a map $\phi: Mat(k, n - k) \to Gr(k, n)$ by

$$\phi(A) := \operatorname{span}(u_i + \sum_{j=1}^{n-k} A_{i,j} u_{k+j} : 1 \le i \le k).$$
(2.2.3)

In other words, the entries of A are the usual coordinates on the big Schubert cell of Gr(k, n)with respect to the basis u_1, \ldots, u_n of \mathbb{R}^n , this Schubert cell being

$$\phi(\operatorname{Mat}(k, n-k)) = \{ X \in \operatorname{Gr}(k, n) : X \cap \operatorname{span}(u_{k+1}, \dots, u_n) = 0 \}.$$

In particular, ϕ is a smooth embedding, and it sends the zero matrix to X_0 . For an example in the case of Gr(2, 4), see Section 2.2.5.

Proposition 2.2.2. The image $\phi(\operatorname{Mat}(k, n-k))$ contains $\operatorname{Gr}_{\geq 0}(k, n)$.

Proof. Let $X \in \operatorname{Gr}_{\geq 0}(k, n)$ be a totally nonnegative subspace. We need to show that $X \cap \operatorname{span}(u_{k+1}, \ldots, u_n) = 0$. Suppose otherwise that there exists a nonzero vector v in this intersection. Extend v to a basis of X, and write this basis as the rows of a $k \times n$ matrix M. Because X is totally nonnegative, the nonzero $k \times k$ minors of M all have the same sign (and at least one minor is nonzero, since M has rank k). Also let M_0 be the $k \times n$ matrix with rows u_1, \ldots, u_k . By Lemma 2.2.1(ii), all $k \times k$ minors of M_0 are nonzero and have the same sign. The vectors u_1, \ldots, u_n are orthogonal, so v is orthogonal to the rows of M_0 . Hence the first column of $M_0 M^T$ is zero, and we obtain $\det(M_0 M^T) = 0$. On the other hand, the

Cauchy–Binet identity implies

$$\det(M_0 M^T) = \sum_{I \in \binom{[n]}{k}} \det((M_0)_I) \det(M_I),$$

where A_I denotes the matrix A restricted to the columns I. Each summand has the same sign and at least one summand is nonzero, contradicting $det(M_0M^T) = 0$.

We have shown that the restriction of ϕ^{-1} to $\operatorname{Gr}_{\geq 0}(k, n)$ yields an embedding

$$\operatorname{Gr}_{\geq 0}(k,n) \hookrightarrow \operatorname{Mat}(k,n-k) \simeq \mathbb{R}^{k(n-k)}$$

whose restriction to $\operatorname{Gr}_{>0}(k, n)$ is smooth.

2.2.2 Flows on Gr(k, n)

For $g \in \operatorname{GL}_n(\mathbb{R})$, we let g act on $\operatorname{Gr}(k, n)$ by taking the subspace X to $g \cdot X := \{g(v) : v \in X\}$. We let $1 \in \operatorname{GL}_n(\mathbb{R})$ denote the identity matrix, and for $x \in \mathfrak{gl}_n(\mathbb{R})$ we let $\exp(x) := \sum_{j=0}^{\infty} \frac{x^j}{j!} \in \operatorname{GL}_n(\mathbb{R})$ denote the matrix exponential of x.

We examine the action of $\exp(tS)$ and $\exp(t\tau)$ on $\operatorname{Gr}(k, n)$.

Lemma 2.2.3. For $X \in \operatorname{Gr}_{\geq 0}(k, n)$ and t > 0, we have $\exp(tS) \cdot X \in \operatorname{Gr}_{>0}(k, n)$.

Proof. We claim that it suffices to prove the following two facts:

- (i) for $X \in \operatorname{Gr}_{\geq 0}(k, n)$ and $t \geq 0$, we have $\exp(tS) \cdot X \in \operatorname{Gr}_{\geq 0}(k, n)$; and
- (ii) for $X \in \operatorname{Gr}_{\geq 0}(k,n) \setminus \operatorname{Gr}_{>0}(k,n)$, we have $\exp(tS) \cdot X \notin \operatorname{Gr}_{\geq 0}(k,n)$ for all t < 0 sufficiently close to zero.

To see why this is sufficient, let $X \in \operatorname{Gr}_{\geq 0}(k, n)$ and t > 0. By part (i), we have $\exp(tS) \cdot X \in \operatorname{Gr}_{\geq 0}(k, n)$, so we just need to show that $\exp(tS) \cdot X \in \operatorname{Gr}_{>0}(k, n)$. Suppose otherwise that $\exp(tS) \cdot X \notin \operatorname{Gr}_{>0}(k, n)$. Then applying part (ii) to $\exp(tS) \cdot X$, we get that $\exp((t + t')S) \cdot X \notin \operatorname{Gr}_{\geq 0}(k, n)$ for t' < 0 sufficiently close to zero. But by part (i), we know that $\exp((t + t')S) \cdot X \in \operatorname{Gr}_{\geq 0}(k, n)$ for all t' in the interval [-t, 0]. This is a contradiction.

Now we prove parts (i) and (ii). We will make use of the operator 1 + tS, which belongs to $\operatorname{GL}_n(\mathbb{R})$ for |t| < 1. Note that if $[M_1 | \cdots | M_n]$ is a $k \times n$ matrix representing X, then a $k \times n$ matrix representing $(1 + tS) \cdot X$ is

$$M' = [M_1 + tM_2 \mid M_2 + tM_3 \mid \dots \mid M_{n-1} + tM_n \mid M_n + (-1)^{k-1}tM_1].$$

We can evaluate the $k \times k$ minors of M' using multilinearity of the determinant. We obtain

$$\Delta_I((1+tS)\cdot X) = \sum_{\epsilon \in \{0,1\}^k} t^{\epsilon_1 + \dots + \epsilon_k} \Delta_{\{i_1 + \epsilon_1, \dots, i_k + \epsilon_k\}}(X) \quad \text{for } I = \{i_1, \dots, i_k\} \subset [n], \quad (2.2.4)$$

where $i_1 + \epsilon_1, \ldots, i_k + \epsilon_k$ are taken modulo n. Therefore $(1 + tS) \cdot X \in \operatorname{Gr}_{\geq 0}(k, n)$ for $X \in \operatorname{Gr}_{\geq 0}(k, n)$ and $t \in [0, 1)$. Since $\exp(tS) = \lim_{j \to \infty} \left(1 + \frac{tS}{j}\right)^j$ and $\operatorname{Gr}_{\geq 0}(k, n)$ is closed, we obtain $\exp(tS) \cdot X \in \operatorname{Gr}_{\geq 0}(k, n)$ for $t \geq 0$. This proves part (i).

To prove part (ii), first note that $\exp(tS) = 1 + tS + O(t^2)$. By (2.2.4), we have

$$\Delta_I(\exp(tS) \cdot X) = \Delta_I(X) + t \sum_{I'} \Delta_{I'}(X) + O(t^2) \quad \text{for } I \in {[n] \choose k}, \qquad (2.2.5)$$

where the sum is over all $I' \in {\binom{[n]}{k}}$ obtained from I by increasing exactly one element by 1 modulo n. If we can find such I and I' with $\Delta_I(X) = 0$ and $\Delta_{I'}(X) > 0$, then $\Delta_I(\exp(tS) \cdot X) < 0$ for all t < 0 sufficiently close to zero, thereby proving part (ii). In order to do this, we introduce the directed graph D with vertex set ${\binom{[n]}{k}}$, where $J \to J'$ is an edge of D if and only if we can obtain J' from J by increasing exactly one element by 1 modulo n. Note that for any two vertices K and K' of D, there exists a directed path from K to K':

- we can get from [k] to any $\{i_1 < \cdots < i_k\}$ by shifting k to i_k , k-1 to i_{k-1} , etc.;
- similarly, we can get from any $\{i_1 < \cdots < i_k\}$ to $\{n-k+1, n-k+2, \ldots, n\};$
- we can get from $\{n k + 1, ..., n\}$ to [k] by shifting n to k, n 1 to k 1, etc.

Now take $K, K' \in {[n] \choose k}$ with $\Delta_K(X) = 0$ and $\Delta_{K'}(X) > 0$, and consider a directed path from K to K'. It goes through an edge $I \to I'$ with $\Delta_I(X) = 0$ and $\Delta_{I'}(X) > 0$, as desired. \Box

Now we consider $\exp(t\tau) = \exp(t(S + S^T))$. Recall that S and S^T are the left and right cyclic shift maps, so by symmetry Lemma 2.2.3 holds with S replaced by S^T . Also, S and S^T commute, so $\exp(t\tau) = \exp(tS) \exp(tS^T)$. We obtain the following.

Corollary 2.2.4. For $X \in \operatorname{Gr}_{\geq 0}(k, n)$ and t > 0, we have $\exp(t\tau) \cdot X \in \operatorname{Gr}_{>0}(k, n)$.

Let us see how $\exp(t\tau)$ acts on matrices $A \in \operatorname{Mat}(k, n-k)$. Note that $\tau(u_i) = \lambda_i u_i$ for $1 \le i \le n$, so $\exp(t\tau)(u_i) = e^{t\lambda_i} u_i$. Therefore $\exp(t\tau)$ acts on the basis of $\phi(A)$ in (2.2.3) by

$$\exp(t\tau)(u_i + \sum_{j=1}^{n-k} A_{i,j}u_{k+j}) = e^{t\lambda_i}(u_i + \sum_{j=1}^{n-k} e^{t(\lambda_{k+j} - \lambda_i)}A_{i,j}u_{k+j})$$

for all $1 \le i \le k$. Thus $\exp(t\tau) \cdot \phi(A) = \phi(f(t, A))$, where by definition $f(t, A) \in \operatorname{Mat}(k, n-k)$ is the matrix with entries

$$(f(t,A))_{i,j} := e^{t(\lambda_{k+j} - \lambda_i)} A_{i,j} \quad \text{for } 1 \le i \le k \text{ and } 1 \le j \le n - k.$$
 (2.2.6)

2.2.3 Proof of Theorem 1.1.1

Consider the map $f : \mathbb{R} \times \operatorname{Mat}(k, n-k) \to \operatorname{Mat}(k, n-k)$ defined by (2.2.6). We claim that f is a contractive flow on $\operatorname{Mat}(k, n-k)$ equipped with the Euclidean norm

$$||A||^2 = \sum_{i=1}^k \sum_{j=1}^{n-k} A_{i,j}^2$$

Indeed, parts (1) and (2) of Definition 2.1.1 hold for f. To see that part (3) holds, note that for any $1 \le i \le k$ and $1 \le j \le n - k$ with $A_{i,j} \ne 0$, we have

$$|(f(t,A))_{i,j}| = |e^{t(\lambda_{k+j}-\lambda_i)}A_{i,j}| = e^{t(\lambda_{k+j}-\lambda_i)}|A_{i,j}| < |A_{i,j}| \quad \text{for } t > 0,$$

using the fact that $\lambda_i \ge \lambda_k > \lambda_{k+1} \ge \lambda_{k+j}$ from Lemma 2.2.1(i). Therefore ||f(t, A)|| < ||A||if $A \ne 0$, verifying part (3).

Let us now apply Lemma 2.1.3 with $\mathbb{R}^N = \operatorname{Mat}(k, n - k)$ and $Q = \phi^{-1}(\operatorname{Gr}_{>0}(k, n))$. We need to know that $\operatorname{Gr}_{\geq 0}(k, n)$ is the closure of $\operatorname{Gr}_{>0}(k, n)$. This was proved by Postnikov [Pos07, Section 17]; it also follows directly from Corollary 2.2.4, since we can express any $X \in \operatorname{Gr}_{\geq 0}(k, n)$ as a limit of totally positive subspaces:

$$X = \lim_{t \to 0+} \exp(t\tau) \cdot X.$$

Therefore $\overline{Q} = \phi^{-1}(\operatorname{Gr}_{\geq 0}(k, n))$. Moreover, $\operatorname{Gr}_{\geq 0}(k, n)$ is closed inside the compact space $\mathbb{P}^{\binom{n}{k}-1}$, and is therefore also compact. So, \overline{Q} is compact (and hence bounded). Finally, the property (2.1.1) in this case is precisely Corollary 2.2.4. We have verified all the hypotheses of Lemma 2.1.3, and conclude that \overline{Q} (and also $\operatorname{Gr}_{\geq 0}(k, n)$) is homeomorphic to a k(n-k)-dimensional closed ball.

2.2.4 Related work

Lusztig [Lus98b, Section 4] used a flow similar to $\exp(t\tau)$ to show that $(G/P)_{\geq 0}$ is contractible. In fact, we used his flow to show that $(G/P)_{\geq 0}$ is a ball in [GKL18]. Our flow can be thought of as an affine (or loop group) analogue of his flow, and is closely related to the whirl matrices of [LP12]. We also remark that Ayala, Kliemann, and San Martin [AKSM04] used the language of control theory to give an alternative development in type A of Lusztig's theory of total positivity. In that context, $\exp(t\tau)$ (t > 0) lies in the interior of the *compres*sion semigroup of $\operatorname{Gr}_{\geq 0}(k, n)$, and X_0 is its attractor.

Marsh and Rietsch defined and studied a superpotential on the Grassmannian in the context of mirror symmetry [MR15, Section 6]. It follows from results of Rietsch [Rie08] (as explained in [Kar18]) that X_0 is, rather surprisingly, also the unique totally nonnegative critical point of the q = 1 specialization of the superpotential. However, the superpotential is not defined on the boundary of $\operatorname{Gr}_{\geq 0}(k, n)$. The precise relationship between τ and the gradient flow of the superpotential remains mysterious.

2.2.5 Example: the case Gr(2, 4)

The matrix $\tau = S + S^T \in \mathfrak{gl}_4(\mathbb{R})$ and an orthogonal basis of real eigenvectors u_1, u_2, u_3, u_4 are

$$\tau = \begin{bmatrix} 0 & 1 & 0 & -1 \\ 1 & 0 & 1 & 0 \\ 0 & 1 & 0 & 1 \\ -1 & 0 & 1 & 0 \end{bmatrix}, \qquad u_1 = (0, 1, \sqrt{2}, 1), \qquad \lambda_1 = \sqrt{2}, \\ u_2 = (-\sqrt{2}, -1, 0, 1), \qquad \lambda_2 = \sqrt{2}, \\ u_3 = (\sqrt{2}, -1, 0, 1), \qquad \lambda_3 = -\sqrt{2} \\ u_4 = (0, 1, -\sqrt{2}, 1), \qquad \lambda_4 = -\sqrt{2} \end{bmatrix}$$

The embedding ϕ : Mat $(2,2) \hookrightarrow Gr(2,4)$ sends the matrix $A = \begin{bmatrix} a & b \\ c & d \end{bmatrix}$ to

$$\phi(A) = X = \begin{bmatrix} u_1 + au_3 + bu_4 \\ u_2 + cu_3 + du_4 \end{bmatrix} = \begin{bmatrix} \sqrt{2}a & 1 - a + b & \sqrt{2} - \sqrt{2}b & 1 + a + b \\ -\sqrt{2} + \sqrt{2}c & -1 - c + d & -\sqrt{2}d & 1 + c + d \end{bmatrix}$$

Above we are identifying $X \in Gr(2,4)$ with a 2 × 4 matrix whose rows form a basis of X. In terms of Plücker coordinates $\Delta_{ij} = \Delta_{\{i,j\}}(X)$, the map ϕ is given by

$$\Delta_{12} = \sqrt{2}(1 - 2a + b - c + ad - bc),$$

$$\Delta_{23} = \sqrt{2}(1 - 2d - b + c + ad - bc), \qquad \Delta_{13} = 2(1 - b - c - ad + bc),$$

$$\Delta_{34} = \sqrt{2}(1 + 2d - b + c + ad - bc), \qquad \Delta_{24} = 2(1 + b + c - ad + bc),$$

$$\Delta_{14} = \sqrt{2}(1 + 2a + b - c + ad - bc),$$

(2.2.7)

and its inverse is given by

$$a = (2\Delta_{14} - 2\Delta_{12})/\delta, \qquad b = (\Delta_{12} - \Delta_{23} - \Delta_{34} + \Delta_{14} - \sqrt{2}\Delta_{13} + \sqrt{2}\Delta_{24})/\delta,$$

$$d = (2\Delta_{34} - 2\Delta_{23})/\delta, \qquad c = (-\Delta_{12} + \Delta_{23} + \Delta_{34} - \Delta_{14} - \sqrt{2}\Delta_{13} + \sqrt{2}\Delta_{24})/\delta,$$

where $\delta = \Delta_{12} + \Delta_{23} + \Delta_{34} + \Delta_{14} + \sqrt{2}\Delta_{13} + \sqrt{2}\Delta_{24}.$

The point $X_0 = \phi(0) = \operatorname{span}(u_1, u_2) \in \operatorname{Gr}_{>0}(2, 4)$ has Plücker coordinates

$$\Delta_{12} = \Delta_{23} = \Delta_{34} = \Delta_{14} = \sqrt{2}, \quad \Delta_{13} = \Delta_{24} = 2,$$

which agrees with Lemma 2.2.1(ii). The image of ϕ is the subset of Gr(2, 4) where $\delta \neq 0$, which we see includes $\operatorname{Gr}_{\geq 0}(2, 4)$, verifying Proposition 2.2.2 in this case. Restricting ϕ^{-1} to $\operatorname{Gr}_{\geq 0}(2, 4)$ gives a homeomorphism onto the subset of \mathbb{R}^4 of points (a, b, c, d) where the 6 polynomials Δ_{ij} in (2.2.7) are nonnegative. By Theorem 1.1.1, these spaces are both homeomorphic to 4-dimensional closed balls. The closures of cells in the cell decomposition of $\operatorname{Gr}_{\geq 0}(2,4)$ are obtained in \mathbb{R}^4 by taking an intersection with the zero locus of some subset of the 6 polynomials. The 0-dimensional cells (corresponding to points of $\operatorname{Gr}_{\geq 0}(2,4)$ with only one nonzero Plücker coordinate) are

$$(a, b, c, d) = (-2, 1, -1, 0), (0, -1, 1, -2), (0, -1, 1, 2), (2, 1, -1, 0), (0, -1, -1, 0), (0, 1, 1, 0).$$

In general, using the embedding ϕ we can describe $\operatorname{Gr}_{\geq 0}(k, n)$ as the subset of $\mathbb{R}^{k(n-k)}$ where some $\binom{n}{k}$ polynomials of degree at most k are nonnegative.

2.3 The cyclically symmetric amplituhedron

Let k, m, n be nonnegative integers with $k + m \leq n$ and m even, and let $S, \tau \in \mathfrak{gl}_n(\mathbb{R})$ be the operators from Section 2.2.1. Let $\lambda_1 \geq \cdots \geq \lambda_n \in \mathbb{R}$ be the eigenvalues of τ corresponding to orthogonal eigenvectors u_1, \ldots, u_n . In this section, we assume that these eigenvectors have norm 1. Recall from Lemma 2.2.1(i) that $\lambda_k > \lambda_{k+1}$. Since m is even, we have $(-1)^{k+m-1} = (-1)^{k-1}$ and $\lambda_{k+m} > \lambda_{k+m+1}$.

Let Z_0 denote the $(k+m) \times n$ matrix whose rows are u_1, \ldots, u_{k+m} . By Lemma 2.2.1(ii), the $(k+m) \times (k+m)$ minors of Z_0 are all positive (perhaps after replacing u_1 with $-u_1$). We may also think of Z_0 as a linear map $\mathbb{R}^n \to \mathbb{R}^{k+m}$. Since the vectors u_1, \ldots, u_n are orthonormal, this map takes u_i to the *i*th unit vector $e_i \in \mathbb{R}^{k+m}$ if $i \leq k+m$, and to 0 if i > k+m. Recall from Section 1.4 that Z_0 induces a map $(Z_0)_{\mathrm{Gr}} : \mathrm{Gr}_{\geq 0}(k,n) \to \mathrm{Gr}(k,k+m)$, whose image is the cyclically symmetric amplituhedron $\mathcal{A}_{n,k,m}(Z_0)$. We remark that if $g \in \mathrm{GL}_{k+m}(\mathbb{R})$,

then $\mathcal{A}_{n,k,m}(gZ_0)$ and $\mathcal{A}_{n,k,m}(Z_0)$ are related by the automorphism g of $\operatorname{Gr}(k, k+m)$, so the topology of $\mathcal{A}_{n,k,m}(Z_0)$ depends only on the row span of Z_0 in $\operatorname{Gr}(k+m,n)$.

Proof of Theorem 1.4.1. We consider the map ϕ : Mat $(k, n-k) \to Gr(k, n)$ defined in (2.2.3). We write each $k \times (n-k)$ matrix $A \in Mat(k, n-k)$ as $[A' \mid A'']$, where A' and A'' are the $k \times m$ and $k \times (n-k-m)$ submatrices of A with column sets $\{1, \ldots, m\}$ and $\{m+1, \ldots, n-k\}$, respectively. We introduce a projection map

$$\pi: \operatorname{Mat}(k, n-k) \to \operatorname{Mat}(k, m), \quad A = [A' \mid A''] \mapsto A'.$$

We claim that there exists an embedding $\gamma : \mathcal{A}_{n,k,m}(Z_0) \hookrightarrow \operatorname{Mat}(k,m)$ making the following diagram commute:

Let $A = [A' | A''] \in \operatorname{Mat}(k, n - k)$ be a matrix such that $\phi(A) \in \operatorname{Gr}_{\geq 0}(k, n)$. Then the element $(Z_0)_{\operatorname{Gr}}(\phi(A))$ of $\operatorname{Gr}(k, k + m)$ is the row span of the $k \times (k + m)$ matrix $[\operatorname{Id}_k | A']$, where Id_k denotes the $k \times k$ identity matrix. Thus $\mathcal{A}_{n,k,m}(Z_0) = (Z_0)_{\operatorname{Gr}}(\operatorname{Gr}_{\geq 0}(k, n))$ lies inside the Schubert cell

$$\{Y \in \operatorname{Gr}(k, k+m) : \Delta_{[k]}(Y) \neq 0\}.$$

Every element Y of this Schubert cell is the row span of $[\mathrm{Id}_k \mid A']$ for a unique A', and we define $\gamma(Y) := A'$. Thus γ embeds $\mathcal{A}_{n,k,m}(Z_0)$ inside $\mathrm{Mat}(k,m)$, and (2.3.1) commutes.

Now we define

$$Q_0 := \pi(\phi^{-1}(\operatorname{Gr}_{>0}(k,n))) \subset \operatorname{Mat}(k,m).$$

We know from Section 2.2.3 that $\phi^{-1}(\operatorname{Gr}_{>0}(k,n))$ is an open subset of $\operatorname{Mat}(k,n)$ whose closure $\phi^{-1}(\operatorname{Gr}_{\geq 0}(k,n))$ is compact. Note that π is an open map (since it is essentially a projection $\mathbb{R}^{k(n-k)} \to \mathbb{R}^{km}$), so Q_0 is an open subset of $\operatorname{Mat}(k,m)$. Note that any open subset of \mathbb{R}^N is a smooth embedded submanifold of dimension N. The closure $\overline{Q_0} = \pi(\phi^{-1}(\operatorname{Gr}_{\geq 0}(k,n)))$ of Q_0 is compact. By (2.3.1), $\overline{Q_0}$ is homeomorphic to $\mathcal{A}_{n,k,m}(Z_0)$.

Let $f: \mathbb{R} \times \operatorname{Mat}(k, n-k) \to \operatorname{Mat}(k, n-k)$ be the map defined by (2.2.6), and define a

similar map $f_0 : \mathbb{R} \times \operatorname{Mat}(k, m) \to \operatorname{Mat}(k, m)$ by

$$f_0(t, A')_{i,j} := e^{t(\lambda_{k+j} - \lambda_i)} A'_{i,j}$$
 for $1 \le i \le k$ and $1 \le j \le m$.

That is, $f_0(t, \pi(A)) = \pi(f(t, A))$ for all $t \in \mathbb{R}$ and $A \in Mat(k, n - k)$. We showed in Section 2.2.3 that f is a contractive flow, so f_0 is also a contractive flow. We also showed that

$$f(t, \phi^{-1}(\operatorname{Gr}_{\geq 0}(k, n))) \subset \phi^{-1}(\operatorname{Gr}_{>0}(k, n)) \quad \text{ for } t > 0,$$

and applying π to both sides shows that

$$f_0(t, \overline{Q_0}) \subset Q_0 \quad \text{for } t > 0.$$

Thus Lemma 2.1.3 applies to Q_0 and f_0 , showing that $\overline{Q_0}$ (and hence $\mathcal{A}_{n,k,m}(Z_0)$) is homeomorphic to a km-dimensional closed ball.

Example 2.3.1. Let k = 1, n = 4, m = 2. We have

Note that this τ differs in the top-right and bottom-left entries from the one in Section 2.2.5, because k is odd rather than even. Also, here the eigenvectors are required to have norm 1. The embedding ϕ : Mat $(1,3) \hookrightarrow \text{Gr}(1,4)$ sends a matrix $A := \begin{bmatrix} a & b & c \end{bmatrix}$ to the line $\phi(A)$ in Gr(1,4) spanned by the vector

$$v = u_1 + au_2 + bu_3 + cu_4 = \frac{1}{2} \left(1 + \sqrt{2}a + c, 1 + \sqrt{2}b - c, 1 - \sqrt{2}a + c, 1 - \sqrt{2}b - c \right).$$

This line gets sent by $(Z_0)_{\text{Gr}}$ to the row span of the matrix $v \cdot Z_0^T = \begin{bmatrix} 1 & a & b \end{bmatrix}$. Finally, γ sends this element of Gr(1,3) to the matrix $\begin{bmatrix} a & b \end{bmatrix}$, so (2.3.1) indeed commutes.

In order for $\phi(A)$ to land in $\operatorname{Gr}_{\geq 0}(1,4)$, the coordinates of v must all have the same sign, and since their sum is 2, they must all be nonnegative:

$$1 + \sqrt{2}a + c \ge 0, \quad 1 + \sqrt{2}b - c \ge 0, \quad 1 - \sqrt{2}a + c \ge 0, \quad 1 - \sqrt{2}b - c \ge 0.$$

These linear inequalities define a tetrahedron in $\mathbb{R}^3 \simeq \operatorname{Mat}(1,3)$ with the four vertices $(0, \pm \sqrt{2}, -1)$, $(\pm \sqrt{2}, 0, 1)$. The projection $\pi = \gamma \circ (Z_0)_{\operatorname{Gr}} \circ \phi$ sends this tetrahedron to a square in $\mathbb{R}^2 \simeq \operatorname{Mat}(1,2)$ with vertices $(0, \pm \sqrt{2})$, $(\pm \sqrt{2}, 0)$. This square is a *km*-dimensional ball, as implied by Theorem 1.4.1. We note that when k = 1, the amplituhedron $\mathcal{A}_{n,k,m}(Z)$ (for any $(k+m) \times n$ matrix Z with positive maximal minors) is a cyclic polytope in the projective space $\operatorname{Gr}(1, m+1) = \mathbb{P}^m$ [Stu88], and is therefore homeomorphic to a *km*-dimensional closed ball. The case of $k \geq 2$ and $Z \neq Z_0$ remains open.

2.4 The space of planar electrical networks

2.4.1 A slice of the totally nonnegative Grassmannian

We recall some background on electrical networks, and refer the reader to [Lam18] and Example 2.4.3 for details. Let $\mathbb{R}^{\binom{2n}{n-1}}$ have basis vectors e_I for $I \in \binom{[2n]}{n-1}$, and let $\mathbb{P}^{\binom{2n}{n-1}-1}$ denote the corresponding projective space. We define

$$[2n]_{\text{odd}} := \{2i - 1 : i \in [n]\}, \quad [2n]_{\text{even}} := \{2i : i \in [n]\}.$$

Let \mathcal{NC}_n denote the set of non-crossing partitions of $[2n]_{\text{odd}}$. Each $\sigma \in \mathcal{NC}_n$ comes with a dual non-crossing partition (or Kreweras complement) $\tilde{\sigma}$ of $[2n]_{\text{even}}$. We call a subset $I \in {[2n] \choose n-1}$ concordant with σ if every part of σ and every part of $\tilde{\sigma}$ contains exactly one element not in I. Let $A_{\sigma} \in \mathbb{R}^{\binom{2n}{n-1}}$ be the sum of e_I over all I concordant with σ , and let \mathcal{H} be the linear subspace of $\mathbb{P}^{\binom{2n}{n-1}-1}$ spanned by the images of A_{σ} for $\sigma \in \mathcal{NC}_n$.

Identifying Gr(n-1, 2n) with its image under the Plücker embedding, we consider the

subvariety $\mathcal{X}_n := \operatorname{Gr}(n-1,2n) \cap \mathcal{H}$. In [Lam18, Theorem 5.8], an embedding

$$\iota: E_n \simeq \mathcal{X}_n \cap \operatorname{Gr}_{\geq 0}(n-1,2n) \hookrightarrow \operatorname{Gr}_{\geq 0}(n-1,2n)$$
(2.4.1)

was constructed, identifying the compactification of the space of planar electrical networks with n boundary vertices E_n with the compact space $\mathcal{X}_n \cap \operatorname{Gr}_{\geq 0}(n-1,2n)$. We will need the following property of $(E_n)_{>0} := \mathcal{X}_n \cap \operatorname{Gr}_{>0}(n-1,2n)$.

Proposition 2.4.1. The space $(E_n)_{>0}$ is diffeomorphic to $\mathbb{R}_{>0}^{\binom{n}{2}}$, and the inclusion $(E_n)_{>0} \hookrightarrow$ $\operatorname{Gr}_{>0}(n-1,2n)$ is a smooth embedding.

Here $\operatorname{Gr}_{>0}(n-1,2n) \subset \operatorname{Gr}(n-1,2n)$ is an open submanifold diffeomorphic to $\mathbb{R}^{(n-1)(n+1)}_{>0}$.

Proof. We recall from [CIM98, Theorem 4] that each point in $(E_n)_{>0} = \Omega_n^+$ is uniquely represented by assigning a positive real number (the *conductance*) to each edge of a wellconnected electrical network Γ with $\binom{n}{2}$ edges. This gives a parametrization $(E_n)_{>0} \simeq \mathbb{R}_{>0}^{\binom{n}{2}}$. The construction $\Gamma \mapsto N(\Gamma)$ of [Lam18, Section 5] sends Γ to a weighted bipartite graph $N(\Gamma)$ embedded into a disk compatibly with the inclusion (2.4.1). The edge weights of $N(\Gamma)$ are monomials in the edge weights of Γ . Furthermore, the underlying bipartite graph G of $N(\Gamma)$ parametrizes $\operatorname{Gr}_{>0}(n-1,2n)$. That is, we can choose a set of (n-1)(n+1) edges of G, so that assigning arbitrary positive edge weights to these edges and weight 1 to the remaining edges induces a parametrization $\operatorname{Gr}_{>0}(n-1,2n) \simeq \mathbb{R}_{>0}^{(n-1)(n+1)}$ (see [Pos07] or [Tal11]). It follows that the inclusion $\mathbb{R}_{>0}^{\binom{n}{2}} \simeq (E_n)_{>0} \hookrightarrow \operatorname{Gr}_{>0}(n-1,2n) \simeq \mathbb{R}_{>0}^{(n-1)(n+1)}$ is a monomial map, and in particular a homomorphism of Lie groups. The result follows.

2.4.2 Operators acting on non-crossing partitions

For each $i \in [2n]$, we define u_i and d_i in $\mathfrak{gl}_{\binom{2n}{n-1}}(\mathbb{R})$ by

$$u_i(e_I) := \begin{cases} e_{I \cup \{i+1\} \setminus \{i\}}, & \text{if } i \in I, i+1 \notin I, \\ 0, & \text{otherwise}, \end{cases} \quad d_i(e_I) := \begin{cases} e_{I \cup \{i-1\} \setminus \{i\}}, & \text{if } i \in I, i-1 \notin I, \\ 0, & \text{otherwise}. \end{cases}$$

Here the indices are taken modulo 2n.

For $i \in [2n]_{\text{odd}}$, we let $\kappa(i) \in \mathcal{NC}_n$ be the non-crossing partition which has two parts, namely $\{i\}$ and $[2n]_{\text{odd}} \setminus \{i\}$. For $i \in [2n]_{\text{even}}$, we let $\mu(i) \in \mathcal{NC}_n$ be the non-crossing partition with n-1 parts, one of which is $\{i-1, i+1\}$ and the rest being singletons. Given $\sigma \in \mathcal{NC}_n$ and $i \in [2n]$, we define the noncrossing partition $\sigma'(i) \in \mathcal{NC}_n$ as the common refinement of σ and $\kappa(i)$ if i is odd, and the common coarsening of σ and $\mu(i)$ if i is even. The following combinatorial lemma is essentially [Lam18, Proposition 5.15], and can be verified directly.

Lemma 2.4.2. For all $i \in [2n]$, we have

$$(u_i + d_i)(A_{\sigma}) = \begin{cases} 0, & \text{if } \sigma = \sigma'(i), \\ A_{\sigma'(i)}, & \text{otherwise.} \end{cases}$$

Example 2.4.3. Let n := 3 and $\sigma := \{\{1,3\}, \{5\}\} \in \mathcal{NC}_n$, so that $\tilde{\sigma} = \{\{2\}, \{4,6\}\}, \sigma'(1) = \sigma'(3) = \{\{1\}, \{3\}, \{5\}\}, \sigma'(2) = \sigma'(5) = \sigma$, and $\sigma'(4) = \sigma'(6) = \{\{1,3,5\}\}$. Abbreviating $e_{\{a,b\}}$ by e_{ab} , we have

$$A_{\sigma} = e_{14} + e_{16} + e_{34} + e_{36}.$$

Note that $\sigma \neq \sigma'(1)$ and

$$(u_1 + d_1)(A_{\sigma}) = (e_{24} + e_{26}) + (e_{46}) = A_{\sigma'(1)},$$

in agreement with Lemma 2.4.2 (since the dual of $\sigma'(1)$ is $\{\{2, 4, 6\}\}$). Similarly, we have $\sigma = \sigma'(2)$ and

$$(u_2 + d_2)(A_{\sigma}) = 0 + 0 = 0$$

We define the operator $\Phi := \sum_{i=1}^{2n} u_i + d_i \in \mathfrak{gl}_{\binom{2n}{n-1}}(\mathbb{R}).$

Lemma 2.4.4. Let $X \in E_n$. We have $\exp(t\tau) \cdot X \in (E_n)_{>0}$ for all t > 0.

Proof. This follows from Corollary 2.2.4, once we show that $\exp(t\tau) \cdot X \in \mathcal{H}$ for $X \in \mathcal{X}_n$ and $t \in \mathbb{R}$. To do this, we identify $\operatorname{Gr}(n-1,2n)$ with its image in $\mathbb{P}^{\binom{2n}{n-1}-1}$ under the Plücker embedding sending $X \in \operatorname{Gr}(n-1,2n)$ to $\sum_{I \in \binom{[2n]}{n-1}} \Delta_I(X) e_I \in \mathbb{P}^{\binom{2n}{n-1}-1}$. Then for
any $X \in \mathcal{X}_n$, we have a smooth curve $t \mapsto \exp(t\tau) \cdot X$ in $\mathbb{P}^{\binom{2n}{n-1}-1}$. As in (2.2.5), we find that

$$\exp(t\tau) \cdot X = X + t\Phi(X) + O(t^2) \quad \text{in } \mathbb{P}^{\binom{2n}{n-1}-1}$$

as $t \to 0$. Therefore $\exp(t\tau) \cdot X$ is an integral curve for the smooth vector field on $\mathbb{P}^{\binom{2n}{n-1}-1}$ defined by the infinitesimal action of Φ . By Lemma 2.4.2, this vector field is tangent to \mathcal{H} , so $\exp(t\tau) \cdot X \in \mathcal{H}$ for all $t \in \mathbb{R}$.

Proof of Theorem 1.5.1. We are identifying $(E_n)_{>0}$ as a subset $\operatorname{Gr}_{>0}(n-1,2n)$ via the smooth embedding of Proposition 2.4.1. In turn, $\operatorname{Gr}_{>0}(n-1,2n)$ is smoothly embedded inside $\operatorname{Mat}(n-1,n+1)$ by the map ϕ^{-1} defined in (2.2.3). Thus $Q := \phi^{-1}((E_n)_{>0}) \subset \operatorname{Mat}(n-1,n+1)$ is a smoothly embedded submanifold of $\operatorname{Mat}(n-1,n+1)$ of dimension $\binom{n}{2}$. The map ϕ^{-1} sends the compact set E_n homeomorphically onto its image $\phi^{-1}(E_n)$. Since $(E_n)_{>0}$ is dense in E_n , we have that $\phi^{-1}(E_n)$ equals the closure \overline{Q} of Q. Let $f : \mathbb{R} \times \operatorname{Mat}(n-1,n+1) \to$ $\operatorname{Mat}(n-1,n+1)$ be the map defined by (2.2.6). We showed in Section 2.2.3 that f is a contractive flow, and Lemma 2.4.4 implies that (2.1.1) holds for our choice of Q and f. Thus Lemma 2.1.3 applies, completing the proof. 3

Regularity theorem for totally nonnegative flag varieties

The goal of this chapter is to prove Theorem 1.2.1 and its special case Theorem 1.1.2. We start by giving a quick explanation of the proof idea in Section 3.1, and then give an outline of the rest of this chapter. A more detailed description of the proof can be found in Section 3.2.

3.1 Stars, links, and the Fomin–Shapiro atlas

Rietsch [Rie06] defined a certain poset (Q_J, \preceq) and proved the decomposition $(G/P)_{\geq 0} = \bigsqcup_{g \in Q_J} \prod_g^{>0}$. She showed that for $h \in Q_J$, the closure $\prod_h^{\geq 0}$ of $\prod_h^{>0}$ is given by $\prod_h^{\geq 0} = \bigsqcup_{g \preceq h} \prod_h^{>0}$. When $(G/P)_{\geq 0}$ is the totally nonnegative Grassmannian $\operatorname{Gr}_{\geq 0}(k, n)$, this gives the positroid cell decomposition of [Pos07].

Given $g \in Q_J$, define the star of g in $(G/P)_{\geq 0}$ by

$$\operatorname{Star}_{g}^{\geq 0} := \bigsqcup_{h \succeq g} \Pi_{h}^{>0}.$$
(3.1.1)

In Section 3.3.1, we define another space $Lk_g^{\geq 0}$ stratified as $Lk_g^{\geq 0} = \bigsqcup_{h \succ g} Lk_{g,h}^{>0}$. We later show in Theorem 3.3.11 that $Lk_g^{\geq 0}$ is a regular CW complex homeomorphic to a closed ball.



Figure 3-1: The map $\bar{\nu}_g$ for the case $G = SL_3$ and P = B from Example 3.1.1.

For each $g \in Q_J$, we construct a (stratification-preserving) homeomorphism

$$\bar{\nu}_g : \operatorname{Star}_g^{\geq 0} \xrightarrow{\sim} \Pi_g^{>0} \times \operatorname{Cone}(\operatorname{Lk}_g^{\geq 0}),$$
(3.1.2)

Here for a topological space A, we denote by $\operatorname{Cone}(A) := (A \times \mathbb{R}_{\geq 0})/(A \times \{0\})$ the open cone over A.

Example 3.1.1. When $G = \operatorname{SL}_n$ and P = B is the standard Borel subgroup, G/B is the complete flag variety consisting of flags in \mathbb{C}^n , and the Weyl group W is the group S_n of permutations of n elements. The face poset Q_J of $(G/B)_{\geq 0}$ is the set of Bruhat intervals $\{(v,w) \in S_n \times S_n \mid v \leq w\}$ in S_n , and the cell $\Pi_{(v,w)}^{>0} \subset (G/B)_{\geq 0}$ indexed by (v,w) has dimension $\ell(w) - \ell(v)$. For example, when n = 3, this gives a cell decomposition of a 3-dimensional ball, see Figure 3-1 (left). For $g := (s_1, s_2s_1)$, $\Pi_g^{>0}$ is an open line segment, and $\operatorname{Star}_g^{\geq 0}$ consists of 4 cells: a line segment $\Pi_g^{>0} = \Pi_{(s_1, s_2s_1)}^{>0}$, two open square faces $\Pi_{(s_1, w_0)}^{>0}$ and $\Pi_{(\operatorname{id}, s_2s_1)}^{>0}$, and an open 3-dimensional ball $\Pi_{(\operatorname{id}, w_0)}^{>0}$. This union is indeed homeomorphic to $\Pi_g^{>0} \times \operatorname{Cone}(\operatorname{Lk}_g^{\geq 0})$ shown in Figure 3-1 (right). Here $\operatorname{Lk}_g^{\geq 0}$ is a closed line segment whose endpoints are $\operatorname{Lk}_{g,(\operatorname{id}, s_2s_1)}^{>0}$, and $\operatorname{Lk}_{g,(\operatorname{id}, s_2s_1)}^{>0}$, and whose interior is $\operatorname{Lk}_{g,(\operatorname{id}, w_0)}^{>0}$.

At the core of our construction lies the notion of a *Fomin–Shapiro atlas*, which is essentially the collection of homeomorphisms $\bar{\nu}_g$ for all $g \in Q_J$. For technical reasons, we require $\bar{\nu}_g$ to be a smooth map defined on an open neighborhood \mathcal{O}_g of $\operatorname{Star}_g^{\geq 0}$, see Section 3.2.1. Our maps are analogous to those introduced in [FS00] for the unipotent radical $U_{\geq 0}$.

In Section 3.3, we prove that constructing a Fomin–Shapiro atlas is sufficient to deduce

that $(G/P)_{\geq 0}$ is a regular CW complex. Our topological argument proceeds by induction on the dimension of a cell. It relies on the generalized Poincaré conjecture [Sma61, Fre82, Per02, Per03a, Per03b] combined with the result of Williams [Wil07] that the face poset Q_J of $(G/P)_{\geq 0}$ is shellable.

In order to construct the Fomin–Shapiro atlas, for each $g \in Q_J$ we give an isomorphism $\bar{\varphi}_u$ between an open dense subset $\mathcal{O}_g \subset G/P$ and a certain subset of the *affine flag variety* \mathcal{G}/\mathcal{B} of the loop group \mathcal{G} associated with G. The map $\bar{\varphi}_u$ sends the projected Richardson stratification [KLS14] of G/P to the affine Richardson stratification of its image inside \mathcal{G}/\mathcal{B} . The hardest part of the proof consists of showing that the subset $\mathcal{O}_g \subset G/P$ contains $\operatorname{Star}_g^{\geq 0}$. See Section 3.2.2 for a more in-depth overview of our proof.

Remark 3.1.2. The map $\bar{\varphi}_u$ generalizes (up to changing signs of some entries) the map of Snider [Sni10] from $\operatorname{Gr}(k, n)$ to all G/P, see Remark 3.9.9. A different approach to give such a generalization is due to He–Knutson–Lu [HKL19], which led them to the notion of a *Bruhat atlas.* See [Ele16] for the definition. We call our map $\bar{\varphi}_u$ an *affine Bruhat atlas* since its target space is always an affine flag variety, while Bruhat atlases of [HKL19] necessarily involve more general *Kac–Moody flag varieties*.

Outline

In Section 3.2, we introduce (abstract) totally nonnegative spaces and define Fomin–Shapiro atlases. We state in Theorem 3.2.4 that every totally nonnegative space that admits a Fomin–Shapiro atlas is a regular CW complex, and prove it in Section 3.3. We give background on G/P in Section 3.4, and study subtraction-free Marsh–Rietsch parametrizations in Section 3.5. We then apply our results on such parametrizations to prove Theorem 3.6.4 that will later imply that the above open subset \mathcal{O}_g contains $\operatorname{Star}_g^{\geq 0}$. We introduce affine Bruhat atlases in Section 3.7 and use them to construct a Fomin–Shapiro atlas for G/P in Section 3.8. Theorem 3.2.5 (which implies our main result Theorem 1.2.1) is proved in Section 3.8.3. Section 3.9 is devoted to specializing our construction to type A (when $G = \operatorname{SL}_n$), with a special focus on the totally nonnegative Grassmannian $\operatorname{Gr}_{\geq 0}(k, n)$. We illustrate many of our constructions by examples in Section 3.9, and we encourage the reader to consult this section while studying other parts of the paper. We discuss some conjectures and further directions in Section 3.10. Finally, we give additional background on Kac–Moody flag varieties in Section 3.A.

3.2 Overview of the proof

We formulate our results in the abstract language of *totally nonnegative spaces*, since we expect that they can be applied in other contexts, see Section 3.10.

3.2.1 Totally nonnegative spaces

We refer the reader to Section 3.3.2 for background on posets and regular CW complexes. For a finite poset (Q, \preceq) , we denote by $\hat{Q} := Q \sqcup \{\hat{0}\}$ the poset obtained from Q by adjoining a minimum $\hat{0}$. Björner showed [Bjö84, Prop. 4.5(a)] that if \hat{Q} is graded, thin, and shellable, then Q is isomorphic to the face poset of some regular CW complex. If \hat{Q} is a graded poset, we let dim : $Q \to \mathbb{Z}_{\geq 0}$ denote the rank function of Q.

Definition 3.2.1. We say that a triple $(\mathcal{Y}, \mathcal{Y}^{\geq 0}, Q)$ is a *totally nonnegative space* (or *TNN* space for short) if the following conditions are satisfied.

(TNN1) (\widehat{Q}, \preceq) is graded, thin, shellable, and contains a unique maximal element $\widehat{1} \in Q$.

(TNN2) \mathcal{Y} is a smooth manifold, stratified into embedded submanifolds $\mathcal{Y} = \bigsqcup_{g \in Q} \mathring{\mathcal{Y}}_g$, and for each $h \in Q$, $\mathring{\mathcal{Y}}_h$ has dimension dim(h) and closure $\mathcal{Y}_h := \bigsqcup_{g \leq h} \mathring{\mathcal{Y}}_g$.

(TNN3) $\mathcal{Y}^{\geq 0}$ is a compact subset of \mathcal{Y} .

(TNN4) For
$$g \in Q, \mathcal{Y}_g^{>0} := \mathring{\mathcal{Y}}_g \cap \mathcal{Y}^{\geq 0}$$
 is a connected component of $\mathring{\mathcal{Y}}_g$ diffeomorphic to $\mathbb{R}_{>0}^{\dim(g)}$.

(TNN5) The closure of $\mathcal{Y}_h^{>0}$ inside \mathcal{Y} equals $\mathcal{Y}_h^{\geq 0} := \bigsqcup_{g \preceq h} \mathcal{Y}_g^{>0}$.

For the case $\mathcal{Y} = G/P$, the smooth submanifolds \mathcal{Y}_g are the open projected Richardson varieties of [KLS14].

Definition 3.2.2. Let $N \ge 0$, and denote by $\|\cdot\|$ the Euclidean norm on \mathbb{R}^N . We say that a pair (Z, ϑ) is a *smooth cone* if $Z \subset \mathbb{R}^N$ is a closed embedded submanifold and $\vartheta : \mathbb{R}_{>0} \times Z \to Z$ a smooth map such that

(SC1) ϑ gives an $(\mathbb{R}_{>0}, \cdot)$ -action on \mathbb{R}^N that restricts to an $(\mathbb{R}_{>0}, \cdot)$ -action on Z.

(SC2) $\frac{\partial}{\partial t} \| \vartheta(t, x) \| > 0$ for all $t \in \mathbb{R}_{>0}$ and $x \in \mathbb{R}^N \setminus \{0\}$.

The map ϑ is a smooth analog of a *contractive flow* from Definition 2.1.1.

For $g \in Q$, define $\operatorname{Star}_g := \bigsqcup_{h \succeq g} \mathring{\mathcal{Y}}_h$ and $\operatorname{Star}_g^{\geq 0} := \bigsqcup_{h \succeq g} \mathcal{Y}_h^{>0}$, cf. (3.1.1).

Definition 3.2.3. We say that a TNN space $(\mathcal{Y}, \mathcal{Y}^{\geq 0}, Q)$ admits a *Fomin–Shapiro atlas* if for each $g \in Q$, there exists an open subset $\mathcal{O}_g \subset \text{Star}_g$, a smooth cone (Z_g, ϑ_g) , and a diffeomorphism

$$\bar{\nu}_g: \mathcal{O}_g \xrightarrow{\sim} (\dot{\mathcal{Y}}_g \cap \mathcal{O}_g) \times Z_g \tag{3.2.1}$$

satisfying the following conditions.

- (FS1) For each $h \succeq g$ we are given $\mathring{Z}_{g,h} \subset Z_g$ such that $Z_g = \bigsqcup_{h \succeq g} \mathring{Z}_{g,h}$ and $\mathring{Z}_{g,g} = \{0\}$.
- (FS2) For all $h \succeq g$ and $t \in \mathbb{R}_{>0}$, we have $\vartheta_g(t, \mathring{Z}_{g,h}) = \mathring{Z}_{g,h}$.
- (FS3) For all $h \succeq g$, we have $\bar{\nu}_g(\mathring{\mathcal{Y}}_h \cap \mathcal{O}_g) = (\mathring{\mathcal{Y}}_g \cap \mathcal{O}_g) \times \mathring{Z}_{g,h}$.
- (FS4) For all $y \in \mathring{\mathcal{Y}}_g \cap \mathcal{O}_g$, we have $\bar{\nu}_g(y) = (y, 0)$.
- (FS5) $\operatorname{Star}_{q}^{\geq 0} \subset \mathcal{O}_{q}.$

Theorem 3.2.4. Suppose that a TNN space $(\mathcal{Y}, \mathcal{Y}^{\geq 0}, Q)$ admits a Fomin–Shapiro atlas. Then $\mathcal{Y}^{\geq 0} = \bigsqcup_{h \in Q} \mathcal{Y}_h^{>0}$ is a regular CW complex. In particular, for each $h \in Q$, $\mathcal{Y}_h^{\geq 0}$ is homeomorphic to a closed ball of dimension dim(h).

Thus Theorem 1.2.1 follows as a corollary of the following result.

Theorem 3.2.5. $(G/P, (G/P)_{\geq 0}, Q_J)$ is a TNN space that admits a Fomin–Shapiro atlas.

3.2.2 Plan of the proof

We give a brief outline of the proof of Theorem 3.2.5. See Section 3.4 for background on G/Pand see Sections 3.A and 3.7 for background on \mathcal{G}/\mathcal{B} . We deduce that $(G/P, (G/P)_{\geq 0}, Q_J)$ is a TNN space from known results in Corollary 3.4.20. In order to construct a Fomin–Shapiro atlas, we consider the (infinite-dimensional) affine flag variety \mathcal{G}/\mathcal{B} associated to G. It is stratified into (finite-dimensional) affine Richardson varieties $\mathcal{G}/\mathcal{B} = \bigsqcup_{\tilde{h} \leq \tilde{f} \in \tilde{W}} \mathcal{R}_{\tilde{h}}^{\tilde{f}}$, where \tilde{W} is the affine Weyl group and \leq denotes its Bruhat order. There exists an order-reversing injective map $\psi : Q_J \to \tilde{W}$, defined by [HL15], see (3.7.7). The set of minimal elements of Q_J equals $\{(u, u) \mid u \in W^J\}$, where W^J is the set of minimal length parabolic coset representatives of the Weyl group, see Section 3.4.6. For each minimal element $f := (u, u) \in$ Q_J, ψ identifies the interval $[f, \hat{1}]$ of Q_J with (the dual of) a certain interval $[\tau_{\lambda}^J, \tau_{u\lambda}] \subset \tilde{W}$. For the case $G/P = \operatorname{Gr}(k, n)$, elements of Q_J are in bijection with Le diagrams of [Pos07], and ψ sends a Le diagram indexing a positroid cell to the corresponding bounded affine permutation of [KLS14], see Example 3.9.6.

In Section 3.7.3, we lift ψ to the geometric level: given a minimal element $f := (u, u) \in Q_J$, we introduce a map $\bar{\varphi}_u : C_u^{(J)} \to \mathcal{G}/\mathcal{B}$ defined on an open dense subset $C_u^{(J)} \subset G/P$. We show in Theorem 3.7.2 that for $g \in Q_J$ such that $g \succeq f$, $\bar{\varphi}_u$ sends $C_u^{(J)} \cap \mathring{\Pi}_g$ isomorphically to the affine Richardson cell $\mathring{\mathcal{R}}_{\psi(g)}^{\psi(f)}$.

For every $\tilde{g} \in \tilde{W}$, we consider an open dense subset $C_{\tilde{g}} \subset \mathcal{G}/\mathcal{B}$ defined by $C_{\tilde{g}} := \tilde{g} \cdot \mathcal{B}_{-} \cdot \mathcal{B}/\mathcal{B}$, as well as affine Schubert and opposite Schubert cells $\hat{\mathcal{X}}^{\tilde{g}} = \bigsqcup_{\tilde{h} \leq \tilde{g}} \hat{\mathcal{R}}_{\tilde{h}}^{\tilde{g}}, \ \hat{\mathcal{X}}_{\tilde{g}} = \bigsqcup_{\tilde{g} \leq \tilde{f}} \hat{\mathcal{R}}_{\tilde{g}}^{\tilde{f}}$. In Proposition 3.8.2, we give a natural isomorphism

$$\mathcal{C}_{\tilde{g}} \xrightarrow{\sim} \mathcal{X}_{\tilde{g}} \times \mathcal{X}^{\tilde{g}}$$
, which restricts to $(\mathcal{C}_{\tilde{g}} \cap \mathcal{R}_{\tilde{h}}^{\tilde{f}}) \xrightarrow{\sim} \mathcal{R}_{\tilde{g}}^{\tilde{f}} \times \mathcal{R}_{\tilde{h}}^{\tilde{g}}$ for all $\tilde{h} \leq \tilde{g} \leq \tilde{f}$. (3.2.2)

A finite-dimensional analog of this map is due to [KWY13], and similar maps have been considered by [KL79, FS00]. The action of ϑ on $\mathring{\mathcal{X}}^{\tilde{g}}$ essentially amounts to multiplying by an element of the affine torus, and thus preserves $\mathring{\mathcal{R}}_{\tilde{h}}^{\tilde{g}}$ for all $\tilde{h} \leq \tilde{g}$.

Let us now fix $g \in Q_J$, and choose some minimal element $f := (u, u) \in Q_J$ such that $f \leq g$. Then the map $\bar{\varphi}_u$ is defined on an open dense subset $C_u^{(J)} \subset G/P$, and let us denote by $\mathcal{O}_g \subset C_u^{(J)}$ the preimage of $\mathcal{C}_{\psi(g)}$ under $\bar{\varphi}_u$. The diffeomorphism (3.2.1) is obtained by

conjugating the isomorphism (3.2.2) by the map $\bar{\varphi}_u$. The smooth cone (Z_g, ϑ_g) is extracted from the corresponding structure on $\mathcal{X}^{\psi(g)}$. As we have already mentioned, the hardest step in the proof consists of showing (FS5). To achieve this, we study subtraction-free parametrizations of partial flag varieties in Section 3.5, and then use them to show that some generalized minors of a particular group element $\zeta_{u,v}^{(J)}(x)$ from Section 3.6 do not vanish for all $x \in \operatorname{Star}_g^{\geq 0}$. The definition of $\zeta_{u,v}^{(J)}(x)$ is quite technical, but we conjecture in Section 3.9 that in the Grassmannian case, these generalized minors specialize to simple functions on $\operatorname{Gr}(k, n)$ that we call *u*-truncated minors. We complete the proof of Theorem 3.2.5 in Section 3.8.3.

3.3 Topological results

Throughout this section, we assume that $(\mathcal{Y}, \mathcal{Y}^{\geq 0}, Q)$ is a TNN space that admits a Fomin– Shapiro atlas. Thus for each $g \in Q$, we have the objects \mathcal{O}_g , Z_g , ϑ_g , and $\bar{\nu}_g$ from Definition 3.2.3. Additionally, we assume some familiarity with basic theory of smooth manifolds, see e.g. [Lee13].

3.3.1 Links

Throughout, we denote the two components of the map $\bar{\nu}_g$ from (3.2.1) by $\bar{\nu}_g = (\bar{\nu}_{g,1}, \bar{\nu}_{g,2})$, where $\bar{\nu}_{g,1} : \mathcal{O}_g \to \mathring{\mathcal{Y}}_g \cap \mathcal{O}_g$ and $\bar{\nu}_{g,2} : \mathcal{O}_g \to Z_g$. We set $\operatorname{Star}_{g,h}^{\geq 0} := \mathcal{Y}_h^{\geq 0} \cap \operatorname{Star}_g^{\geq 0} = \bigsqcup_{g \preceq g' \preceq h} \mathcal{Y}_{g'}^{>0}$.

Definition 3.3.1. Let $g \leq h$ and assume that $Z_g \subset \mathbb{R}^{N_g}$. Denote

$$Z_{g}^{\geq 0} := \bar{\nu}_{g,2} \left(\operatorname{Star}_{g}^{\geq 0} \right), \qquad \qquad Z_{g,h}^{\geq 0} := \bar{\nu}_{g,2} \left(\operatorname{Star}_{g,h}^{\geq 0} \right), \qquad \qquad Z_{g,h}^{\geq 0} := Z_{g}^{\geq 0} \cap \mathring{Z}_{g,h}$$
$$S_{g} := \{ x \in \mathbb{R}^{N_{g}} : \|x\| = 1 \}, \qquad \operatorname{Lk}_{g,h}^{\geq 0} := Z_{g,h}^{\geq 0} \cap S_{g}, \qquad \qquad \operatorname{Lk}_{g,h}^{\geq 0} := Z_{g,h}^{\geq 0} \cap S_{g}.$$

Notice that $\operatorname{Lk}_{g,g}^{>0} \subset S_g \cap \mathring{Z}_{g,g}$, but $\mathring{Z}_{g,g} = \{0\}$ by (FS1), thus we have a stratification

$$\operatorname{Lk}_{g,h}^{\geq 0} = \bigsqcup_{g \prec g' \preceq h} \operatorname{Lk}_{g,g'}^{> 0}.$$
(3.3.1)

Recall that $\operatorname{Cone}(A) := (A \times \mathbb{R}_{\geq 0})/(A \times \{0\})$ is the open cone over A. We denote by $c := (*, 0) \in \operatorname{Cone}(A)$ its *cone point*.

Lemma 3.3.2. Let $g \prec h \in Q$.

- (i) For all $x \in \mathcal{O}_g$, we have $x \in \mathcal{Y}_h^{>0}$ if and only if $\bar{\nu}_g(x) \in \mathcal{Y}_g^{>0} \times Z_{q,h}^{>0}$.
- (ii) $Z_{g,h}^{>0}$ is an embedded submanifold of Z_g that intersects S_g transversely.
- (iii) $Lk_{g,h}^{>0}$ is a contractible smooth manifold of dimension $\dim(h) \dim(g) 1$.
- (iv) $\operatorname{Lk}_{a,h}^{\geq 0}$ is a compact subset of Z_g .

Before we prove these properties, let us state some simple preliminary results on smooth manifolds. Given smooth manifolds A, B and a smooth map $f : A \to B$, a point $a \in A$ is called a *regular point* of f if the differential of f at a is surjective. Similarly, $b \in B$ is called a *regular value* of f if $f^{-1}(b)$ consists of regular points. In this case $f^{-1}(b)$ is a closed embedded submanifold of A of dimension dim $(A) - \dim(B)$.

Lemma 3.3.3. Suppose that A, B are smooth manifolds and $B' \subset B$ is such that $A \times B'$ is an embedded submanifold of $A \times B$. Then B' is an embedded submanifold of B.

Proof. Choose $a \in A$. Clearly a is a regular value of the projection map $A \times B' \to A$, therefore $\{a\} \times B'$ is an embedded submanifold of $\{a\} \times B$. The projection map $\{a\} \times B \to B$ is a diffeomorphism.

Proof of Lemma 3.3.2. (i): We prove this more generally for $g \leq h$. The set $\operatorname{Star}_{g}^{\geq 0}$ is a connected since it contains a connected dense subset $\mathcal{Y}_{1}^{>0}$. Therefore $\bar{\nu}_{g,1}(\operatorname{Star}_{g}^{\geq 0})$ is a connected subset of $\mathring{\mathcal{Y}}_{g} \cap \mathcal{O}_{g}$. By (FS4), it contains $\mathcal{Y}_{g}^{>0}$, therefore $\bar{\nu}_{g,1}(\operatorname{Star}_{g}^{\geq 0}) = \mathcal{Y}_{g}^{>0}$ by (TNN4). By definition, $\bar{\nu}_{g,2}(\operatorname{Star}_{g,h}^{\geq 0}) = Z_{g,h}^{\geq 0}$, thus $\bar{\nu}_{g}(\operatorname{Star}_{g,h}^{\geq 0}) \subset \mathcal{Y}_{g}^{>0} \times Z_{g,h}^{\geq 0}$. By (FS3), we get $\bar{\nu}_{g}(\mathcal{Y}_{h}^{>0}) \subset \mathcal{Y}_{g}^{>0} \times Z_{g,h}^{>0}$. In particular, $Z_{g,h}^{>0} = \bar{\nu}_{g,2}(\mathcal{Y}_{h}^{>0})$ is a connected subset of $\mathring{Z}_{g,h}$. Let C be the connected component of $\mathring{Z}_{g,h}$ containing $Z_{g,h}^{>0}$. By (FS3), $\bar{\nu}_{g}^{-1}(\mathcal{Y}_{g}^{>0} \times C)$ is a connected subset of $\mathring{\mathcal{Y}}_{h} \cap \mathcal{O}_{g}$, which contains $\mathcal{Y}_{h}^{>0}$ as we have just shown. Therefore we must have $\bar{\nu}_{g}^{-1}(\mathcal{Y}_{g}^{>0} \times C) = \mathcal{Y}_{h}^{>0}$ by (TNN4), which shows that $Z_{g,h}^{>0} = C$ is a connected component of $\mathring{Z}_{g,h}$. Thus indeed $\bar{\nu}_{g}(\mathcal{Y}_{h}^{>0}) = \mathcal{Y}_{g}^{>0} \times Z_{g,h}^{>0}$.

(ii): Applying Lemma 3.3.3 to $A := \mathcal{Y}_g^{>0}$ and $B := Z_g$, and $B' := Z_{g,h}^{>0}$, we get that $Z_{g,h}^{>0}$ is an embedded submanifold of Z_g by (i). Moreover, it follows from (FS2) that $\vartheta_g(t, Z_{g,h}^{>0}) = Z_{g,h}^{>0}$ for all $t \in \mathbb{R}_{>0}$ since $Z_{g,h}^{>0}$ is a connected component of $\mathring{Z}_{g,h}$. Thus 1 is a regular value of the restriction $\|\cdot\| : Z_{g,h}^{>0} \to \mathbb{R}_{>0}$, so the manifolds S_g and $Z_{g,h}^{>0}$ intersect transversely inside \mathbb{R}^{N_g} . (iii): By (ii), $\operatorname{Lk}_{g,h}^{>0} = Z_{g,h}^{>0} \cap S_g$ is an embedded submanifold of Z_g of dimension dim $(h) - \dim(g) - 1$. To show that it is contractible, note that $\mathcal{Y}_h^{>0}$ is contractible, therefore so is $\bar{\nu}_g(\mathcal{Y}_h^{>0}) = \mathcal{Y}_g^{>0} \times Z_{g,h}^{>0}$. We see that $Z_{g,h}^{>0}$ is contractible since $\{x\} \times Z_{g,h}^{>0}$ is a retract of $\mathcal{Y}_g^{>0} \times Z_{g,h}^{>0}$ for any $x \in \mathcal{Y}_g^{>0}$, cf. [Hat02, Ex. 0.9].

For each $x \in \mathbb{R}^{N_g} \setminus \{0\}$, the function $t \mapsto \|\vartheta_g(t,x)\|$ is a diffeomorphism $\mathbb{R}_{>0} \xrightarrow{\sim} \mathbb{R}_{>0}$ by Lemma 2.1.2, so there exists a unique $t_x \in \mathbb{R}_{>0}$ such that $\|\vartheta_g(t_x,x)\| = 1$, and we get a continuous map $\mathbb{R}^{N_g} \setminus \{0\} \to S_g$ sending $x \mapsto \vartheta_g(t_x,x)$. It gives a retraction $Z_{g,h}^{>0} \to \mathrm{Lk}_{g,h}^{>0}$, finishing the proof of (iii).

(iv): Clearly $\operatorname{Star}_{g,h}^{\geq 0} = \mathcal{Y}_h^{\geq 0} \cap \operatorname{Star}_g^{\geq 0} = \mathcal{Y}_h^{\geq 0} \cap \mathcal{O}_g$ is a closed subset of \mathcal{O}_g . Thus $\bar{\nu}_g(\operatorname{Star}_{g,h}^{\geq 0})$ is a closed subset of $\mathcal{Y}_g^{>0} \times Z_g$. Since $\bar{\nu}_g(\operatorname{Star}_{g,h}^{\geq 0}) = \mathcal{Y}_g^{>0} \times Z_{g,h}^{\geq 0}$, we get that $Z_{g,h}^{\geq 0}$ is a closed subset of Z_g . It follows that $\operatorname{Lk}_{g,h}^{\geq 0} = Z_{g,h}^{\geq 0} \cap S_g$ is a closed bounded subset of Z_g , which is closed in \mathbb{R}^{N_g} by Definition 3.2.2.

Proposition 3.3.4. Let $g \prec h \in Q$.

- (i) There exists a homeomorphism $Z_{g,h}^{\geq 0} \xrightarrow{\sim} \text{Cone}(\text{Lk}_{g,h}^{\geq 0})$ sending 0 to the cone point c. For $g \prec g' \preceq h$, it sends $Z_{g,g'}^{>0}$ to $\text{Lk}_{g,g'}^{>0} \times \mathbb{R}_{>0}$.
- (ii) We have a homeomorphism

$$\operatorname{Star}_{g,h}^{\geq 0} \xrightarrow{\sim} \mathcal{Y}_g^{>0} \times \operatorname{Cone}(\operatorname{Lk}_{g,h}^{\geq 0}).$$
 (3.3.2)

Proof. (i): Consider the map $\xi: Z_{g,h}^{\geq 0} \to \operatorname{Cone}(\operatorname{Lk}_{g,h}^{\geq 0})$ sending $0 \mapsto c$ and $x \mapsto \left(\vartheta_g(t_x, x), \frac{1}{t_x}\right)$ for $x \in Z_{g,h}^{\geq 0} \setminus \{0\}$, where t_x was defined in the proof of Lemma 3.3.2(iii). We claim that ξ is a homeomorphism. We have already shown that $\vartheta_g(t, Z_{g,g'}^{>0}) = Z_{g,g'}^{>0}$ for all $g \preceq g'$. In particular, for $x \in Z_{g,h}^{\geq 0}$, we find that $\xi(t_x, x) \in \operatorname{Cone}(\operatorname{Lk}_{g,h}^{\geq 0})$. It is easy to see that ξ is a bijection whose inverse sends $c \mapsto 0$ and $(y, t) \mapsto \vartheta_g(t, y)$ for $(y, t) \in \operatorname{Cone}(\operatorname{Lk}_{g,h}^{\geq 0}) \setminus \{c\} = \operatorname{Lk}_{g,h}^{\geq 0} \times \mathbb{R}_{>0}$. Clearly ξ is continuous on $Z_{g,h}^{\geq 0} \setminus \{0\}$ while its inverse is continuous on $\operatorname{Lk}_{g,h}^{\geq 0} \times \mathbb{R}_{>0}$.

Suppose that $(x_n)_{n\geq 0}$ is a sequence of elements of $Z_{g,h}^{\geq 0} \setminus \{0\}$ converging to 0. We claim that $t_{x_n} \to \infty$ as $n \to \infty$. Otherwise, after passing to a subsequence, we may assume that $t_{x_n} \leq R$ for some $R \in \mathbb{R}_{>0}$ and all $n \geq 0$. The limit of $\vartheta_g(R, x_n)$ is $\vartheta_g(R, 0) = 0$, which gives a contradiction since $\|\vartheta_g(R, x_n)\| \geq \|\vartheta_g(t_{x_n}, x_n)\| = 1$ for all $n \geq 0$. This shows that ξ is continuous at 0. Suppose now that $((y_n, t_n))_{n\geq 0}$ is a sequence of elements of $\operatorname{Lk}_{g,h}^{\geq 0} \times \mathbb{R}_{>0}$ that converges to c, which is equivalent to $t_n \to 0$. Let $x_n := \vartheta_g(t_n, y_n) = \xi^{-1}(y_n, t_n)$. Suppose that $||x_n||$ does not converge to 0. The function $D(t) := \max_{x \in S_g} ||\vartheta_g(t, x)||$ is strictly increasing in t by compactness of S_g and (SC2). Since for each $x \in S_g$, $\lim_{t\to 0^+} ||\vartheta_g(t, x)|| = 0$ by Lemma 2.1.2, we get $\lim_{t\to 0^+} D(t) = 0$ by compactness of S_g (more precisely, by *Dini's theorem*). This gives a contradiction, showing that ξ^{-1} is continuous at c.

(ii): By Lemma 3.3.2(i), $\bar{\nu}_g$ restricts to a homeomorphism $\operatorname{Star}_{g,h}^{\geq 0} \xrightarrow{\sim} \mathcal{Y}_g^{\geq 0} \times Z_{g,h}^{\geq 0}$. By (i), $Z_{g,h}^{\geq 0}$ is homeomorphic to $\operatorname{Cone}(\operatorname{Lk}_{g,h}^{\geq 0})$.

Our next aim is to analyze the local structure of the space $Lk_{g,h}^{\geq 0}$. For two topological spaces A and B and $a \in A, b \in B$, we say that there is a *local homeomorphism between* (A, a) and (B, b) if there exist open neighborhoods $a \in U \subset A, b \in V \subset B$, and a homeomorphism $U \xrightarrow{\sim} V$ sending a to b.

Lemma 3.3.5. Let $g \prec p \preceq h \in Q$, $x_p \in Lk_{g,p}^{>0}$, and $d := \dim(p) - \dim(g) - 1$. Then there exists a local homeomorphism between $(Lk_{g,h}^{\geq 0}, x_p)$ and $(Z_{p,h}^{\geq 0} \times \mathbb{R}^d, (0,0))$.

Proof. Choose some $x_g \in \mathcal{Y}_g^{>0}$ and consider the open subset $H_p \subset Z_g$ defined by $H_p := \{x \in Z_g \mid \bar{\nu}_g^{-1}(x_g, x) \in \mathcal{O}_p\}$. Introduce a map

$$\theta_{g,p}: H_p \cap S_g \to Z_p, \quad x \mapsto \bar{\nu}_{p,2}(\bar{\nu}_q^{-1}(x_g, x)). \tag{3.3.3}$$

Since $x_p \in Lk_{g,p}^{>0} \subset Z_{g,p}^{>0}$ and $x_g \in \mathcal{Y}_g^{>0}$, we get $\bar{x}_p := \bar{\nu}_g^{-1}(x_g, x_p) \in \mathcal{Y}_p^{>0}$ by Lemma 3.3.2(i). By (FS5), $\mathcal{Y}_p^{>0} \subset \operatorname{Star}_p^{\geq 0} \subset \mathcal{O}_p$, thus $x_p \in H_p$. Since H_p is open in Z_g , $H_p \cap S_g$ is an open subset of $Z_g \cap S_g$, which is nonempty since it contains x_p . Moreover, since $\bar{\nu}_{p,2}(\mathcal{Y}_p^{>0}) = Z_{p,p}^{>0} = \{0\}$ by Lemma 3.3.2(i), we find that $\theta_{g,p}(x_p) = 0$.

We claim that x_p is a regular point of $\theta_{g,p}$. Clearly the differential of $\bar{\nu}_{p,2} : \mathcal{O}_p \to Z_p$ is surjective at \bar{x}_p , and its kernel is the tangent space of $\mathring{\mathcal{Y}}_p$ at \bar{x}_p . Recall from (TNN4) and (FS5) that $\mathcal{Y}_p^{>0}$ is a connected component of $\mathring{\mathcal{Y}}_p \cap \mathcal{O}_p$, and it contains $\bar{x}_p = \bar{\nu}_g^{-1}(x_g, x_p)$ as we have shown above. Therefore x_p is a regular point of $\theta_{g,p}$ if and only if the manifolds $\mathcal{Y}_p^{>0}$ and $F := \bar{\nu}_g^{-1}(\{x_g\} \times (H_p \cap S_g))$ intersect transversely at \bar{x}_p . By Lemma 3.3.2(i), we have $\bar{\nu}_g(\mathcal{Y}_p^{>0}) = \mathcal{Y}_g^{>0} \times Z_{g,p}^{>0}$, and clearly $\bar{\nu}_g(F) = \{x_g\} \times (H_p \cap S_g)$. These two manifolds intersect transversely at (x_g, x_p) by Lemma 3.3.2(ii). We have shown that x_p is a regular point of $\theta_{g,p}$. By the Submersion theorem (see e.g. [Kos93, Cor. A(1.3)]), there exist local coordinates centered at $x_p \in H_p \cap S_g$ and $0 \in Z_p$ in which $\theta_{g,p}$ is just the canonical projection $\mathbb{R}^{\dim(H_p \cap S_g)} \to \mathbb{R}^{\dim(Z_p)}$. Recall that Q contains a unique maximal element $\hat{1}$, and we have $\dim(Z_g) = \operatorname{codim}(g) := \dim(\hat{1}) - \dim(g)$. Thus $\dim(H_p \cap S_g) = \operatorname{codim}(g) - 1$, $\dim(Z_p) = \operatorname{codim}(p)$, and $\dim(H_p \cap S_g) - \dim(Z_p) = d$. We have shown that there exist open neighborhoods $x_p \in U \subset H_p \cap S_g$ and $0 \in V \subset Z_p$ and a diffeomorphism $\beta : U \xrightarrow{\sim} V \times \mathbb{R}^d$ sending x_p to (0, 0).

In order to complete the proof, we need to show that the image $\beta(U \cap \operatorname{Lk}_{g,h}^{\geq 0})$ equals $(V \cap Z_{p,h}^{\geq 0}) \times \mathbb{R}^d$. We may assume that U is connected. Suppose we are given $x \in U$ and let $r \in Q$ be such that $x' := \bar{\nu}_g^{-1}(x_g, x) \in \mathring{\mathcal{Y}}_r$. Since $U \subset H_p$, x' belongs to $\mathcal{O}_p \subset \operatorname{Star}_p$ by Definition 3.2.3, and therefore $p \preceq r$. By Lemma 3.3.2(i), we have $x \in U \cap \operatorname{Lk}_{g,r}^{>0}$ if and only if $x' \in \mathcal{Y}_r^{>0}$. On the other hand, $\bar{\nu}_{p,1}(\bar{\nu}_g^{-1}(\{x_g\} \times U))$ is a connected subset of $\mathring{\mathcal{Y}}_p \cap \mathcal{O}_p$ that contains $\bar{\nu}_{p,1}(\bar{x}_p) \in \mathcal{Y}_p^{>0}$. Thus $\bar{\nu}_{p,1}(\bar{\nu}_g^{-1}(x_g, U)) \subset \mathcal{Y}_p^{>0}$ by (TNN4). It follows that $x' \in \mathcal{Y}_r^{>0}$ if and only if $\theta_{g,p}(x) = \bar{\nu}_{p,2}(x')$ belongs to $Z_{p,r}^{>0}$. The result follows by taking the union over all $p \preceq r \preceq h$.

3.3.2 Topological background

Regular CW complexes

We refer to [Hat02, LW69] for background on CW complexes.

Definition 3.3.6. Let X be a Hausdorff space. We call a finite disjoint union $X = \bigsqcup_{\alpha \in P} X_{\alpha}$ a *regular CW complex* if it satisfies the following two properties.

(CW1) For each $\alpha \in P$, there exists a homeomorphism from the closure $\overline{X_{\alpha}}$ to a closed ball \overline{B} which sends X_{α} to the interior of \overline{B} ;

(CW2) For each $\beta \in P$, $\overline{X_{\beta}}$ equals $\bigsqcup_{\alpha \in P'} X_{\alpha}$ for some $P' \subset P$.

The property (CW2) is often omitted from the definition of a regular CW complex, but is necessary in order to apply the arguments of [Bjö84]. We remark that the cell decomposition of $\mathcal{Y}^{\geq 0}$ satisfies (CW2) by (TNN5).

Posets

We review the notions of thinness and shellability (see e.g. [Bjö80]), though we will not need them in our arguments. Let (P, \leq) be a finite poset. We say that P is thin if for all $x \leq y \leq z$ in P (where \leq denotes a covering relation of P), there exists a unique $y' \neq y$ with x < y' < z. We say that a pure d-dimensional simplicial complex Δ is shellable if its maximal faces can be ordered as F_1, \ldots, F_n so that for $2 \leq k \leq n$, $F_k \cap (\bigcup_{1 \leq i < k} F_i)$ is a nonempty union of (d-1)-dimensional faces of F_k . We say that P is graded if all maximal chains in P have the same length l, in which case we denote by ht(P) := l the height of P. Each graded poset Pgives rise to a pure (ht(P) - 1)-dimensional simplicial complex called the order complex Δ_P of P. The vertices of Δ_P are the elements of P and the faces of Δ_P are the chains in P. We say that P is shellable if its order complex Δ_P is shellable. For $x \leq z \in P$, we denote by $[x, z] := \{y \in P \mid x \leq y \leq z\}$ the corresponding interval.

Proposition 3.3.7 ([Bjö80, Prop. 4.2]). If a poset is shellable then so are each of its intervals.

See [Bjö84, §2,3] for the proof of the following result.

Theorem 3.3.8 ([LW69, DK74, Bjö84]). Suppose that X is a regular CW complex with face poset P. If $P \sqcup \{\hat{0}, \hat{1}\}$ is graded, thin, and shellable, then X is homeomorphic to a sphere of dimension ht(P) - 1.

Poincaré conjecture

Recall that an *n*-dimensional topological manifold C with boundary is a Hausdorff space such that every point $x \in C$ has an open neighborhood homeomorphic either to \mathbb{R}^n or to $\mathbb{R}_{\geq 0} \times \mathbb{R}^{n-1}$. In the latter case, we say that x belongs to the boundary ∂C of C.

The following is a well known consequence of the *(generalized) Poincaré conjecture* due to Smale [Sma61], Freedman [Fre82], and Perelman [Per02, Per03a, Per03b]. We refer to [Dav08, Thm. 10.3.3(ii)] for this formulation.

Theorem 3.3.9 ([Sma61, Fre82, Per02, Per03a, Per03b]). Let C be a compact contractible ndimensional topological manifold with boundary, such that its boundary ∂C is homeomorphic to an (n-1)-dimensional sphere. Then C is homeomorphic to a closed n-dimensional ball. For $n \ge 6$, Theorem 3.3.9 can be proved using the topological h-cobordism theorem [Mil65, KS77]. We sketch another standard argument for deducing Theorem 3.3.9 from the Poincaré conjecture when n is arbitrary. The boundary of C is collared by [Bro62, Thm. 2]. Thus we can attach the (collared) boundary of a closed n-dimensional ball to the (collared) boundary of C, obtaining a topological manifold C'. By van Kampen's theorem, C' is simply connected. It is easy to see from the Mayer–Vietoris sequence that C' is a homology sphere. Thus C' must be homeomorphic to a sphere by the Poincaré conjecture. Therefore C is homeomorphic to a closed ball by Brown's Schoenflies theorem [Bro60].

Proposition 3.3.10. Suppose that C is a topological manifold with boundary ∂C . Then C is homotopy equivalent to its interior $int(C) := C \setminus \partial C$.

Proof. By [Bro62, Thm. 2], there exists an open subset $\partial C \subset U \subset C$ that is homeomorphic to $\partial C \times [0, 1)$, which shows the result.

3.3.3 Link induction

Theorem 3.3.11. Let $(\mathcal{Y}, \mathcal{Y}^{\geq 0}, Q)$ be a TNN space that admits a Fomin–Shapiro atlas. Suppose that $g \prec h \in Q$. Then $Lk_{g,h}^{\geq 0}$ is homeomorphic to a closed ball of dimension $d := \dim(h) - \dim(g) - 1$.

Proof. We proceed by induction on d. For the base case d = 0, we see by (3.3.1) that $Lk_{g,h}^{\geq 0} = Lk_{g,h}^{>0}$, which is a 0-dimensional contractible manifold by Lemma 3.3.2(iii). Thus $Lk_{g,h}^{\geq 0}$ is a point and we are done with the base case.

Assume now that d > 0. We claim that $Lk_{g,h}^{\geq 0}$ is a topological manifold with boundary

$$\partial \operatorname{Lk}_{g,h}^{\geq 0} = \bigsqcup_{g \prec g' \prec h} \operatorname{Lk}_{g,g'}^{\geq 0}.$$
(3.3.4)

Let $x \in Lk_{g,h}^{\geq 0}$. By (3.3.1), we have $x \in Lk_{g,g'}^{>0}$ for a unique $g \prec g' \preceq h$. If g' = h, then x has an open neighborhood in $Lk_{g,h}^{\geq 0}$ homeomorphic to \mathbb{R}^d by Lemma 3.3.2(iii). If $g' \prec h$, then by Lemma 3.3.5 we have a local homeomorphism $(Lk_{g,h}^{\geq 0}, x) \xrightarrow{\sim} (Z_{g',h}^{\geq 0} \times \mathbb{R}^d, (0,0))$, where $d' := \dim(g') - \dim(g) - 1$. By Proposition 3.3.4(i), we have a homeomorphism $Z_{g',h}^{\geq 0} \xrightarrow{\sim} \text{Cone}(Lk_{g',h}^{\geq 0})$ which sends 0 to the cone point c. By the induction hypothesis, $Lk_{g',h}^{\geq 0}$

is homeomorphic to a (d - d' - 1)-dimensional closed ball, and so we have a homeomorphism $\operatorname{Cone}(\operatorname{Lk}_{g',h}^{\geq 0}) \xrightarrow{\sim} \mathbb{R}_{\geq 0} \times \mathbb{R}^{d-d'-1}$ which sends c to (0,0). Composing gives a local homeomorphism $(\operatorname{Lk}_{g,h}^{\geq 0}, x) \xrightarrow{\sim} (\mathbb{R}_{\geq 0} \times \mathbb{R}^{d-d'-1} \times \mathbb{R}^{d'}, (0,0,0))$. Thus indeed $\operatorname{Lk}_{g,h}^{\geq 0}$ is a topological manifold with boundary given by (3.3.4).

By Lemma 3.3.2(iv), $Lk_{g,h}^{\geq 0}$ is compact. By Lemma 3.3.2(iii) and Proposition 3.3.10, $Lk_{g,h}^{\geq 0}$ is contractible. Thus $Lk_{g,h}^{\geq 0}$ is a compact contractible topological manifold with boundary. In addition, the boundary $\partial Lk_{g,h}^{\geq 0}$ is a regular CW complex by the induction hypothesis. Its face poset is the interval $(g,h) := [g,h] \setminus \{g,h\}$ in Q. The interval [g,h] is graded, thin, and shellable by (TNN1) and Proposition 3.3.7, thus $\partial Lk_{g,h}^{\geq 0}$ is homeomorphic to a (d-1)-dimensional sphere by Theorem 3.3.8. We are done by Theorem 3.3.9.

Proof of Theorem 3.2.4. The proof follows the same structure as the proof of Theorem 3.3.11. We proceed by induction on dim(h). If dim(h) = 0 then $\mathcal{Y}_h^{\geq 0} = \mathcal{Y}_h^{>0}$ is a point by (TNN4), which finishes the base case.

Let dim(h) > 0. We want to show that $\mathcal{Y}_h^{\geq 0}$ is a topological manifold with boundary

$$\partial \mathcal{Y}_{h}^{\geq 0} = \bigsqcup_{g \prec h} \mathcal{Y}_{g}^{\geq 0}. \tag{3.3.5}$$

Let $x \in \mathcal{Y}_h^{\geq 0}$. By (TNN5), we have $x \in \mathcal{Y}_g^{>0}$ for a unique $g \leq h$. If g = h then x has an open neighborhood in $\mathcal{Y}_h^{\geq 0}$ homeomorphic to $\mathbb{R}^{\dim(h)}$ by (TNN4).

If $g \prec h$ then $\operatorname{Star}_{g}^{\geq 0}$ is an open subset of $\mathcal{Y}^{\geq 0}$, since its complement is $\bigcup_{g' \not\equiv g} \mathcal{Y}_{g'}^{\geq 0}$, which is closed by (TNN5). Thus $\operatorname{Star}_{g,h}^{\geq 0}$ is an open neighborhood of x in $\mathcal{Y}_{h}^{\geq 0}$. By Proposition 3.3.4(ii), (TNN4), and Theorem 3.3.11, $\operatorname{Star}_{g,h}^{\geq 0}$ is homeomorphic to $\mathbb{R}_{\geq 0} \times \mathbb{R}^{\dim(h)-1}$. This shows that $\mathcal{Y}_{h}^{\geq 0}$ is a topological manifold with boundary given by (3.3.5).

By (TNN3) and (TNN5), $\mathcal{Y}_{h}^{\geq 0}$ is compact. By (TNN4) and Proposition 3.3.10, $\mathcal{Y}_{h}^{\geq 0}$ is contractible. Thus $\mathcal{Y}_{h}^{\geq 0}$ is a compact contractible topological manifold with boundary. In addition, the boundary $\partial \mathcal{Y}_{h}^{\geq 0}$ is a regular CW complex by the induction hypothesis. Its face poset is the interval $(\hat{0}, h)$ in \hat{Q} . The interval $[\hat{0}, h]$ is graded, thin, and shellable by (TNN1) and Proposition 3.3.7, thus $\partial \mathcal{Y}_{h}^{\geq 0}$ is a (d-1)-dimensional sphere by Theorem 3.3.8. We are done by Theorem 3.3.9.

Remark 3.3.12. We note that Theorems 3.2.5 and 3.3.11 imply the result of Hersh [Her14] (see Corollary 1.3.1) that the link of the identity in the Bruhat decomposition of $U_{\geq 0}$ is a regular CW complex. (Recall that U is the unipotent radical of the standard Borel subgroup $B \subset G$.) Indeed, let $B_{-} \subset G$ denote the opposite Borel subgroup. Then the natural inclusion $U \hookrightarrow G/B_{-}$ sends U to the opposite Schubert cell $\operatorname{Star}_{(\operatorname{id},\operatorname{id})}$ indexed by $\operatorname{id} \in W$, and the composition of this map with $\overline{\nu}_{(\operatorname{id},\operatorname{id})}$ sends the link of the identity in $\overline{U}_{\geq 0}^{w}$ homeomorphically to $\operatorname{Lk}_{(\operatorname{id},\operatorname{id}),(\operatorname{id},w)}^{\geq 0}$ for all $w \in W$.

3.4 G/P: preliminaries

We give some background on partial flag varieties. Throughout, \mathbb{K} denotes an algebraically closed field of characteristic 0, and $\mathbb{K}^* := \mathbb{K} \setminus \{0\}$ denotes its multiplicative group. We work with a simple algebraic group G; the case of a general semisimple group reduces to the simple case by taking products.

3.4.1 Pinnings

We recall some standard notions that can be found in e.g. [Lus94, §1]. Suppose that G is a simple and simply connected algebraic group over \mathbb{K} , with $T \subset G$ a maximal torus. Let B, B_- be opposite Borel subgroups satisfying $T = B \cap B_-$. We identify G with its set of \mathbb{K} -valued points. When $\mathbb{K} = \mathbb{C}$, we assume that G and T are split over \mathbb{R} , and denote by $G(\mathbb{R}) \subset G$ and $T(\mathbb{R}) \subset T$ the sets of their \mathbb{R} -valued points. (Thus what was denoted by Gin Section 1.2 is from now on denoted by $G(\mathbb{R})$.)

Let $X(T) := \text{Hom}(T, \mathbb{K}^*)$ be weight lattice, and for a weight $\gamma \in X(T)$ and $a \in T$, we denote the value of γ on a by a^{γ} . Let $\Phi \subset X(T)$ be the set of roots. We have a decomposition $\Phi = \Phi^+ \sqcup \Phi^-$ of Φ as a union of positive and negative roots corresponding to the choice of B, see [Hum75, §27.3]. For $\alpha \in \Phi$, we write $\alpha > 0$ if $\alpha \in \Phi^+$ and $\alpha < 0$ if $\alpha \in \Phi^-$. Let $\{\alpha_i\}_{i \in I}$ be the simple roots corresponding to the choice of Φ^+ . For every $i \in I$, we have a homomorphism $\phi_i : \text{SL}_2 \to G$, and denote

$$x_{i}(t) := \phi_{i} \begin{pmatrix} 1 & t \\ 0 & 1 \end{pmatrix}, \quad y_{i}(t) := \phi_{i} \begin{pmatrix} 1 & 0 \\ t & 1 \end{pmatrix}, \quad \dot{s}_{i} := \phi_{i} \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix} = y_{i}(1)x_{i}(-1)y_{i}(1).$$
(3.4.1)

The data $(T, B, B_{-}, x_i, y_i; i \in I)$ is called a *pinning* for G. Let $W := N_G(T)/T$ be the Weyl group, and for $i \in I$, let $s_i \in W$ be represented by \dot{s}_i above. Then W is generated by $\{s_i\}_{i\in I}$, and $(W, \{s_i\}_{i\in I})$ is a finite Coxeter group. For $w \in W$, we denote by $\ell(w)$ the minimal n such that $w = s_{i_1} \cdots s_{i_n}$ for some $i_1, \ldots, i_n \in I$. When $n = \ell(w)$, $\mathbf{i} := (i_1, \ldots, i_n)$ is called a *reduced word* for w. The representatives $\{\dot{s}_i\}_{i\in I}$ satisfy the braid relations, so we set $\dot{w} := \dot{s}_{i_1} \cdots \dot{s}_{i_n} \in G$, and this representative does not depend on the choice of \mathbf{i} .

Let $Y(T) := \operatorname{Hom}(\mathbb{K}^*, T)$ be the *coweight lattice*. For $i \in I$, we have a simple coroot $\alpha_i^{\vee}(t) := \phi_i \begin{pmatrix} t & 0 \\ 0 & t^{-1} \end{pmatrix} \in Y(T)$. Denote by $\langle \cdot, \cdot \rangle : X(T) \times Y(T) \to \mathbb{Z}$ the natural pairing so that for $\gamma \in X(T), \beta \in Y(T)$, and $t \in \mathbb{K}^*$, we have $(\beta(t))^{\gamma} = t^{\langle \gamma, \beta \rangle}$. Let $\{\omega_i\}_{i \in I} \subset X(T)$ be the fundamental weights. They form a dual basis to $\{\alpha_i^{\vee}\}_{i \in I} : \langle \omega_i, \alpha_j^{\vee} \rangle = \delta_{ij}$ for $i, j \in I$.

The Weyl group W acts on T by conjugation, which induces an action on Y(T), X(T), and Φ . For $\gamma \in X(T)$, $t \in \mathbb{K}^*$, $a \in T$, and $w \in W$, we have [FZ99, Eqns. (1.2) and (2.5)]

$$(\dot{w}^{-1}a\dot{w})^{\gamma} = a^{w\gamma}, \quad ax_i(t)a^{-1} = x_i(a^{\alpha_i}t), \quad ay_i(t)a^{-1} = y_i(a^{-\alpha_i}t).b$$
 (3.4.2)

Following [BZ97, Eqns. (1.6) and (1.7)] (see also [FZ99, Eqns. (2.1) and (2.2)]), we define two involutive anti-automorphisms $x \mapsto x^T$ and $x \mapsto x^{\iota}$ of G by

$$(x_i(t))^T = y_i(t),$$
 $(y_i(t))^T = x_i(t),$ $\dot{w}^T = \dot{w}^{-1},$ $a^T = a$ (3.4.3)

$$(x_i(t))^{\iota} = x_i(t),$$
 $(y_i(t))^{\iota} = y_i(t),$ $\dot{w}^{\iota} = \dot{z},$ $a^{\iota} = a^{-1},$ (3.4.4)

for all $i \in I$, $t \in \mathbb{K}^*$, $a \in T$, and $w \in W$, where $z := w^{-1}$. We note that when $z = w^{-1} \in W$ and $\mathbf{i} = (i_1, \ldots, i_n)$ is a reduced word for w then $\dot{w}^{-1} = \dot{s}_{i_n}^{-1} \cdots \dot{s}_{i_1}^{-1}$ while $\dot{z} = \dot{s}_{i_n} \cdots \dot{s}_{i_1}$.

3.4.2 Subgroups of U

We say that a subset $\Theta \subset \Phi$ is *bracket closed* if whenever $\alpha, \beta \in \Theta$ are such that $\alpha + \beta \in \Phi$, we have $\alpha + \beta \in \Theta$. For a bracket closed subset $\Theta \subset \Phi^+$, define $U(\Theta) \subset U$ to be the subgroup generated by $\{U_{\alpha} \mid \alpha \in \Theta\}$, where U_{α} is a *root subgroup* of *G*, see [Hum75, Theorem 26.3]. For a bracket closed subset $\Theta \subset \Phi^-$, let $U_{-}(\Theta) := U(-\Theta)^T \subset U_{-}$.

Given closed subgroups H_1, \ldots, H_n of an algebraic group H, we say that H_1, \cdots, H_n directly span H if the multiplication map $H_1 \times \cdots \times H_n \to H$ is a biregular isomorphism.

Lemma 3.4.1 ([Hum75, Prop. 28.1]). Let $\Theta \subset \Phi^+$ be a bracket closed subset.

- (i) If $\Theta = \bigsqcup_{i=1}^{n} \Theta_i$ and $\Theta, \Theta_1, \ldots, \Theta_n \subset \Phi^+$ are bracket closed then $U(\Theta)$ is directly spanned by $U(\Theta_1), \ldots, U(\Theta_n)$.
- (ii) In particular, $U(\Theta)$ is directly spanned by $\{U_{\alpha} \mid \alpha \in \Theta\}$ in any order, and therefore $U(\Theta) \cong \mathbb{K}^{|\Theta|}$.

For $\alpha \in \Phi$ and $w \in W$, we have $\dot{w}U_{\alpha}\dot{w}^{-1} = U_{w\alpha}$. For $w \in W$, define $\operatorname{Inv}(w) := (w^{-1}\Phi^{-}) \cap \Phi^{+}$. The subsets $\operatorname{Inv}(w)$ and $\Phi^{+} \setminus \operatorname{Inv}(w)$ are bracket closed [Hum75, §28.1], and

$$U(\text{Inv}(w)) = \dot{w}^{-1} U_{-} \dot{w} \cap U.$$
(3.4.5)

3.4.3 Bruhat projections

Let $\Theta \subset \Phi^+$ be bracket closed, and let $w \in W$. Define $\Theta_1 := \Theta \cap \operatorname{Inv}(w)$ and $\Theta_2 := \Theta \setminus \operatorname{Inv}(w)$. Thus both sets are bracket closed and

$$\dot{w}U(\Theta)\dot{w}^{-1}\cap U_{-}=U_{-}(w\Theta_{1}),\quad \dot{w}U(\Theta)\dot{w}^{-1}\cap U=U(w\Theta_{2}).$$

Denote $U_1 := U_-(w\Theta_1)$ and $U_2 := U(w\Theta_2)$. Then by Lemma 3.4.1(i), the multiplication map gives isomorphisms $\mu_{12} : U_1 \times U_2 \to \dot{w}U(\Theta)\dot{w}^{-1}$ and $\mu_{21} : U_2 \times U_1 \to \dot{w}U(\Theta)\dot{w}^{-1}$. Denote by $\nu_1 : \dot{w}U(\Theta)\dot{w}^{-1} \to U_1$ and $\nu_2 : \dot{w}U(\Theta)\dot{w}^{-1} \to U_2$ the second component of μ_{21}^{-1} and μ_{12}^{-1} , respectively. In other words, given $g \in \dot{w}U(\Theta)\dot{w}^{-1}$, $\nu_1(g)$ is the unique element in $U_1 \cap U_2g$ and $\nu_2(g)$ is the unique element in $U_2 \cap U_1g$. **Lemma 3.4.2** ([KWY13, Lemma 2.2]). The map (ν_1, ν_2) : $\dot{w}U(\Theta)\dot{w}^{-1} \rightarrow U_1 \times U_2$ is a biregular isomorphism.

3.4.4 Commutation relations

Let $a, b \in W$ be such that $\ell(ab) = \ell(a) + \ell(b)$. Then

$$\operatorname{Inv}(b) \subset \operatorname{Inv}(ab), \quad b^{-1}\operatorname{Inv}(a) \subset \Phi^+, \quad \text{and} \quad \operatorname{Inv}(ab) = (b^{-1}\operatorname{Inv}(a)) \sqcup \operatorname{Inv}(b).$$
(3.4.6)

Thus by Lemma 3.4.1(i), the multiplication map gives an isomorphism

$$\dot{b}^{-1}U(\operatorname{Inv}(a))\dot{b} \times U(\operatorname{Inv}(b)) \xrightarrow{\sim} U(\operatorname{Inv}(ab)).$$
 (3.4.7)

We will later need the following trivial consequences of (3.4.7): if $\ell(ab) = \ell(a) + \ell(b)$ then

$$\dot{b}^{-1} \cdot (U_{-} \cap \dot{a}^{-1}U\dot{a}) \subset (U_{-} \cap (\dot{a}\dot{b})^{-1}U\dot{a}\dot{b}) \cdot \dot{b}^{-1},$$
(3.4.8)

$$(U \cap \dot{a}^{-1}U_{-}\dot{a}) \cdot \dot{b} \subset \dot{b} \cdot (U \cap (\dot{a}\dot{b})^{-1}U_{-}\dot{a}\dot{b}).$$
(3.4.9)

Multiplying both sides of (3.4.9) by \dot{b}^{-1} on the left, we get $\dot{b}^{-1}U(\text{Inv}(a))\dot{b} \subset U(\text{Inv}(ab))$, which follows from (3.4.6). Eq. (3.4.8) follows from (3.4.9) by applying the map $x \mapsto x^T$, see (3.4.3).

Lemma 3.4.3. Let $\alpha \in \Phi^+$ and $i \in I$ be such that $\alpha \neq \alpha_i$. Let $\Psi \subset \Phi$ denote the set of all roots of the form $m\alpha - r\alpha_i$ for integers m > 0, $r \ge 0$. Then Ψ is a bracket closed subset of Φ^+ , and for all $t \in \mathbb{K}$ we have $y_i(t)U_{\alpha}y_i(-t) \subset U(\Psi)$.

Proof. Let $x \in U_{\alpha}$ and $x' := \dot{s}_i^{-1} x \dot{s}_i \in U_{s_i \alpha}$. Recall from [BB05, Lemma 4.4.3] that s_i permutes $\Phi^+ \setminus \{\alpha_i\}$ (in particular, $s_i \alpha > 0$). Write

$$y_i(t) \cdot x \cdot y_i(-t) = \dot{s}_i x_i(-t) \dot{s}_i^{-1} \cdot x \cdot \dot{s}_i x_i(t) \dot{s}_i^{-1} = \dot{s}_i x_i(-t) \cdot x' \cdot x_i(t) \dot{s}_i^{-1}.$$

Denote by $\Psi' \subset \Phi$ the set of all roots of the form $ms_i\alpha + r\alpha_i$ for integers m, r > 0. It is clear that $\Psi' \subset \Phi^+ \setminus \{\alpha_i, s_i\alpha\}$ is a bracket closed subset. By [Hum75, Lemma 32.5], we have $\begin{aligned} x_i(-t)x'x_i(t)x'^{-1} &\in U(\Psi'), \text{ so } x_i(-t)x'x_i(t) \in U(\Psi')x'. \text{ Thus } \Psi'' &:= s_i\Psi' \text{ is also a bracket} \\ \text{closed subset of } \Phi^+ \setminus \{\alpha_i, \alpha\}, \text{ and we have } \dot{s}_i U(\Psi')x'\dot{s}_i^{-1} &= U(\Psi'')x. \text{ Clearly, } \Psi = \Psi'' \sqcup \{\alpha\}. \end{aligned}$ We thus have $y_i(t)U_{\alpha}y_i(-t) \subset U(\Psi'')U_{\alpha} = U(\Psi). \Box$

3.4.5 Flag variety and Bruhat decomposition

Let G/B be the *flag variety* of G (over \mathbb{K}). We recall some standard properties of the Bruhat decomposition that can be found in e.g. [Hum75, §28]. Define open Schubert, opposite Schubert, and Richardson varieties:

$$\mathcal{X}^w = B\dot{w}B/B, \quad \mathcal{X}_v = B_-\dot{v}B/B, \quad \mathring{R}_{v,w} := \mathcal{X}_v \cap \mathcal{X}^w \quad \text{(for } v \le w \in W\text{)}. \tag{3.4.10}$$

Recall the Bruhat and Birkhoff decompositions:

$$G = \bigsqcup_{w \in W} B\dot{w}B = \bigsqcup_{v \in W} B_{-}\dot{v}B, \quad \text{where}$$
(3.4.11)

$$B_{-}\dot{v}B \cap B\dot{w}B = \emptyset$$
 and $\mathcal{X}_{v} \cap \mathcal{X}^{w} = \emptyset$ for $v \not\leq w \in W$. (3.4.12)

Let X_v denote the (Zariski) closure of \mathcal{X}_v . Similarly, let X^w denote the closure of \mathcal{X}^w , and then $R_{v,w} = X_v \cap X^w$ is the closure of $\mathring{R}_{v,w}$ in G/B. We have

$$X_{v} = \bigsqcup_{v \le v'} \mathcal{X}_{v'}, \qquad \qquad X^{w} = \bigsqcup_{w' \le w} \mathcal{X}^{w'}, \qquad (3.4.13)$$

For any $w \in W$, $i \in I$, and $t \in \mathbb{K}^*$, we have

$$x_i(t) \in B_-\dot{s}_i B_-, \quad y_i(t) \in B\dot{s}_i B,$$
 (3.4.15)

$$B\dot{s}_{i}B \cdot B\dot{w}B \subset \begin{cases} B\dot{s}_{i}\dot{w}B, & \text{if } s_{i}w > w, \\ B\dot{s}_{i}\dot{w}B \sqcup B\dot{w}B, & \text{if } s_{i}w < w, \end{cases}$$
(3.4.16)

$$B_{-}\dot{s}_{i}B_{-} \cdot B_{-}\dot{w}B \subset \begin{cases} B_{-}\dot{s}_{i}\dot{w}B, & \text{if } s_{i}w < w, \\ B_{-}\dot{s}_{i}\dot{w}B \sqcup B_{-}\dot{w}B, & \text{if } s_{i}w > w, \end{cases}$$
(3.4.17)

$$B\dot{v}B \cdot B\dot{w}B \subset B\dot{v}\dot{w}B$$
 for $v \in W$ such that $\ell(vw) = \ell(v) + \ell(w)$. (3.4.18)

For $\mathbf{t} = (t_1, \ldots, t_n) \in (\mathbb{K}^*)^n$ and a reduced word $\mathbf{i} = (i_1, \ldots, i_n)$ for $w \in W$, define

$$\mathbf{x}_{\mathbf{i}}(\mathbf{t}) := x_{i_1}(t_1) \cdots x_{i_n}(t_n), \text{ and } \mathbf{y}_{\mathbf{i}}(\mathbf{t}) := y_{i_1}(t_1) \cdots y_{i_n}(t_n).$$
 (3.4.19)

It follows from (3.4.15), (3.4.16), and (3.4.3) that

$$\mathbf{x}_{\mathbf{i}}(\mathbf{t}) \in B_{-}\dot{w}B_{-}, \quad \mathbf{y}_{\mathbf{i}}(\mathbf{t}) \in B\dot{w}B.$$
 (3.4.20)

3.4.6 Parabolic subgroup W_J of W

Let $J \subset I$, and denote by $W_J \subset W$ the subgroup generated by $\{s_i\}_{i \in J}$. Let w_J be the longest element of W_J , and $w^J := w_0 w_J$ be the maximal element of W^J . Let $\Phi_J \subset \Phi$ consist of roots that are linear combinations of $\{\alpha_i\}_{i \in J}$. Define

$$\Phi_J^+ := \Phi_J \cap \Phi^+, \quad \Phi_J^- := \Phi_J \cap \Phi^-, \quad \Phi_+^{(J)} := \Phi^+ \setminus \Phi_J^+, \quad \Phi_-^{(J)} := \Phi^- \setminus \Phi_J^-.$$

The sets Φ_J^+ , $\Phi_+^{(J)}$, Φ_J^- , $\Phi_-^{(J)}$ are clearly bracket closed, so consider subgroups

$$U_J = U(\Phi_J^+), \quad U_J^- = U_-(\Phi_J^-), \quad U^{(J)} = U(\Phi_+^{(J)}), \quad U_-^{(J)} = U_-(\Phi_-^{(J)}).$$

In fact, we have

$$\Phi_J^+ = \text{Inv}(w_J), \quad \Phi_+^{(J)} = \text{Inv}(w^J), \quad \dot{w}_J U_J^- \dot{w}_J^{-1} = U_J.$$
(3.4.21)

Let $W_{\text{max}}^J := \{ww_J \mid w \in W^J\}$. By [BB05, Prop. 2.4.4], every $w \in W$ admits a unique parabolic factorization $w = w_1w_2$ for $w_1 \in W^J$ and $w_2 \in W_J$, and this factorization is length-additive. We state some standard facts on parabolic factorizations for later use.

Lemma 3.4.4.

- (i) If $u \in W^J$ and $s_i u < u$ then $s_i u \in W^J$.
- (ii) Given $u \in W^J$ and $r, r' \in W_J$, we have $ur \leq ur'$ if and only if $r \leq r'$.

Proof. Part (i) follows since if $s_i u \notin W^J$ then $s_i u s_j < s_i u$ for some $j \in J$, therefore $s_i u s_j < s_i u < u < u s_j$, which contradicts $\ell(u s_j) = \ell(s_i u s_j) \pm 1$. For (ii), see [BB05, Exercise 2.21].

Lemma 3.4.5. For any $w \in W^J$, we have $\operatorname{Inv}(w) \subset \Phi^{(J)}_+$. In particular, $w\Phi^+_J \subset \Phi^+$, $\dot{w}U_J\dot{w}^{-1} \subset U$, and $\dot{w}U_J^-\dot{w}^{-1} \subset U^-$.

Proof. Let $\alpha \in \Phi^+$ be a positive root. Then it can be written as $\alpha = \sum_{i \in I} c_i \alpha_i$ for $c_i \in \mathbb{Z}_{\geq 0}$. Since $w \in W^J$, we have $w\alpha_i > 0$ for all $i \in J$. Thus if $w\alpha < 0$, we must have $c_i \neq 0$ for some $i \notin J$, so $\alpha \in \Phi^{(J)}_+$.

Lemma 3.4.6 ([He09]). Let $x, y \in W$.

- (i) The set {uv | u ≤ x, v ≤ y} contains a unique maximal element, denoted x * y. The set {xv | v ≤ y} contains a unique minimal element, denoted x ⊲ y.
- (ii) There exist elements $u' \leq x$ and $v' \leq y$ such that x * y = xv' = u'y, and these factorizations are both length-additive.
- (iii) If $x' \leq x$, then $x' * y \leq x * y$ and $x' \triangleleft y \leq x \triangleleft y$.
- (iv) If xy is length-additive, then x * y = xy and $(xy) \triangleleft y^{-1} = x$.

The operation * is known as the *Demazure product*.

Proof. The first three parts were shown in [He09, $\S1.3$], with the caveat that our \triangleleft is the 'mirror image' of his \triangleright . Part (iv) follows from the definitions of \ast and \triangleleft .

Definition 3.4.7. Let $Q_J = \{(v, w) \in W \times W^J \mid v \leq w\}$. We define an order relation \preceq on Q_J as follows: for $(v, w), (v', w') \in Q_J$, we write $(v, w) \preceq (v', w')$ if and only if there exists $r \in W_J$ such that vr is length-additive and $v' \leq vr \leq wr \leq w'$. For $(v, w) \in Q_J$, denote

$$Q_J^{\succeq(v,w)} := \{ (v',w') \in Q_J \mid (v,w) \preceq (v',w') \}, \quad Q_J^{\preceq(v,w)} := \{ (v',w') \in Q_J \mid (v',w') \preceq (v,w) \}.$$

Lemma 3.4.8.

- (i) Suppose that $(v, w), (v', w') \in Q_J, r \in W_J, and v' \leq vr \leq wr \leq w'$. Then $(v, w) \preceq (v', w')$.
- (ii) Let $(u, u), (v, w), (v', w') \in Q_J$. Then $(u, u) \preceq (v, w) \preceq (v', w')$ if and only if

$$v' \le vr' \le ur \le wr' \le w'$$
 for some $r, r' \in W_J$ such that vr' is length-additive.
(3.4.22)

Proof. (i): By Lemma 3.4.6, there exists $r' \leq r$ such that $v * r = vr' \geq vr$, and vr' is length-additive. We have $vr' \leq wr'$ by Lemma 3.4.6(iii), and $wr' \leq wr$ by Lemma 3.4.4(ii). Therefore $v' \leq vr \leq vr' \leq wr' \leq wr \leq w'$, so $(v, w) \preceq (v', w')$.

(ii) (\Rightarrow): Suppose that $(u, u) \preceq (v, w) \preceq (v', w')$. Then by Definition 3.4.7, there exist $r', r'' \in W_J$ such that vr' is length-additive, $v' \leq vr' \leq wr' \leq w'$, and $v \leq ur'' \leq w$. Define $r \in W_J$ by the equality (ur'') * r' = ur. Then applying *r' on the right to $v \leq ur'' \leq w$, by Lemma 3.4.6(iii)–(iv), we obtain $vr' \leq ur \leq wr'$. Therefore (3.4.22) holds.

(ii) (\Leftarrow): Suppose that (3.4.22) holds. Then $(v, w) \preceq (v', w')$. Define $r'' \in W_J$ by the equality $(ur) \triangleleft r'^{-1} = ur''$. Then applying $\triangleleft (r')^{-1}$ on the right to $vr' \leq ur \leq wr'$, by Lemma 3.4.6(iii)–(iv), we obtain $v \leq ur'' \leq w$. Therefore $(u, u) \preceq (v, w)$.

Remark 3.4.9. By Lemma 3.4.8(i), Definition 3.4.7 remains unchanged if we omit the condition that vr is length-additive. It follows that Q_J coincides with the poset studied in [HL15, §2.4]. Therefore by [HL15, Appendix], Q_J is also isomorphic to the posets studied in [Rie06, GY09, KLS13].

3.4.7 The partial flag variety G/P

Fix $J \subset I$ as before. Let $P \subset G$ be the subgroup generated by B and $\{y_i(t) \mid t \in \mathbb{K}^*, i \in J\}$. We denote by G/P the partial flag variety corresponding to J, and let $\pi_J : G/B \to G/P$ be the natural projection map. Let $L_J \subset P$ be the Levi subgroup of P. It is generated by T and $\{x_i(t), y_i(t) \mid i \in J, t \in \mathbb{K}^*\}$. Let P_- be the parabolic subgroup opposite to P, with $L_J = P \cap P_-$.

For $(v, w) \in Q_J$ we introduce $\mathring{\Pi}_{v,w} := \pi_J(\mathring{R}_{v,w}) \subset G/P$, and define the projected Richardson variety $\Pi_{v,w} \subset G/P$ to be the closure of $\mathring{\Pi}_{v,w}$ in the Zariski topology. By [KLS14, Prop. 3.6], we have

$$G/P = \bigsqcup_{(v,w)\in Q_J} \mathring{\Pi}_{v,w}, \quad \text{and} \quad \Pi_{v,w} = \bigsqcup_{(v',w')\in Q_J^{\preceq(v,w)}} \mathring{\Pi}_{v',w'}. \tag{3.4.23}$$

Let now $\mathbb{K} = \mathbb{C}$. The varieties \mathcal{X}^w , \mathcal{X}_v , X_w , X^w , $\mathring{R}_{v,w}$, and $R_{v,w}$ are defined over \mathbb{R} . The map π_J is defined over \mathbb{R} as well, thus so are $\mathring{\Pi}_{v,w}$ and $\Pi_{v,w}$. We let $(G/B)_{\mathbb{R}} := \{gB \mid g \in G(\mathbb{R})\} \subset G/B$, $\mathring{R}^{\mathbb{R}}_{v,w} := (G/B)_{\mathbb{R}} \cap \mathring{R}_{v,w}$, and $R^{\mathbb{R}}_{v,w} := (G/B)_{\mathbb{R}} \cap R_{v,w}$. Additionally, set $(G/P)_{\mathbb{R}} := \{xP \mid x \in G(\mathbb{R})\} \subset G/P$, $\mathring{\Pi}^{\mathbb{R}}_{v,w} := \mathring{\Pi}_{v,w} \cap (G/P)_{\mathbb{R}}$, and $\Pi^{\mathbb{R}}_{v,w} := \Pi_{v,w} \cap (G/P)_{\mathbb{R}}$. It follows that the decomposition (3.4.23) is valid over \mathbb{R} :

$$(G/P)_{\mathbb{R}} = \bigsqcup_{(v,w)\in Q_J} \mathring{\Pi}^{\mathbb{R}}_{v,w}, \qquad \qquad \Pi^{\mathbb{R}}_{v,w} = \bigsqcup_{(v',w')\in Q_J^{\leq (v,w)}} \mathring{\Pi}^{\mathbb{R}}_{v',w'}. \tag{3.4.24}$$

3.4.8 Total positivity

Assume $\mathbb{K} = \mathbb{C}$ in this subsection. Recall that for each $i \in I$, we have one-parameter subgroups $x_i(t), y_i(t), \alpha_i^{\vee}(t)$ (for $t \in \mathbb{C}$), see (3.4.1).

Definition 3.4.10 ([Lus94]). Let $G_{\geq 0} \subset G(\mathbb{R})$ be the submonoid generated by $x_i(t), y_i(t), \alpha^{\vee}(t)$ for $t \in \mathbb{R}_{>0}$. Define $(G/B)_{\geq 0}$ to be the closure of $(G_{\geq 0}/B) \subset (G/B)_{\mathbb{R}}$ in the analytic topology. For $v \leq w \in W$, let $R_{v,w}^{\geq 0}$ denote the closure of $R_{v,w}^{>0} := \mathring{R}_{v,w} \cap (G/B)_{\geq 0}$ inside $(G/B)_{\geq 0}$.

Definition 3.4.11 ([Lus98a, Rie99]). Set $(G/P)_{\geq 0} := \pi_J((G/B)_{\geq 0})$. For $(v, w) \in Q_J$, let $\Pi_{v,w}^{\geq 0}$ denote the closure of $\Pi_{v,w}^{>0} := \pi_J(R_{v,w}^{>0})$ inside $(G/P)_{\geq 0}$.

Thus we denote by $\Pi_{v,w}^{>0}$ what was denoted by $\Pi_{(v,w)}^{>0}$ in Example 3.1.1. We have decompositions

$$(G/P)_{\geq 0} = \bigsqcup_{(v,w)\in Q_J} \Pi_{v,w}^{>0}, \qquad \qquad \Pi_{v,w}^{\geq 0} = \bigsqcup_{(v',w')\in Q_J^{\leq (v,w)}} \Pi_{v',w'}^{>0}. \tag{3.4.25}$$

As a special case of (3.4.25) for $J = \emptyset$, we have

$$(G/B)_{\geq 0} = \bigsqcup_{v \leq w} R_{v,w}^{>0}, \qquad \qquad R_{v,w}^{\geq 0} = \bigsqcup_{v \leq v' \leq w' \leq w} R_{v',w'}^{>0}. \tag{3.4.26}$$

Lemma 3.4.12. (Assume $\mathbb{K} = \mathbb{C}$.) Let $(v, w) \in Q_J$ and $r \in W_J$ be such that vr is lengthadditive. Then

$$\overset{\circ}{\Pi}_{v,w} = \pi_J(\overset{\circ}{R}_{v,w}) = \pi_J(\overset{\circ}{R}_{vr,wr}), \qquad \Pi_{v,w}^{>0} = \pi_J(R_{v,w}^{>0}) = \pi_J(R_{vr,wr}^{>0}), \qquad (3.4.27)$$

$$\Pi_{v,w} = \pi_J(R_{v,w}) = \pi_J(R_{vr,wr}), \qquad \Pi_{v,w}^{\ge 0} = \pi_J(R_{v,w}^{\ge 0}) = \pi_J(R_{vr,wr}^{\ge 0}). \tag{3.4.28}$$

Proof. By [KLS13, Lemma 3.1], we have $\pi_J(\mathring{R}_{v,w}) = \pi_J(\mathring{R}_{vr,wr}) = \mathring{\Pi}_{v,w}$, and π_J restricts to isomorphisms $\mathring{R}_{v,w} \xrightarrow{\sim} \mathring{\Pi}_{v,w}$, $\mathring{R}_{vr,wr} \xrightarrow{\sim} \mathring{\Pi}_{v,w}$. Thus $\pi_J(R_{v,w}^{>0}) = \pi_J(R_{vr,wr}^{>0}) = \Pi_{v,w}^{>0}$ follows from the equality $\pi_J((G/B)_{\geq 0}) = (G/P)_{\geq 0}$, proving (3.4.27). To show (3.4.28), note that $R_{a,b}$ and $R_{a,b}^{\geq 0}$ are compact for any $a \leq b$, and therefore their images under π_J are closed. \Box

Recall the definition of $\mathbf{x}_i(\mathbf{t})$ and $\mathbf{y}_i(\mathbf{t})$ from (3.4.19). Choose a reduced word $\mathbf{i} = (i_1, \ldots, i_n)$ for $w \in W$ and define

$$U_{>0}(w) := \{ \mathbf{x}_{\mathbf{i}}(\mathbf{t}) \mid \mathbf{t} \in \mathbb{R}_{>0}^n \}, \quad U_{>0}^-(w) := \{ \mathbf{y}_{\mathbf{i}}(\mathbf{t}) \mid \mathbf{t} \in \mathbb{R}_{>0}^n \}.$$

Let $U_{\geq 0} \subset U(\mathbb{R})$ (resp., $U_{\geq 0}^- \subset U_-(\mathbb{R})$) be the submonoid generated by $x_i(t)$ (resp., by $y_i(t)$) for $t \in \mathbb{R}_{>0}$. Then $U_{\geq 0} = \bigsqcup_{w \in W} U_{>0}(w)$ and $U_{\geq 0}^- = \bigsqcup_{w \in W} U_{>0}^-(w)$. We have $U_{>0}(w) = U_{\geq 0} \cap B_- \dot{w}B_-$ and $U_{>0}^-(w) = U_{\geq 0}^- \cap B\dot{w}B$, and these sets do not depend on the choice of the reduced word **i** for w, see [Lus94, Prop. 2.7].

3.4.9 MR-parametrizations

Assume that \mathbb{K} is algebraically closed. Given $w \in W$, an expression \mathbf{w} for w is a sequence $\mathbf{w} = (w_{(0)}, \ldots, w_{(n)})$ such that $w_{(0)} = \mathrm{id}$, $w_{(n)} = w$, and for $j = 1, \ldots, n$, either $w_{(j)} = w_{(j-1)}$ or $w_{(j)} = w_{(j-1)}s_{i_j}$ for some $i_j \in I$. In the latter case we require $w_{(j-1)} < w_{(j)}$, unlike [MR04]. We denote $J^+_{\mathbf{w}} := \{1 \leq j \leq n \mid w_{(j-1)} < w_{(j)}\}$ and $J^\circ_{\mathbf{w}} := \{1 \leq j \leq n \mid w_{(j-1)} = w_{(j)}\}$ so that $J^+_{\mathbf{w}} \sqcup J^\circ_{\mathbf{w}} = \{1, 2, \ldots, n\}$. Every reduced word $\mathbf{i} = (i_1, \ldots, i_n)$ for w gives rise to a reduced expression $\mathbf{w} = \mathbf{w}(\mathbf{i}) = (w_{(0)}, \ldots, w_{(n)})$ with $w_{(j)} = w_{(j-1)}s_{i_j}$ for $j = 1, \ldots, n$.

Lemma 3.4.13 ([MR04, Lemma 3.5]). Let $v \le w \in W$, and consider a reduced expression $\mathbf{w} = (w_{(0)}, \ldots, w_{(n)})$ for w corresponding to a reduced word $\mathbf{i} = (i_1, \ldots, i_n)$. Then there exists a unique positive subexpression \mathbf{v} for v inside \mathbf{w} , *i.e.*, an expression $\mathbf{v} = (v_{(0)}, \ldots, v_{(n)})$ for v such that for $j = 1, \ldots, n$, we have $v_{(j-1)} < v_{(j-1)}s_{i_j}$. This positive subexpression can be constructed inductively by setting $v_{(n)} := v$ and

$$v_{(j-1)} := \begin{cases} v_{(j)}s_{i_j}, & \text{if } v_{(j)}s_{i_j} < v_{(j)}, \\ v_{(j)}, & \text{otherwise,} \end{cases} & \text{for } j = n, \dots, 1. \end{cases}$$
(3.4.29)

Corollary 3.4.14. In the above setting, if $v_{(1)} = s_i$ for some $i \in I$ then $v \not\leq s_i w$.

Proof. Indeed, if $v \leq s_i w < w$ then there exists a positive subexpression $\mathbf{v}' = (v'_{(0)}, \ldots, v'_{(n-1)})$ for v inside $\mathbf{w}(\mathbf{i}')$, where $\mathbf{i}' = (i_2, \ldots, i_n)$. By (3.4.29), we have $v'_{(j)} = v_{(j+1)}$ for $j = 0, 1, \ldots, n-1$, which contradicts the fact that $v'_{(0)} = 1$ while $v_{(1)} = s_i$.

For $w \in W$, let $\operatorname{Red}(w) := \{ \mathbf{w} \mid \mathbf{w} \text{ is a reduced expression for } w \}$. For $v \leq w \in W$, let

 $\operatorname{Red}(v, w) := \{ (\mathbf{v}, \mathbf{w}) \mid \mathbf{w} \in \operatorname{Red}(w), \ \mathbf{v} \text{ is a positive subexpression for } v \text{ inside } \mathbf{w} \}.$

Thus for all $v \leq w$, the sets $\operatorname{Red}(w)$ and $\operatorname{Red}(v, w)$ have the same cardinality. Let $v \leq w \in W$ and $(\mathbf{v}, \mathbf{w}) \in \operatorname{Red}(v, w)$. Given a collection $\mathbf{t} = (t_k)_{k \in J^\circ_{\mathbf{v}}} \in (\mathbb{K}^*)^{J^\circ_{\mathbf{v}}}$, define

$$\mathbf{g}_{\mathbf{v},\mathbf{w}}(\mathbf{t}) := g_1 \cdots g_n, \quad \text{where} \quad g_k := \begin{cases} y_{i_k}(t_k), & \text{if } k \in J_{\mathbf{v}}^{\circ}, \\ \dot{s}_{i_k}, & \text{if } k \in J_{\mathbf{v}}^{+}. \end{cases}$$
(3.4.30)

MR-parametrizations of $(G/B)_{\geq 0}$

In this section, we assume $\mathbb{K} = \mathbb{C}$. Let v, w, \mathbf{v} , and \mathbf{w} be as above. Define a subset $G_{\mathbf{v},\mathbf{w}}^{>0} \subset G(\mathbb{R})$ by

$$G_{\mathbf{v},\mathbf{w}}^{>0} := \{ \mathbf{g}_{\mathbf{v},\mathbf{w}}(\mathbf{t}) \mid \mathbf{t} \in \mathbb{R}_{>0}^{J_{\mathbf{v}}^{\circ}} \}.$$

Theorem 3.4.15 ([MR04, Theorem 11.3]). The map $G(\mathbb{R}) \to (G/B)_{\mathbb{R}}$ sending $g \mapsto gB$ restricts to an isomorphism of real semialgebraic varieties

$$G^{>0}_{\mathbf{v},\mathbf{w}} \xrightarrow{\sim} R^{>0}_{v,w}$$

Proposition 3.4.16 ([Lus94, Prop. 8.12]). We have $G_{\geq 0} \cdot (G/B)_{\geq 0} \subset (G/B)_{\geq 0}$.

Lemma 3.4.17. Suppose that $g \in G_{\geq 0}$ and $x \in G$ are such that $xB \in R^{>0}_{v,w}$ for some $v \leq w \in W$. Then $gxB \in R^{>0}_{v',w'}$ for some $v' \leq v \leq w \leq w'$.

Proof. By Proposition 3.4.16, we have $gxB \in (G/B)_{\geq 0}$, so it suffices to show that $gx \in B\dot{w}'B \cap B_-\dot{v}'B$ for some $v' \leq v \leq w \leq w'$. Note that we have $x \in B\dot{w}B \cap B_-\dot{v}B$. By Definition 3.4.10, it is enough to consider the cases $g = x_i(t)$ and $g = y_i(t)$ for $i \in I$ and $t \in \mathbb{R}_{>0}$.

Suppose that $g = y_i(t)$. We clearly have $gx \in B_-\dot{v}B$. If $s_iw > w$ then by (3.4.16) we have $gx \in B\dot{s}_i\dot{w}B$. Thus we may assume that $s_iw < w$. By Theorem 3.4.15, we can also assume $x = \mathbf{g}_{\mathbf{v},\mathbf{w}}(\mathbf{t}) = g_1 \cdots g_n$ for $\mathbf{t} \in \mathbb{R}_{>0}^{J_{\mathbf{v}}^\circ}$ and some choice of $(\mathbf{v},\mathbf{w}) \in \operatorname{Red}(v,w)$ such that $\mathbf{w} = (w_{(0)}, \ldots, w_{(n)})$ satisfies $w_{(1)} = s_i$. Let $\mathbf{v} = (v_{(0)}, \ldots, v_{(n)})$. If $v_{(1)} \neq s_i$ then $g_1 = y_i(t')$ so $gx \in G_{\mathbf{v},\mathbf{w}}^{>0}$ and we are done. If $v_{(1)} = s_i$ then by Corollary 3.4.14 we have $v \not\leq s_iw$. Recall that $gx \in B_-\dot{v}B$ and by (3.4.16), $gx \in B\dot{s}_i\dot{w}B \sqcup B\dot{w}B$. But $B_-\dot{v}B \cap B\dot{s}_i\dot{w}B = \emptyset$ by (3.4.12). Therefore we must have $gx \in B\dot{w}B$, finishing the proof in this case.

The case $g = x_i(t)$ follows analogously using a "dual" Marsh–Rietsch parametrization [Rie06, Section 3.4], where for $(\mathbf{v}, \mathbf{w}) \in \text{Red}(v, w)$, every element of $R_{ww_0, vw_0}^{>0}$ is parametrized as

$$g_1 \cdots g_n \dot{w}_0 B, \quad \text{where} \quad g_k := \begin{cases} x_{i_k}(t_k), & \text{if } k \in J^\circ_{\mathbf{v}}, \\ \dot{s}^{-1}_{i_k}, & \text{if } k \in J^+_{\mathbf{v}}. \end{cases}$$

We will use the following consequence of Theorem 3.4.15 in Section 3.9.11

Corollary 3.4.18 (cf. [KLS14, Prop. 3.3]). Let $u \in W^J$, $r \in W_J$, and $v \in W$ be such that $v \leq ur$. Then

$$\pi_J(R_{v,ur}^{>0}) = \pi_J(R_{v \triangleleft r^{-1}, u}^{>0}) = \Pi_{v \triangleleft r^{-1}, u}^{>0}.$$

Proof. Let $\mathbf{i} = (i_1, \ldots, i_n)$ be a reduced word for w := ur, such that $(i_{\ell(u)+1}, \ldots, i_n)$ is a reduced word for r. Let $(\mathbf{v}, \mathbf{w}) \in \operatorname{Red}(v, w)$ be such that \mathbf{w} corresponds to \mathbf{i} . Then it is clear from Lemma 3.4.13 that after setting $\mathbf{v}' := (v_{(0)}, \ldots, v_{(\ell(u))})$ and $\mathbf{u} := (w_{(0)}, \ldots, w_{(\ell(u))})$, we get $(\mathbf{v}', \mathbf{u}) \in \operatorname{Red}(v \triangleleft r^{-1}, u)$. Moreover, the indices $i_{\ell(u)+1}, \ldots, i_n$ clearly belong to J, so if $g_1 \ldots g_n \in G_{\mathbf{v},\mathbf{w}}^{>0}$ then $g_1 \ldots g_{\ell(u)} \in G_{\mathbf{v}',\mathbf{u}}^{>0}$ and $\pi_J(g_1 \ldots g_n B) = \pi_J(g_1 \ldots g_{\ell(u)} B)$. We are done by Theorem 3.4.15.

3.4.10 G/P is a TNN space

We show that the triple $((G/P)_{\mathbb{R}}, (G/P)_{\geq 0}, Q_J)$ is a TNN space in the sense of Definition 3.2.1. We start by recalling several well known facts.

Proposition 3.4.19.

- (i) The poset $\widehat{Q}_J := Q_J \sqcup \{\widehat{0}\}$ is graded, thin, and shellable.
- (ii) $(G/P)_{\mathbb{R}}$ is a smooth manifold, and each $\mathring{\Pi}_{v,w}^{\mathbb{R}}$ is a smooth embedded submanifold of $(G/P)_{\mathbb{R}}$.
- (iii) For $(v, w) \in Q_J$, $\Pi_{v,w}^{>0}$ is a connected component of $\mathring{\Pi}_{v,w}^{\mathbb{R}}$.

Proof. For (i), see [Wil07]. For (ii), $(G/P)_{\mathbb{R}}$ is a smooth manifold because it is a homogeneous space of a real Lie group, and $\mathring{\Pi}_{v,w}^{\mathbb{R}}$ is a smooth embedded manifold because it is the set of real points of a smooth algebraic subvariety $\mathring{\Pi}_{v,w}$ of G/P, see [KLS14, Cor. 3.2] or [Lus98a, Rie06]. Part (iii) is due to [Rie99].

Corollary 3.4.20. $((G/P)_{\mathbb{R}}, (G/P)_{\geq 0}, Q_J)$ is a TNN space.

Proof. Let us check each part of Definition 3.2.1.

(TNN1): Follows from Proposition 3.4.19(i). The maximal element $\hat{1} \in Q_J$ is given by (id, w^J), see Section 3.4.6.

(TNN2): Follows from Proposition 3.4.19(ii) and (3.4.24).

- (TNN3): Clear since $(G/P)_{\mathbb{R}}$ is compact and $\Pi_{v,w}^{\geq 0} \subset G/P$ is closed.
- (TNN4): Follows from Proposition 3.4.19(iii) combined with Theorem 3.4.15.

(TNN5): This result is due to [Rie06], see (3.4.25).

3.4.11 Gaussian decomposition

Assume \mathbb{K} is algebraically closed. Let us define

$$G_0^{\mp} := B_- B, \quad G_0^{\pm} := B B_-.$$

For $i \in I$, let $\Delta_i^{\mp} : G_0^{\mp} \to \mathbb{K}$ and $\Delta_i^{\pm} : G_0^{\pm} \to \mathbb{K}$ be defined as follows. Given $(x_-, x_0, x_+) \in U_- \times T \times U$, we have $x_- x_0 x_+ \in G_0^{\mp}$ and $x_+ x_0 x_- \in G_0^{\pm}$, and we set $\Delta_i^{\mp}(x_- x_0 x_+) := x_0^{\omega_i}$, $\Delta_i^{\pm}(x_+ x_0 x_-) := x_0^{w_0 \omega_i}$. For a finite set A, let $\mathbb{P}_{n-1}^{(2n)-1}$ denote the (|A|-1)-dimensional projective space over \mathbb{K} , with coordinates indexed by elements of A.

Lemma 3.4.21.

(i) The multiplication map gives biregular isomorphisms:

$$U_- \times T \times U \xrightarrow{\sim} G_0^{\mp}, \quad U \times T \times U_- \xrightarrow{\sim} G_0^{\pm}.$$

- (ii) The maps Δ_i^{\mp} and Δ_i^{\pm} extend to regular functions $G \to \mathbb{K}$.
- (iii) We have $G_0^{\mp} = \{x \in G \mid \Delta_i^{\mp}(x) \neq 0 \ \forall i \in I\}$ and $G_0^{\pm} = \{x \in G \mid \Delta_i^{\pm}(x) \neq 0 \ \forall i \in I\}.$
- (iv) Fix $i \in I$ and let $W\omega_i := \{w\omega_i \mid w \in W\}$ denote the W-orbit of the corresponding fundamental weight. Then there exists a regular map $\Delta_i^{\text{flag}} : G/B \to \mathbb{P}^{\binom{2n}{n-1}-1}W\omega_i}$ such that for $w \in W$ and $x \in G$, the $w\omega_i$ -th coordinate of $\Delta_i^{\text{flag}}(xB)$ equals $\Delta_i^{\mp}(\dot{w}^{-1}x)$.

Proof. For (i), see [Hum75, Prop. 28.5]. Parts (ii) and (iii) are well known when $\mathbb{K} = \mathbb{C}$, see [FZ99, Prop. 2.4 and Cor. 2.5]. We give a proof for arbitrary algebraically closed \mathbb{K} , using a standard argument that relies on representation theory. We refer to [Hum75, §31] for the necessary notation and background.

We have $G_0^{\pm} = \dot{w}_0^{-1} G_0^{\mp} \dot{w}_0$ and $\Delta_i^{\pm} (\dot{w}_0^{-1} g \dot{w}_0) = \Delta_i^{\mp} (g)$ for all $g \in G_0^{\mp}$. Thus it suffices to give a proof for Δ_i^{\mp} and G_0^{\mp} . For $i \in I$, there exists a regular function $c_{\omega_i} : G \to \mathbb{K}$ that

coincides with Δ_i^{\mp} on G_0^{\mp} , see [Hum75, §31.4]. This shows (ii). Explicitly, c_{ω_i} is given as follows: consider the highest weight module V_{ω_i} for G, and let $v_+ \in V_{\omega_i}$ be its highest weight vector. We have a direct sum of vector spaces $V_{\omega_i} = \mathbb{K}v_+ \oplus V'$, where V' is spanned by weight vectors of weight other than ω_i . Let $r^+ : V_{\omega_i} \to \mathbb{K}$ denote the linear function such that $r^+(v_+) = 1$ and $r^+(V') = \{0\}$, then $c_{\omega_i}(g) := r^+(gv_+)$ for all $g \in G$. The decomposition $V_{\omega_i} = \mathbb{K}v_+ \oplus V'$ is such that for $(x_-, x_0, x_+) \in U_- \times T \times U$ and $w \in W$, we have $x_+v_+ = v_+$, $x_0v_+ = Mv_+$ for some $M \in \mathbb{K}^*$, $x_-v_+ \in v_+ + V'$, $x_-V' \subset V'$, and $wv_+ \in V'$ if $w\omega_i \neq \omega_i$. Thus if $g \in G_0^{\mp}$ then $c_{\omega_i}(g) \neq 0$ for all $i \in I$. Conversely, if $x \notin G_0^{\pm}$ then by (3.4.11), there exists a unique $w \neq id \in W$ such that $g \in U_-\dot{w}TU$. For $i \in I$ such that $w\omega_i \neq \omega_i$, we get $c_{\omega_i}(g) = 0$. This proves (iii). For (iv), let $V_{\omega_i} = V_1 \oplus V_2$ where V_1 is spanned by all weight vectors of weight in $W\omega_i$, and V_2 is spanned by the remaining weight vectors. Let $\pi_1 : V_{\omega_i} \to V_1$ denote the projection along V_2 . It follows that for all $g \in G$, $\pi_1(gv_+) \neq 0$. Then Δ_i^{flag} is the natural morphism $G/B \to \mathbb{P}^{\binom{2n}{n-1}-1}(V_1)$, sending gB to $[\pi_1(gv_+)]$.

Lemma 3.4.22. Define $G_0^{(J)} := P_- P_-$.

- (i) We have $G_0^{(J)} = P_- B$ and $P = \bigsqcup_{r \in W_J} B\dot{r} B$.
- (ii) For $p \in P$, we have $pU^{(J)}p^{-1} = U^{(J)}$. Similarly, for $p \in P_{-}$, we have $pU^{(J)}_{-}p^{-1} = U^{(J)}_{-}$. In particular, for $p \in L_J$, we have $pU^{(J)}p^{-1} = U^{(J)}_{-}$ and $pU^{(J)}_{-}p^{-1} = U^{(J)}_{-}$.
- (iii) The multiplication map gives a biregular isomorphism $U_{-}^{(J)} \times L_J \times U^{(J)} \xrightarrow{\sim} G_0^{(J)}$. In particular, every element $x \in G_0^{(J)}$ can be uniquely factorized as $[x]_{-}^{(J)} \cdot [x]_J \cdot [x]_{+}^{(J)} \in U_{-}^{(J)} \cdot L_J \cdot U^{(J)}$. The map $G_0^{(J)} \to L_J$ sending $x \mapsto [x]_J$ satisfies $[p_-xp_+]_J = [p_-]_J[x]_J[p_+]_J$ for all $x \in G_0^{(J)}$, $p_- \in P_-$, and $p_+ \in P$.
- (iv) The map $b \to [b]_J$ gives group homomorphisms $U \to U_J$ and $U_- \to U_J^-$, sending

$$x_i(t) \mapsto [x_i(t)]_J = \begin{cases} x_i(t), & \text{if } i \in J, \\ 1, & \text{otherwise,} \end{cases} \quad y_i(t) \mapsto [y_i(t)]_J = \begin{cases} y_i(t), & \text{if } i \in J, \\ 1, & \text{otherwise,} \end{cases}$$

Proof. By [Hum75, §30.2], $U^{(J)}$ is the *unipotent radical* (in particular, a normal subgroup) of P and $U_{-}^{(J)}$ is the unipotent radical of P_{-} . This shows (ii). It follows that $P = L_J U^{(J)} = L_J B$, therefore $G_0^{(J)} = P_- B$. By [Hum75, §30.1] and (3.4.11), $P = \bigsqcup_{r \in W_J} B\dot{r}B$, which proves (i).

By [Bor91, Prop. 14.21(iii)], the multiplication map gives a biregular isomorphism $U_{-}^{(J)} \times$

 $P \to G_0^{(J)}$. By [Hum75, §30.2], the multiplication map gives a biregular isomorphism $L_J \times U^{(J)} \to P$. Thus we get a biregular isomorphism $U_-^{(J)} \times L_J \times U^{(J)} \xrightarrow{\sim} G_0^{(J)}$. It is clear from the definition that $[p_-xp_+]_J = [p_-]_J[x]_J[p_+]_J$, since we can factorize $p_- = [p_-]_-^{(J)}[p_-]_J$ and $p_+ = [p_+]_J[p_+]_+^{(J)}$. Thus we are done with (iii), and (iv) follows by repeatedly applying (iii).

3.4.12 Affine charts

For $u \in W^J$, denote $C_u^{(J)} := \dot{u}G_0^{(J)}/P \subset G/P$. The following maps are biregular isomorphisms for $u \in W^J$ and $v, w \in W$, see [Bor91, Prop. 14.21(iii)], [Spr98, Prop. 8.5.1(ii)], and [FH91, Cor. 23.60]:

$$\dot{u}U_{-}^{(J)}\dot{u}^{-1} \xrightarrow{\sim} C_{u}^{(J)}, \qquad \qquad g^{(J)} \mapsto g^{(J)}\dot{u}P, \qquad (3.4.31)$$

$$\dot{v}U_{-}\dot{v}^{-1}\cap U_{-} \xrightarrow{\sim} \mathcal{X}_{v}, \qquad g \mapsto g\dot{v}B, \qquad (3.4.32)$$

$$\dot{w}U_-\dot{w}^{-1}\cap U \xrightarrow{\sim} \mathcal{X}^w, \qquad g \mapsto g\dot{w}B.$$
 (3.4.33)

As a consequence of (3.4.32) and (3.4.33), we get

$$B_{-}\dot{v}B = (\dot{v}U_{-} \cap U_{-}\dot{v}) \cdot B, \quad B\dot{w}B = (\dot{w}U_{-} \cap U\dot{w}) \cdot B.$$
 (3.4.34)

The isomorphism in (3.4.31) identifies an open dense subset $C_u^{(J)}$ of G/P with the group $\dot{u}U_{-}^{(J)}\dot{u}^{-1}$. We now combine this with Lemma 3.4.2.

Definition 3.4.23. Let $U_1^{(J)} := \dot{u}U_-^{(J)}\dot{u}^{-1} \cap U$ and $U_2^{(J)} := \dot{u}U_-^{(J)}\dot{u}^{-1} \cap U_-$. For $x \in \dot{u}G_0^{(J)}$, consider the element $g^{(J)} \in \dot{u}U_-^{(J)}\dot{u}^{-1}$ such that $g^{(J)}\dot{u} \in xP \cap \dot{u}U_-^{(J)}$, unique by (3.4.31). Further, let $h_1^{(J)}, g_1^{(J)} \in U_1^{(J)}$ and $h_2^{(J)}, g_2^{(J)} \in U_2^{(J)}$ be the elements such that $h_2^{(J)}g^{(J)} = g_1^{(J)}$ and $h_1^{(J)}g^{(J)} = g_2^{(J)}$. By (3.4.31), the map $x \mapsto g^{(J)}$ is regular, and the map $g^{(J)} \to (g_1^{(J)}, g_2^{(J)}, h_1^{(J)}, h_2^{(J)})$ is regular by Lemma 3.4.2. Let us denote by $\kappa : \dot{u}G_0^{(J)} \to U_2^{(J)}$ the map $x \mapsto \kappa_x := h_2^{(J)}$. It descends to a regular map $\kappa : C_u^{(J)} \to U_2^{(J)}$ sending $xP \mapsto \kappa_x$.

3.5 Subtraction-free parametrizations

We study subtraction-free analogs of Marsh–Rietsch parametrizations of $(G/B)_{\geq 0}$.

3.5.1 Subtraction-free subsets

Given some fixed collection \mathbf{t} of variables of size $|\mathbf{t}|$, let $\mathbb{R}[\mathbf{t}]$ be the ring of polynomials in \mathbf{t} , and $\mathbb{R}_{>0}[\mathbf{t}] \subset \mathbb{R}[\mathbf{t}]$ be the semiring of nonzero polynomials in \mathbf{t} with positive real coefficients. Let $\mathcal{F} := \mathbb{R}(\mathbf{t})$ be the field of rational functions in \mathbf{t} . Define

$$\mathcal{F}_{\mathrm{sf}}^* := \{ R(\mathbf{t})/Q(\mathbf{t}) \mid R(\mathbf{t}), Q(\mathbf{t}) \in \mathbb{R}_{>0}[\mathbf{t}] \}, \quad \mathcal{F}_{\mathrm{sf}} := \{ 0 \} \sqcup \mathcal{F}_{\mathrm{sf}}^*,$$
$$\mathcal{F}^{\diamond} := \{ R(\mathbf{t})/Q(\mathbf{t}) \mid R(\mathbf{t}) \in \mathbb{R}[\mathbf{t}], \ Q(\mathbf{t}) \in \mathbb{R}_{>0}[\mathbf{t}] \}.$$

We call elements of \mathcal{F}_{sf} subtraction-free rational expressions in **t**. In this section, we assume that $\mathbb{K} = \overline{\mathcal{F}}$ is the algebraic closure of \mathcal{F} .

Definition 3.5.1. Let $T^{\text{sf}} \subset T$ be the submonoid generated by $\alpha_i^{\vee}(t)$ for $i \in I$ and $t \in \mathcal{F}_{\text{sf}}^*$. Let $G^{\diamond} \subset G$ be the subgroup generated by

$$\{x_i(t), y_i(t) \mid i \in I, t \in \mathcal{F}^\diamond\} \cup \{\dot{w} \mid w \in W\} \cup T^{\mathrm{sf}}.$$

We define subgroups $U^{\diamond} := U \cap G^{\diamond}$, $U_{-}^{\diamond} := U_{-} \cap G^{\diamond}$, $B^{\mathrm{sf}} := T^{\mathrm{sf}}U^{\diamond} = U^{\diamond}T^{\mathrm{sf}}$ and $B_{-}^{\mathrm{sf}} = T^{\mathrm{sf}}U_{-}^{\diamond} = U_{-}^{\diamond}T^{\mathrm{sf}}$ (cf. Lemma 3.5.2 below). We also put $U^{\diamond}(\Theta) := U^{\diamond} \cap U(\Theta)$ (resp., $U_{-}^{\diamond}(\Theta) := U_{-}^{\diamond} \cap U_{-}(\Theta)$) for a bracket closed subset Θ of Φ^{+} (resp., of Φ^{-}). Given a reduced word **i** for $w \in W$, define

$$U_{\rm sf}(w) := \{ \mathbf{x}_{\mathbf{i}}(\mathbf{t}') \mid \mathbf{t}' \in (\mathcal{F}_{\rm sf}^*)^n \}, \quad U_{\rm sf}^-(w) := \{ \mathbf{y}_{\mathbf{i}}(\mathbf{t}') \mid \mathbf{t}' \in (\mathcal{F}_{\rm sf}^*)^n \}.$$
(3.5.1)

These subsets do not depend on the choice of i, see [BZ97].

For two subsets H_1, H_2 of G, we say that H_1 commutes with H_2 if $H_1 \cdot H_2 = H_2 \cdot H_1$. We say that H_1 commutes with $g \in G$ if $H_1 \cdot g = g \cdot H_1$.

Lemma 3.5.2. T^{sf} commutes with B^{sf} , U, U_{-} , $U^{\diamond}(\Theta)$, $U_{-}^{\diamond}(\Theta)$, $U_{sf}(w)$, $U_{sf}^{-}(w)$ and \dot{w} .

Proof. It follows from (3.4.2) that T^{sf} commutes with B^{sf} , U, U_{-} , $U_{\text{sf}}(w)$, $U_{\text{sf}}^{-}(w)$ and \dot{w} . For $U^{\diamond}(\Theta)$, $U_{-}^{\diamond}(\Theta)$, we use a generalization of (3.4.2): for $\alpha \in \Phi^{+}$, $i \in I$, and $w \in W$ such that $w\alpha_{i} = \alpha$, write $x_{\alpha}(t) := \dot{w}x_{i}(t)\dot{w}^{-1} \in U^{\diamond}(\{\alpha\})$ and $y_{\alpha}(t) := \dot{w}y_{i}(t)\dot{w}^{-1} \in U_{-}^{\diamond}(\{-\alpha\})$ for $t \in \mathcal{F}^{\diamond}$. Then (3.4.2) implies $ax_{\alpha}(t)a^{-1} = x_{\alpha}(a^{\alpha}t)$ and $ay_{\alpha}(t)a^{-1} = y_{\alpha}(a^{-\alpha}t)$. Let us now introduce subtraction-free analogs of MR parametrizations. Let $v \leq w \in W$ and $(\mathbf{v}, \mathbf{w}) \in \operatorname{Red}(v, w)$. Recall that for $\mathbf{t}' = (t'_k)_{k \in J^\circ_{\mathbf{v}}} \in (\mathbb{K}^*)^{J^\circ_{\mathbf{v}}}$, $\mathbf{g}_{\mathbf{v},\mathbf{w}}(\mathbf{t}') = g_1 \cdots g_n$ is defined in (3.4.30). Define $G^{\mathrm{sf}}_{\mathbf{v},\mathbf{w}} := {\mathbf{g}_{\mathbf{v},\mathbf{w}}(\mathbf{t}') \mid \mathbf{t}' \in (\mathcal{F}^*_{\mathrm{sf}})^{J^\circ_{\mathbf{v}}}} \subset G^\circ$. The following result is closely related to [MR04, Lemma 11.8].

Lemma 3.5.3. Let $v \leq w \in W$ and $(\mathbf{v}, \mathbf{w}) \in \operatorname{Red}(v, w)$. Let $\mathbf{g}_{\mathbf{v}, \mathbf{w}}(\mathbf{t}')$ be as in (3.4.30) for $\mathbf{t}' \in (\mathcal{F}_{\mathrm{sf}}^*)^{J_{\mathbf{v}}^\circ}$. Then for each $k = 0, 1, \ldots, n$ and for all $x \in U^\circ \cap \dot{v}_{(k)}^{-1} U_- \dot{v}_{(k)}$, we have

$$g_1 \cdots g_k \cdot x \cdot g_{k+1} \cdots g_n \in g_1 \cdots g_n \cdot U^\diamond. \tag{3.5.2}$$

Proof. We prove this by induction on k. For k = n, the result is trivial, so suppose that k < n. Let $x \in U^{\diamond} \cap \dot{v}_{(k)}^{-1}U_{-}\dot{v}_{(k)}$. If $g_{k+1} = \dot{s}_i$ for some $i \in I$ then $\ell(v_{(k+1)}) = \ell(v_{(k)}) + \ell(s_i)$, so we use (3.4.9) to show that $x \cdot g_{k+1} = g_{k+1} \cdot x'$ for some $x' \in U \cap \dot{v}_{(k+1)}^{-1}U_{-}\dot{v}_{(k+1)}$. Since $x' = \dot{s}_i^{-1}x\dot{s}_i$ and each term belongs to G^{\diamond} , we see that $x' \in U^{\diamond} \cap \dot{v}_{(k+1)}^{-1}U_{-}\dot{v}_{(k+1)}$, so we are done by induction.

Suppose now that $g_{k+1} = y_i(t)$ for some $i \in I$ and $t \in \mathcal{F}^*_{sf}$. Write

$$x \cdot g_{k+1} = g_{k+1} \cdot g_{k+1}^{-1} x g_{k+1} = g_{k+1} \cdot y_i(-t) x y_i(t).$$

By (3.4.5), $U^{\diamond} \cap \dot{v}_{(k)}^{-1}U_{-}\dot{v}_{(k)} = U^{\diamond}(\operatorname{Inv}(v_{(k)}))$. Clearly again $y_i(-t)xy_i(t) \in G^{\diamond}$, and we claim that $y_i(-t)xy_i(t) \in U(\operatorname{Inv}(v_{(k)}))$ for all $x \in U(\operatorname{Inv}(v_{(k)}))$. First, using Lemma 3.4.1(ii), we can assume that $x \in U_{\alpha}$ for some $\alpha \in \operatorname{Inv}(v_{(k)})$. Since $v_{(k)}s_i > v_{(k)}$, we have $\alpha_i \notin \operatorname{Inv}(v_{(k)})$, so $\alpha \neq \alpha_i$. Let $\Psi = \{m\alpha - r\alpha_i\} \subset \Phi^+$ be the set of roots as in Lemma 3.4.3. Our goal is to show that $\Psi \subset \operatorname{Inv}(v_{(k)})$. Let $\gamma := m\alpha - r\alpha_i \in \Psi$ for some m > 0 and $r \ge 0$. We now show that $\gamma \in \operatorname{Inv}(v_{(k)})$, which is equivalent to saying that $v_{(k)}\gamma < 0$. Indeed, $v_{(k)}\gamma = mv_{(k)}\alpha - rv_{(k)}\alpha_i$. Since $\alpha \in \operatorname{Inv}(v_{(k)})$, $v_{(k)}\alpha < 0$. Since $\alpha_i \notin \operatorname{Inv}(v_{(k)})$, $v_{(k)}\alpha_i > 0$. Thus $v_{(k)}\gamma < 0$, because $-v_{(k)}\gamma$ is a positive linear combination of positive roots. We have shown that $\Psi \subset \operatorname{Inv}(v_{(k)})$, thus by Lemma 3.4.3, we find $y_i(-t)xy_i(t) \in U(\operatorname{Inv}(v_{(k)}))$. Since $v_{(k)} = v_{(k+1)}$, we get

$$y_i(-t)xy_i(t) \in U^{\diamond}(\operatorname{Inv}(v_{(k)})) = U^{\diamond} \cap \dot{v}_{(k)}^{-1}U_-\dot{v}_{(k)} = U^{\diamond} \cap \dot{v}_{(k+1)}^{-1}U_-\dot{v}_{(k+1)},$$

and we are done by induction.

-		

Proposition 3.5.4. For $v \leq w \in W$, the set $G_{\mathbf{v},\mathbf{w}}^{\mathrm{sf}} \cdot U^{\diamond} \subset G^{\diamond}$ does not depend on the choice of $(\mathbf{v},\mathbf{w}) \in \mathrm{Red}(v,w)$. In other words: let $(\mathbf{v}_0,\mathbf{w}_0), (\mathbf{v}_1,\mathbf{w}_1) \in \mathrm{Red}(v,w)$. Then for any $\mathbf{t}_0 \in (\mathcal{F}_{\mathrm{sf}}^*)^{J_{\mathbf{v}_0}^{\diamond}}$ there exists $\mathbf{t}_1 \in (\mathcal{F}_{\mathrm{sf}}^*)^{J_{\mathbf{v}_1}^{\diamond}}$ and $x \in U^{\diamond}$ such that $\mathbf{g}_{\mathbf{v}_0,\mathbf{w}_0}(\mathbf{t}_0) = \mathbf{g}_{\mathbf{v}_1,\mathbf{w}_1}(\mathbf{t}_1) \cdot x$.

Proof. Recall that for each $\mathbf{w}_0 \in \text{Red}(w)$ there exists a unique positive subexpression \mathbf{v}_0 for v such that $(\mathbf{v}_0, \mathbf{w}_0) \in \text{Red}(v, w)$. We need to show that choosing a different reduced expression \mathbf{w}_1 for w results in a subtraction-free coordinate change $\mathbf{t}_0 \mapsto \mathbf{t}_1$ of the parameters in Theorem 3.4.15. Any two reduced expressions for w are related by a sequence of braid moves, so it suffices to assume that \mathbf{w}_0 and \mathbf{w}_1 differ in a single braid move.

The explicit formulae for the corresponding coordinate transformations can be found in the proof of [Rie08, Prop. 7.2], however, an extra step is needed to show that those formulae indeed give the correct coordinate transformations. More precisely, suppose that Φ' is a root subsystem of Φ of rank 2 (i.e., Φ' is of type $A_1 \times A_1$, A_2 , B_2 , or G_2), and let W' be its Weyl group. Then it was checked in the proof of [Rie08, Prop. 7.2] that for any $v' \leq w' \in W'$, any $(\mathbf{v}'_0, \mathbf{w}'_0), (\mathbf{v}'_1, \mathbf{w}'_1) \in \text{Red}(v', w')$, and any $\mathbf{t}'_0 \in (\mathcal{F}^*_{sf})^{J^\circ_{\mathbf{v}'_0}}$, there exists $\mathbf{t}'_1 \in (\mathcal{F}^*_{sf})^{J^\circ_{\mathbf{v}'_1}}$ and $x \in U$ such that $\mathbf{g}_{\mathbf{v}'_0, \mathbf{w}'_0}(\mathbf{t}'_0) = \mathbf{g}_{\mathbf{v}'_1, \mathbf{w}'_1}(\mathbf{t}'_1) \cdot x$.

Since $\mathbf{g}_{\mathbf{v}'_0,\mathbf{w}'_0}(\mathbf{t}'_0)$ and $\mathbf{g}_{\mathbf{v}'_1,\mathbf{w}'_1}(\mathbf{t}'_1)$ belong to G^{\diamond} , we must have $x \in U^{\diamond}$. Note that the only non-trivial cases to check are the ones where $w' = (s_i s_j)^{m_{ij}}$ is the longest element of W' and $v' \in W'$ is arbitrary. If Φ' is of type G_2 then we must have $\Phi = \Phi'$ and $w' = w = w_0$, so we are done. Using a computer algebra system [Sag16], we were able to additionally check in each of the remaining cases (i.e. Φ' being of type $A_1 \times A_1$, A_2 , or B_2) that $x \in \dot{v}'^{-1}U_-\dot{v}'$.

Let us now complete the proof of Proposition 3.5.4 (as well as of [Rie08, Prop. 7.2]). Suppose that \mathbf{w}_0 and \mathbf{w}_1 differ in a braid move along a subword $g_{k+1} \cdots g_{k+m}$ of $g_1 \cdots g_n$. Here $g_{k+1} \cdots g_{k+m} = \mathbf{g}_{\mathbf{v}'_0, \mathbf{w}'_0}(\mathbf{t}'_0)$ as above. Applying a move from [Rie08], we transform $g_{k+1} \cdots g_{k+m}$ into $g'_{k+1} \cdots g'_{k+m} x$ for some $x \in U$ and $g'_{k+1} \cdots g'_{k+m} = \mathbf{g}_{\mathbf{v}'_1, \mathbf{w}'_1}(\mathbf{t}'_1)$. Thus

$$g_1 \cdots g_n = g_1 \cdots g_k \cdot g'_{k+1} \cdots g'_{k+m} \cdot x \cdot g_{k+m+1} \cdots g_n.$$

We have checked that $x \in U^{\diamond}$, and if k + m < n then $x \in U^{\diamond} \cap v'^{-1}U_{-}v'$, where $v' \in W$ satisfies $v_{(k+m)} = v_{(k)} \cdot v'$ and $\ell(v_{(k+m)}) = \ell(v_{(k)}) + \ell(v')$. It follows by (3.4.6) that $x \in$ $U^{\diamond} \cap v_{(k+m)}^{-1} U_{-} v_{(k+m)}$, so by Lemma 3.5.3, we have

$$g_1 \cdots g_n \in g_1 \cdots g_k \cdot g'_{k+1} \cdots g'_{k+m} \cdot g_{k+m+1} \cdots g_n \cdot U^\diamond.$$

Definition 3.5.5. From now on we denote $R_{v,w}^{\text{sf}} := G_{\mathbf{v},\mathbf{w}}^{\text{sf}} B^{\text{sf}} \subset G^{\diamond}$. By Proposition 3.5.4, the set $R_{v,w}^{\text{sf}}$ does not depend on the choice of $(\mathbf{v}, \mathbf{w}) \in \text{Red}(v, w)$. As we discuss in Section 3.5.4, $R_{v,w}^{\text{sf}}$ is the "subtraction-free" analog of $R_{v,w}^{>0}$.

3.5.2 Collision moves

Assume $\mathbb{K} = \overline{\mathcal{F}}$. By [FZ99, Eq. (2.13)], for each $t \in \mathcal{F}_{sf}^*$ there exist $t_+ \in \mathcal{F}_{sf}^*$, $a_+ \in T^{sf}$, and $t_- \in \mathcal{F}^{\diamond}$ satisfying

$$\dot{s}_i x_i(t) = a_+ x_i(t_-) y_i(t_+), \quad x_i(t) \dot{s}_i = y_i(t_+) x_i(t_-) a_+, \tag{3.5.3}$$

$$\dot{s}_i^{-1}y_i(t) = a_+ y_i(t_-) x_i(t_+), \quad y_i(t)\dot{s}_i^{-1} = x_i(t_+) y_i(t_-) a_+.$$
(3.5.4)

(Here, each of the four moves yields different t_+, a_+, t_- .) By [FZ99, Eq. (2.11)], for each $t, t' \in \mathcal{F}_{sf}^*$ there exist $t_+, t'_+ \in \mathcal{F}_{sf}^*$ and $a_+ \in T^{sf}$ satisfying

$$x_i(t)y_i(t') = y_i(t'_+)x_i(t_+)a_+, \quad y_i(t')x_i(t) = x_i(t_+)y_i(t'_+)a_+.$$
(3.5.5)

We also have [FZ99, Eq. (2.9)]

$$x_i(t)y_j(t') = y_j(t')x_i(t), \text{ for } i \neq j.$$
 (3.5.6)

As a direct consequence of (3.5.5), (3.5.6), and Lemma 3.5.2, for any $v, w \in W$ we get

$$U_{\rm sf}(v) \cdot U_{\rm sf}^{-}(w) \cdot T^{\rm sf} = U_{\rm sf}^{-}(w) \cdot U_{\rm sf}(v) \cdot T^{\rm sf}.$$
(3.5.7)

Lemma 3.5.6.
(i) Let $w \in W$. Then

$$B_{-}^{\rm sf} \cdot \dot{w}^{-1} \cdot U_{\rm sf}^{-}(w) = B_{-}^{\rm sf} \cdot U_{\rm sf}(w^{-1}) \quad and \quad U_{\rm sf}^{-}(w) \cdot \dot{w}^{-1} \cdot B_{-}^{\rm sf} = U_{\rm sf}(w^{-1}) \cdot B_{-}^{\rm sf}.$$
(3.5.8)

(ii) If $v, w \in W$ are such that $\ell(vw) = \ell(v) + \ell(w)$ then

$$\dot{w}^{-1}\dot{v}^{-1} \cdot U_{\rm sf}^{-}(v) \subset B_{-}^{\rm sf} \cdot \dot{w}^{-1} \cdot U_{\rm sf}(v^{-1}).$$
(3.5.9)

(iii) Let $w_1, \ldots, w_k \in W$ be such that $\ell(w_1 \cdots w_k) = \ell(w_1) + \cdots + \ell(w_k)$. Then for any $h \in U_{\mathrm{sf}}^-(w_1 \cdots w_k)$ there exist $b_1 \in U_{\mathrm{sf}}(w_1^{-1}), \ldots, b_k \in U_{\mathrm{sf}}(w_k^{-1})$ such that for each $1 \leq i \leq k$, we have

$$\dot{w}_i^{-1} \cdots \dot{w}_1^{-1} \cdot h \in B_-^{\text{sf}} \cdot b_i \cdots b_1.$$
(3.5.10)

(iv) Let $v \leq w \in W$. Then

$$\dot{v}^{-1} \cdot U_{\rm sf}^{-}(w) \subset B_{-}^{\rm sf} \cdot U_{\rm sf}(v^{-1}).$$
 (3.5.11)

Proof. Let us prove the following claim: if $vv_1 = w$ and $\ell(w) = \ell(v) + \ell(v_1)$ then

$$\dot{v}^{-1}U_{\rm sf}^{-}(w) \subset T^{\rm sf} \cdot (U_{-}^{\diamond} \cap \dot{v}^{-1}U\dot{v}) \cdot U_{\rm sf}^{-}(v_{1}) \cdot U_{\rm sf}(v^{-1}).$$
(3.5.12)

We prove this by induction on $\ell(v)$. If $\ell(v) = 0$ then v = id and (3.5.12) is trivial. Otherwise there exists an $i \in I$ such that $v' := s_i v < v$ and thus $w' := s_i w < w$. Let $\mathbf{y}_i(\mathbf{t}') \in U^-_{\text{sf}}(w)$. Using (3.5.4), we see that for some $t_1 \in \mathcal{F}^*_{\text{sf}}, t_+ \in \mathcal{F}^*_{\text{sf}}$ and $t_- \in \mathcal{F}^\diamond$,

$$\dot{v}^{-1} \cdot \mathbf{y}_{\mathbf{i}}(\mathbf{t}') \in \dot{v}'^{-1} \cdot \dot{s}_{i}^{-1} y_{i}(t_{1}) \cdot U_{\mathrm{sf}}^{-}(w') \subset T^{\mathrm{sf}} \dot{v}'^{-1} \cdot y_{i}(t_{-}) x_{i}(t_{+}) \cdot U_{\mathrm{sf}}^{-}(w').$$

By (3.5.7), $x_i(t_+) \cdot U_{sf}^-(w') \subset T^{sf} \cdot U_{sf}^-(w') \cdot U_{sf}(s_i)$. Clearly $s_i v' > v'$, so $y' := \dot{v}'^{-1} y_i(t_-) \dot{v}' \in U_-$. On the other hand, $\dot{v}y'\dot{v}^{-1} = \dot{s}_i^{-1} y_i(t_-) \dot{s}_i = x_i(-t_-) \in U$. Thus $y' \in U_- \cap \dot{v}^{-1}U\dot{v}$, and it is also clear that $y' \in G^{\diamond}$. We have shown that

$$\dot{v}^{-1} \cdot \mathbf{y}_{\mathbf{i}}(\mathbf{t}') \in T^{\mathrm{sf}} \cdot y' \cdot \dot{v}'^{-1} \cdot U_{\mathrm{sf}}^{-}(w') \cdot U_{\mathrm{sf}}(s_{i}) \subset T^{\mathrm{sf}} \cdot (U_{-}^{\diamond} \cap \dot{v}^{-1} U \dot{v}) \cdot \dot{v}'^{-1} \cdot U_{\mathrm{sf}}^{-}(w') \cdot U_{\mathrm{sf}}(s_{i}).$$
(3.5.13)

We have $v'v_1 = w'$, so by induction,

$$\dot{v}'^{-1} \cdot U_{\rm sf}^{-}(w') \subset T^{\rm sf} \cdot (U_{-}^{\diamond} \cap \dot{v}'^{-1}U\dot{v}') \cdot U_{\rm sf}^{-}(v_{1}) \cdot U_{\rm sf}(v'^{-1}).$$

Since $U_{\rm sf}(v'^{-1}) \cdot U_{\rm sf}(s_i) = U_{\rm sf}(v^{-1})$, we have shown that

$$\dot{v}^{-1}\mathbf{y}_{\mathbf{i}}(\mathbf{t}') \in T^{\mathrm{sf}} \cdot (U_{-}^{\diamond} \cap \dot{v}^{-1}U\dot{v}) \cdot (U_{-}^{\diamond} \cap \dot{v}'^{-1}U\dot{v}') \cdot U_{\mathrm{sf}}^{-}(v_{1}) \cdot U_{\mathrm{sf}}(v^{-1}).$$

By (3.4.6) applied to $a = s_i$, b = v', ab = v, we get $\operatorname{Inv}(v') \subset \operatorname{Inv}(v)$, so $(U_{-}^{\diamond} \cap \dot{v}'^{-1}U\dot{v}') \subset (U_{-}^{\diamond} \cap \dot{v}^{-1}U\dot{v})$, and we have finished the proof of (3.5.12). Combining (3.5.12) with (3.4.8), we obtain (3.5.9). Next, (3.5.10) can be shown by induction: the case k = 0 is trivial. For $k \geq 1$, we can write $h = h_1 \cdots h_k \in U_{\mathrm{sf}}^-(w_1) \cdots U_{\mathrm{sf}}^-(w_k)$. By (3.5.9), we have

$$\dot{w}_i^{-1}\cdots\dot{w}_1^{-1}\cdot h_1\cdots h_k\in B_-^{\mathrm{sf}}\cdot\dot{w}_i^{-1}\cdots\dot{w}_2^{-1}\cdot b_1'\cdot h_2\cdots h_k$$

for some $b'_1 \in U_{\rm sf}(w_1)$ that does not depend on *i*. Using (3.5.7), we write $b'_1 \cdot h_2 \cdots h_k = h'_2 \cdots h'_k \cdot b_1 \in U_{\rm sf}^-(w_2) \cdots U_{\rm sf}^-(w_k) \cdot U_{\rm sf}(w_1)$, and then proceed by induction. Let us state several further corollaries of (3.5.12):

$$\dot{w}^{-1} \cdot U_{\rm sf}^{-}(w) \subset T^{\rm sf} \cdot (U_{-}^{\diamond} \cap \dot{w}^{-1}U\dot{w}) \cdot U_{\rm sf}(w^{-1}), \qquad (3.5.14)$$

$$U_{\rm sf}^{-}(w) \cdot \dot{w}^{-1} \subset U_{\rm sf}(w^{-1}) \cdot (U_{-}^{\diamond} \cap \dot{w}U\dot{w}^{-1}) \cdot T^{\rm sf}, \qquad (3.5.15)$$

$$\dot{w} \cdot U_{\rm sf}(w^{-1}) \subset (U^{\diamond} \cap \dot{w}U_{-}\dot{w}^{-1}) \cdot U_{\rm sf}^{-}(w) \cdot T^{\rm sf}.$$
 (3.5.16)

Indeed, specializing (3.5.12) to v = w, we obtain (3.5.14). Eq. (3.5.15) is obtained from (3.5.14) by replacing w with $z := w^{-1}$ and then applying the involution $x \mapsto x^{\iota}$ of (3.4.4), while (3.5.16) is obtained from (3.5.15) by applying the involution $x \mapsto x^T$ of (3.4.3).

To show (3.5.8), observe that the inclusion $B_{-}^{\text{sf}} \cdot \dot{w}^{-1} \cdot U_{\text{sf}}^{-}(w) \subset B_{-}^{\text{sf}} \cdot U_{\text{sf}}(w^{-1})$ follows from (3.5.14). To show the reverse inclusion, we use (3.5.16) to write

$$B_{-}^{\mathrm{sf}} \cdot U_{\mathrm{sf}}(w^{-1}) = B_{-}^{\mathrm{sf}} \cdot \dot{w}^{-1} \cdot \dot{w} \cdot U_{\mathrm{sf}}(w^{-1}) \subset B_{-}^{\mathrm{sf}} \cdot \dot{w}^{-1} \cdot (U^{\diamond} \cap \dot{w}U_{-}\dot{w}^{-1}) \cdot U_{\mathrm{sf}}^{-}(w).$$

Since $\dot{w}^{-1} \cdot (U^{\diamond} \cap \dot{w}U_{-}\dot{w}^{-1}) \subset U_{-}^{\diamond}\dot{w}^{-1}$, we obtain $B_{-}^{\mathrm{sf}} \cdot \dot{w}^{-1} \cdot U_{\mathrm{sf}}^{-}(w) = B_{-}^{\mathrm{sf}} \cdot U_{\mathrm{sf}}(w^{-1})$, which is the first part of (3.5.8). The second part follows by applying the involution $x \mapsto x^{\iota}$ of (3.4.4).

It remains to show (3.5.11). We argue by induction on $\ell(w)$, and the base case $\ell(w) = 0$ is clear. Suppose that $v \leq w$, and let $w' := s_i w < w$ for some $i \in I$. If $v' := s_i v < v$ then by the same argument as in the proof of (3.5.13), we get

$$\dot{v}^{-1} \cdot U_{\mathrm{sf}}^{-}(w) \subset B_{-}^{\mathrm{sf}} \cdot \dot{v}'^{-1} \cdot U_{\mathrm{sf}}^{-}(w') \cdot U_{\mathrm{sf}}(s_i).$$

Since $v' \leq w'$, we can apply the induction hypothesis to write $\dot{v}'^{-1} \cdot U_{\rm sf}^{-}(w') \subset B_{-}^{\rm sf} \cdot U_{\rm sf}(v'^{-1})$. We thus obtain

$$\dot{v}^{-1} \cdot U_{\mathrm{sf}}^{-}(w) \subset B_{-}^{\mathrm{sf}} \cdot U_{\mathrm{sf}}(v'^{-1}) \cdot U_{\mathrm{sf}}(s_i) = B_{-}^{\mathrm{sf}} \cdot U_{\mathrm{sf}}(v^{-1}),$$

finishing the induction step in the case $s_i v < v$. But if $s_i v > v$ then $\dot{v}^{-1} y_i(t_1) \dot{v} \in U_-^{\diamond}$, so in this case we have $\dot{v}^{-1} U_{\rm sf}^-(w) \subset U_-^{\diamond} \cdot \dot{v}^{-1} \cdot U_{\rm sf}^-(w')$, and the result follows by applying the induction hypothesis to the pair $v \leq w'$.

3.5.3 Alternative parametrizations for the top cell

The following two lemmas are subtraction-free versions of [Rie06, Lemmas 4.2 and 4.3].

Lemma 3.5.7. Let $v \in W$. Then we have

$$R_{v,w_0}^{\mathrm{sf}} = U_{\mathrm{sf}}(vw_0) \cdot \dot{w}_0 \cdot B^{\mathrm{sf}}.$$

Proof. Recall from Definition 3.5.5 that $R_{v,w}^{\text{sf}} = G_{\mathbf{v},\mathbf{w}}^{\text{sf}} \cdot B^{\text{sf}}$. We have $w = w_0$, so choose a reduced expression \mathbf{w}_0 for w_0 that ends with v. With this choice, $G_{\mathbf{v},\mathbf{w}_0}^{\text{sf}} = U_{\text{sf}}^-(w_0v^{-1}) \cdot \dot{v}$. Thus we can write

$$R_{v,w_0}^{\rm sf} = G_{\mathbf{v},\mathbf{w}_0}^{\rm sf} \cdot B^{\rm sf} = U_{\rm sf}^-(w_0v^{-1}) \cdot \dot{v} \cdot B^{\rm sf} = U_{\rm sf}^-(w_0v^{-1}) \cdot \dot{v}\dot{w}_0^{-1} \cdot \dot{w}_0 \cdot B^{\rm sf}.$$

Let $z := w_0 v^{-1}$. Using (3.5.8) and $B_{-}^{\text{sf}} \cdot \dot{w}_0 = \dot{w}_0 \cdot B^{\text{sf}}$, we have

$$U_{\rm sf}^{-}(w_0v^{-1})\cdot \dot{v}\dot{w}_0^{-1}\cdot \dot{w}_0\cdot B^{\rm sf} = U_{\rm sf}^{-}(z)\cdot \dot{z}^{-1}\cdot \dot{w}_0\cdot B^{\rm sf} = U_{\rm sf}(z^{-1})\cdot \dot{w}_0\cdot B^{\rm sf}.$$

Combining the above equations, we find $R_{v,w_0}^{\text{sf}} = U_{\text{sf}}(z^{-1}) \cdot \dot{w}_0 \cdot B^{\text{sf}}$, and it remains to note that $z^{-1} = vw_0^{-1} = vw_0$.

Lemma 3.5.8. Let $v \leq w \in W$. Then we have

$$U_{\rm sf}(v^{-1}) \cdot U_{\rm sf}^{-}(w_0 w^{-1}) \cdot R_{v,w}^{\rm sf} = R_{{\rm id},w_0}^{\rm sf} = U_{\rm sf}^{-}(w_0) \cdot B^{\rm sf}.$$
 (3.5.17)

Proof. It follows from the definition of $G_{\mathbf{v},\mathbf{w}}^{\mathrm{sf}}$ that if w'w is length-additive then $U_{\mathrm{sf}}^{-}(w')R_{v,w}^{\mathrm{sf}} = R_{v,w'w}^{\mathrm{sf}}$. Applying this to $w' = w_0w^{-1}$, we get $U_{\mathrm{sf}}^{-}(w_0w^{-1}) \cdot R_{v,w}^{\mathrm{sf}} = R_{v,w_0}^{\mathrm{sf}}$. By Lemma 3.5.7, we have $R_{v,w_0}^{\mathrm{sf}} \cdot B^{\mathrm{sf}} = U_{\mathrm{sf}}(vw_0) \cdot \dot{w}_0 \cdot B^{\mathrm{sf}}$. Thus $U_{\mathrm{sf}}(v^{-1}) \cdot U_{\mathrm{sf}}(vw_0) \cdot \dot{w}_0 \cdot B^{\mathrm{sf}} = U_{\mathrm{sf}}(w_0) \cdot \dot{w}_0 \cdot B^{\mathrm{sf}}$, so applying Lemma 3.5.7 again, we find $U_{\mathrm{sf}}(w_0) \cdot \dot{w}_0 \cdot B^{\mathrm{sf}} = R_{\mathrm{id},w_0}^{\mathrm{sf}} \cdot B^{\mathrm{sf}}$. The result follows since $R_{\mathrm{id},w_0}^{\mathrm{sf}} = U_{\mathrm{sf}}^{-}(w_0) \cdot B^{\mathrm{sf}}$.

3.5.4 Evaluation

We explain the relationship between $R_{v,w}^{\mathrm{sf}}$ and $R_{v,w}^{>0}$. Given $\mathbf{t}' \in \mathbb{R}_{>0}^{|\mathbf{t}|}$, we denote by $\mathrm{eval}_{\mathbf{t}'}$: $\mathcal{F}_{\mathrm{sf}} \to \mathbb{R}_{>0}$ the evaluation homomorphism (of semifields) sending $f(\mathbf{t}) \to f(\mathbf{t}')$. It extends to a well defined group homomorphism $\mathrm{eval}_{\mathbf{t}'}: G^{\diamond} \to G(\mathbb{R})$, and it follows from Theorem 3.4.15 that $\{\mathrm{eval}_{\mathbf{t}'}(g)B \mid g \in R_{v,w}^{\mathrm{sf}}\} = R_{v,w}^{>0}$ as subsets of $(G/B)_{\mathbb{R}}$. It is clear that the following diagram is commutative.

Here solid arrows denote regular maps, and dashed arrows denote maps defined on a subset $\mathcal{F}' \subset \mathcal{F}$ given by $\mathcal{F}' := \{R(\mathbf{t})/Q(\mathbf{t}) \mid R(\mathbf{t}), Q(\mathbf{t}) \in \mathbb{R}[\mathbf{t}], Q(\mathbf{t}') \neq 0\}$. Since the diagram (3.5.18) is commutative, it follows that the images $\Delta_i^{\mp}(G^{\diamond})$ and $\Delta_i^{\pm}(G^{\diamond})$ belong to \mathcal{F}' .

Let $\mathbf{t} = (\mathbf{t}', \mathbf{t}'')$. Observe that any $f(\mathbf{t}', \mathbf{t}'') \in \mathcal{F}_{\mathrm{sf}}^*$ gives rise to a continuous function

$$\mathbb{R}_{>0}^{|\mathbf{t}'|} \times \mathbb{R}_{>0}^{|\mathbf{t}''|} \to \mathbb{R}_{>0}.$$

Lemma 3.5.9. Suppose that $f(\mathbf{t}', \mathbf{t}'') \in \mathcal{F}_{sf}^*$ is such that the corresponding function $\mathbb{R}_{>0}^{|\mathbf{t}'|} \times \mathbb{R}_{>0}^{|\mathbf{t}''|} \to \mathbb{R}_{>0}$ extends to a continuous function $\mathbb{R}_{>0}^{|\mathbf{t}'|} \times \mathbb{R}_{\geq 0}^{|\mathbf{t}''|} \to \mathbb{R}_{\geq 0}$. Then $\lim_{\mathbf{t}''\to 0} f(\mathbf{t}', \mathbf{t}'')$ can be represented (as a function $\mathbb{R}_{>0}^{|\mathbf{t}'|} \to \mathbb{R}_{\geq 0}$) by a subtraction-free rational expression in \mathbf{t}' .

Proof. By induction, it is enough to prove this when $|\mathbf{t}''| = 1$, where $\mathbf{t}'' = t''$ is a single variable. In this case, $f(\mathbf{t}', t'') = R(\mathbf{t}', t'')/Q(\mathbf{t}', t'')$ where R and Q have positive coefficients. Let us consider R and Q as polynomials in t'' only. After dividing R and Q by $(t'')^k$ for some k, we may assume that one of them is not divisible by t''. Then Q cannot be divisible by t'', otherwise f would not give rise to a continuous function $\mathbb{R}_{>0}^{|\mathbf{t}'|} \times \mathbb{R}_{\geq 0}^{|\mathbf{t}''|} \to \mathbb{R}_{\geq 0}$. We can write $Q(\mathbf{t}', t'') = Q_1(\mathbf{t}', t'')t'' + Q_2(\mathbf{t}')$ and $R(\mathbf{t}', t'') = R_1(\mathbf{t}', t'')t' + R_2(\mathbf{t}')$, where R_1, R_2, Q_1, Q_2 are polynomials with nonnegative coefficients and $Q_2(\mathbf{t}') \neq 0$. Thus $\lim_{t''\to 0} f(\mathbf{t}', t'')$ can be represented by $R_2(\mathbf{t}')/Q_2(\mathbf{t}')$, which is a subtraction free rational expression in \mathbf{t}' .

Lemma 3.5.10. (Assume $\mathbb{K} = \mathbb{C}$.) Suppose that $a \leq b \leq c \in W$. Then we cannot have $\Delta^{\mp}(\dot{b}^{-1}x) = 0$ for all $x \in G(\mathbb{R})$ such that $xB \in R^{>0}_{a,b}$.

Proof. Suppose that $\Delta^{\mp}(\dot{b}^{-1}x) = 0$ for all $x \in G(\mathbb{R})$ such that $xB \in R_{a,b}^{>0}$. Consider the map $\Delta_i^{\text{flag}} : G/B \to \mathbb{P}^{\binom{2n}{n-1}-1}W^{W_{\omega_i}}$ from Lemma 3.4.21(iv). We get that the $b\omega_i$ -th coordinate of Δ_i^{flag} is identically zero on $R_{a,c}^{>0}$. Therefore it must be zero on the Zariski closure of $R_{a,c}^{>0}$ inside G/B. It is well known that $R_{a,c}^{>0}$ is Zariski dense in $\mathring{R}_{a,c}$, so the closure of $R_{a,c}^{>0}$ is $R_{a,c}$. By (3.4.14), $R_{a,c}$ contains $\dot{b}B = \mathring{R}_{b,b}$, thus $\Delta_i^{\mp}(\dot{b}^{-1}\dot{b})$ must be zero. We get a contradiction since by definition $\Delta_i^{\mp}(\dot{b}^{-1}\dot{b}) = 1$.

3.5.5 Applications to the flag variety

We use the machinery developed in the previous sections to obtain some natural statements about $(G/B)_{\geq 0}$.

Lemma 3.5.11. (Assume $\mathbb{K} = \overline{\mathcal{F}}$.) Suppose that $a \leq c \in W$ and $b \in W$. Then for any $x \in R_{a,c}^{sf}$ and $i \in I$,

$$\Delta_i^{\mp}(\dot{b}^{-1}x) \in \mathcal{F}_{\mathrm{sf}}.\tag{3.5.19}$$

Moreover, if $a \leq b \leq c$ then

$$\Delta_i^{\mp}(\dot{b}^{-1}x) \in \mathcal{F}_{\mathrm{sf}}^*, \quad and \quad x \in \dot{b}B_-B.$$
(3.5.20)

Proof. Let $\mathbf{t} = (\mathbf{t}_1, \mathbf{t}_2, \mathbf{t}_3)$ for $|\mathbf{t}_1| = \ell(a)$, $|\mathbf{t}_2| = \ell(w_0) - \ell(c)$, $|\mathbf{t}_3| = \ell(c) - \ell(a)$. Choose reduced words \mathbf{i} for a^{-1} and \mathbf{j} for w_0c^{-1} , and let $(\mathbf{a}, \mathbf{c}) \in \text{Red}(a, c)$. Suppose that $x \in \mathbf{g}_{\mathbf{a}, \mathbf{c}}(\mathbf{t}_3)B^{\text{sf}}$ and let

$$g := \mathbf{x}_{\mathbf{i}}(\mathbf{t}_1) \cdot \mathbf{y}_{\mathbf{j}}(\mathbf{t}_2) \cdot \mathbf{g}_{\mathbf{a},\mathbf{c}}(\mathbf{t}_3) \in U_{\mathrm{sf}}(a^{-1}) \cdot U_{\mathrm{sf}}^{-}(w_0 c^{-1}) \cdot R_{a,c}^{\mathrm{sf}}.$$

By Lemma 3.5.8, $g \in U_{\rm sf}^-(w_0) \cdot B^{\rm sf} = U_{\rm sf}^-(b) \cdot U_{\rm sf}^-(b^{-1}w_0) \cdot B^{\rm sf}$. By (3.5.8), we have $\dot{b}^{-1} \cdot U_{\rm sf}^-(b) \subset B_-^{\rm sf} \cdot U_{\rm sf}(b^{-1})$. Therefore

$$\dot{b}^{-1}g \in B^{\mathrm{sf}}_{-} \cdot U_{\mathrm{sf}}(b^{-1}) \cdot U^{-}_{\mathrm{sf}}(b^{-1}w_0) \cdot B^{\mathrm{sf}}.$$

By (3.5.7), we get $\dot{b}^{-1}g \in B^{\mathrm{sf}}_{-} \cdot U^{-}_{\mathrm{sf}}(b^{-1}w_{0}) \cdot U_{\mathrm{sf}}(b^{-1}) \cdot B^{\mathrm{sf}} = B^{\mathrm{sf}}_{-} \cdot B^{\mathrm{sf}}$, and by definition, $\Delta^{\mp}_{i}(y) \in \mathcal{F}^{*}_{\mathrm{sf}}$ for any $y \in B^{\mathrm{sf}}_{-} \cdot B^{\mathrm{sf}}$. Since Δ^{\mp}_{i} is a regular function on G by Lemma 3.4.21(ii), the function $f(\mathbf{t}_{1}, \mathbf{t}_{2}, \mathbf{t}_{3}) := \Delta^{\mp}_{i}(\dot{b}^{-1}g) \in \mathcal{F}^{*}_{\mathrm{sf}}$ extends to a continuous function on $\mathbb{R}^{|\mathbf{t}_{1}|}_{\geq 0} \times \mathbb{R}^{|\mathbf{t}_{2}|}_{\geq 0} \times \mathbb{R}^{|\mathbf{t}_{3}|}_{\geq 0}$. Therefore by Lemma 3.5.9, $\lim_{\mathbf{t}_{1},\mathbf{t}_{2}\to 0} f(\mathbf{t}_{1},\mathbf{t}_{2},\mathbf{t}_{3})$ is a subtraction-free rational expression in \mathbf{t}_{3} . Since $\lim_{\mathbf{t}_{1},\mathbf{t}_{2}\to 0} g = \mathbf{g}_{\mathbf{a},\mathbf{c}}(\mathbf{t}_{3})$, we get that $\Delta^{\mp}_{i}(\dot{b}^{-1}\mathbf{g}_{\mathbf{a},\mathbf{c}}(\mathbf{t}_{3})) \in \mathcal{F}_{\mathrm{sf}}$. Since $x \in \mathbf{g}_{\mathbf{a},\mathbf{c}}(\mathbf{t}_{3})B^{\mathrm{sf}}$, (3.5.19) follows.

Suppose now that $a \leq b \leq c$. We would like to show (3.5.20), thus let us assume that for some $i \in I$ and for $x \in R_{a,c}^{\text{sf}}$, we have $\Delta_i^{\mp}(\dot{b}^{-1}x) = 0$. Let $\mathbf{t}' \in (\mathcal{F}_{\text{sf}}^*)^{|\mathbf{t}|}$ and $(\mathbf{a}, \mathbf{c}) \in \text{Red}(a, c)$ be such that $x \in \mathbf{g}_{\mathbf{a},\mathbf{c}}(\mathbf{t}')B^{\text{sf}}$, and let $y(\mathbf{t}) := \mathbf{g}_{\mathbf{a},\mathbf{c}}(\mathbf{t})$. Then we have $\Delta_i^{\mp}(\dot{b}^{-1}y(\mathbf{t})) \in \mathcal{F}_{\text{sf}}$ by (3.5.19). If $\Delta_i^{\mp}(\dot{b}^{-1}y(\mathbf{t}))$ was a nonzero rational function in \mathbf{t} then clearly substituting $\mathbf{t} \mapsto \mathbf{t}'$ for $\mathbf{t}' \in (\mathcal{F}_{\text{sf}}^*)^{|\mathbf{t}|}$ would also produce a nonzero rational function. Since substituting $\mathbf{t} \mapsto \mathbf{t}'$ yields $\Delta_i^{\mp}(\dot{b}^{-1}x) = 0$, we must have $\Delta_i^{\mp}(\dot{b}^{-1}y(\mathbf{t})) = 0$. Therefore $\Delta_i^{\mp}(\dot{b}^{-1}x') = 0$ for all $x' \in R_{a,c}^{\text{sf}}$.

Let now $\mathbf{t}' \in \mathbb{R}_{>0}^{|\mathbf{t}|}$. Recall from Section 3.5.4 that the image of $R_{a,c}^{\mathrm{sf}}$ in $(G/B)_{\mathbb{R}}$ under the map eval_{t'} equals $R_{a,c}^{>0}$, thus by (3.5.18), $\Delta_i^{\mp}(\dot{b}^{-1}x') = 0$ for all $x' \in G(\mathbb{R})$ such that $x'B \in R_{a,c}^{>0}$. This contradicts Lemma 3.5.10, hence $\Delta_i^{\mp}(\dot{b}^{-1}x) \in \mathcal{F}_{\mathrm{sf}}^*$, and therefore $x \in \dot{b}B_-B$ follows from Lemma 3.4.21(iii), finishing the proof of (3.5.20). **Corollary 3.5.12.** (Assume $\mathbb{K} = \mathbb{C}$.) Suppose that $a \leq c \in W$ and $b \in W$. Then for any $(\mathbf{a}, \mathbf{c}) \in \operatorname{Red}(a, c)$ and $\mathbf{t}' \in \mathbb{R}^{J_{\mathbf{a}}^{\circ}}_{>0}$, we have

$$\Delta_i^{\mp}(\dot{b}^{-1}\mathbf{g}_{\mathbf{a},\mathbf{c}}(\mathbf{t}')) \ge 0. \tag{3.5.21}$$

Moreover, if $a \leq b \leq c$ then

$$\Delta_i^{\mp}(\dot{b}^{-1}\mathbf{g}_{\mathbf{a},\mathbf{c}}(\mathbf{t}')) > 0, \quad and \quad R_{a,c}^{>0} \subset \dot{b}B_-B/B.$$
(3.5.22)

Proof. By (3.5.19), we know that $\Delta_i^{\mp}(\dot{b}^{-1}\mathbf{g}_{\mathbf{a},\mathbf{c}}(\mathbf{t})) \in \mathcal{F}_{\mathrm{sf}}$ for all $i \in I$. Evaluating at $\mathbf{t} = \mathbf{t}'$ (cf. Section 3.5.4), we find that $\Delta_i^{\mp}(\dot{b}^{-1}\mathbf{g}_{\mathbf{a},\mathbf{c}}(\mathbf{t}')) \geq 0$ for all $i \in I$, showing (3.5.21). Similarly, (3.5.22) follows from (3.5.20).

Proposition 3.5.13. (Assume $\mathbb{K} = \overline{\mathcal{F}}$.) For all $v, w, v', w' \in W$ and $x \in U_{\mathrm{sf}}(v') \cdot T^{\mathrm{sf}} \cdot U_{\mathrm{sf}}^{-}(w')$, we have $\Delta_{i}^{\pm}(\dot{v}x\dot{w}^{-1}) \in \mathcal{F}_{\mathrm{sf}}$.

Proof. Let $\mathbf{t} = (\mathbf{t}_1, \mathbf{t}_2, \mathbf{t}'_1, \mathbf{t}'_2)$ with $|\mathbf{t}_1| = \ell(v')$, $|\mathbf{t}_2| = \ell(w')$, $|\mathbf{t}'_1| = \ell(w_0) - \ell(v')$, and $|\mathbf{t}'_2| = \ell(w_0) - \ell(w')$. Let $\mathbf{t}_v := (\mathbf{t}'_1, \mathbf{t}_1)$ and $\mathbf{t}_w := (\mathbf{t}_2, \mathbf{t}'_2)$. Choose reduced words \mathbf{i}, \mathbf{j} for w_0 such that \mathbf{i} ends with a reduced word for v' and \mathbf{j} starts with a reduced word for w'. Set $g = g(\mathbf{t}_1, \mathbf{t}_2, \mathbf{t}_v, \mathbf{t}_w) := \mathbf{x}_{\mathbf{i}}(\mathbf{t}_v) \cdot a \cdot \mathbf{y}_{\mathbf{j}}(\mathbf{t}_w)$ for some arbitrary element $a \in T^{\mathrm{sf}}$. We get

$$\dot{v}g\dot{w}^{-1} \in \dot{v} \cdot U_{\rm sf}(w_0) \cdot T^{\rm sf} \cdot U_{\rm sf}^{-}(w_0) \cdot \dot{w}^{-1} \subset \dot{v} \cdot U_{\rm sf}(v^{-1}) \cdot U_{\rm sf}(vw_0) \cdot T^{\rm sf} \cdot U_{\rm sf}^{-}(w_0w^{-1}) \cdot U_{\rm sf}^{-}(w) \cdot \dot{w}^{-1}.$$

By (3.5.16), (3.5.7), and (3.5.8), we get $\dot{v}g\dot{w}^{-1} \in B^{\mathrm{sf}} \cdot U_{\mathrm{sf}}^{-}(v) \cdot U_{\mathrm{sf}}(w^{-1}) \cdot B_{-}^{\mathrm{sf}}$. By (3.5.7), we can permute $U_{\mathrm{sf}}^{-}(v)$ and $U_{\mathrm{sf}}(w^{-1})$, showing $\dot{v}g\dot{w}^{-1} \in B^{\mathrm{sf}} \cdot B_{-}^{\mathrm{sf}}$. Thus $\Delta_{i}^{\pm}(\dot{v}g\dot{w}^{-1}) \in \mathcal{F}_{\mathrm{sf}}^{*}$. It gives rise to a continuous function on $\mathbb{R}_{>0}^{|\mathbf{t}_{1}|} \times \mathbb{R}_{\geq 0}^{|\mathbf{t}_{2}|} \times \mathbb{R}_{\geq 0}^{|\mathbf{t}_{2}|}$, so sending $\mathbf{t}_{1}', \mathbf{t}_{2}' \to 0$ via Lemma 3.5.9 and varying $\mathbf{t}_{1}, \mathbf{t}_{2}$, and a, we get $\Delta_{i}^{\pm}(\dot{v}x\dot{w}^{-1}) \in \mathcal{F}_{\mathrm{sf}}$ for all $x \in U_{\mathrm{sf}}(v') \cdot T^{\mathrm{sf}} \cdot U_{\mathrm{sf}}^{-}(w')$.

3.6 Bruhat projections and total positivity

In this section, we prove a technical result (Theorem 3.6.4) which will be later used to finish the proof of Theorem 3.2.5. Assume \mathbb{K} is algebraically closed and fix $u \in W^J$.

3.6.1 The map $\zeta_{u,v}^{(J)}$

Retain the notation from Definition 3.4.23. Given $v \in W$ and $u \in W^J$, let us introduce a subset

$$G_{u,v}^{(J)} := \{ x \in \dot{u} G_0^{(J)} \mid \kappa_x x \in \dot{v} G_0^{(J)} \} \subset G.$$
(3.6.1)

Note that if $x \in G_{u,v}^{(J)}$ then $xP \subset G_{u,v}^{(J)}$, see Lemma 3.6.2(iii) below.

Definition 3.6.1. Define a map $\eta : G_{u,v}^{(J)} \to L_J$ sending $x \in G_{u,v}^{(J)}$ to $\eta(x) := [\dot{v}^{-1}\kappa_x x]_J$. Also define a map $\pi_{\dot{u}P_-} : \dot{u}G_0^{(J)} \to \dot{u}P_-$ sending $x \in \dot{u}G_0^{(J)}$ to the unique element $\pi_{\dot{u}P_-}(x) \in \dot{u}P_- \cap xU^{(J)}$. Explicitly (cf. Lemma 3.4.22(iii)), we put

$$\pi_{\dot{u}P_{-}}(x) := \dot{u}[\dot{u}^{-1}x]_{-}^{(J)}[\dot{u}^{-1}x]_{J} = x \cdot ([\dot{u}^{-1}x]_{+}^{(J)})^{-1}.$$
(3.6.2)

Finally, define $\zeta_{u,v}^{(J)} : G_{u,v}^{(J)} \to G$ by $\zeta_{u,v}^{(J)}(x) := \pi_{\dot{u}P_{-}}(x) \cdot \eta(x)^{-1}$.

Lemma 3.6.2.

- (i) The maps κ and $\pi_{\dot{u}P_{-}}$ are regular on $\dot{u}G_{0}^{(J)}$.
- (ii) The maps η and $\zeta_{u,v}^{(J)}$ are regular on $G_{u,v}^{(J)} \subset \dot{u}G_0^{(J)}$.
- (iii) If $x \in \dot{u}G_0^{(J)}$ and $x' \in xP$ then $\kappa_{x'} = \kappa_x$.
- (iv) If $x \in G_{u,v}^{(J)}$ and $x' \in xP$ then $\zeta_{u,v}^{(J)}(x) = \zeta_{u,v}^{(J)}(x')$.

Proof. Parts (i) and (ii) are clear since each map is a composition of regular maps. Part (iii) follows from Definition 3.4.23, since by construction the map κ starts by applying the isomorphism in (3.4.31), which gives a regular map $C_u^{(J)} \to \dot{u} U_-^{(J)} \dot{u}^{-1}$. To prove (iv), suppose that $x \in G_{u,v}^{(J)}$ and $x' \in xP$ is given by x' = xp for $p \in P$. Then $\pi_{\dot{u}P_-}(x') = \pi_{\dot{u}P_-}(x)[p]_J$ by Lemma 3.4.22(iii). By (iii), $\kappa_{x'} = \kappa_x$, and

$$\eta(x') = [\dot{v}^{-1}\kappa_{x'}x']_J = [\dot{v}^{-1}\kappa_x x]_J[p]_J = \eta(x)[p]_J, \quad \text{thus}$$

$$\zeta_{u,v}^{(J)}(x') = \pi_{\dot{u}P_-}(x') \cdot \eta(x')^{-1} = \pi_{\dot{u}P_-}(x)[p]_J \cdot [p]_J^{-1}\eta(x)^{-1} = \zeta_{u,v}^{(J)}(x). \quad \Box$$

Lemma 3.6.3. *Let* $x \in \dot{u}P_{-}$ *.*

(i) We have $\pi_{\dot{u}P_{-}}(x) = x$.

(ii) If
$$x \in G_{u,v}^{(J)}$$
 then $\zeta_{u,v}^{(J)}(x) = x\eta(x)^{-1}$.

Proof. Both parts are clear from Definition 3.6.1.

The ultimate goal of this section is to prove the following result.

Theorem 3.6.4. (Assume $\mathbb{K} = \mathbb{C}$.) Let $(u, u) \preceq (v, w) \preceq (v', w') \in Q_J$ and $x \in G$ be such that $xB \in R_{v',w'}^{>0}$. Then $x \in G_{u,v}^{(J)}$ and $\zeta_{u,v}^{(J)}(x) \in BB_-\dot{w}$.

3.6.2 Properties of κ

We further investigate the element $\kappa_x x$. Denote $\tilde{u} := u w_J \in W^J_{\max}$.

Lemma 3.6.5. The groups $U^{(J)}$, $U_1^{(J)}$, $U_2^{(J)}$ from Definition 3.4.23 satisfy

$$\dot{u}U_{-}^{(J)}\dot{u}^{-1} = \dot{\tilde{u}}U_{-}^{(J)}\dot{\tilde{u}}^{-1}, \qquad (3.6.3)$$

$$U_1^{(J)} = \dot{u} U_-^{(J)} \dot{u}^{-1} \cap U = \dot{u} U_- \dot{u}^{-1} \cap U, \qquad (3.6.4)$$

$$U_2^{(J)} = \dot{u}U_-^{(J)}\dot{u}^{-1} \cap U_- = \dot{\tilde{u}}U_-\dot{\tilde{u}}^{-1} \cap U_-.$$
(3.6.5)

Proof. By Lemma 3.4.22(ii), we see that $\dot{w}_J U_-^{(J)} \dot{w}_J^{-1} = U_-^{(J)}$, which shows (3.6.3). For (3.6.4), $U_1^{(J)} = \dot{u} U_-^{(J)} \dot{u}^{-1} \cap U$ is just the definition. By Lemma 3.4.5, we have $\dot{u} U_J^- \dot{u}^{-1} \subset U_-$, so (3.6.4) follows from (3.4.5).

For (3.6.5), observe that $w_J \Phi_J^+ = \Phi_J^-$, so $\tilde{u} \Phi_J^+ \subset \Phi^-$ by (3.4.6). We thus have $\dot{\tilde{u}} U_- \dot{\tilde{u}}^{-1} = (\dot{\tilde{u}} U_J^- \dot{\tilde{u}}^{-1}) \cdot (\dot{\tilde{u}} U_-^{(J)} \dot{\tilde{u}}^{-1})$ where $(\dot{\tilde{u}} U_J^- \dot{\tilde{u}}^{-1}) \subset U$, and hence $\dot{\tilde{u}} U_- \dot{\tilde{u}}^{-1} \cap U_- = \dot{\tilde{u}} U_-^{(J)} \dot{\tilde{u}}^{-1} \cap U_- = U_2^{(J)}$ by the definition of $U_2^{(J)}$.

Lemma 3.6.6. For $x \in \dot{u}G_0^{(J)}$, there exists a unique element $h \in U_2^{(J)}$ such that $hx \in U_1^{(J)}\dot{u}P$, and we have $h = \kappa_x$.

Proof. Let $g^{(J)} \in U^{(J)}$ and $p \in P$ be such that $g^{(J)}\dot{u} = xp$. We first show that such an $h \in U_2^{(J)}$ exists. By Definition 3.4.23, κ_x is an element of $U_2^{(J)}$ such that $\kappa_x g^{(J)} \in U_1^{(J)}$. In particular, $\kappa_x x = \kappa_x g^{(J)} \dot{u} p^{-1} \in U_1^{(J)} \dot{u} P$, which shows the existence. To show the uniqueness, observe that the action of $\dot{u} U_{-}^{(J)} \dot{u}^{-1}$ on $\dot{u} G_0^{(J)} / P \subset G / P$ is free by (3.4.31), and in particular the action of $U_2^{(J)}$ is also free.

Lemma 3.6.7. If $x \in \dot{u}G_0^{(J)} \cap B\dot{u}\dot{r}B$ for some $r \in W_J$ then $\kappa_x = 1$.

Proof. By Lemma 3.6.6, it suffices to show that $B\dot{u}\dot{r}B \subset U_1^{(J)}uP$. Write

$$B\dot{u}\dot{r}B \subset B\dot{u}P \subset (B\dot{u}B) \cdot P$$

By (3.4.34), $B\dot{u}B \subset (\dot{u}U_{-} \cap U\dot{u}) \cdot B$, therefore we find

$$B\dot{u}\dot{r}B \subset (\dot{u}U_{-} \cap U\dot{u}) \cdot P = (\dot{u}U_{-}\dot{u}^{-1} \cap U)\dot{u}P = U_{1}^{(J)}\dot{u}P,$$

where the last equality follows from (3.6.4).

Lemma 3.6.8. Let $a \in T$.

- (i) The subgroups $\dot{u}U^{(J)}\dot{u}^{-1}$, $U_1^{(J)}$, and $U_2^{(J)}$ are preserved under conjugation by a.
- (ii) If $x \in \dot{u}G_0^{(J)}$ then $ax \in \dot{u}G_0^{(J)}$ and $\kappa_{ax}ax = a\kappa_x x$.
- (iii) (Assume $\mathbb{K} = \mathbb{C}$.) For each $w \in W$, there exists $\rho_w^{\vee} \in Y(T)$ such that for all $x \in \dot{w}B_-B$, $\lim_{t\to 0} \rho_w^{\vee}(t) \cdot xB = \dot{w}B$ in G/B. If $w \in W^J$ then for all $x \in \dot{w}G_0^{(J)}$, $\lim_{t\to 0} \rho_w^{\vee}(t) \cdot xP = \dot{w}P$ in G/P.

Proof. Since $\dot{u} \in N_G(T)$, there exists $b \in T$ such that $a\dot{u} = \dot{u}b$. Thus $a\dot{u}U^{(J)}\dot{u}^{-1}a^{-1} = \dot{u}bU^{(J)}b^{-1}\dot{u}^{-1} = \dot{u}U^{(J)}\dot{u}^{-1}$, which shows (i), and (ii) is a simple consequence of (i). To show (iii), assume $\mathbb{K} = \mathbb{C}$ and choose $\rho^{\vee} \in Y(T)$ such that $\langle \alpha_i, \rho^{\vee} \rangle < 0$ for all $i \in I$. Then $\lim_{t\to 0} \rho^{\vee}(t)y\rho^{\vee}(t)^{-1} = 1$ for all $y \in U_-$, in particular, for all $y \in U_-^{(J)}$. Set $\rho_w^{\vee} := w^{-1}\rho^{\vee}$, thus for $t \in \mathbb{C}^*$, $\rho_w^{\vee}(t) = \dot{w}\rho^{\vee}(t)\dot{w}^{-1}$, see (3.4.2). Every $x \in \dot{w}B_-B$ belongs to $\dot{w}yB$ for some $y \in U_-$, so $\rho_w^{\vee}(t) \cdot x \cdot B = \dot{w}\rho^{\vee}(t)y\rho^{\vee}(t)^{-1} \cdot B \to \dot{w}B$ as $t \to 0$. Similarly, if $w \in W^J$ then every $x \in \dot{w}G_0^{(J)}$ belongs to $\dot{w}yP$ for some $y \in U_-^{(J)}$ by (3.4.31), so $\rho_w^{\vee}(t) \cdot xP \to \dot{w}P$ as $t \to 0$.

Lemma 3.6.9. Suppose that $v'' \leq ur \leq w''$ for some $v'', w'' \in W$ and $r \in W_J$, and let $x \in G$.

- (i) (Assume $\mathbb{K} = \overline{\mathcal{F}}$.) If $x \in R_{v'',w''}^{sf}$ then $x \in \dot{u}G_0^{(J)}$.
- (ii) (Assume $\mathbb{K} = \mathbb{C}$.) If $xB \in R^{>0}_{v'',w''}$ then $x \in \dot{u}G^{(J)}_0$ and $\kappa_x xB \in R^{>0}_{v'',ur_w}$ for some $r_w \in W_J$ such that $r_w \ge r$.

Proof. When $\mathbb{K} = \overline{\mathcal{F}}$, (3.5.20) shows that $R_{v'',w''}^{\mathrm{sf}} \subset \dot{u}\dot{r}B_{-}B \subset \dot{u}P_{-}B$, and by Lemma 3.4.22(i), $P_{-}B = G_{0}^{(J)}$, which shows (i). Similarly (for $\mathbb{K} = \mathbb{C}$), by Corollary 3.5.12, we have $x \in \dot{u}\dot{r}B_{-}B$ for any $x \in R_{v'',w''}^{>0}$, so $R_{v'',w''}^{>0} \subset \dot{u}G_{0}^{(J)}$.

Assume now that $\mathbb{K} = \mathbb{C}$ and $xB \in R_{v'',w''}^{>0}$. Let $p \in P$ and $g^{(J)} \in \dot{u}U_{-}^{(J)}\dot{u}^{-1}$ be such that $xp = g^{(J)}\dot{u}$. Then $\kappa_x xp = g_1^{(J)}\dot{u}$ for $g_1^{(J)} \in U_1^{(J)}$. By (3.6.4), $U_1^{(J)}\dot{u} \subset U\dot{u} \subset B\dot{u}B$. By Lemma 3.4.22(i), we have $p^{-1} \in B\dot{r}_w B$ for some $r_w \in W_J$. We get $\kappa_x x = g_1^{(J)}\dot{u} \cdot p^{-1} \in$ $B\dot{u}B \cdot B\dot{r}_w B \subset B\dot{u}\dot{r}_w B$ by (3.4.18). On the other hand, $\kappa_x \in U_-$ and $x \in B_-v''B$, so $\kappa_x x \in B_-v''B$. Therefore $\kappa_x xB \in \mathring{R}_{v'',ur_w}$.

We now show $r_w \ge r$. By (3.5.22), $x \in \dot{u}\dot{r}B_-B$, so by Lemma 3.6.8(iii), we have $\rho_{ur}^{\vee}(t) \cdot xB \to \dot{u}\dot{r}B$ as $t \to 0$ in G/B. Since $\dot{u}\dot{r} \in \dot{u}G_0^{(J)}$, κ is regular at $\dot{u}\dot{r}B$, and by Lemma 3.6.7, we have $\kappa_{\dot{u}\dot{r}} = 1$. Thus $\kappa_{\rho_{ur}^{\vee}(t)x}\rho_{ur}^{\vee}(t)xB \to \dot{u}\dot{r}B$ as $t \to 0$. By Lemma 3.6.8(ii), $\kappa_{\rho_{ur}^{\vee}(t)x}\rho_{ur}^{\vee}(t)xB = \rho_{ur}^{\vee}(t) \cdot \kappa_x xB$, which belongs to \mathring{R}_{v'',ur_w} for all $t \in \mathbb{C}^*$. We see that the closure of \mathring{R}_{v'',ur_w} contains $\dot{u}\dot{r}B$, thus $v'' \le ur \le ur_w$ by (3.4.14), so $r \le r_w$ by Lemma 3.4.4(ii).

Finally, we show $\kappa_x xB \in (G/B)_{\geq 0}$. First, clearly the map κ is defined over \mathbb{R} , thus $\kappa_x xB \in (G/B)_{\mathbb{R}}$. Consider the subset $R_{v'',[\tilde{u},w_0]}^{>0} := \bigsqcup_{w''\geq \tilde{u}} R_{v'',w''}^{>0} \subset (G/B)_{\geq 0}$. It contains $R_{v'',w_0}^{>0}$ as an open dense subset, and therefore $R_{v'',[\tilde{u},w_0]}^{>0}$ is connected. We have already shown that for any $x' \in R_{v'',[\tilde{u},w_0]}^{>0}$, $\kappa_{x'}x'B \in \mathring{R}_{v'',\tilde{u}}^{\mathbb{R}}$ (because we have $r_w \geq r = w_J$). Thus the image of the set $R_{v'',[\tilde{u},w_0]}^{>0}$ under the map $x' \mapsto \kappa_{x'}x'$ must lie inside a single connected component of $\mathring{R}_{v'',\tilde{u}}^{\mathbb{R}}$. However, if $x' \in R_{v'',\tilde{u}}^{>0} \subset R_{v'',[\tilde{u},w_0]}^{>0}$ then $\kappa_{x'} = 1$ by Lemma 3.6.7, so in this case $\kappa_{x'}x' \in R_{v'',\tilde{u}}^{>0}$. We conclude that the image of $R_{v'',[\tilde{u},w_0]}^{>0}$ is contained inside $R_{v'',\tilde{u}}^{>0} \subset (G/B)_{\geq 0}$. It follows by continuity that for arbitrary $v'' \leq ur \leq w''$ and $x \in R_{v'',w''}^{>0}$, we have $\kappa_x xB \in (G/B)_{\geq 0}$.

We will use the following consequence of Lemma 3.6.9(ii) in Section 3.9.11.

Corollary 3.6.10. (Assume $\mathbb{K} = \mathbb{C}$.) In the notation of Lemma 3.6.9(ii), we have $\kappa_x x P \in \Pi_{\bar{v}',u}^{>0}$ for $\bar{v}'' := v'' \triangleleft r_w^{-1}$.

Proof. Indeed, Lemma 3.6.9(ii) says that $\kappa_x x B \in R^{>0}_{v'',ur_w}$, so applying Corollary 3.4.18, we find that $\pi_J(\kappa_x x B) = \kappa_x x P \in \Pi^{>0}_{\overline{v}'',u}$.

3.6.3 **Proof via subtraction-free parametrizations**

In this section, we fix some set **t** of variables and assume $\mathbb{K} = \overline{\mathcal{F}}$. Also fix $u \in W^J$ and recall that $\tilde{u} = uw_J \in W^J_{\text{max}}$.

By Definition 3.4.23, the map κ is defined on $\dot{u}G_0^{(J)}$. By Lemma 3.6.9(i), we have $R_{v'',w''}^{\text{sf}} \subset \dot{u}G_0^{(J)}$ whenever $v'' \leq ur \leq w''$ for some $r \in W_J$. In particular, κ is defined on $U_{\text{sf}}^-(w'') \subset R_{\text{id},w''}^{\text{sf}}$ for all $w'' \geq \tilde{u}$.

Proposition 3.6.11. Let $q \in W$ be such that $\ell(\tilde{u}q) = \ell(\tilde{u}) + \ell(q)$. Then for $h \in U^-_{sf}(\tilde{u}q)$, we have $\kappa_h h \in U^-_{sf}(\tilde{u})$.

Proof. Write $h \in U_{sf}^{-}(\tilde{u}q) = U_{sf}^{-}(\tilde{u}) \cdot U_{sf}^{-}(q)$. Using (3.5.8), we find

$$h \in \dot{\tilde{u}} \cdot \dot{\tilde{u}}^{-1} \cdot U_{\mathrm{sf}}^{-}(\tilde{u}) \cdot U_{\mathrm{sf}}^{-}(q) \subset \dot{\tilde{u}} \cdot B_{-}^{\mathrm{sf}} \cdot U_{\mathrm{sf}}(\tilde{u}^{-1}) \cdot U_{\mathrm{sf}}^{-}(q).$$

By (3.5.7), $B_{-}^{sf} \cdot U_{sf}(\tilde{u}^{-1}) \cdot U_{sf}^{-}(q) = B_{-}^{sf} \cdot U_{sf}^{-}(q) \cdot U_{sf}(\tilde{u}^{-1}) \subset B_{-}^{sf} \cdot U_{sf}(\tilde{u}^{-1})$. Writing $B_{-}^{sf} \subset U_{-} \cdot T^{sf}$, we get

$$h \in \dot{\tilde{u}} \cdot U_{-} \cdot T^{\mathrm{sf}} \cdot U_{\mathrm{sf}}(\tilde{u}^{-1}) = T^{\mathrm{sf}} \cdot \dot{\tilde{u}} U_{-} \dot{\tilde{u}}^{-1} \cdot \dot{\tilde{u}} \cdot U_{\mathrm{sf}}(\tilde{u}^{-1}).$$

Applying (3.5.16), we find

$$h \in T^{\mathrm{sf}} \cdot \dot{\tilde{u}} U_{-} \dot{\tilde{u}}^{-1} \cdot T^{\mathrm{sf}} \cdot (U^{\diamond} \cap \dot{\tilde{u}} U_{-} \dot{\tilde{u}}^{-1}) \cdot U_{\mathrm{sf}}^{-} (\tilde{u}) \subset \dot{\tilde{u}} U_{-} \dot{\tilde{u}}^{-1} \cdot T^{\mathrm{sf}} \cdot U_{\mathrm{sf}}^{-} (\tilde{u})$$

Let $g \in \dot{\tilde{u}}U_-\dot{\tilde{u}}^{-1}$ be such that $h \in g \cdot T^{\mathrm{sf}} \cdot U_{\mathrm{sf}}^-(\tilde{u})$. Recall from (3.6.5) that $U_2^{(J)} = \dot{\tilde{u}}U_-\dot{\tilde{u}}^{-1} \cap U_-$. By Lemma 3.4.1(i), there exists $h' \in U_2^{(J)}$ such that $h'g \in \dot{\tilde{u}}U_-\dot{\tilde{u}}^{-1} \cap U$. Thus

$$h'h \in (\dot{\tilde{u}}U_{-}\dot{\tilde{u}}^{-1} \cap U) \cdot T^{\mathrm{sf}} \cdot U_{\mathrm{sf}}^{-}(\tilde{u}) \subset U \cdot T^{\mathrm{sf}} \cdot U_{\mathrm{sf}}^{-}(\tilde{u}).$$

But observe that both h and h' belong to U_- . Since the factorization of h'h as an element of $U \cdot T \cdot U_-$ is unique by Lemma 3.4.21(i), it follows that $h'h \in U_{\rm sf}^-(\tilde{u})$. By (3.4.20), $U_{\rm sf}^-(\tilde{u}) \subset B\dot{\tilde{u}}B$. By Lemma 3.6.7, $\kappa_{h'h} = 1$, so $\kappa_h = h'$, thus $\kappa_h h \in U_{\rm sf}^-(\tilde{u})$.

Corollary 3.6.12. For $q \in W$ such that $\ell(\tilde{u}q) = \ell(\tilde{u}) + \ell(q)$ and $v \leq \tilde{u}$, we have $R_{id,\tilde{u}q}^{sf} \subset G_{u,v}^{(J)}$. *Proof.* As we have already mentioned, Lemma 3.6.9(i) shows that $R_{id,\tilde{u}q}^{sf} \subset \dot{u}G_0^{(J)}$. Let $x \in R_{\mathrm{id},\tilde{u}q}^{\mathrm{sf}} = U_{\mathrm{sf}}^{-}(\tilde{u}q) \cdot B^{\mathrm{sf}}$, and let $b \in B^{\mathrm{sf}}$ and $h \in U_{\mathrm{sf}}^{-}(\tilde{u}q)$ be such that x = hb. By Lemma 3.6.2(iii), we have $\kappa_x = \kappa_h$. By Proposition 3.6.11, $\kappa_h h \in U_{\mathrm{sf}}^{-}(\tilde{u})$, therefore $\kappa_x x \in U_{\mathrm{sf}}^{-}(\tilde{u}) \cdot B^{\mathrm{sf}} = R_{\mathrm{id},\tilde{u}}^{\mathrm{sf}}$. By (3.5.20), we get $\kappa_x x \in \dot{v}B_-B$.

Corollary 3.6.12 shows that the map $\zeta_{u,v}^{(J)}$ is defined on the whole $R_{\mathrm{id},\tilde{u}q}^{\mathrm{sf}}$.

Lemma 3.6.13. Suppose that $u_0 \in W^J$ and $v_0 \leq \tilde{u}_0 := u_0 w_J$. Let $h \in U^-_{sf}(\tilde{u}_0)$, and let $b_u, b_v \in U$ be such that $\dot{\tilde{u}}_0^{-1}h \in B_- \cdot b_u$ and $\dot{v}_0^{-1}h \in B_- \cdot b_v$. Then $[b_u b_v^{-1}]_J \in U_{sf}(r)$ for some $r \in W_J$.

Proof. First, recall from Lemma 3.4.21(i) and (3.5.11) that b_u and b_v are uniquely defined and satisfy $b_u \in U_{\rm sf}(\tilde{u}_0^{-1}), b_v \in U_{\rm sf}(v_0^{-1})$. Let $h = h_1h_2$ for $h_1 \in U_{\rm sf}^-(u_0)$ and $h_2 \in U_{\rm sf}^-(w_J)$. Our first goal is to show that $[b_u]_J \in U_J$ satisfies (and is uniquely defined by) $\dot{w}_J^{-1}h_2 \in B_- \cdot [b_u]_J$. Letting $b'_u \in U_J$ be uniquely defined by $\dot{w}_J^{-1}h_2 \in B_- \cdot b'_u$, we thus need to show that $[b_u]_J = b'_u$.

By (3.5.9), there exists $d \in U_{\rm sf}(u_0^{-1})$ such that

$$\dot{w}_J^{-1} \dot{u}_0^{-1} h_1 \in B^{\mathrm{sf}}_- \cdot \dot{w}_J^{-1} \cdot d.$$

Since $d \in U$, we can use Lemma 3.4.22(iii) to factorize it as $d = [d]_J [d]_+^{(J)}$. Since $h_2 \in U_J^- \subset L_J$, Lemma 3.4.22(ii) shows that there exists $d' \in U^{(J)}$ such that $[d]_+^{(J)}h_2 = h_2d'$. Since $[d]_J \in U_J$ by Lemma 3.4.22(iv), (3.4.21) shows that $\dot{w}_J^{-1}[d]_J \in U_- \dot{w}_J^{-1}$. Combining the pieces together, we get

$$\dot{\tilde{u}}_0^{-1}h = \dot{w}_J^{-1}\dot{u}_0^{-1}h_1h_2 \in B_-^{\mathrm{sf}} \cdot \dot{w}_J^{-1} \cdot [d]_J[d]_+^{(J)} \cdot h_2 \subset B_- \cdot \dot{w}_J^{-1}h_2d' = B_- \cdot b'_ud'.$$

On the other hand, $\dot{\tilde{u}}_0^{-1}h \in B_- \cdot b_u$, so $b_u = b'_u d'$, where $b'_u \in U_J$ and $d' \in U^{(J)}$. It follows that $[b_u]_J = b'_u$, and thus we have shown that $\dot{w}_J^{-1}h_2 \in B_- \cdot [b_u]_J$.

We now prove the result by induction on $\ell(u_0)$. When $\ell(u_0) = 0$, we have $\tilde{u}_0 = w_J$ and $v_0 \in W_J$. Thus there exists $v_1 \in W_J$ such that $w_J = v_0 \cdot v_1$ with $\ell(w_J) = \ell(v_0) + \ell(v_1)$. We have $b_u, b_v \in U_J$, so $[b_u b_v^{-1}]_J = b_u b_v^{-1}$ by Lemma 3.4.22(iv). By (3.5.10), there exist $b_0 \in U_{\rm sf}(v_0^{-1})$ and $b_1 \in U_{\rm sf}(v_1^{-1})$ such that

$$\dot{v}_0^{-1}h \in B_-^{\mathrm{sf}} \cdot b_0, \quad \dot{w}_J^{-1}h \in B_-^{\mathrm{sf}} \cdot b_1b_0.$$

In particular, we have $b_v = b_0$ and $b_u = b_1 b_0$. Thus $[b_u b_v^{-1}]_J = b_1 \in U_{sf}(v_1^{-1})$, and we are done with the base case.

Assume $\ell(u_0) > 0$, and let $i \in I$ be such that $u_1 := s_i u_0 < u_0$. By Lemma 3.4.4(i), $u_1 \in W^J$, so denote $\tilde{u}_1 := u_1 w_J \in W^J_{\text{max}}$. Let $h \in U^-_{\text{sf}}(\tilde{u}_0)$ be factorized as $h = h_i h'_1 h_2$ for $h_i = y_i(t) \in U^-_{\text{sf}}(s_i), h'_1 \in U^-_{\text{sf}}(u_1)$, and $h_2 \in U^-_{\text{sf}}(w_J)$.

Suppose that $s_i v_0 > v_0$, in which case we have $v_0 \leq \tilde{u}_1$. Let $h' := h'_1 h_2$ and $b'_u \in U$ be defined by $\dot{\tilde{u}}_1^{-1}h' \in B_- \cdot b'_u$. Since $s_i v_0 > v_0$, we see that $\dot{v}_0^{-1}h_i \in B_- \cdot \dot{v}_0^{-1}$, so $\dot{v}_0^{-1}h' \in B_- \cdot \dot{v}_0^{-1}h = B_- \cdot b_v$. By the induction hypothesis applied to $v_0 \leq \tilde{u}_1$ and $h' \in U_{sf}^-(\tilde{u}_1)$, we have $[b'_u b_v^{-1}]_J \in U_{sf}(r)$ for some $r \in W_J$. On the other hand, we have shown above that $[b_u]_J$ satisfies $\dot{w}_J^{-1}h_2 \in B_- \cdot [b_u]_J$. But since $h' = h'_1h_2$ for $h_2 \in U_{sf}^-(w_J)$, we get that $[b'_u]_J$ satisfies $\dot{w}_J^{-1}h_2 \in B_- \cdot [b'_u]_J$, thus $[b_u]_J = [b'_u]_J$. Therefore using Lemma 3.4.22(iv), we get

$$[b_u b_v^{-1}]_J = [b_u]_J [b_v^{-1}]_J = [b'_u]_J [b_v^{-1}]_J = [b'_u b_v^{-1}]_J \in U_{\rm sf}(r),$$

finishing the induction step in the case $s_i v_0 > v_0$.

Suppose now that $v_1 := s_i v_0 < v_0$. Let $h = h_i h'_1 h_2 \in U_{sf}^-(\tilde{u}_0)$ be as above. By (3.5.8), $\dot{s}_i^{-1} h_i \in B_-^{sf} \cdot U_{sf}(s_i)$, so let $d_i \in U_{sf}(s_i)$ be such that $\dot{s}_i^{-1} h_i \in B_-^{sf} \cdot d_i$. By (3.5.7), $U_{sf}(s_i) \cdot U_{sf}^-(\tilde{u}_1) = U_{sf}^-(\tilde{u}_1) \cdot U_{sf}(s_i)$, so let $b_i \in U_{sf}(s_i)$ and $h' \in U_{sf}^-(\tilde{u}_1)$ be such that $d_i h'_1 h_2 = h' b_i$. We check using (3.5.9) that

$$\dot{\tilde{u}}_{0}^{-1}h \in B_{-}^{\mathrm{sf}} \cdot \dot{\tilde{u}}_{1}^{-1}h' \cdot b_{i}, \quad \dot{v}_{0}^{-1}h \in B_{-}^{\mathrm{sf}} \cdot \dot{v}_{1}^{-1}h' \cdot b_{i}.$$
(3.6.6)

Let $b'_u, b'_v \in U$ be defined by $\dot{\tilde{u}}_1^{-1}h' \in B_- \cdot b'_u$ and $\dot{v}_1^{-1}h' \in B_- \cdot b'_v$. Then by the induction hypothesis applied to $v_1 \leq \tilde{u}_1$ and $h' \in U_{\rm sf}^-(\tilde{u}_1)$, we find $[b'_u b'_v^{-1}]_J \in U_{\rm sf}(r)$ for some $r \in W_J$. But it is clear from (3.6.6) that $b_u = b'_u b_i$ and $b_v = b'_v b_i$. Therefore $[b_u b_v^{-1}]_J \in U_{\rm sf}(r)$. \Box

Theorem 3.6.14. For all $v \leq \tilde{u}$, $w \in W^J$, $i \in I$, and $x \in R_{id,w_0}^{sf}$, we have

$$\Delta_{i}^{\pm}(\zeta_{u,v}^{(J)}(x)\dot{w}^{-1}) \in \mathcal{F}_{\rm sf}.$$
(3.6.7)

Proof. Let $q \in W$ be such that $w_0 = \tilde{u}q$, thus $\ell(\tilde{u}q) = \ell(\tilde{u}) + \ell(q)$. Let $x \in R^{\text{sf}}_{\text{id},w_0} = U^-_{\text{sf}}(w_0) \cdot B^{\text{sf}}$ be written as $x = h \cdot b$, where $h = h_1 h_2 h_3 \in U^-_{\text{sf}}(w_0)$ for $h_1 \in U^-_{\text{sf}}(u)$, $h_2 \in U^-_{\text{sf}}(w_0)$

 $U_{\mathrm{sf}}^{-}(w_J), h_3 \in U_{\mathrm{sf}}^{-}(q)$, and $b \in B^{\mathrm{sf}}$. By (3.5.10), there exist $b_1 \in U_{\mathrm{sf}}(u^{-1}), b_2 \in U_{\mathrm{sf}}(w_J)$, and $b_3 \in U_{\mathrm{sf}}(q^{-1})$ such that

$$\dot{u}^{-1}h \in B^{\mathrm{sf}}_{-} \cdot b_1, \quad \dot{\tilde{u}}^{-1}h \in B^{\mathrm{sf}}_{-} \cdot b_2 b_1, \quad \dot{w}^{-1}_0 h \in B^{\mathrm{sf}}_{-} \cdot b_3 b_2 b_1.$$
(3.6.8)

Let $x' := hb_1^{-1}$. We have $x' = xb^{-1}b_1^{-1} \in xB \subset xP$, therefore $x' \in G_{u,v}^{(J)}$ and $\zeta_{u,v}^{(J)}(x') = \zeta_{u,v}^{(J)}(x)$ by Lemma 3.6.2(iv). On the other hand, by (3.6.8), $x' \in \dot{u}B_-^{\mathrm{sf}} \subset \dot{u}P_-$, so Lemma 3.6.3(ii) implies $\zeta_{u,v}^{(J)}(x') = x'\eta(x')^{-1}$.

Let us now compute $\eta(x') = [\dot{v}^{-1}\kappa_{x'}x']_J$. By Lemma 3.6.2(iii), $\kappa_x = \kappa_{x'} = \kappa_h$, and by Proposition 3.6.11, $\kappa_h h \in U_{sf}^-(\tilde{u})$. Thus by (3.5.11), $\dot{v}^{-1}\kappa_h h \in B_-^{sf} \cdot U_{sf}(v^{-1})$, so let $d_0 \in B_-^{sf}$ and $b_0 \in U_{sf}(v^{-1})$ be such that $\dot{v}^{-1}\kappa_h h = d_0b_0$. By definition, $\kappa_h \in U_2^{(J)}$, so by (3.6.5), $\dot{\tilde{u}}^{-1}\kappa_h\dot{\tilde{u}} \in U_-$, and therefore using (3.6.8) we find

$$\dot{\tilde{u}}^{-1}\kappa_h h = \dot{\tilde{u}}^{-1}\kappa_h \dot{\tilde{u}} \cdot \dot{\tilde{u}}^{-1} h \in U_- \cdot \dot{\tilde{u}}^{-1} h \subset B_- \cdot b_2 b_1.$$

We can now apply Lemma 3.6.13: we have $v \leq \tilde{u}$, $\kappa_h h \in U_{sf}^-(\tilde{u})$, $\dot{\tilde{u}}^{-1}\kappa_h h \in B_- \cdot b_2 b_1$, and $\dot{v}^{-1}\kappa_h h \in B_- \cdot b_0$. Let $b_u := b_2 b_1 \in U$ and $b_v := b_0 \in U$. By Lemma 3.6.13, $[b_u b_v^{-1}]_J = [b_2 b_1 b_0^{-1}]_J \in U_{sf}(r)$ for some $r \in W_J$.

Recall that $\dot{v}^{-1}\kappa_h h = d_0 b_0$ for $d_0 \in B^{\mathrm{sf}}_-$ and $b_0 \in U_{\mathrm{sf}}(v^{-1})$. Thus

$$\eta(x') = [\dot{v}^{-1}\kappa_{x'}x']_J = [\dot{v}^{-1}\kappa_h x']_J = [\dot{v}^{-1}\kappa_h h b_1^{-1}]_J = [d_0 b_0 b_1^{-1}]_J.$$

By Lemma 3.4.22(iii), we get $[d_0b_0b_1^{-1}]_J = [d_0]_J[b_0b_1^{-1}]_J$. Thus

$$\zeta_{u,v}^{(J)}(x) = \zeta_{u,v}^{(J)}(x') = x'\eta(x')^{-1} = x'[b_0b_1^{-1}]_J^{-1}[d_0]_J^{-1}.$$

By (3.6.8), we have $\dot{w}_0^{-1}x' \in B_-^{\text{sf}} \cdot b_3b_2$, so $x' \in B^{\text{sf}}\dot{w}_0b_3b_2$. Using Lemma 3.4.22(iv), we thus get

$$\zeta_{u,v}^{(J)}(x) = x'[b_0b_1^{-1}]_J^{-1}[d_0]_J^{-1} \in B^{\mathrm{sf}} \cdot \dot{w}_0b_3[b_2b_1b_0^{-1}]_J[d_0]_J^{-1}$$

We are interested in the element $\zeta_{u,v}^{(J)}(x)\dot{w}^{-1}$. We know that $d_0 \in B_-^{\mathrm{sf}}$, thus $[d_0]_J \in T^{\mathrm{sf}}U_J^-$,

and by Lemma 3.4.5, $\dot{w}[d_0]_J \dot{w}^{-1} \in T^{\mathrm{sf}} \cdot U_-$. Hence

$$\zeta_{u,v}^{(J)}(x)\dot{w}^{-1} \in B^{\mathrm{sf}} \cdot \dot{w}_0 b_3 [b_2 b_1 b_0^{-1}]_J [d_0]_J^{-1} \dot{w}^{-1} \subset B^{\mathrm{sf}} \cdot \dot{w}_0 b_3 [b_2 b_1 b_0^{-1}]_J \dot{w}^{-1} \cdot T^{\mathrm{sf}} \cdot U_-.$$

In particular, $\Delta_i^{\pm}(\zeta_{u,v}^{(J)}(x)\dot{w}^{-1}) \in \mathcal{F}_{sf}$ if and only if $\Delta_i^{\pm}(\dot{w}_0b_3[b_2b_1b_0^{-1}]_J\dot{w}^{-1}) \in \mathcal{F}_{sf}$. Recall that $b_3 \in U_{sf}(q^{-1})$ and $[b_2b_1b_0^{-1}]_J \in U_{sf}(r)$ for some $r \in W_J$. Thus $b_3[b_2b_1b_0^{-1}]_J \in U_{sf}(q^{-1}r)$, so we are done by Proposition 3.5.13.

Proof of Theorem 3.6.4. Our strategy will be very similar to the one we used in the proof of Corollary 3.5.12.

Fix $(u, u) \leq (v, w) \leq (v', w') \in Q_J$. Let $\mathbf{t} = (\mathbf{t}_1, \mathbf{t}_2, \mathbf{t}_3)$ with $|\mathbf{t}_1| = \ell(v')$, $|\mathbf{t}_2| = \ell(w_0) - \ell(w')$, and $|\mathbf{t}_3| := \ell(w') - \ell(v')$, and assume $\mathbb{K} = \overline{\mathcal{F}}$. Choose reduced words \mathbf{i} for v'^{-1} and \mathbf{j} for $w_0 w'^{-1}$, and let $(\mathbf{v}', \mathbf{w}') \in \operatorname{Red}(v', w')$. Suppose that $x \in \mathbf{g}_{\mathbf{v}', \mathbf{w}'}(\mathbf{t}_3) \cdot B^{\mathrm{sf}}$. Then

$$g(\mathbf{t}_1, \mathbf{t}_2, \mathbf{t}_3) := \mathbf{x}_{\mathbf{i}}(\mathbf{t}_1) \cdot \mathbf{y}_{\mathbf{j}}(\mathbf{t}_2) \cdot \mathbf{g}_{\mathbf{v}', \mathbf{w}'}(\mathbf{t}_3) \in U_{\mathrm{sf}}(v'^{-1}) \cdot U_{\mathrm{sf}}^{-}(w_0 w'^{-1}) \cdot R_{v', w'}^{\mathrm{sf}}(w_0 w'^{-1}) \cdot R_{v', w'}^{\mathrm{sf}}(w_0 w'^{-1}) \cdot R_{v', w'}^{\mathrm{sf}}(\mathbf{t}_3)$$

By Lemma 3.5.8, we have $g(\mathbf{t}_1, \mathbf{t}_2, \mathbf{t}_3) \in R_{\mathrm{id},w_0}^{\mathrm{sf}}$. Thus by Theorem 3.6.14, for all $i \in I$ we have $\Delta_i^{\pm}(\zeta_{u,v}^{(J)}(g(\mathbf{t}_1, \mathbf{t}_2, \mathbf{t}_3))\dot{w}^{-1}) \in \mathcal{F}_{\mathrm{sf}}$. Denote by $f(\mathbf{t}_1, \mathbf{t}_2, \mathbf{t}_3) := \Delta_i^{\pm}(\zeta_{u,v}^{(J)}(g(\mathbf{t}_1, \mathbf{t}_2, \mathbf{t}_3))\dot{w}^{-1})$ the corresponding subtraction-free rational expression, which yields a continuous function $\mathbb{R}_{\geq 0}^{|\mathbf{t}_1|} \times \mathbb{R}_{>0}^{|\mathbf{t}_2|} \times \mathbb{R}_{>0}^{|\mathbf{t}_3|} \to \mathbb{R}_{\geq 0}$. We claim that f extends to a continuous function $\mathbb{R}_{\geq 0}^{|\mathbf{t}_1|} \times \mathbb{R}_{\geq 0}^{|\mathbf{t}_2|} \times \mathbb{R}_{>0}^{|\mathbf{t}_3|} \to \mathbb{R}_{\geq 0}$. Indeed, fix some $(\mathbf{t}_1', \mathbf{t}_2', \mathbf{t}_3') \in \mathbb{R}_{\geq 0}^{|\mathbf{t}_1|} \times \mathbb{R}_{\geq 0}^{|\mathbf{t}_2|} \times \mathbb{R}_{>0}^{|\mathbf{t}_3|}$ and let $\mathbb{K} = \mathbb{C}$. The element $x' := g(\mathbf{t}_1', \mathbf{t}_2', \mathbf{t}_3')$ (obtained by evaluating at $(\mathbf{t}_1', \mathbf{t}_2', \mathbf{t}_3')$, see Section 3.5.4) belongs to $G_{\geq 0} \cdot \mathbb{R}_{v',w'}^{>0}$, and by Lemma 3.4.17 there exist $v'', w'' \in W$ such that $v'' \leq v' \leq w' \leq w''$ and $x' \in \mathbb{R}_{v'',w''}^{>0}$. Recall from Lemma 3.4.8(ii) that we have

$$v'' \le v' \le vr' \le ur \le wr' \le w'$$

for some $r', r \in W_J$ such that $\ell(vr') = \ell(v) + \ell(r')$. In particular, by Lemma 3.6.9(ii), $x' \in \dot{u}G_0^{(J)}$ and $\kappa_{x'}x' \in R_{v'',ur_w}^{>0}$ for some $r_w \in W_J$ such that $r_w \ge r$. By Corollary 3.5.12, $\kappa_{x'}x' \in \dot{v}\dot{r}'B_-B \subset \dot{v}G_0^{(J)}$, which shows that $x' \in G_{u,v}^{(J)}$. The map $\zeta_{u,v}^{(J)}$ is therefore regular at x' by Lemma 3.6.2(ii). The map Δ_i^{\pm} is regular on G by Lemma 3.4.21(ii), so in particular it is regular at $\zeta_{u,v}^{(J)}(x')\dot{w}^{-1}$. We have shown that the map $x'' \mapsto \Delta_i^{\pm}(\zeta_{u,v}^{(J)}(x'')\dot{w}^{-1})$ is regular at $x' = g(\mathbf{t}'_1, \mathbf{t}'_2, \mathbf{t}'_3)$ for all $(\mathbf{t}'_1, \mathbf{t}'_2, \mathbf{t}'_3) \in \mathbb{R}_{\geq 0}^{|\mathbf{t}_1|} \times \mathbb{R}_{\geq 0}^{|\mathbf{t}_2|} \times \mathbb{R}_{>0}^{|\mathbf{t}_3|}$. Thus the map $f(\mathbf{t}_1, \mathbf{t}_2, \mathbf{t}_3)$ extends to a continuous function $\mathbb{R}_{\geq 0}^{|\mathbf{t}_1|} \times \mathbb{R}_{\geq 0}^{|\mathbf{t}_2|} \times \mathbb{R}_{>0}^{|\mathbf{t}_3|} \to \mathbb{R}_{\geq 0}$. By Lemma 3.5.9, we find that $f(0, 0, \mathbf{t}_3) := \lim_{\mathbf{t}_1, \mathbf{t}_2 \to 0} f(\mathbf{t}_1, \mathbf{t}_2, \mathbf{t}_3)$ belongs to $\mathcal{F}_{\mathrm{sf}}$, i.e., can be represented by a subtraction-free rational expression in the variables \mathbf{t}_3 . On the other hand, it is clear that $f(0, 0, \mathbf{t}_3) = \Delta_i^{\pm}(\zeta_{u,v}^{(J)}(\mathbf{g}_{\mathbf{v}',\mathbf{w}'}(\mathbf{t}_3))\dot{w}^{-1})$.

Our next goal is to show that $f(0, 0, \mathbf{t}_3) \in \mathcal{F}_{\mathrm{sf}}^*$. Indeed, suppose otherwise that $f(0, 0, \mathbf{t}_3) = 0$ (as an element of \mathcal{F}). By Lemma 3.6.2(iv), $\zeta_{u,v}^{(J)}$ descends to a regular map $G_{u,v}^{(J)}/P \to G$ (still assuming $\mathbb{K} = \mathbb{C}$). Therefore the map $\bar{f}: G_{u,v}^{(J)}/P \to \mathbb{C}$ sending $x'P \mapsto \Delta_i^{\pm}(\zeta_{u,v}^{(J)}(x')\dot{w}^{-1})$ is also regular. If $f(0, 0, \mathbf{t}_3) = 0$ then \bar{f} vanishes on $\pi_J(R_{v',w'}^{>0}) = \prod_{v',w'}^{>0}$, and therefore it vanishes on its Zariski closure, which is $\prod_{v',w'}$. We have $\pi_J(R_{v,w}^{>0}) = \prod_{v,w}^{>0} \subset \prod_{v',w'}$, thus $\bar{f}(x) = 0$ for any $x \in G_{u,v}^{(J)}$ such that $xB \in R_{v,w}^{>0}$. Let us show that this leads to a contradiction.

Let $x \in G$ be such that $xB \in R_{v,w}^{>0}$. By (3.4.27), there exists $x' \in xP$ such that $x'B \in R_{vr',wr'}^{>0}$. By Lemma 3.6.9(ii), we have $x' \in \dot{u}G_0^{(J)}$, and thus $x \in \dot{u}G_0^{(J)}$. Having $xB \in R_{v,w}^{>0}$ implies $x \in B_-\dot{v}B \cap B\dot{w}B$. Since $\kappa_x \in U_2^{(J)} \subset U_-$, we have $\kappa_x x \in B_-\dot{v}B$. By (3.4.34), $B_-\dot{v}B = (\dot{v}U_- \cap U_-\dot{v})B \subset \dot{v}B_-B$, thus $\kappa_x x \in \dot{v}B_-B$, and therefore $x \in G_{u,v}^{(J)}$. Moreover, $\dot{v}^{-1}\kappa_x x \in B_-B$, thus $\eta(x) = [\dot{v}^{-1}\kappa_x x]_J \in U_J^-TU_J$. On the other hand, $\pi_{\dot{u}P_-}(x) \in xU^{(J)} \subset xB \subset B\dot{w}B$, see Definition 3.6.1. Thus

$$\zeta_{u,v}^{(J)}(x) = \pi_{\dot{u}P_{-}}(x)\eta(x)^{-1} \in B\dot{w}B \cdot U_{J}TU_{J}^{-} = B\dot{w}B \cdot U_{J}^{-}.$$

Recall that because $w \in W^J$, we have $U_J^- \dot{w}^{-1} \subset \dot{w}^{-1} U_-$ by Lemma 3.4.5. Hence

$$\zeta_{u,v}^{(J)}(x)\dot{w}^{-1} \in B\dot{w}B \cdot U_J^- \cdot \dot{w}^{-1} \subset B\dot{w}B\dot{w}^{-1}B_-.$$

By (3.4.34) (after taking inverses of both sides), $B\dot{w}B = B \cdot (U_-\dot{w} \cap \dot{w}U)$, so

$$\zeta_{u,v}^{(J)}(x)\dot{w}^{-1} \in B \cdot (U_- \cap \dot{w}U\dot{w}^{-1}) \cdot B_- \subset B \cdot B_-.$$

In particular, $\Delta_i^{\pm}(\zeta_{u,v}^{(J)}(x)\dot{w}^{-1}) \neq 0$ for all $i \in I$. This gives a contradiction, showing $f(0,0,\mathbf{t}_3) \in \mathcal{F}_{\mathrm{sf}}^*$. But then evaluating f at any $\mathbf{t}_3' \in \mathbb{R}_{>0}^{\ell(w')-\ell(v')}$ yields a positive real number. We have shown that $\Delta_i^{\pm}(\zeta_{u,v}^{(J)}(x)\dot{w}^{-1}) \neq 0$ for all $x \in G$ such that $xB \in R_{v',w'}^{>0}$. We are done

3.7 Affine Bruhat atlas for the projected Richardson stratification

In this section, we embed the stratification (3.4.23) of G/P inside the affine Richardson stratification of the affine flag variety. Throughout, we work over $\mathbb{K} = \mathbb{C}$.

3.7.1 Loop groups and affine flag varieties

Recall that G is a simple and simply connected algebraic group. Let $\mathcal{A} := \mathbb{C}[z, z^{-1}]$ and $\mathcal{A}_+, \mathcal{A}_- \subset \mathcal{A}$ denote the subrings given by $\mathcal{A}_+ := \mathbb{C}[z], \mathcal{A}_- := \mathbb{C}[z^{-1}]$. Then we have ring homomorphisms $\bar{\operatorname{ev}}_0 : \mathcal{A}_+ \to \mathbb{C}$ (resp., $\bar{\operatorname{ev}}_\infty : \mathcal{A}_- \to \mathbb{C}$), sending a polynomial in z (resp., in z^{-1}) to its constant term. Let $\mathcal{G} := G(\mathcal{A})$ denote the polynomial loop group of G.

Remark 3.7.1. The group \mathcal{G} is closely related to the *(minimal) affine Kac–Moody group* \mathcal{G}^{\min} associated to G, introduced by Kac–Peterson [KP83, PK83]. Below we state many standard results on \mathcal{G} without proof. We refer the reader unfamiliar with Kac–Moody groups to Section 3.A, where we give some background and explain how to derive these statements from Kumar's book [Kum02].

We introduce opposite Iwahori subgroups

$$\mathcal{B} := \{ g(z) \in G(\mathcal{A}_{+}) \mid \bar{\text{ev}}_{0}(g) \in B \}, \qquad \mathcal{B}_{-} := \{ g(z^{-1}) \in G(\mathcal{A}_{-}) \mid \bar{\text{ev}}_{\infty}(g) \in B_{-} \}$$

of \mathcal{G} , and denote by

$$\mathcal{U} := \{ g(z) \in G(\mathcal{A}_{+}) \mid \bar{\operatorname{ev}}_{0}(g) \in U \}, \qquad \mathcal{U}_{-} := \{ g(z^{-1}) \in G(\mathcal{A}_{-}) \mid \bar{\operatorname{ev}}_{\infty}(g) \in U_{-} \}$$

their unipotent radicals. There exists a tautological embedding $G \hookrightarrow \mathcal{G}$, and we treat G as a subset of \mathcal{G} .

We let $\mathcal{T} := \mathbb{C}^* \times T \subset \mathbb{C}^* \ltimes G$ be the affine torus, where \mathbb{C}^* acts on \mathcal{G} via loop rotation, see Section 3.8.2. The *affine root system* Δ of \mathcal{G} is the subset of $X(\mathcal{T}) := \operatorname{Hom}(\mathcal{T}, \mathbb{C}^*) \cong$ $X(T) \oplus \mathbb{Z}\delta$ given by

$$\Delta = \Delta_{\rm re} \sqcup \Delta_{\rm im}, \quad \text{where} \quad \Delta_{\rm re} := \{\beta + j\delta \mid \beta \in \Phi, \ j \in \mathbb{Z}\}, \quad \Delta_{\rm im} := \{j\delta \mid j \in \mathbb{Z} \setminus \{0\}\}$$

are the real and imaginary roots, and the set of *positive roots* $\Delta^+ \subset \Delta$ has the form

$$\Delta^{+} = \{j\delta \mid j > 0\} \sqcup \{\beta + j\delta \mid \beta \in \Phi, \ j > 0\} \sqcup \{\beta \mid \beta \in \Phi^{+}\}.$$

$$(3.7.1)$$

We let $\Delta_{\rm re}^+ := \Delta^+ \cap \Delta_{\rm re}$ and $\Delta_{\rm re}^- := \Delta^- \cap \Delta_{\rm re}$. For each $\alpha \in \Delta_{\rm re}^+$ (resp., $\alpha \in \Delta_{\rm re}^-$), we have a one-parameter subgroup $\mathcal{U}_{\alpha} \subset \mathcal{U}$ (resp., $\mathcal{U}_{\alpha} \subset \mathcal{U}_-$). The group \mathcal{U} (resp., \mathcal{U}_-) is generated by $\{\mathcal{U}_{\alpha}\}_{\alpha \in \Delta_{\rm re}^+}$ (resp., $\{\mathcal{U}_{\alpha}\}_{\alpha \in \Delta_{\rm re}^-}$), and for each $\alpha \in \Delta_{\rm re}$, we fix a group isomorphism $x_{\alpha} : \mathbb{C} \xrightarrow{\sim} \mathcal{U}_{\alpha}$.

Let $Q_{\Phi}^{\vee} := \bigoplus_{i \in I} \mathbb{Z} \alpha_i^{\vee}$ denote the coroot lattice of Φ . The affine Weyl group $\tilde{W} = W \ltimes Q_{\Phi}^{\vee}$ is a semidirect product of W and Q_{Φ}^{\vee} , i.e., as a set we have $\tilde{W} = W \ltimes Q_{\Phi}^{\vee}$, and the product rule is given by $(w_1, \lambda_1) \cdot (w_2, \lambda_2) := (w_1 w_2, \lambda_1 + w_1 \lambda_2)$. For $\lambda \in Q_{\Phi}^{\vee}$, we denote the element $(\mathrm{id}, \lambda) \in \tilde{W}$ by τ_{λ} . The group \tilde{W} is isomorphic to $N_{\mathbb{C}^* \ltimes \mathcal{G}}(\mathcal{T})/\mathcal{T}$, and for $f \in \tilde{W}$, we choose a representative $\dot{f} \in \mathcal{G}$ of f in $N_{\mathbb{C}^* \ltimes \mathcal{G}}(\mathcal{T})$, with the assumption that for $w \in W$, the representative $\dot{w} \in G \subset \mathcal{G}$ is given by (3.4.1). Thus \tilde{W} is a Coxeter group with generators $s_0 \sqcup \{s_i\}_{i \in I}$, length function $\ell : \tilde{W} \to \mathbb{Z}_{\geq 0}$, and affine Bruhat order \leq . The group \tilde{W} acts on Δ , and for $\alpha \in \Phi$, $\beta \in \Delta_{\mathrm{re}}$, $\lambda \in Q_{\Phi}^{\vee}$, and $w \in W$, we have

$$w\tau_{\lambda}w^{-1} = \tau_{w\lambda}, \quad \tau_{\lambda}\alpha = \alpha + \langle \lambda, \alpha \rangle \delta, \quad \tau_{\lambda}\delta = \delta, \quad \dot{\tau}_{\lambda}\mathcal{U}_{\beta}\dot{\tau}_{\lambda}^{-1} = \mathcal{U}_{\tau_{\lambda}\beta}.$$
 (3.7.2)

Let \mathcal{G}/\mathcal{B} denote the affine flag variety of G. This is an ind-variety that is isomorphic to the flag variety of the corresponding affine Kac-Moody group \mathcal{G}^{\min} , see Section 3.A.4. For each $h, f \in \tilde{W}$ we have Schubert cells $\mathring{\mathcal{X}}^f := \mathcal{B}_f \mathcal{B}/\mathcal{B}$ and opposite Schubert cells $\mathring{\mathcal{X}}_h := \mathcal{B}_- \dot{h} \mathcal{B}/\mathcal{B}$. If $h \not\leq f \in \tilde{W}$ then $\mathring{\mathcal{X}}^f \cap \mathring{\mathcal{X}}_h = \emptyset$. For $h \leq f$, we denote $\mathring{\mathcal{R}}_h^f := \mathring{\mathcal{X}}_h \cap \mathring{\mathcal{X}}^f$. For all $g \in \tilde{W}$, we have

$$\mathring{\mathcal{X}}^{g} = \bigsqcup_{h \le g} \mathring{\mathcal{R}}^{g}_{h}, \quad \mathring{\mathcal{X}}_{g} = \bigsqcup_{g \le f} \mathring{\mathcal{R}}^{f}_{g}, \quad \mathcal{X}^{g} := \bigsqcup_{h \le g} \mathring{\mathcal{X}}^{h}, \quad \mathcal{X}_{g} := \bigsqcup_{g \le f} \mathring{\mathcal{X}}_{f}. \tag{3.7.3}$$

For $g \in \tilde{W}$, let

$$\mathcal{C}_g := \dot{g}\mathcal{B}_-\mathcal{B}/\mathcal{B}, \quad \mathcal{U}_1(g) := \dot{g}\mathcal{U}_-\dot{g}^{-1}\cap\mathcal{U}, \quad \text{and} \quad \mathcal{U}_2(g) := \dot{g}\mathcal{U}_-\dot{g}^{-1}\cap\mathcal{U}_-.$$
 (3.7.4)

As we explain in Section 3.A.5, the map $x \mapsto x \dot{g} \mathcal{B}$ gives biregular isomorphisms

$$\dot{g}\mathcal{U}_{-}\dot{g}^{-1} \xrightarrow{\sim} \mathcal{C}_{g}, \quad \mathcal{U}_{1}(g) \xrightarrow{\sim} \mathring{\mathcal{X}}^{g}, \quad \mathcal{U}_{2}(g) \xrightarrow{\sim} \mathring{\mathcal{X}}_{g}.$$
 (3.7.5)

Let $\mathcal{U}^{(I)} \subset \mathcal{U}$ be the subgroup generated by $\{\mathcal{U}_{\alpha}\}_{\alpha \in \Delta_{\mathrm{re}}^+ \setminus \Phi^+}$. Similarly, let $\mathcal{U}_{-}^{(I)} \subset \mathcal{U}_{-}$ be the subgroup generated by $\{\mathcal{U}_{\alpha}\}_{\alpha \in \Delta_{\mathrm{re}}^- \setminus \Phi^-}$. For $x \in G \subset \mathcal{G}$, we have

$$x \cdot \mathcal{U}^{(I)} \cdot x^{-1} = \mathcal{U}^{(I)}, \quad x \cdot \mathcal{U}^{(I)}_{-} \cdot x^{-1} = \mathcal{U}^{(I)}_{-}.$$
 (3.7.6)

3.7.2 Combinatorial Bruhat atlas for G/P

We fix an element $\lambda \in Q_{\Phi}^{\vee}$ such that $\langle \lambda, \alpha_i \rangle = 0$ for $i \in J$ and $\langle \lambda, \alpha_i \rangle \in \mathbb{Z}_{<0}$ for $i \in I \setminus J$. Thus λ is anti-dominant and the stabilizer of λ in W is equal to W_J . Following [HL15], define a map

$$\psi: Q_J \to \tilde{W}, \quad (v, w) \mapsto v\tau_\lambda w^{-1}.$$
 (3.7.7)

By [HL15, Thm. 2.2], the map ψ gives an order-reversing bijection between Q_J and a subposet of \tilde{W} . More precisely, let $\tau_{\lambda}^J := \tau_{\lambda} (w^J)^{-1}$, and recall from (3.7.2) that $u\tau_{\lambda}u^{-1} = \tau_{u\lambda}$. By [HL15, §2.3], for all $(v, w) \in Q_J$ we have

$$v\tau_{\lambda}w^{-1} = v \cdot \tau_{\lambda}^{J} \cdot w^{J}w^{-1}, \quad \ell(v\tau_{\lambda}w^{-1}) = \ell(v) + \ell(\tau_{\lambda}^{J}) + \ell(w^{J}w^{-1}), \quad (3.7.8)$$

see Figure 3-2 for an example. By [HL15, Thm. 2.2], for all $u \in W^J$ we have

$$\psi(Q_J^{\succeq(u,u)}) = \{ g \in \tilde{W} \mid \tau_\lambda^J \le g \le \tau_{u\lambda} \}, \quad \text{and}$$
(3.7.9)

$$\psi(Q_J) = \{ g \in \tilde{W} \mid \tau_{\lambda}^J \le g \le \tau_{w\lambda} \text{ for some } w \in W^J \}.$$
(3.7.10)

3.7.3 Bruhat atlas for the projected Richardson stratification of G/P

Let $u \in W^J$. Recall that $\lambda \in Q_{\Phi}^{\vee}$ has been fixed. We further assume that the representatives $\dot{\tau}_{\lambda}$ and $\dot{\tau}_{u\lambda}$ satisfy the identity $\dot{u}\dot{\tau}_{\lambda}\dot{u}^{-1} = \dot{\tau}_{u\lambda}$.

Our goal is to construct a geometric lifting of the map ψ . Recall the maps $x \mapsto g_1^{(J)}$ and $x \mapsto g_2^{(J)}$ from Definition 3.4.23. We define maps

$$\varphi_u : C_u^{(J)} \to \mathcal{G}, \qquad xP \mapsto g_1^{(J)} \dot{u} \cdot \dot{\tau}_\lambda \cdot (g_2^{(J)} \dot{u})^{-1} = g_1^{(J)} \cdot \dot{\tau}_{u\lambda} \cdot (g_2^{(J)})^{-1}, \quad \text{and} \qquad (3.7.11)$$

$$\bar{\varphi}_u: C_u^{(J)} \to \mathcal{G}/\mathcal{B}, \qquad xP \mapsto \varphi_u(xP) \cdot \mathcal{B}.$$
 (3.7.12)

The main result of this section is the following theorem.

Theorem 3.7.2.

(1) The map $\bar{\varphi}_u$ is a biregular isomorphism

$$\bar{\varphi}_u: C_u^{(J)} \xrightarrow{\sim} \mathcal{X}_{\tau_\lambda^J} \cap \mathring{\mathcal{X}}^{\tau_{u\lambda}} = \bigsqcup_{(v,w) \in Q_I^{\succeq (u,u)}} \mathring{\mathcal{R}}_{v\tau_\lambda w^{-1}}^{\tau_{u\lambda}},$$

and for all $(v, w) \succeq (u, u)$ in Q_J , $\overline{\varphi}_u$ restricts to a biregular isomorphism

$$\bar{\varphi}_u: C_u^{(J)} \cap \overset{\circ}{\Pi}_{v,w} \xrightarrow{\sim} \overset{\circ}{\mathcal{R}}_{v\tau_\lambda w^{-1}}^{\tau_{u\lambda}}.$$

(2) Suppose that $(u, u) \preceq (v, w) \preceq (v', w')$ in Q_J . Then

$$\bar{\varphi}_u\left(\Pi_{v',w'}^{>0}\right) \subset \mathcal{C}_{v\tau_\lambda w^{-1}}$$

The remainder of this section will be devoted to the proof of Theorem 3.7.2.

3.7.4 An alternative definition of $\bar{\varphi}_u$

Recall the notation from Definition 3.4.23, and that we have fixed $u \in W^J$ and $\lambda \in Q_{\Phi}^{\vee}$ satisfying $\langle \lambda, \alpha_i \rangle = 0$ for $i \in J$ and $\langle \lambda, \alpha_i \rangle \in \mathbb{Z}_{<0}$ for $i \in I \setminus J$. We list the rules for conjugating elements of $G \subset \mathcal{G}$ by $\dot{\tau}_{\lambda}$.

Lemma 3.7.3. We have

$$\dot{\tau}_{\lambda} \cdot p = p \cdot \dot{\tau}_{\lambda} \quad \text{for all } p \in L_J,$$

$$(3.7.13)$$

$$\dot{\tau}_{\lambda} \cdot U^{(J)} \cdot \dot{\tau}_{\lambda}^{-1} \subset \mathcal{U}_{-}^{(I)}, \quad \dot{\tau}_{\lambda} \cdot U_{-}^{(J)} \cdot \dot{\tau}_{\lambda}^{-1} \subset \mathcal{U}^{(I)}, \qquad (3.7.14)$$

$$\dot{\tau}_{\lambda}^{-1} \cdot U^{(J)} \cdot \dot{\tau}_{\lambda} \subset \mathcal{U}^{(I)}, \quad \dot{\tau}_{\lambda}^{-1} \cdot U_{-}^{(J)} \cdot \dot{\tau}_{\lambda} \subset \mathcal{U}_{-}^{(I)}, \tag{3.7.15}$$

$$\dot{\tau}_{u\lambda} \cdot U_2^{(J)} \cdot \dot{\tau}_{u\lambda}^{-1} \subset \mathcal{U}^{(I)}, \quad \dot{\tau}_{u\lambda}^{-1} \cdot U_1^{(J)} \cdot \dot{\tau}_{u\lambda} \subset \mathcal{U}_-^{(I)}.$$
(3.7.16)

Proof. Recall that L_J is generated by T, U_J , and U_J^- , and since $\tau_\lambda \alpha = \alpha$ for all $\alpha \in \Phi_J$, we see that (3.7.13) follows from (3.7.2). By (3.7.2), we find $\tau_\lambda \alpha \in \Delta_{\rm re}^+ \setminus \Phi^+$ for $\alpha \in \Phi_-^{(J)}$ and $\tau_\lambda \alpha \in \Delta_{\rm re}^- \setminus \Phi^-$ for $\alpha \in \Phi_+^{(J)}$, which shows (3.7.14). Similarly, $\tau_\lambda^{-1} \alpha \in \Delta_{\rm re}^+ \setminus \Phi^+$ for $\alpha \in \Phi_+^{(J)}$ and $\tau_\lambda^{-1} \alpha \in \Delta_{\rm re}^- \setminus \Phi^-$ for $\alpha \in \Phi_-^{(J)}$, which shows (3.7.15).

To show (3.7.16), we use (3.7.6), (3.7.14), (3.7.15), and $U_1^{(J)}, U_2^{(J)} \subset \dot{u} U_-^{(J)} \dot{u}^{-1}$ to get

$$\dot{\tau}_{u\lambda} \cdot U_2^{(J)} \cdot \dot{\tau}_{u\lambda}^{-1} = \dot{u}\dot{\tau}_{\lambda}\dot{u}^{-1} \cdot U_2^{(J)} \cdot \dot{u}\dot{\tau}_{\lambda}^{-1}\dot{u}^{-1} \subset \dot{u}\dot{\tau}_{\lambda} \cdot U_{-}^{(J)} \cdot \dot{\tau}_{\lambda}^{-1}\dot{u}^{-1} \subset \dot{u}\mathcal{U}^{(I)}\dot{u}^{-1} = \mathcal{U}^{(I)},$$

$$\dot{\tau}_{u\lambda}^{-1} \cdot U_1^{(J)} \cdot \dot{\tau}_{u\lambda} = \dot{u}\dot{\tau}_{\lambda}^{-1}\dot{u}^{-1} \cdot U_1^{(J)} \cdot \dot{u}\dot{\tau}_{\lambda}\dot{u}^{-1} \subset \dot{u}\dot{\tau}_{\lambda}^{-1} \cdot U_{-}^{(J)} \cdot \dot{\tau}_{\lambda}\dot{u}^{-1} \subset \dot{u}\mathcal{U}_{-}^{(I)}\dot{u}^{-1} = \mathcal{U}_{-}^{(I)}.$$

The map $\bar{\varphi}_u$ can alternatively be characterized as follows. Recall from Definition 3.4.23 that we have a regular map $\kappa : \dot{u}G_0^{(J)} \to U_2^{(J)}$ that descends to a regular map $\kappa : C_u^{(J)} \to U_2^{(J)}$ by Lemma 3.6.2(iii). Recall also from Lemma 3.4.22(i) that $\dot{u}G_0^{(J)} = \dot{u}P_- \cdot B$.

Lemma 3.7.4. Let $x \in \dot{u}P_{-}$. Then

$$\bar{\varphi}_u(xP) = \kappa_x x \cdot \dot{\tau}_\lambda \cdot x^{-1} \cdot \mathcal{B}. \tag{3.7.17}$$

Proof. We continue using the notation of Definition 3.4.23. Let $p \in L_J$ and $g^{(J)} \in \dot{u}U_-^{(J)}\dot{u}^{-1}$ be such that $xp = g^{(J)}\dot{u}$. Note that $g_2^{(J)}\dot{u} = h_1^{(J)}g^{(J)}\dot{u} = h_1^{(J)}xp$, and since $h_1^{(J)} \in U_1^{(J)} \subset U \subset U$

 \mathcal{B} , we see that $(g_2^{(J)}\dot{u})^{-1}\cdot\mathcal{B} = (xp)^{-1}\cdot\mathcal{B}$. On the other hand, $\kappa_x xp = h_2^{(J)}g^{(J)}\dot{u} = g_1^{(J)}\dot{u}$. Since p commutes with $\dot{\tau}_{\lambda}$ by (3.7.13), we find

$$\bar{\varphi}_u(xP) = g_1^{(J)} \dot{u} \cdot \dot{\tau}_\lambda \cdot (g_2^{(J)} \dot{u})^{-1} \cdot \mathcal{B} = \kappa_x xp \cdot \dot{\tau}_\lambda \cdot (xp)^{-1} \cdot \mathcal{B} = \kappa_x x \cdot \dot{\tau}_\lambda \cdot x^{-1} \cdot \mathcal{B}.$$

3.7.5 The affine Richardson cell of $\bar{\varphi}_u$

Lemma 3.7.5. We have

$$C_{u}^{(J)} = \bigsqcup_{(v,w)\in Q_{J}^{\succeq(u,u)}} (C_{u}^{(J)} \cap \mathring{\Pi}_{v,w}).$$
(3.7.18)

Proof. The torus T acts on G/P by left multiplication and preserves the sets $C_u^{(J)}$ and $\mathring{\Pi}_{v,w}$ for all $(v,w) \in Q_J$. By (3.4.23), $\Pi_{v,w}$ contains $\dot{u}P$ if and only if $(u,u) \preceq (v,w)$. Suppose that $xP \in C_u^{(J)} \cap \mathring{\Pi}_{v,w}$ for some $(v,w) \in Q_J$. Then $TxP/P \subset C_u^{(J)}$, and by Lemma 3.6.8(iii), the closure of this set contains $\dot{u}P$. On the other hand, the closure of this set is contained inside $\Pi_{v,w}$, thus $(u,u) \preceq (v,w)$.

Lemma 3.7.6. Let $(v, w) \in Q_J^{\succeq(u,u)}$. Then

$$\bar{\varphi}_u(C_u^{(J)} \cap \mathring{\Pi}_{v,w}) \subset \mathring{\mathcal{R}}_{v\tau_\lambda w^{-1}}^{\tau_{u\lambda}}.$$
(3.7.19)

Proof. Let $x \in \dot{u}G_0^{(J)}$ be such that $xP \in \mathring{\Pi}_{v,w}$. Let us first show that $\bar{\varphi}_u(xP) \in \mathring{\mathcal{X}}^{\tau_{u\lambda}}$. By (3.7.12), we have

$$\bar{\varphi}_u(xP) = g_1^{(J)} \cdot \dot{\tau}_{u\lambda} \cdot (g_2^{(J)})^{-1} \cdot \dot{\tau}_{u\lambda}^{-1} \cdot \dot{\tau}_{u\lambda} \cdot \mathcal{B}.$$
(3.7.20)

Observe that $g_1^{(J)} \in U_1^{(J)} \subset U$ and by (3.7.16), $\dot{\tau}_{u\lambda} \cdot (g_2^{(J)})^{-1} \cdot \dot{\tau}_{u\lambda}^{-1} \in \mathcal{U}^{(I)}$, so by (3.7.6), we have

$$\varphi_u(xP) \cdot \dot{\tau}_{u\lambda}^{-1} \in \mathcal{U}, \quad \text{thus} \quad \varphi_u(xP) \in \mathcal{B} \cdot \dot{\tau}_{u\lambda} \cdot \mathcal{B}.$$
 (3.7.21)

This proves that $\bar{\varphi}_u(xP) \in \overset{\circ}{\mathcal{X}}^{\tau_{u\lambda}}$.

We now show $\bar{\varphi}_u(xP) \in \mathring{\mathcal{X}}_{v\tau_\lambda w^{-1}}$. Recall that $\mathring{\Pi}_{v,w} = \pi_J(\mathring{R}_{v,w})$, so assume that $x \in B_-\dot{v}B \cap B\dot{w}B$. Since $\dot{u}G_0^{(J)} = \dot{u}P_-B$ by Lemma 3.4.22(i), we may assume that $x \in \dot{u}P_-$, in which case $\bar{\varphi}_u(xP)$ is given by (3.7.17). We have $\kappa_x x \in B_-\dot{v}B$ and $x^{-1} \in B\dot{w}^{-1}B$, so it

suffices to show

$$B_{-}\dot{v}B\cdot\dot{\tau}_{\lambda}\cdot B\dot{w}^{-1}B\subset\mathcal{B}_{-}\cdot\dot{v}\dot{\tau}_{\lambda}\dot{w}^{-1}\cdot\mathcal{B}.$$
(3.7.22)

Clearly we have

$$B_{-}\dot{v}B\cdot\dot{\tau}_{\lambda}\cdot B\dot{w}^{-1}B\subset \mathcal{B}_{-}\cdot\dot{v}\cdot U^{(J)}\cdot U_{J}\cdot\dot{\tau}_{\lambda}\cdot U^{(J)}\cdot U_{J}\cdot\dot{w}^{-1}\cdot\mathcal{B}.$$

By (3.7.13) and Lemma 3.4.22(ii), U_J can be moved to the right past $\dot{\tau}_{\lambda}$ and $U^{(J)}$. We can then move $U^{(J)}$ to the left past $\dot{\tau}_{\lambda}$ using (3.7.14), which gives

$$B_{-}\dot{v}B\cdot\dot{\tau}_{\lambda}\cdot B\dot{w}^{-1}B\subset \mathcal{B}_{-}\cdot\dot{v}\cdot U^{(J)}\cdot\mathcal{U}_{-}^{(I)}\cdot\dot{\tau}_{\lambda}\cdot U_{J}\cdot\dot{w}^{-1}\cdot\mathcal{B}.$$

By (3.7.6), $\mathcal{U}_{-}^{(I)}$ can be moved to the left past $\dot{v} \cdot U^{(J)}$, and then $U^{(J)}$ can be moved to the right past $\dot{\tau}_{\lambda}$ using (3.7.15), yielding

$$B_{-}\dot{v}B\cdot\dot{\tau}_{\lambda}\cdot B\dot{w}^{-1}B\subset \mathcal{B}_{-}\cdot\dot{v}\cdot\dot{\tau}_{\lambda}\cdot\mathcal{U}^{(I)}\cdot U_{J}\cdot\dot{w}^{-1}\cdot\mathcal{B}.$$

By (3.7.6), $\mathcal{U}^{(I)}$ can be moved to the right past $U_J \cdot \dot{w}^{-1}$. Since $w \in W^J$, Lemma 3.4.5 shows that $U_J \cdot \dot{w}^{-1} \subset \dot{w}^{-1}U$, so (3.7.22) follows.

3.7.6 Proof of Theorem **3.7.2**(1)

Observe that $\mathcal{X}_{\tau_{\lambda}^{J}} \cap \mathring{\mathcal{X}}^{\tau_{u\lambda}} = \bigsqcup_{(v,w) \in Q_{J}^{\succeq(u,u)}} \mathring{\mathcal{R}}_{v\tau_{\lambda}w^{-1}}^{\tau_{u\lambda}}$ by (3.7.3) and (3.7.9). By (3.7.19), $\bar{\varphi}_{u}(C_{u}^{(J)}) \subset \mathcal{X}_{\tau_{\lambda}^{J}} \cap \mathring{\mathcal{X}}^{\tau_{u\lambda}}$. Let us identify $\mathring{\mathcal{X}}^{\tau_{u\lambda}}$ with the affine variety $\mathcal{U}_{1}(\tau_{u\lambda})$ via (3.7.5), and denote by $\bar{\varphi}_{u}^{\dagger} : C_{u}^{(J)} \to \mathcal{U}_{1}(\tau_{u\lambda})$ the composition of (3.7.5) and $\bar{\varphi}_{u}$.

We claim that $\bar{\varphi}_{u}^{\dagger}$ gives a biregular isomorphism between $C_{u}^{(J)}$ and a closed subvariety of $\mathcal{U}_{1}(\tau_{u\lambda})$. Let $x \in \dot{u}G_{0}^{(J)}$ and let $g^{(J)}, g_{1}^{(J)}, g_{2}^{(J)}$ be as in Definition 3.4.23. Let $y := \varphi_{u}(xP) \cdot \dot{\tau}_{u\lambda}^{-1}$, so $\bar{\varphi}_{u}(xP) = y \cdot \dot{\tau}_{u\lambda} \cdot \mathcal{B}$. Thus $\bar{\varphi}_{u}^{\dagger}(xP) = y$ if and only if $y \in \mathcal{U}_{1}(\tau_{u\lambda})$. By (3.7.21), we have $y \in \mathcal{U}$. Hence in order to prove $y \in \mathcal{U}_{1}(\tau_{u\lambda})$, we need to show $y \in \dot{\tau}_{u\lambda}\mathcal{U}_{-}\dot{\tau}_{u\lambda}^{-1}$. Conjugating both sides by $\dot{\tau}_{u\lambda}$, we get

$$\dot{\tau}_{u\lambda}^{-1} \cdot y \cdot \dot{\tau}_{u\lambda} = \dot{\tau}_{u\lambda}^{-1} g_1^{(J)} \dot{\tau}_{u\lambda} \cdot (g_2^{(J)})^{-1},$$

which belongs to \mathcal{U}_{-} since $(g_{2}^{(J)})^{-1} \in U_{2}^{(J)} \subset U_{-}$ by definition and $\dot{\tau}_{u\lambda}^{-1}g_{1}^{(J)}\dot{\tau}_{u\lambda} \in \mathcal{U}_{-}^{(I)}$ by (3.7.16). Thus $y \in \mathcal{U}_{1}(\tau_{u\lambda})$ and $\bar{\varphi}_{u}^{\dagger}(xP) = y$. By Lemma 3.4.2, we may identify $C_{u}^{(J)}$ with $U_{1}^{(J)} \times U_{2}^{(J)}$, so let $\bar{\varphi}_{u}^{\dagger} : U_{1}^{(J)} \times U_{2}^{(J)} \to \mathcal{U}_{1}(\tau_{u\lambda})$ be the map sending $(g_{1}^{(J)}, g_{2}^{(J)}) \mapsto y :=$ $g_{1}^{(J)} \cdot \dot{\tau}_{u\lambda}(g_{2}^{(J)})^{-1}\dot{\tau}_{u\lambda}^{-1}$.

Let $\Theta_1 := u \Phi_-^{(J)} \cap \Phi^+$ and $\Theta_2 := u \Phi_-^{(J)} \cap \Phi^-$, thus $U_1^{(J)} = U(\Theta_1)$, $U_2^{(J)} = U_-(\Theta_2)$, and $\Theta_1 \sqcup \Theta_2 = u \Phi_-^{(J)}$. By the proof of (3.7.16), $\tau_{u\lambda}\Theta_2 \subset \Delta_{\rm re}^+ \setminus \Phi^+$ and $\tau_{u\lambda}^{-1}\Theta_1 \subset \Delta_{\rm re}^-$, thus $\Theta_1 \sqcup \tau_{u\lambda}\Theta_2 \subset \operatorname{Inv}(\tau_{u\lambda}^{-1})$. Let $\Theta_3 \subset \Delta_{\rm re}^+$ be defined by $\Theta_3 := \operatorname{Inv}(\tau_{u\lambda}^{-1}) \setminus (\Theta_1 \sqcup \Theta_2)$. By Lemma 3.A.1, the multiplication map gives a biregular isomorphism

$$\mathcal{U}(\Theta_1) \times \mathcal{U}(\tau_{u\lambda}\Theta_2) \times \prod_{\alpha \in \Theta_3} \mathcal{U}_{\alpha} \xrightarrow{\sim} \mathcal{U}(\operatorname{Inv}(\tau_{u\lambda}^{-1})) = \mathcal{U}_1(\tau_{u\lambda}), \qquad (3.7.23)$$

where $\mathcal{U}(\Theta)$ denotes the subgroup generated by $\{\mathcal{U}_{\alpha}\}_{\alpha\in\Theta}$. In particular, $\mathcal{U}(\Theta_1) \cdot \mathcal{U}(\tau_{u\lambda}\Theta_2)$ is a closed subvariety of $\mathcal{U}_1(\tau_{u\lambda})$ isomorphic to $\mathbb{C}^{|\Theta_1|+|\Theta_2|} = \mathbb{C}^{\ell(w^J)}$. Observe that $\mathcal{U}(\tau_{u\lambda}\Theta_2) = \dot{\tau}_{u\lambda}U_2^{(J)}\dot{\tau}_{u\lambda}^{-1}$, thus $\bar{\varphi}_u^{\dagger}$ essentially coincides with the restriction of the map (3.7.23) to $\mathcal{U}(\Theta_1) \times \mathcal{U}(\tau_{u\lambda}\Theta_2) \times \{1\}$. We have thus shown that $\bar{\varphi}_u^{\dagger}$ gives a biregular isomorphism between $U_1^{(J)} \times U_2^{(J)}$ and a closed $\ell(w^J)$ -dimensional subvariety of $\mathcal{U}_1(\tau_{u\lambda})$. Therefore $\bar{\varphi}_u$ gives a biregular isomorphism between $C_u^{(J)}$ and a closed $\ell(w^J)$ -dimensional subvariety of $\hat{\mathcal{X}}^{\tau_{u\lambda}}$, and by (3.7.8) and Proposition 3.A.2, $\mathcal{X}_{\tau_\lambda} \cap \hat{\mathcal{X}}^{\tau_{u\lambda}}$ is a closed irreducible subvariety of $\hat{\mathcal{X}}^{\tau_{u\lambda}}$, and by (3.7.8) and Proposition 3.A.2, it has dimension $\ell(w^J)$. Since $\bar{\varphi}_u(C_u^{(J)}) \subset \mathcal{X}_{\tau_\lambda} \cap \hat{\mathcal{X}}^{\tau_{u\lambda}}$, it follows that $\bar{\varphi}_u(C_u^{(J)}) = \mathcal{X}_{\tau_\lambda} \cap \hat{\mathcal{X}}^{\tau_{u\lambda}}$. We are done with the proof of Theorem 3.7.2(1).

Remark 3.7.7. Alternatively, the proof of Theorem 3.7.2(1) could be deduced from *Deodhar*type parametrizations [Had84, Had85, BD94] of $\mathring{\mathcal{R}}_{v\tau_{\lambda}w^{-1}}^{\tau_{u\lambda}}$, by observing that any reduced word for $\tau_{u\lambda}$ that is compatible with the length-additive factorization $\tau_{u\lambda} = u \cdot \tau_{\lambda}^{J} \cdot w^{J} u^{-1}$ in (3.7.8) contains a unique reduced subword for τ_{λ}^{J} .

3.7.7 Proof of Theorem 3.7.2(2)

We use the notation and results from Section 3.6. Let $x \in G$ be such that $xP \in \Pi_{v',w'}^{>0}$. Since $\Pi_{v',w'}^{>0} = \pi_J(R_{v',w'}^{>0})$, we may assume that $xB \in R_{v',w'}^{>0}$. Then $x \in \dot{u}G_0^{(J)}$ by Lemma 3.6.9(ii), so $\bar{\varphi}_u(xP)$ is defined. In addition, by Lemma 3.4.22(i) we may assume that $x \in \dot{u}P_-$. By

definition, $\bar{\varphi}_u(xP) \in \mathcal{C}_{v\tau_\lambda w^{-1}}$ if and only if $\dot{w}\dot{\tau}_\lambda^{-1}\dot{v}^{-1}\bar{\varphi}_u(xP) \in \mathcal{B}_-\mathcal{B}/\mathcal{B}$. By (3.7.17), this is equivalent to

$$\dot{w}\dot{\tau}_{\lambda}^{-1}\dot{v}^{-1}\cdot\kappa_{x}x\cdot\dot{\tau}_{\lambda}\cdot x^{-1}\in\mathcal{B}_{-}\mathcal{B}.$$
(3.7.24)

By Theorem 3.6.4, $x \in G_{u,v}^{(J)}$, so $\dot{v}^{-1}\kappa_x x \in G_0^{(J)}$. Let us factorize $y := \dot{v}^{-1}\kappa_x x$ as $y = [y]_{-}^{(J)}[y]_0[y]_{+}^{(J)}$ using Lemma 3.4.22(iii). By (3.7.13) and (3.7.15), we get

$$\dot{w}\dot{\tau}_{\lambda}^{-1}\dot{v}^{-1}\cdot\kappa_{x}x\cdot\dot{\tau}_{\lambda}\cdot x^{-1} = \dot{w}\cdot\dot{\tau}_{\lambda}^{-1}[y]_{-}^{(J)}\dot{\tau}_{\lambda}\cdot\dot{\tau}_{\lambda}^{-1}[y]_{+}^{(J)}\dot{\tau}_{\lambda}\cdot x^{-1} \in \dot{w}\cdot\mathcal{U}_{-}^{(I)}\cdot[y]_{0}\cdot\mathcal{U}^{(I)}\cdot x^{-1}.$$

Using (3.7.6), we can move $\mathcal{U}_{-}^{(I)}$ to the left and $\mathcal{U}^{(I)}$ to the right, so we see that (3.7.24) is equivalent to $\dot{w}[y]_0 x^{-1} \in \mathcal{B}_- \mathcal{B}$. By Definition 3.6.1, we have $[y]_0 = \eta(x)$, and by Lemma 3.6.3(ii), we have $\zeta_{u,v}^{(J)}(x) = x\eta(x)^{-1} = x[y]_0^{-1}$. By Theorem 3.6.4, $\zeta_{u,v}^{(J)}(x) \in BB_-\dot{w}$, and after taking inverses, we obtain $\dot{w}[y]_0 x^{-1} \in B_- \mathcal{B} \subset \mathcal{B}_- \mathcal{B}$, finishing the proof.

3.8 From Bruhat atlas to Fomin–Shapiro atlas

We use Theorem 3.7.2 to prove Theorem 3.2.5.

3.8.1 Affine Bruhat projections

We first define the affine flag variety version of the map $\bar{\nu}_g$ from (3.2.1). We will need some results on Gaussian decomposition inside \mathcal{G} , see Section 3.A.5 for a proof.

Lemma 3.8.1. Let $\mathcal{G}_0 := \mathcal{B}_- \cdot \mathcal{B}_-$

(i) The multiplication map gives a biregular isomorphism of ind-varieties:

$$\mathcal{U}_{-} \times \mathcal{T} \times \mathcal{U} \xrightarrow{\sim} \mathcal{G}_{0}. \tag{3.8.1}$$

For $x \in \mathcal{G}_0$, we denote by $[x]_- \in \mathcal{U}_-$, $[x]_0 \in \mathcal{T}$, and $[x]_+ \in \mathcal{U}$ the unique elements such that $x = [x]_-[x]_0[x]_+$.

(ii) For $g \in \tilde{W}$, the multiplication map gives biregular isomorphisms of ind-varieties:

$$\mu_{12}: \mathcal{U}_1(g) \times \mathcal{U}_2(g) \xrightarrow{\sim} \dot{g}\mathcal{U}_- \dot{g}^{-1}, \quad \mu_{21}: \mathcal{U}_2(g) \times \mathcal{U}_1(g) \xrightarrow{\sim} \dot{g}\mathcal{U}_- \dot{g}^{-1}.$$
(3.8.2)

The group $\dot{g}\mathcal{U}_{-}\dot{g}^{-1}$, as well as its subgroups $\mathcal{U}_{1}(g)$ and $\mathcal{U}_{2}(g)$, act on \mathcal{C}_{g} . The following result, which we state for the polynomial loop group \mathcal{G} , holds in Kac-Moody generality.

Proposition 3.8.2. Let $g \in \tilde{W}$.

- (i) For $x \in \mathcal{G}$ such that $x\mathcal{B} \in \mathcal{C}_g$, there exist unique elements $y_1 \in \mathcal{U}_1(g)$ and $y_2 \in \mathcal{U}_2(g)$ such that $y_1 x\mathcal{B} \in \overset{\circ}{\mathcal{X}_g}$ and $y_2 x\mathcal{B} \in \overset{\circ}{\mathcal{X}^g}$.
- (ii) The map $\tilde{\nu}_g : \mathcal{C}_g \xrightarrow{\sim} \mathring{\mathcal{X}}_g \times \mathring{\mathcal{X}}^g$ sending $x\mathcal{B} \mapsto (y_1 x\mathcal{B}, y_2 x\mathcal{B})$ is a biregular isomorphism of ind-varieties.
- (iii) For all $h, f \in \tilde{W}$ satisfying $h \leq g \leq f$, the map $\tilde{\nu}_g$ restricts to a biregular isomorphism $\mathcal{C}_g \cap \mathring{\mathcal{R}}_h^f \xrightarrow{\sim} \mathring{\mathcal{R}}_g^f \times \mathring{\mathcal{R}}_h^g$ of finite-dimensional varieties.

Proof. Let us first prove an affine analog of Lemma 3.4.2. Let $\nu_1 : \dot{g}\mathcal{U}_{-}\dot{g}^{-1} \to \mathcal{U}_2(g), \nu_2 : \dot{g}\mathcal{U}_{-}\dot{g}^{-1} \to \mathcal{U}_1(g)$ denote the second component of μ_{12}^{-1} and μ_{21}^{-1} (cf. (3.8.2)), respectively, and let $\nu := (\nu_1, \nu_2) : \dot{g}\mathcal{U}_{-}\dot{g}^{-1} \to \mathcal{U}_2(g) \times \mathcal{U}_1(g)$. We claim that ν is a biregular isomorphism. By Lemma 3.8.1(ii), ν is a regular morphism. Let us now compute the inverse of ν . Given $x_1 \in \mathcal{U}_1(g)$ and $x_2 \in \mathcal{U}_2(g)$, we claim that there exist unique $y_1 \in \mathcal{U}_1(g)$ and $y_2 \in \mathcal{U}_2(g)$ such that $y_1x_2 = y_2x_1$. Indeed, this equation is equivalent to $y_2^{-1}y_1 = x_1x_2^{-1}$, so we must have $y_2 = [x_1x_2^{-1}]_{-}^{-1}$ and $y_1 = [x_1x_2^{-1}]_{+}$. Clearly, $\nu^{-1}(x_2, x_1) = y_1x_2 = y_2x_1$, and by Lemma 3.8.1(i), the map ν^{-1} is regular. Applying (3.7.5) finishes the proof of (i) and (ii).

We now prove (iii). Observe that if $x\mathcal{B} \in \mathcal{C}_g \cap \mathring{\mathcal{R}}_h^f$ for some $h \leq f \in \tilde{W}$ then $x \in \mathcal{B}_h \dot{\mathcal{B}} \cap \mathcal{B}_h \dot{\mathcal{B}}$. Let y_1, y_2 be as in (ii). Then $y_1 \in \mathcal{U}_1(g) \subset \mathcal{U}$, so $y_1 x \in \mathcal{B}_h \dot{\mathcal{B}}$. Similarly, $y_2 \in \mathcal{U}_2(g) \subset \mathcal{U}_-$, so $y_2 x \in \mathcal{B}_- \dot{h} \mathcal{B}$. It follows that if $x\mathcal{B} \in \mathcal{C}_g \cap \mathring{\mathcal{R}}_h^f$ then $\tilde{\nu}_g(x\mathcal{B}) \in \mathring{\mathcal{R}}_h^g \times \mathring{\mathcal{R}}_g^f$. In particular, we must have $h \leq g \leq f$, and we are done by (3.7.3).

3.8.2 Torus action

Recall that $\mathcal{T} = \mathbb{C}^* \times T$ is the affine torus. The group \mathbb{C}^* acts on \mathcal{G} via loop rotation as follows. For $t \in \mathbb{C}^*$, we have $t \cdot g(z) = g(tz)$. We form the semidirect product $\mathbb{C}^* \ltimes \mathcal{G}$ with multiplication given by $(t_1, x_1(z)) \cdot (t_2, x_2(z)) := (t_1t_2, x_1(z)x_2(t_1z))$, for $(t_1, x_1(z)), (t_2, x_2(z)) \in \mathbb{C}^* \times \mathcal{G}$. Let $Y(\mathcal{T}) := \operatorname{Hom}(\mathbb{C}^*, \mathcal{T}) \cong \mathbb{Z}d \oplus Y(T)$. For $\lambda \in Y(\mathcal{T}), t \in \mathbb{C}^*, t' \in \mathbb{C}$, and $\alpha \in \Delta_{\operatorname{re}}$, we have

$$\lambda(t)x_{\alpha}(t')\lambda(t)^{-1} = x_{\alpha}(t^{\langle\lambda,\alpha\rangle}t').$$
(3.8.3)

where $x_{\alpha} : \mathbb{C} \xrightarrow{\sim} \mathcal{U}_{\alpha}$ is as in Section 3.7.1, and $\langle \cdot, \cdot \rangle : Y(\mathcal{T}) \times X(\mathcal{T}) \to \mathbb{Z}$ extends the pairing from Section 3.4.1 in such a way that $\langle d, \delta \rangle = 1$ and $\langle d, \alpha_i \rangle = \langle \alpha_i^{\vee}, \delta \rangle = 0$ for $i \in I$.

Let $g \in \tilde{W}$ and denote $N := \ell(g)$. If $\operatorname{Inv}(g) = \{\alpha^{(1)}, \ldots, \alpha^{(N)}\}$ then by Lemma 3.A.1, the map $\mathbf{x}_g : \mathbb{C}^N \to \mathcal{U}_1(g)$ given by

$$\mathbf{x}_{g}(t_{1},\ldots,t_{N}) := x_{\alpha^{(1)}}(t_{1})\cdots x_{\alpha^{(N)}}(t_{N})$$
(3.8.4)

is a biregular isomorphism. For $\mathbf{t} = (t_1, \ldots, t_N) \in \mathbb{C}^N$, define $\|\mathbf{t}\| := (|t_1|^2 + \cdots + |t_N|^2)^{\frac{1}{2}} \in \mathbb{R}_{\geq 0}$, and let $\|\cdot\| : \mathcal{U}_1(g) \to \mathbb{R}_{\geq 0}$ be defined by $\|y\| := \|\mathbf{x}_g^{-1}(y)\|$. Identifying $\mathcal{U}_1(g)$ with $\hat{\mathcal{X}}^g$ via (3.7.5), we get a function $\|\cdot\| : \hat{\mathcal{X}}^g \to \mathbb{R}_{\geq 0}$.

We say that $\tilde{\rho} \in Y(\mathcal{T})$ is a regular dominant integral coweight if $\langle \tilde{\rho}, \delta \rangle \in \mathbb{Z}_{>0}$ and $\langle \tilde{\rho}, \alpha_i \rangle \in \mathbb{Z}_{>0}$ for all $i \in I$. In this case, we have $\langle \tilde{\rho}, \alpha \rangle \in \mathbb{Z}_{>0}$ for any $\alpha \in \Delta_{\mathrm{re}}^+$. Let us choose such a coweight $\tilde{\rho}$, and define $\vartheta_g : \mathbb{R}_{>0} \times \mathcal{G}/\mathcal{B} \to \mathcal{G}/\mathcal{B}$ by $\vartheta_g(t, x\mathcal{B}) := \tilde{\rho}(t)x\mathcal{B}$.

It follows from (3.8.3) that if $g \in \tilde{W}$ and $y \in \mathcal{U}_1(g)$ is such that $\mathbf{x}_g^{-1}(y) = (t_1, \ldots, t_N)$ then there exist $k_1, \ldots, k_N \in \mathbb{Z}_{>0}$ satisfying

$$\|\vartheta_g(t, y\dot{g}\mathcal{B})\| = \left(t^{k_1}|t_1|^2 + \dots + t^{k_N}|t_N|^2\right)^{\frac{1}{2}} \quad \text{for all } t \in \mathbb{R}_{>0}.$$
 (3.8.5)

3.8.3 Proof of Theorem 3.2.5

By Corollary 3.4.20, $((G/P)_{\mathbb{R}}, (G/P)_{\geq 0}, Q_J)$ is a TNN space in the sense of Definition 3.2.1. Thus it suffices to construct a Fomin–Shapiro atlas.

Let $(u, u) \leq (v, w) \in Q_J$, and denote f := (u, u), g := (v, w). Thus we have $\psi(f) = \tau_{u\lambda}$ and $\psi(g) = v\tau_{\lambda}w^{-1}$. Moreover, for the maximal element $\hat{1} = (\mathrm{id}, w^J) \in Q_J$, we have $\psi(\hat{1}) = \tau_{\lambda}^J$. By Theorem 3.7.2(1), the map $\bar{\varphi}_u$ gives an isomorphism $C_u^{(J)} \xrightarrow{\sim} \mathcal{X}_{\psi(\hat{1})} \cap \mathring{\mathcal{X}}^{\psi(f)}$. Let $\mathcal{O}_g^{\mathbb{C}} \subset C_u^{(J)}$ be the preimage of $\mathcal{C}_{\psi(g)} \cap \mathcal{X}_{\psi(\hat{1})} \cap \mathring{\mathcal{X}}^{\psi(f)}$ under $\bar{\varphi}_u$, and denote by $\mathcal{O}_g := \mathcal{O}_g^{\mathbb{C}} \cap (G/P)_{\mathbb{R}}$. Since $\mathcal{C}_{\psi(g)}$ is open in \mathcal{G}/\mathcal{B} , we see that $\mathcal{O}_g^{\mathbb{C}}$ is open in $C_u^{(J)}$ which is open in G/P, so \mathcal{O}_g is an open subset of $(G/P)_{\mathbb{R}}$. By Theorem 3.7.2(2), \mathcal{O}_g contains $\mathrm{Star}_g^{\geq 0}$, which shows (FS5). Moreover, we claim that $\mathcal{O}_g \subset \mathrm{Star}_g$. Indeed, if $h \succeq f$ but $h \nsucceq g$ then $\psi(h) \nleq \psi(g)$. The map $\bar{\varphi}_u$ sends $\mathring{\Pi}_h \cap C_u^{(J)}$ to $\mathring{\mathcal{R}}_{\psi(h)}^{\psi(f)}$, which does not intersect $\mathcal{C}_{\psi(g)}$ by (3.A.3).

We now define the smooth cone (Z_g, ϑ_g) . Throughout, we identify $\mathring{\mathcal{X}}^{\psi(g)}$ with \mathbb{C}^{N_g} for

 $N_g := \ell(\psi(g))$ via (3.8.4). We set $Z_g^{\mathbb{C}} := \mathcal{X}_{\psi(\hat{1})} \cap \mathring{\mathcal{X}}^{\psi(g)}$ and $\mathring{Z}_{g,h}^{\mathbb{C}} := \mathring{\mathcal{R}}_{\psi(h)}^{\psi(g)}$ for $g \leq h \in Q_J$. We let $Z_g := Z_g^{\mathbb{C}} \cap \mathbb{R}^{N_g}$ and $\mathring{Z}_{g,h} := \mathring{Z}_{g,h}^{\mathbb{C}} \cap \mathbb{R}^{N_g}$ denote the corresponding sets of real points. Thus (FS1) follows trivially. The action ϑ_g restricts to \mathbb{R}^{N_g} , and by (3.8.5), it satisfies (SC2). As we discussed in Section 3.8.2, the action of ϑ_g also preserves Z_g (showing (SC1)) and $\mathring{Z}_{g,h}$ (showing (FS2)).

Finally, we define a map $\bar{\nu}_g : \mathcal{O}_g^{\mathbb{C}} \to (\mathring{\Pi}_g \cap \mathcal{O}_g^{\mathbb{C}}) \times \mathbb{C}^{N_g}$ as follows. Let $\tilde{\nu}_g = (\tilde{\nu}_{g,1}, \tilde{\nu}_{g,2}) :$ $\mathcal{C}_g \xrightarrow{\sim} \mathring{\mathcal{X}}_g \times \mathring{\mathcal{X}}^g$ be the map from Proposition 3.8.2. We let $\bar{\nu}_{g,2} := \tilde{\nu}_{g,2} \circ \bar{\varphi}_u$, so it sends $\mathcal{O}_g^{\mathbb{C}} \to \mathcal{C}_{\psi(g)} \to \mathring{\mathcal{X}}^{\psi(g)} \cong \mathbb{C}^{N_g}$. By Proposition 3.8.2(iii), the image of $\bar{\nu}_{g,2}$ is precisely $Z_g^{\mathbb{C}}$. We also let $\bar{\nu}_{g,1} := \bar{\varphi}_u^{-1} \circ \tilde{\nu}_{g,1} \circ \bar{\varphi}_u$, thus it sends

$$\mathcal{O}_g^{\mathbb{C}} \xrightarrow{\sim} \mathcal{C}_{\psi(g)} \cap \mathcal{X}_{\psi(\hat{1})} \cap \mathring{\mathcal{X}}^{\psi(f)} \to \mathring{\mathcal{R}}_{\psi(g)}^{\psi(f)} \xrightarrow{\sim} \mathring{\Pi}_g \cap \mathcal{O}_g^{\mathbb{C}}.$$

It follows from Theorem 3.7.2(1) and Proposition 3.8.2 that $\bar{\nu}_g := (\bar{\nu}_{g,1}, \bar{\nu}_{g,2})$ gives a biregular isomorphism $\mathcal{O}_g^{\mathbb{C}} \xrightarrow{\sim} (\mathring{\Pi}_g \cap \mathcal{O}_g^{\mathbb{C}}) \times Z_g^{\mathbb{C}}$. All maps in the definition of $Z_g^{\mathbb{C}}$ are defined over \mathbb{R} , thus $\bar{\nu}_g$ gives a smooth embedding $\mathcal{O}_g \to (\mathring{\Pi}_g^{\mathbb{R}} \cap \mathcal{O}_g) \times \mathbb{R}^{N_g}$ with image $(\mathring{\Pi}_g^{\mathbb{R}} \cap \mathcal{O}_g) \times Z_g$. By Lemma 3.3.3, we find that Z_g is an embedded submanifold of \mathbb{R}^{N_g} , so we get a diffeomorphism

$$\bar{\nu}_g: \mathcal{O}_g \xrightarrow{\sim} (\mathring{\Pi}_g^{\mathbb{R}} \cap \mathcal{O}_g) \times Z_g.$$

By Theorem 3.7.2(1) and Proposition 3.8.2(iii), we find that for $h \succeq g$, $\bar{\nu}_g$ sends $\mathring{\Pi}_h \cap \mathcal{O}_g$ to $(\mathring{\Pi}_g \cap \mathcal{O}_g) \times \mathring{Z}_{g,h}$, showing (FS3). When $xP \in \mathring{\Pi}_g \cap \mathcal{O}_g$, we have $\bar{\varphi}_u(xP) \in \mathring{\mathcal{R}}_{\psi(g)}^{\psi(f)}$, so clearly $\tilde{\nu}_{g,1}(\bar{\varphi}_u(xP)) = \bar{\varphi}_u(xP)$ and $\tilde{\nu}_{g,2}(\bar{\varphi}_u(xP)) \in \mathring{\mathcal{R}}_{\psi(g)}^{\psi(g)}$. Thus $\bar{\nu}_{g,1}(xP) = x$ and $\bar{\nu}_{g,2}(xP) =$ 0, showing (FS4). We have thus completed all requirements of Definitions 3.2.1, 3.2.2, and 3.2.3.

3.9 The case $G = SL_n$

In this section, we illustrate our construction in type A. We mostly focus on the case when G/P is the Grassmannian $\operatorname{Gr}(k,n)$ so that $(G/P)_{\geq 0}$ is the totally nonnegative Grassmannian $\operatorname{Gr}_{\geq 0}(k,n)$. Throughout, we assume $\mathbb{K} = \mathbb{C}$.

3.9.1 Preliminaries

Fix an integer $n \ge 1$ and recall the notation $[n] := \{1, 2, \dots, n\}, \binom{[n]}{k} := \{S \subset [n] : |S| = k\}.$

Let $G = SL_n$ be the group of $n \times n$ matrices over \mathbb{C} of determinant 1. We have subgroups $B, B_-, T, U, U_- \subset G$ consisting of upper triangular, lower triangular, diagonal, upper unitriangular, and lower unitriangular matrices of determinant 1, respectively. The Weyl group W is the group S_n of permutations of [n], and for $i \in I = [n - 1]$, $s_i \in W$ is the simple transposition of elements i and i + 1. If $w \in W$ is written as a product $w = s_{i_1} \dots s_{i_l}$ then the action of w on [n] is given by $w(j) := s_{i_1}(\dots(s_{i_l}(j))\dots)$ for $j \in [n]$. For $S \subset [n]$, we set $wS := \{w(j) \mid j \in S\}$. For example, if n = 3 and $w = s_2s_1$ then w(1) = 3, w(2) = 1, w(3) = 2, and $w\{1,3\} = \{2,3\}$.

For $i \in [n-1]$, the homomorphism $\phi_i : \operatorname{SL}_2 \to G$ just sends a matrix $A \in \operatorname{SL}_2$ to the $n \times n$ matrix $\phi_i(A) \in \operatorname{SL}_n$ which has a 2×2 block equal to A in rows and columns i, i+1. Thus if n = 3 then $\dot{s}_1 = \begin{bmatrix} 0 & -1 & 0 \\ 1 & 0 & 0 \\ 0 & 0 & 1 \end{bmatrix}$, $\dot{s}_2 = \begin{bmatrix} 1 & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 1 & 0 \end{bmatrix}$, and for $w = s_2 s_1$, $\dot{w} = \begin{bmatrix} 0 & -1 & 0 \\ 0 & 0 & -1 \\ 1 & 0 & 0 \end{bmatrix}$. In general, given $w \in S_n$, \dot{w} contains a ± 1 in row w(j) and column j for each $j \in [n]$, and the sign of this entry is -1 if and only if the number of ± 1 's strictly below and to the left of it is odd. In other words, the (w(j), j)-th entry of \dot{w} equals $(-1)^{\#\{i < j | w(i) > w(j)\}}$.

For $x \in SL_n$, x^T is just the matrix transpose of x, and x^{ι} defined in (3.4.4) has (i, j)-th entry equal to the determinant of the submatrix obtained from x by deleting row j and column i.

For $i \in [n-1]$, the function $\Delta_i^{\mp} : \mathrm{SL}_n \to \mathbb{C}$ is the top-left $i \times i$ principal minor, while $\Delta_i^{\pm} : \mathrm{SL}_n \to \mathbb{C}$ is the bottom-right $i \times i$ principal minor. The subset $G_0^{\mp} = B_- B$ consists precisely of matrices $x \in \mathrm{SL}_n$ all of whose top-left principal minors are nonzero, in agreement with Lemma 3.4.21(iii). We denote $\Delta_n^{\mp}(x) = \Delta_n^{\pm}(x) := \det x = 1$.

3.9.2 Flag variety

The group B acts on $G = \operatorname{SL}_n$ by right multiplication, and G/B is the complete flag variety in \mathbb{C}^n . It consists of flags $\{0\} = V_0 \subset V_1 \subset \cdots \subset V_n = \mathbb{C}^n$ in \mathbb{C}^n such that dim $V_i = i$ for $i \in [n]$. For a matrix $x \in \operatorname{SL}_n$, the element $xB \in G/B$ gives rise to a flag $V_0 \subset V_1 \subset \cdots \subset V_n$ such that V_i is the span of columns $1, \ldots i$ of x. For $k \in [n], S \in {[n] \choose k}$, and $x \in \operatorname{SL}_n$, we denote by Δ_S^{flag} the determinant of the $k \times k$ submatrix of x with row set S and column set [k]. Thus for each $k \in [n]$, we have a map $\Delta_k^{\text{flag}} : G/B \to \mathbb{CP}^{\binom{n}{k}-1}$ sending xB to $(\Delta_S^{\text{flag}}(x))_{S \in \binom{[n]}{k}}$. Here $\binom{[n]}{k}$ is identified with the set $W\omega_k$ from Lemma 3.4.21(iv).

3.9.3 Partial flag variety

For $J \subset [n]$, we have a parabolic subgroup $P \subset G$, and the partial flag variety G/P consists of partial flags $\{0\} = V_0 \subset V_{j_1} \subset \cdots \subset V_{j_l} \subset V_n = \mathbb{C}^n$, where $\{j_1 < \cdots < j_l\} := ([n-1] \setminus J)$ and dim $V_{j_i} = j_i$ for $i \in [l]$. The projection $\pi_J : G/B \to G/P$ sends a flag (V_0, V_1, \ldots, V_n) to $(V_0, V_{j_1}, \ldots, V_{j_l}, V_n)$. When $J = \emptyset$, we have P = B and G/P = G/B. We will focus on the "opposite" special case.

Unless otherwise stated, we assume that $J = [n-1] \setminus \{k\}$ for some fixed $k \in [n-1]$.

In this case, G/P is the *(complex) Grassmannian* $\operatorname{Gr}(k, n)$, which we will identify with the space of $n \times k$ full rank matrices modulo column operations. Let us write matrices in SL_n in block form $\left[\frac{A \mid B}{C \mid D}\right]$, where A is of size $k \times k$ and D is of size $(n - k) \times (n - k)$. For a matrix $x = \left[\frac{A \mid B}{C \mid D}\right] \in \operatorname{SL}_n$, we denote by $[x] := \left[\frac{A}{C}\right]$ the $n \times k$ submatrix consisting of the first k columns of x. Thus every $x \in \operatorname{SL}_n$ gives rise to an element xP of G/P which is a k-dimensional subspace $V_k \subset \mathbb{C}^n$ equal to the column span of [x]. The map Δ_k^{flag} in this case is the classical $Pl \ddot{u} cker embedding \Delta_k^{\text{flag}} : \operatorname{Gr}(k, n) \hookrightarrow \mathbb{CP}^{\binom{n}{k}-1}$, cf. Section 2.2.

The set W^J from Section 3.4.6 consists of *Grassmannian permutations*: we have $w \in W^J$ if and only if w = id or every reduced word for w ends with s_k . Equivalently, $w \in W^J$ if and only if $w(1) < \cdots < w(k)$ and $w(k+1) < \cdots < w(n)$, so the map $w \mapsto w[k]$ gives a bijection $W^J \to {[n] \choose k}$. The parabolic subgroup W_J (generated by $\{s_j\}_{j\in J}$) consists of permutations $w \in S_n$ such that w[k] = [k], and the longest element $w_J \in W_J$ is given by $(w_J(1), \ldots, w_J(n)) = (k, \ldots, 1, n, \ldots, k+1)$. The maximal element w^J of W^J is given by $(w^J(1), \ldots, w^J(n)) = (n - k + 1, \ldots, n, 1, \ldots, n - k)$. We have

$$U_J = \left\{ \begin{bmatrix} U_k & 0 \\ 0 & U_{n-k} \end{bmatrix} \right\}, \quad U_-^{(J)} = \left\{ \begin{bmatrix} I_k & 0 \\ \hline C & I_{n-k} \end{bmatrix} \right\}, \quad L_J = \left\{ \begin{bmatrix} A & 0 \\ \hline 0 & D \end{bmatrix} \right\}, \quad P = \left\{ \begin{bmatrix} A & B \\ \hline 0 & D \end{bmatrix} \right\},$$

where U_r is an $r \times r$ upper unitriangular matrix, I_r is an $r \times r$ identity matrix, $A \in SL_k$, $D \in SL_{n-k}$, and B, C are arbitrary $k \times (n-k)$ and $(n-k) \times k$ matrices, respectively.

3.9.4 Affine charts

We have $G_0^{(J)} := \{x \in G \mid \Delta_{[k]}^{\text{flag}}(x) \neq 0\}$, and for $x = \begin{bmatrix} A \mid B \\ C \mid D \end{bmatrix} \in G_0^{(J)}$ (such that $\Delta_{[k]}^{\text{flag}}(x) = \det A \neq 0$), the factorization $x = [x]_{-}^{(J)}[x]_{0}^{(J)}[x]_{+}^{(J)}$ from Lemma 3.4.22(iii) is given by

$$\begin{bmatrix} A & B \\ \hline C & D \end{bmatrix} = \begin{bmatrix} I_k & 0 \\ \hline CA^{-1} & I_{n-k} \end{bmatrix} \cdot \begin{bmatrix} A & 0 \\ \hline 0 & D - CA^{-1}B \end{bmatrix} \cdot \begin{bmatrix} I_k & A^{-1}B \\ \hline 0 & I_{n-k} \end{bmatrix}.$$
 (3.9.1)

The matrix $D - CA^{-1}B$ is called the Schur complement of D in x.

For $u \in W^J$, the set $C_u^{(J)} \subset G/P$ consists of elements xP such that $\Delta_{u[k]}^{\text{flag}}(x) \neq 0$. The (inverse of the) isomorphism (3.4.31) essentially amounts to computing the reduced column echelon form of an $n \times k$ matrix: if $x \in G$ is such that $xP \in C_u^{(J)}$ is sent to $g^{(J)} \in \dot{u}U_-^{(J)}\dot{u}^{-1}$ via (3.4.31) then the $n \times k$ matrices [x| and $[g^{(J)}\dot{u}|$ have the same column span, and the submatrix of $[g^{(J)}\dot{u}|$ with row set u[k] is the $k \times k$ identity matrix. Let us say that an $n \times k$ matrix M is in u[k]-echelon form if its submatrix with row set u[k] is the $k \times k$ identity matrix.

The matrices $g_1^{(J)}\dot{u}$ and $g_2^{(J)}\dot{u}$ from Definition 3.4.23 are obtained from $g^{(J)}\dot{u}$ simply by replacing some entries with 0. Explicitly, let $(M_{i,j}) := \left[g^{(J)}\dot{u}\right|, (M'_{i,j}) := \left[g_1^{(J)}\dot{u}\right|$, and $(M''_{i,j}) := \left[g_2^{(J)}\dot{u}\right|$ be the corresponding $n \times k$ matrices. Thus $M_{i,j} = \delta_{i,u(j)}$ for all $i \in u[k]$ and $j \in [k]$, and we have

$$M'_{i,j} = \begin{cases} M_{i,j}, & \text{if } i \le u(j), \\ 0, & \text{otherwise,} \end{cases} \qquad M''_{i,j} = \begin{cases} M_{i,j}, & \text{if } i \ge u(j), \\ 0, & \text{otherwise,} \end{cases} \text{ for all } i \in [n] \text{ and } j \in [k]. \end{cases}$$

The operation $M \mapsto M'$, which we call *u*-truncation, will play an important role.

Example 3.9.1. Let G/P = Gr(2, 4) and $u = s_3 s_2 \in W^J$, so $u[k] = \{1, 4\}$. We have

$$x = g^{(J)}\dot{u} = \begin{bmatrix} 1\\x_1 & x_2 & -1\\x_3 & x_4 & -1 \end{bmatrix}, \quad [x] = \begin{bmatrix} 1\\x_1 & x_2\\x_3 & x_4\\1 \end{bmatrix}, \quad \left[g_1^{(J)}\dot{u}\right] = \begin{bmatrix} 1\\x_2\\x_4\\1 \end{bmatrix}, \quad \left[g_2^{(J)}\dot{u}\right] = \begin{bmatrix} 1\\x_1\\x_3\\1 \end{bmatrix}.$$

3.9.5 Positroid varieties

We review the background on *positroid varieties* inside Gr(k, n), which were introduced in [KLS13], building on the work of Postnikov [Pos07]. Let \tilde{S}_n be the group of affine permu*tations*, i.e., bijections $f : \mathbb{Z} \to \mathbb{Z}$ such that f(i+n) = f(i) + n for all $i \in \mathbb{Z}$. We have a function av : $\tilde{S}_n \to \mathbb{Z}$ sending f to av $(f) := \frac{1}{n} \sum_{i=1}^n (f(i) - i)$, which is an integer for all $f \in \tilde{S}_n$. For $j \in \mathbb{Z}$, denote $\tilde{S}_{j,n} := \{ f \in \tilde{S}_n \mid \operatorname{av}(f) = j \}$. Every $f \in \tilde{S}_n$ is determined by the sequence $f(1),\ldots,f(n)$, thus we write f in window notation as $f = [f(1),\ldots,f(n)]$. For $\lambda \in \mathbb{Z}^n$, define $\tau_{\lambda} \in \tilde{S}_n$ by $\tau_{\lambda} := [d_1, \ldots, d_n]$, where $d_i = i + n\lambda_i$ for all $i \in [n]$. Let Bound $(k, n) \subset \tilde{S}_{k,n}$ be the set of bounded affine permutations, which consists of all $f \in \tilde{S}_n$ satisfying $\operatorname{av}(f) = k$ and $i \leq f(i) \leq i + n$ for all $i \in \mathbb{Z}$. The subset $\tilde{S}_{0,n}$ is a Coxeter group with generators $s_1, \ldots, s_{n-1}, s_n = s_0$, where for $i \in [n], s_i : \mathbb{Z} \to \mathbb{Z}$ sends $i \mapsto i+1, i+1 \mapsto i$, and $j \mapsto j$ for all $j \not\equiv i, i+1 \pmod{n}$. We let \leq denote the Bruhat order on $\tilde{S}_{0,n}$, and $\ell : \tilde{S}_{0,n} \to \mathbb{Z}_{\geq 0}$ denote the length function. The sets $\tilde{S}_{0,n}$ and $\tilde{S}_{k,n}$ are in bijection sending $(i \mapsto f(i))$ to $(i \mapsto f(i)+k)$, and thus we get a poset structure and a length function on $\tilde{S}_{k,n}$. When $f \leq g$, we write $g \leq^{\text{op}} f$, and we will be interested in the poset $(\text{Bound}(k, n), \leq^{\text{op}})$, which has a unique maximal element $\tau_k := [1+k, 2+k, \dots, n+k]$. It is known that Bound(k, n) is a lower order ideal of $(\tilde{S}_{k,n}, \leq^{\text{op}})$. We fix $\lambda = 1^k 0^{n-k} := (1, \ldots, 1, 0, \ldots, 0) \in \mathbb{Z}^n$ (with k 1's). Then $\tau_{\lambda} = [1+n, \dots, k+n, k+1, \dots, n]$ is one of the $\binom{[n]}{k}$ minimal elements of (Bound(k, n), \leq^{op}). The group S_n is naturally a subset of $\tilde{S}_{0,n}$, and we have $\tau_k = \tau_\lambda (w^J)^{-1} = \tau_\lambda^J$, where τ_λ^J was introduced in Section 3.7.2.

Given an $n \times k$ matrix M and $i \in [n]$, we let M_i denote the *i*-th row of M. We extend this to all $i \in \mathbb{Z}$ in such a way that $M_{i+n} = (-1)^{k-1}M_i$ for all $i \in \mathbb{Z}$. Thus we view M as a periodic $\mathbb{Z} \times k$ matrix. (The sign $(-1)^{k-1}$ is chosen so that if $M \in \operatorname{Gr}_{\geq 0}(k, n)$, then the matrix with rows M_i, \ldots, M_{i+n-1} belongs to $\operatorname{Gr}_{\geq 0}(k, n)$ for all $i \in \mathbb{Z}$, see Section 3.9.11.) Every $n \times k$ matrix M of rank k gives rise to a map $f_M : \mathbb{Z} \to \mathbb{Z}$ sending $i \in \mathbb{Z}$ to the minimal $j \geq i$ such that M_i belongs to the linear span of M_{i+1}, \ldots, M_j . It is easy to see that $f_M \in \operatorname{Bound}(k, n)$ and f_M depends only on the column span of M. For $h \in \operatorname{Bound}(k, n)$, the *(open) positroid variety* $\mathring{\Pi}_h \subset \operatorname{Gr}(k, n)$ is the subset $\mathring{\Pi}_h := \{M \in \operatorname{Gr}(k, n) \mid f_M = h\}$. Its Zariski closure inside $\operatorname{Gr}(k, n)$ is $\Pi_h = \bigsqcup_{g \leq \operatorname{oph}} \mathring{\Pi}_g$. For $h \in \text{Bound}(k, n)$, define the *Grassmann necklace* $\mathcal{I}_h = (I_a)_{a \in \mathbb{Z}}$ of h by

$$I_a := \{h(i) \mid i < a, \ h(i) \ge a\} \quad \text{for } a \in \mathbb{Z}.$$
(3.9.2)

Then I_a is a k-element subset of [a, a+n), where for $a \leq b \in \mathbb{Z}$ we set $[a, b) := \{a, a+1, \ldots, b-1\}$. For $a \leq b \in \mathbb{Z}$ and $M \in \operatorname{Gr}(k, n)$, define $\operatorname{rk}(M; a, b)$ to be the rank of the submatrix of M with row set [a, b). For $a, b \in \mathbb{Z}$ and $h \in \tilde{S}_n$, define $r_{a,b}(h) := \#\{i < a \mid h(i) \geq b\}$. We describe two well known alternative characterizations of open positroid varieties, see [KLS13].

Proposition 3.9.2. Let $h \in \text{Bound}(k, n)$ and let $\mathcal{I}_h = (I_a)_{a \in \mathbb{Z}}$ be its Grassmann necklace.

- (i) The set H
 _h consists of all M ∈ Gr(k, n) such that for each a ∈ Z, I_a is the lexicographically minimal k-element subset S of [a, a + n) such that the rows (M_i)_{i∈S} are linearly independent.
- (ii) For $M \in Gr(k, n)$, we have $M \in \overset{\circ}{\Pi}_h$ if and only if

$$k - \operatorname{rk}(M; a, b) = r_{a,b}(h) \quad \text{for all } a \le b \in \mathbb{Z}.$$
(3.9.3)

We use window notation for Grassmann necklaces as well, i.e., we write $\mathcal{I}_h = [I_1, \ldots, I_n]$. Recall that we have fixed $\lambda = 1^k 0^{n-k} \in \mathbb{Z}^n$. For $(v, w) \in Q_J$, define $f_{v,w} \in \tilde{S}_n$ by

$$f_{v,w} := v\tau_{\lambda}w^{-1}. \tag{3.9.4}$$

Theorem 3.9.3 ([KLS13]). The map $(v, w) \mapsto f_{v,w}$ gives a poset isomorphism

$$(Q_J, \preceq) \xrightarrow{\sim} (\text{Bound}(k, n), \leq^{\text{op}}).$$

For $(v,w) \in Q_J$, we have $\mathring{\Pi}_{v,w} = \mathring{\Pi}_{f_{v,w}}$ and $\Pi_{v,w} = \Pi_{f_{v,w}}$ as subsets of $G/P = \operatorname{Gr}(k,n)$.

Example 3.9.4. There are *n* positroid varieties of codimension 1, each given by the condition $\Delta_{\{i-k+1,\dots,i\}}^{\text{flag}} = 0$ for some $i \in [n]$. The top element $(\text{id}, w^J) \in Q_J$ covers *n* elements, namely, (s_i, w^J) for $i \in [n-1]$ together with $(\text{id}, s_{n-k}w^J)$. For $i \in [n-1]$, $v = s_i$, and $w = w^J$, we have $f_{v,w} = s_i \tau_{\lambda}^J$, which corresponds to the variety $\Delta_{\{i-k+1,\dots,i\}}^{\text{flag}} = 0$. For the remaining pair $v = \text{id}, w = s_{n-k}w^J$, we have $f_{v,w} = \tau_{\lambda}^J s_{n-k}$, which corresponds to the cell $\Delta_{\{n-k+1,\dots,n\}}^{\text{flag}} = 0$.



Figure 3-2: A Le diagram (bottom left) and the labeling of its squares by simple transpositions (top left). The result of applying the bijection of Theorem 3.9.3 (right). See Example 3.9.6 for details.

Example 3.9.5. It is easy to see directly from (3.9.4) and (3.9.2) that the first element of the Grassmann necklace of $f_{v,w}$ is $I_1 = v[k]$. Similarly, $w[k] = \{i \in [n] \mid f_{v,w}(i) > n\}$.

Example 3.9.6. Elements of Bound(k, n) and Q_J are in bijection with *Le diagrams* of [Pos07]. The bijection between Q_J and the set of Le diagrams is described in [Pos07, §19]: a pair $(v, w) \in Q_J$ gives rise to a Le diagram whose *shape* is a Young diagram inside a $k \times (n - k)$ rectangle, corresponding to the set w[k]. The squares of the Le diagram correspond to the terms in a reduced expression for w, as shown in Figure 3-2 (top left): the box with coordinates (i, j) in matrix notation is labeled by s_{k+j-i} . The terms in the positive subexpression for v inside w correspond to the squares of the Le diagram that are *not* filled with dots, see Figure 3-2 (bottom left). Thus the bijection of Theorem 3.9.3 can be pictorially represented as in Figure 3-2 (right). We refer to [Pos07, §19] or [Wil07, Appendix A] for the precise description. For the example in Figure 3-2, we have $v = s_1$, $w = s_2s_1s_4s_3s_2$, and $f_{v,w} = [3, 4, 7, 5, 6]$ in window notation, which is obtained by following the strands in Figure 3-2 (right) starting from the top.

3.9.6 Polynomial loop group

We explain how the construction in Section 3.7 applies to the case $G/P = \operatorname{Gr}(k, n)$. Recall that $\mathcal{A} := \mathbb{C}[z, z^{-1}]$. Let $\operatorname{GL}_n(\mathcal{A})$ denote the *polynomial loop group of* GL_n , consisting of $n \times n$ matrices with entries in \mathcal{A} whose determinant is a nonzero Laurent monomial in z, i.e., an invertible element of \mathcal{A} . (We use $\operatorname{GL}_n(\mathcal{A})$ instead of $\operatorname{SL}_n(\mathcal{A})$ as the constructions are combinatorially more elegant.) We have a group homomorphism val : $\operatorname{GL}_n(\mathcal{A}) \to \mathbb{Z}$ sending $x \in \operatorname{GL}_n(\mathcal{A})$ to $j \in \mathbb{Z}$ such that det $x = cz^{-j}$ for some $c \in \mathbb{C}^*$, and we let $\operatorname{GL}_n^{(j)}(\mathcal{A}) :=$ $\{x \in \operatorname{GL}_n(\mathcal{A}) \mid \operatorname{val} x = j\}$. The subgroups $\operatorname{GL}_n(\mathcal{A}_+)$ and $\operatorname{GL}_n(\mathcal{A}_-)$ are contained inside the group $\operatorname{GL}_n^{(0)}(\mathcal{A})$ of matrices whose determinant belongs to \mathbb{C}^* . We have subgroups $U(\mathcal{A}_+) := \operatorname{ev}_0^{-1}(U), U_-(\mathcal{A}_-) := \operatorname{ev}_\infty^{-1}(U_-), B(\mathcal{A}_+) := \operatorname{ev}_0^{-1}(B)$ and $B_-(\mathcal{A}_-) := \operatorname{ev}_\infty^{-1}(B_-)$ of $\operatorname{GL}_n^{(0)}(\mathcal{A})$. Thus in the notation of Section 3.7 for $G = \operatorname{SL}_n$, we have $\mathcal{G} = \operatorname{SL}_n(\mathcal{A}) \subsetneq \operatorname{GL}_n^{(0)}(\mathcal{A}),$ $\mathcal{B} = \operatorname{SL}_n(\mathcal{A}) \cap B(\mathcal{A}_+) \subsetneq B(\mathcal{A}_+), \mathcal{U} = U(\mathcal{A}_+),$ and $\mathcal{U}_- = U_-(\mathcal{A}_-).$

To each matrix $x \in \operatorname{GL}_n(\mathcal{A})$, we associate a $\mathbb{Z} \times \mathbb{Z}$ matrix $\tilde{x} = (\tilde{x}_{i,j})_{i,j \in \mathbb{Z}}$ that is uniquely defined by the conditions

- 1. $\tilde{x}_{i,j} = \tilde{x}_{i+n,j+n}$ for all $i, j \in \mathbb{Z}$, and
- 2. the entry $x_{i,j}(z)$ equals the finite sum $\sum_{d \in \mathbb{Z}} \tilde{x}_{i,j+dn} z^d$ for all $i, j \in [n]$.

One can check that if $x = x_1 x_2$, then $\tilde{x} = \tilde{x}_1 \tilde{x}_2$. With this identification, the subgroups \mathcal{U} , \mathcal{U}_- , $B(\mathcal{A}_+)$, and $B_-(\mathcal{A}_-)$ have a very natural meaning. For example, $x \in \operatorname{GL}_n(\mathcal{A})$ belongs to \mathcal{U} if and only if $\tilde{x}_{i,j} = 0$ for i > j and $\tilde{x}_{i,i} = 1$ for all $i \in \mathbb{Z}$. Similarly, $B(\mathcal{A}_+)$ consists of all elements $x \in \operatorname{GL}_n(\mathcal{A})$ such that $\tilde{x}_{i,j} = 0$ for i > j and $\tilde{x}_{i,i} \neq 0$ for all $i \in \mathbb{Z}$.

To each affine permutation $f \in \tilde{S}_{k,n}$, we associate an element $\dot{f} \in \operatorname{GL}_n(\mathcal{A})$ so that the corresponding $\mathbb{Z} \times \mathbb{Z}$ matrix \tilde{f} satisfies $\tilde{f}_{i,j} = 1$ if i = f(j) and $\tilde{f}_{i,j} = 0$ otherwise, for all $i, j \in \mathbb{Z}$. In other words, if for $i, j \in [n]$ there exists $d \in \mathbb{Z}$ such that f(j) = i + dn then $\dot{f}_{i,j}(z) := z^{-d}$, and otherwise $\dot{f}_{i,j}(z) := 0$. Observe that val $\dot{f} = k$ for all $f \in \tilde{S}_{k,n}$, thus $\dot{f} \in \operatorname{GL}_n^{(k)}(\mathcal{A})$. Recall that we have fixed $\lambda = 1^k 0^{n-k} \in \mathbb{Z}^n$. We obtain $\dot{\tau}_{\lambda} = \operatorname{diag}\left(\frac{1}{z}, \ldots, \frac{1}{z}, 1, \ldots, 1\right)$ with k entries equal to $\frac{1}{z}$, and for $u \in W^J$, we therefore get $\dot{\tau}_{u\lambda} = \operatorname{diag}(c_1, \ldots, c_n)$, where $c_i = \frac{1}{z}$ for $i \in u[k]$ and $c_i = 1$ for $i \notin u[k]$.

3.9.7 Affine flag variety

The quotient $\operatorname{GL}_n^{(k)}(\mathcal{A})/B(\mathcal{A}_+)$ is isomorphic to the affine flag variety \mathcal{G}/\mathcal{B} of Section 3.7 for the case $G = \operatorname{SL}_n$. Indeed, $\operatorname{GL}_n^{(0)}(\mathcal{A})$ acts simply transitively on $\operatorname{GL}_n^{(k)}(\mathcal{A})$ and we
clearly have $\operatorname{GL}_{n}^{(0)}(\mathcal{A})/B(\mathcal{A}_{+}) \cong \mathcal{G}/\mathcal{B}$. For $f \leq^{\operatorname{op}} h \in \tilde{S}_{k,n}$ and $g \in \tilde{S}_{k,n}$, we have subsets $\mathring{\mathcal{X}}^{f}, \mathring{\mathcal{X}}_{h}, \mathring{\mathcal{R}}_{h}^{f}, \mathcal{C}_{g} \subset \operatorname{GL}_{n}^{(k)}(\mathcal{A})/B(\mathcal{A}_{+})$ defined by

$$\mathring{\mathcal{X}}^{f} := B(\mathcal{A}_{+}) \cdot \dot{f} \cdot B(\mathcal{A}_{+}) / B(\mathcal{A}_{+}), \qquad \qquad \mathring{\mathcal{X}}_{h} := B_{-}(\mathcal{A}_{-}) \cdot \dot{h} \cdot B(\mathcal{A}_{+}) / B(\mathcal{A}_{+}),$$

$$\mathring{\mathcal{R}}_{h}^{f} := \mathring{\mathcal{X}}_{h} \cap \mathring{\mathcal{X}}^{f}, \qquad \qquad \mathcal{C}_{g} := \dot{g} \cdot B_{-}(\mathcal{A}_{-}) \cdot B(\mathcal{A}_{+}) / B(\mathcal{A}_{+}).$$

Let us now calculate the map φ_u from (3.7.11). Recall that it sends $xP \in C_u^{(J)}$ to $g_1^{(J)} \cdot \dot{\tau}_{u\lambda} \cdot (g_2^{(J)})^{-1}$. Assuming as before that $x = g^{(J)}\dot{u} \in \dot{u}U_-^{(J)}$, consider the corresponding $n \times k$ matrix $(M_{i,j}) := [x]$ in u[k]-echelon form.

Proposition 3.9.7. The matrix $y := \varphi_u(xP) \in \operatorname{GL}_n^{(k)}(\mathcal{A})$ is given for all $i, j \in [n]$ by

$$y_{i,j}(z) = \begin{cases} \delta_{i,j}, & \text{if } j \notin u[k], \\ -M_{i,s}, & \text{if } i > j \text{ and } j = u(s) \text{ for some } s \in [k], \\ \frac{M_{i,s}}{z}, & \text{if } i \leq j \text{ and } j = u(s) \text{ for some } s \in [k]. \end{cases}$$

$$(3.9.5)$$

Proof. This follows by directly computing the product $g_1^{(J)} \cdot \dot{\tau}_{u\lambda} \cdot (g_2^{(J)})^{-1}$.

Example 3.9.8. In the notation of Example 3.9.1, we have

$$y = g_1^{(J)} \cdot \dot{\tau}_{u\lambda} \cdot (g_2^{(J)})^{-1} = \begin{bmatrix} 1 & & \\ 1 & & \\ & 1 & \\ & & 1 \end{bmatrix} \cdot \begin{bmatrix} \frac{1}{z} & & \\ & 1 & \\ & & \frac{1}{z} \end{bmatrix} \cdot \begin{bmatrix} 1 & & & \\ -x_1 & 1 & & \\ -x_3 & 1 & 1 \\ & & & 1 \end{bmatrix} = \begin{bmatrix} \frac{1}{z} & & & \\ -x_1 & 1 & \frac{x_2}{z} \\ -x_3 & 1 & \frac{x_4}{z} \\ & & & \frac{1}{z} \end{bmatrix} .$$
(3.9.6)

Remark 3.9.9. The map $\bar{\varphi}_u : xP \mapsto g_1^{(J)} \cdot \dot{\tau}_{u\lambda} \cdot (g_2^{(J)})^{-1} \cdot B(\mathcal{A}_+)$ is a slight variation of a similar embedding of [Sni10] which we denote $\bar{\varphi}'_u$. We have $\bar{\varphi}'_u(xP) = g_1^{(J)} \cdot \dot{\tau}_{u\lambda} \cdot g_2^{(J)} \cdot B(\mathcal{A}_+)$, and the corresponding matrix $y' = \varphi'_u(xP) := g_1^{(J)} \cdot \dot{\tau}_{u\lambda} \cdot g_2^{(J)}$ is given by (3.9.5) except that $-M_{i,s}$ should be replaced by $M_{i,s}$. Thus y' is obtained from y by substituting $z \mapsto -z$ and then changing the signs of all columns in u[k]. In particular, y' and y are related by an element of the affine torus from Section 3.8.2. Proposition 3.9.14 below is due to Snider [Sni10]. Theorem 3.7.2(1) generalizes Snider's result to arbitrary G/P.

We give a standard convenient characterization of $\mathring{\mathcal{X}}_h$ using *lattices*. For each $x \in \mathrm{GL}_n(\mathcal{A})$ and column $a \in \mathbb{Z}$, we introduce a Laurent polynomial $x_a(t) \in \mathbb{C}[t, t^{-1}]$ defined by $x_a(t) :=$ $\sum_{i\in\mathbb{Z}} \tilde{x}_{i,a}t^i$, and an infinite-dimensional linear subspace $L_a(x) \subset \mathbb{C}[t, t^{-1}]$ given by $L_a(x) :=$ Span $\{x_j(t) \mid j < a\}$, where Span denotes the space of all finite linear combinations. For $b \in \mathbb{Z}$, define another linear subspace $E_b \subset \mathbb{C}[t, t^{-1}]$ by $E_b :=$ Span $\{t^i \mid i \geq b\}$. Finally, for $a, b \in \mathbb{Z}$, define $r_{a,b}(x) \in \mathbb{Z}$ to be the dimension of $L_a(x) \cap E_b$. In other words, $r_{a,b}(x)$ is the dimension of the space of $\mathbb{Z} \times 1$ vectors that have zeros in rows $b - 1, b - 2, \ldots$ and can be obtained as finite linear combinations of columns $a - 1, a - 2, \ldots$ of \tilde{x} . Recall from Section 3.9.5 that for $a, b \in \mathbb{Z}$ and $h \in \tilde{S}_n$, we denote $r_{a,b}(h) := \#\{i < a \mid h(i) \geq b\}$.

Lemma 3.9.10. Let $x \in \operatorname{GL}_n^{(d)}(\mathcal{A})$ and $h \in \tilde{S}_{d,n}$ for some $d \in \mathbb{Z}$. Then

$$x \cdot B(\mathcal{A}_+) \in \check{\mathcal{X}}_h$$
 if and only if $r_{a,b}(x) = r_{a,b}(h)$ for all $a, b \in \mathbb{Z}$. (3.9.7)

Proof. It is clear that $r_{a,b}(x) = r_{a,b}(h)$ when $x = \dot{h}$. One can check that $r_{a,b}(y_-xy_+) = r_{a,b}(x)$ for all $x \in \operatorname{GL}_n^{(d)}(\mathcal{A}), y_- \in B_-(\mathcal{A}_-), y_+ \in B(\mathcal{A}_+)$, and $a, b \in \mathbb{Z}$. This proves (3.9.7) since $\operatorname{GL}_n^{(d)}(\mathcal{A})/B(\mathcal{A}_+) = \bigsqcup_{h \in \tilde{S}_{d,n}} \mathring{\mathcal{X}}_h$ by (3.A.2).

Remark 3.9.11. A *lattice* \mathcal{L} is usually defined (see e.g. [Kum02, §13.2.13]) to be a free $\mathbb{C}[[z]]$ -submodule of $\mathbb{C}((t)) \cong \mathbb{C}((z))^n$ (where $z = t^n$) satisfying $\mathcal{L} \otimes_{\mathbb{C}[[z]]} \mathbb{C}((z)) \cong \mathbb{C}((z))^n$. The $\mathbb{C}[[z]]$ -submodule generated by our $L_a(x)$ gives a lattice $\mathcal{L}_a(x)$ in the usual sense.

Definition 3.9.12. Suppose we are given an $n \times k$ matrix M in u[k]-echelon form. Recall that we have defined the row M_a for all $a \in \mathbb{Z}$ in such a way that $M_{a+n} = (-1)^{k-1}M_a$. For $a \in \mathbb{Z}$ and $j \in [k]$, denote by $\theta_{a,j}^u \in [a, a+n)$ the unique integer that is equal to u(j) modulo n. Define the *u*-truncation $M^{\operatorname{tr}_u^a}$ of M to be the $[a, a+n) \times k$ matrix $M^{\operatorname{tr}_u^a} = (M_{i,j}^{\operatorname{tr}_u^a})$ such that for $i \in [a, a+n)$ and $j \in [k]$, the entry $M_{i,j}^{\operatorname{tr}_u^a}$ is equal to $M_{i,j}$ if $i \leq \theta_{a,j}^u$ and to 0 otherwise, see Example 3.9.18. Thus $M^{\operatorname{tr}_u^a}$ is obtained from the matrix with rows M_a, \ldots, M_{a+n-1} by setting an entry to 0 if it is below the corresponding ± 1 in the same column, and we label its rows by $a, \ldots, a+n-1$ rather than by $1, \ldots, n$. For example, if $x = g^{(J)}\dot{u}$ and M = [x]then $M^{\operatorname{tr}_u^1} = \left[g_1^{(J)}\dot{u}\right]$, cf. Example 3.9.1.

Lemma 3.9.13. Let $x = g^{(J)}\dot{u} \in \dot{u}U_{-}^{(J)}$, M := [x], and $y := \varphi_u(xP)$. Then for all $a \in \mathbb{Z}$,

the space $L_a(y)$ has a basis

$$\{t^i \mid i < a\} \sqcup \{P_1(t), \dots, P_k(t)\}, \quad where \quad P_s(t) := \sum_{i=a}^{a+n-1} M_{i,s}^{\operatorname{tr}_u^a} t^i \quad for \ s \in [k].$$
(3.9.8)

Proof. For a subset $S \subset \mathbb{Z}$, define $S + n\mathbb{Z} := \{j + in \mid j \in S, i \in \mathbb{Z}\}$. The space $L_a(y)$ is the span of $y_j(t)$ for all j < a. If $j \notin u[k] + n\mathbb{Z}$ then $y_j(t) = t^j$ by definition. If $j \in u[k] + n\mathbb{Z}$ then $y_{j-n}(t) = t^j + \sum_{j-n < i < j} c_i t^i$, where c_i is zero for $i \in u[k] + n\mathbb{Z}$. It follows that $L_a(y)$ contains t^i for all i < a. Moreover, the only indices j < a such that $y_j(t) \notin \text{Span}\{t^i \mid i < a\}$ are those that belong to $[a - n, a) \cap (u[k] + n\mathbb{Z})$. Let $j \in [a - n, a) \cap (u[k] + n\mathbb{Z})$ be such an index, and let $s \in [k]$ be the unique index such that $u(s) \in j + n\mathbb{Z}$. Then clearly $y_j(t) \pm P_s(t) \in \text{Span}\{t^i \mid i < a\}$, where the sign depends on the parity of $\frac{j-u(s)}{n} \in \mathbb{Z}$. Thus $P_s(t) \in L_a(y)$ for all $s \in [k]$, and $L_a(y)$ is the span of $\{t^i \mid i < a\} \sqcup \{P_1(t), \ldots, P_k(t)\}$. Since the Laurent polynomials $P_s(t)$ have different degrees, they must be linearly independent. \Box

We give an alternative proof of Theorem 3.7.2(1) for the case G/P = Gr(k, n).

Proposition 3.9.14. For $h \in \text{Bound}(k, n)$ such that $\tau_{u\lambda} \leq {}^{\text{op}} h$, the map $\overline{\varphi}_u$ gives isomorphisms

$$\bar{\varphi}_u: C_u^{(J)} \xrightarrow{\sim} \mathring{\mathcal{X}}^{\tau_{u\lambda}}, \quad \bar{\varphi}_u: C_u^{(J)} \cap \mathring{\Pi}_h \xrightarrow{\sim} \mathring{\mathcal{R}}_h^{\tau_{u\lambda}}.$$

Proof. It is clear from (3.9.5) that we have a biregular isomorphism $U_1^{(J)} \times U_2^{(J)} \xrightarrow{\sim} \mathcal{U}_1(\tau_{u\lambda})$ sending $(g_1^{(J)}, g_2^{(J)}) \mapsto g_1^{(J)} \cdot \dot{\tau}_{u\lambda} (g_2^{(J)})^{-1} \dot{\tau}_{u\lambda}^{-1}$. Thus the map $(g_1^{(J)}, g_2^{(J)}) \mapsto g_1^{(J)} \cdot \dot{\tau}_{u\lambda} \cdot (g_2^{(J)})^{-1} \cdot B(\mathcal{A}_+)$ gives a parametrization of $\mathcal{X}^{\tau_{u\lambda}}$, see (3.7.5). Since $C_u^{(J)} = \bigsqcup_{h \in \text{Bound}(k,n)} (C_u^{(J)} \cap \mathring{\Pi}_h)$, let us fix $h \in \text{Bound}(k,n)$ and $x = g^{(J)}\dot{u} \in \dot{u}U_-^{(J)}$. Denote M := [x| and $y := \varphi_u(xP)$. By (3.9.3), we have $M \in \mathring{\Pi}_h$ if and only if $k - \operatorname{rk}(M; a, b) = r_{a,b}(h)$ for all $a \leq b \in \mathbb{Z}$. By (3.9.7), we have $y \cdot B(\mathcal{A}_+) \in \mathring{\mathcal{X}}_h$ if and only if $r_{a,b}(y) = r_{a,b}(h)$ for all $a, b \in \mathbb{Z}$. If a > b then $r_{a,b}(y) = r_{a,b+1}(y) + 1$ by (3.9.8) and $r_{a,b}(h) = r_{a,b+1}(h) + 1$ since $h \in \operatorname{Bound}(k, n)$ satisfies $h^{-1}(b) \leq b$, so $h^{-1}(b) < a$. We have shown that $y \cdot B(\mathcal{A}_+) \in \mathring{\mathcal{X}}_h$ if and only if $r_{a,b}(y) = r_{a,b}(h)$ for all $a \leq b \in \mathbb{Z}$.

$$r_{a,b}(y) + \operatorname{rk}(M; a, b) = k \quad \text{for all } a \le b \in \mathbb{Z}.$$
(3.9.9)

By (3.9.8), $r_{a,b}(y)$ is the dimension of $\text{Span}\{P_1(t), \ldots, P_k(t)\} \cap E_b$. By the rank-nullity theorem, $k - r_{a,b}(y)$ is the rank of the submatrix of $M^{\text{tr}_a^a}$ with row set [a, b), which is obtained by downward row operations from the submatrix of M with row set [a, b). This shows (3.9.9).

Remark 3.9.15. By Theorem 3.7.2(1), the image of $\bar{\varphi}_u$ is $\mathcal{X}_{\tau_\lambda} \cap \mathcal{X}^{\tau_{u\lambda}}$, where $\tau_\lambda^J = \tau_\lambda(w^J)^{-1}$. But recall from Section 3.9.5 that $\tau_\lambda(w^J)^{-1} = \tau_k$, and since \mathcal{X}_{τ_k} is dense in $\mathrm{GL}_n^{(k)}(\mathcal{A})/B(\mathcal{A}_+)$, we find that $\mathcal{X}_{\tau_\lambda} \cap \mathcal{X}^{\tau_{u\lambda}} = \mathcal{X}^{\tau_{u\lambda}}$.

Example 3.9.16. Suppose that $x = g^{(J)}\dot{u}$ is given in Example 3.9.1, then $y = \varphi_u(xP)$ is the matrix from Example 3.9.8. It is clear that $y \in B(\mathcal{A}_+) \cdot \dot{\tau}_{u\lambda}$ regardless of the values of x_1, x_2, x_3, x_4 , and therefore $y \cdot B(\mathcal{A}_+)$ belongs to $\mathring{\mathcal{X}}^{\tau_{u\lambda}}$. We can try to factorize y as an element of $B_-(\mathcal{A}_-) \cdot \dot{\tau}_k \cdot B(\mathcal{A}_+)$:

$$y = \begin{bmatrix} 1 & \frac{x_2}{(x_1x_4 - x_2x_3)z} & \frac{x_4}{x_3z} \\ \frac{1}{x_2} & \frac{x_4}{x_1x_4 - x_2x_3} & 1 \end{bmatrix} \cdot \begin{bmatrix} \frac{1}{z} \\ \frac{1}{z} \\ 1 \\ 1 \end{bmatrix} \cdot \begin{bmatrix} \frac{x_1x_4 - x_2x_3}{x_2} & -\frac{x_4}{x_2} & 1 \\ \frac{x_3}{x_1x_4 - x_2x_3} & -\frac{x_1}{x_1x_4 - x_2x_3} \\ \frac{1}{x_3} \\ -x_1z & z & x_2 \end{bmatrix}.$$

This factorization makes sense only when all denominators on the right-hand side are nonzero, which shows that $y \cdot B(\mathcal{A}_+) \in \mathring{\mathcal{R}}_{\tau_k}^{\tau_{u\lambda}}$ whenever the minors $\Delta_{12}^{\text{flag}}(x) = x_2$, $\Delta_{23}^{\text{flag}} = x_1 x_4 - x_2 x_3$, and $\Delta_{34}^{\text{flag}} = x_3$ are nonzero. Observe also that $\Delta_{14}^{\text{flag}}(x) = 1$. Thus $y \cdot B(\mathcal{A}_+) \in \mathring{\mathcal{R}}_{\tau_k}^{\tau_{u\lambda}}$ precisely when $xP \in \mathring{\Pi}_{\tau_k}$, where $\tau_k = [3, 4, 5, 6]$ in window notation. If $x_2 = 0$ then $xP \in \mathring{\Pi}_h$ for h = [2, 4, 5, 7]. In this case, we have

$$\dot{h} = \begin{bmatrix} \frac{1}{z} \\ 1 \\ 1 \\ 1 \end{bmatrix}, \quad y|_{x_2=0} = \begin{bmatrix} 1 & -\frac{1}{x_1 z} & \frac{x_4}{x_3 z} \\ 1 \\ -\frac{x_3}{x_1 x_4} & \frac{1}{x_4} \end{bmatrix} \cdot \begin{bmatrix} \frac{1}{z} \\ 1 \\ 1 \\ 1 \end{bmatrix} \cdot \begin{bmatrix} -x_1 & 1 \\ \frac{x_3}{x_1 x_4} & -\frac{1}{x_4} \\ \frac{1}{x_3} \\ -x_3 z & z & x_4 \end{bmatrix}$$

Therefore $y|_{x_2=0}$ belongs to $\mathring{\mathcal{R}}_h^{\tau_{u\lambda}}$ whenever $x_1, x_3, x_4 \neq 0$. Observe that the Grassmann necklace of h is given by $\mathcal{I}_h = [\{1,3\}, \{2,3\}, \{3,4\}, \{4,5\}]$ in window notation, and the corresponding flag minors of $x|_{x_2=0}$ are given by $\Delta_{13}^{\text{flag}} = x_4$, $\Delta_{23}^{\text{flag}} = x_1x_4$, $\Delta_{34}^{\text{flag}} = x_3$, $\Delta_{14}^{\text{flag}} = 1$, in agreement with Proposition 3.9.14.

3.9.8 Preimage of C_q

For this section, we fix $\tau_{u\lambda} \leq^{\operatorname{op}} g \in \operatorname{Bound}(k, n)$. We would like to understand the preimage of $\left(\mathring{\mathcal{X}}^{\tau_{u\lambda}} \cap \mathcal{C}_{g}\right) \subset \operatorname{GL}_{n}^{(k)}(\mathcal{A})/B(\mathcal{A}_{+})$ under the map $\bar{\varphi}_{u}$. For a set $S \subset [a, a + n)$ of size k, define $\Delta_{S}^{\operatorname{tr}_{u}^{a}}(M)$ to be the determinant of the $k \times k$ submatrix of $M^{\operatorname{tr}_{u}^{a}}$ with row set S. Let $\mathcal{I}_{g} = (I_{a})_{a \in \mathbb{Z}}$ be the Grassmann necklace of g.

Proposition 3.9.17. Suppose that $xP \in C_u^{(J)}$ and let $M := \left[g^{(J)}\dot{u}\right]$. Then $\bar{\varphi}_u(xP) \in \mathcal{C}_g$ if and only if $\Delta_{I_a}^{\operatorname{tr}_a^u}(M) \neq 0$ for all $a \in [n]$.

Proof. Let $h \in \tilde{S}_n$ be the unique element such that $\dot{g}^{-1}\bar{\varphi}_u(xP)$ belongs to \mathring{X}_h , thus $\bar{\varphi}_u(xP) \in \mathcal{C}_g$ if and only if $h = \mathrm{id}$. Since val $\varphi_u(xP) = k$ and val $\dot{g}^{-1} = -k$, we get $h \in \tilde{S}_{0,n}$. Hence $h = \mathrm{id}$ if and only if $r_{a,a}(h) = 0$ for all $a \in \mathbb{Z}$. Let $y := \varphi_u(xP)$ and $y' := \dot{g}^{-1}y$. Then for $a \in \mathbb{Z}$, we get $L_a(y') = g^{-1}L_a(y)$, where g^{-1} acts on $\mathbb{C}[t, t^{-1}]$ by a linear map sending $t^j \mapsto t^{g^{-1}(j)}$. In particular, $L_a(y') \cap E_a = (g^{-1}L_a(y)) \cap E_a$ has the same dimension as $L_a(y) \cap gE_a$. Let us denote $H_a := \{t^i \mid i \geq a\}$, so $E_a = \mathrm{Span}(H_a)$ and $gE_a = \mathrm{Span}(gH_a)$. Since $g(i) \geq i$ for all $i \in \mathbb{Z}$, it follows from (3.9.2) that $gH_a = H_a \setminus \{t^j\}_{j \in I_a}$. Therefore by (3.9.8), $L_a(y) \cap gE_a = \{0\}$ if and only if $\mathrm{Span}\{P_j(t)\}_{j \in [k]} \cap \mathrm{Span}(H_a \setminus \{t^j\}_{j \in I_a}) = \{0\}$, which happens precisely when the submatrix of $M^{\mathrm{tr}_a^n}$ with row set I_a is nonsingular, i.e., $\Delta_{I_a}^{\mathrm{tr}_a^n}(M) \neq 0$.

Example 3.9.18. Suppose that x is the matrix from Example 3.9.1, then $y := \varphi_u(xP)$ is given in Example 3.9.8. We have

$$M = \begin{bmatrix} 1 \\ x_1 & x_2 \\ x_3 & x_4 \\ 1 \end{bmatrix}, \quad M^{\text{tr}_u^1} = \begin{bmatrix} 1 \\ x_2 \\ x_4 \\ 1 \end{bmatrix}, \quad M^{\text{tr}_u^2} = \begin{bmatrix} x_1 & x_2 \\ x_3 & x_4 \\ -1 \\ -1 \end{bmatrix}, \quad M^{\text{tr}_u^3} = \begin{bmatrix} x_3 & x_4 \\ -1 \\ -1 \end{bmatrix}, \quad M^{\text{tr}_u^4} = \begin{bmatrix} -1 & 1 \\ -1 \\ -1 \end{bmatrix}.$$

Suppose that g = [2, 4, 5, 7] as in Example 3.9.16, then its Grassmann necklace is $\mathcal{I}_g = [\{1, 3\}, \{2, 3\}, \{3, 4\}, \{4, 5\}]$ in window notation. This gives

$$\Delta_{13}^{\mathrm{tr}_{u}^{1}}(M) = x_{4}, \quad \Delta_{23}^{\mathrm{tr}_{u}^{2}}(M) = x_{1}x_{4} - x_{2}x_{3}, \quad \Delta_{34}^{\mathrm{tr}_{u}^{3}}(M) = x_{3}, \quad \Delta_{45}^{\mathrm{tr}_{u}^{4}}(M) = 1.$$
(3.9.10)

On the other hand, recall from Example 3.9.16 that $\dot{g} = \begin{bmatrix} 1 & \frac{1}{z} \\ 1 & & \\ & 1 \end{bmatrix}$. Since $y \in \mathcal{C}_g$ if and only

if $\dot{g}^{-1}y \in B_{-}(\mathcal{A}_{-}) \cdot B(\mathcal{A}_{+})$, we can factorize it as

$$\dot{g}^{-1}y = \begin{bmatrix} -x_1 & 1 & \frac{x_2}{z} \\ & \frac{1}{z} \\ 1 & -x_3z & z & x_4 \end{bmatrix} = \begin{bmatrix} 1 & \frac{x_2}{x_4z} \\ -\frac{x_3}{x_1x_4 - x_2x_3} & 1 & \frac{1}{x_4z} \\ -\frac{x_4}{x_1x_4 - x_2x_3} & \frac{x_4}{x_3} & 1 \\ & & 1 \end{bmatrix} \cdot \begin{bmatrix} -\frac{x_1x_4 - x_2x_3}{x_4} & 1 & -\frac{x_2}{x_4} \\ & \frac{x_3}{x_1x_4 - x_2x_3} & -\frac{x_1}{x_1x_4 - x_2x_3} \\ & & \frac{1}{x_3} \\ -x_3z & & z & x_4 \end{bmatrix} .$$
(3.9.11)

Again, this is valid only when the denominators in the right-hand side are nonzero. We thus see that $\dot{g}^{-1}y$ belongs to $B_{-}(\mathcal{A}_{-}) \cdot B(\mathcal{A}_{+})$ precisely when all minors in (3.9.10) are nonzero, in agreement with Proposition 3.9.17.

3.9.9 Fomin–Shapiro atlas

The computation in (3.9.11) can now be used to find the maps $\bar{\nu}_g$ and ϑ_g . As in Section 3.8.3, denote by $\mathcal{O}_g \subset C_u^{(J)}$ the preimage of $\mathcal{C}_g \cap \mathring{\mathcal{X}}^{\tau_{u\lambda}}$ under $\bar{\varphi}_u$. Thus for our running example, \mathcal{O}_g is the subset of $C_u^{(J)}$ where all minors in (3.9.10) are nonzero. We are interested in the map $\bar{\nu}_g = (\bar{\nu}_{g,1}, \bar{\nu}_{g,2}) : \mathcal{O}_g \to (\mathring{\Pi}_g \cap \mathcal{O}_g) \times Z_g$ from (3.2.1), defined in Section 3.8.3. The first component is $\bar{\nu}_{g,1} = \bar{\varphi}_u^{-1} \circ \tilde{\nu}_{g,1} \circ \bar{\varphi}_u$, where $\tilde{\nu}_g : \mathcal{C}_g \cap \mathring{\mathcal{X}}^{\tau_{u\lambda}} \xrightarrow{\sim} \mathring{\mathcal{R}}_g^{\tau_{u\lambda}} \times \mathring{\mathcal{X}}^g$ is the map from Proposition 3.8.2(ii). In order to compute it, we consider the factorization $\dot{g}^{-1}y =$ $y_- \cdot y_+ \in \mathcal{U}_- \cdot B(\mathcal{A}_+)$ from (3.9.11). The group $\mathcal{U}_1(g)$ is 1-dimensional since $\ell(g) = 1$, and the corresponding element $y_1 \in \mathcal{U}_1(g)$ from Proposition 3.8.2(ii) can be computed by factorizing $\dot{g}y_-\dot{g}^{-1}$ as an element of $\mathcal{U}_1(g) \cdot \mathcal{U}_2(g)$:

$$\dot{g}y_{-}\dot{g}^{-1} = \begin{bmatrix} 1 - \frac{x_4}{(x_1x_4 - x_2x_3)z} & \frac{x_4}{x_3z} \\ 1 & \frac{x_2}{x_4} \\ -\frac{x_3}{x_1x_4 - x_2x_3} & \frac{1}{x_4} & 1 \end{bmatrix} = \begin{bmatrix} 1 \\ 1 & \frac{x_2}{x_4} \\ 1 \\ 1 \end{bmatrix} \cdot \begin{bmatrix} 1 - \frac{x_4}{(x_1x_4 - x_2x_3)z} & \frac{x_4}{x_3z} \\ 1 \\ -\frac{x_3}{x_1x_4 - x_2x_3} & \frac{1}{x_4} & 1 \end{bmatrix}, \quad y_1 = \begin{bmatrix} 1 \\ 1 - \frac{x_2}{x_4} \\ 1 \\ 1 \end{bmatrix}.$$

Therefore the map $\tilde{\nu}_{g,1}$ sends $y \cdot B(\mathcal{A}_+)$ from (3.9.6) to

$$y_1 y \cdot B(\mathcal{A}_+) = \begin{bmatrix} \frac{1}{z} & & \\ -\frac{x_1 x_4 - x_2 x_3}{x_4} & 1 - \frac{x_2}{x_4} \\ -x_3 & 1 & \frac{x_4}{z} \\ & & & \frac{1}{z} \end{bmatrix} \cdot B(\mathcal{A}_+) = \begin{bmatrix} \frac{1}{z} & & \\ -\frac{x_1 x_4 - x_2 x_3}{x_4} & 1 \\ -x_3 & 1 & \frac{x_4}{z} \\ & & & \frac{1}{z} \end{bmatrix} \cdot B(\mathcal{A}_+).$$

Applying $\bar{\varphi}_u^{-1}$ to the right-hand side, we see that the map $\bar{\nu}_{g,1}$ is given by

$$\bar{\nu}_{g,1}: \mathcal{O}_g \to \overset{\circ}{\Pi}_g \cap \mathcal{O}_g, \quad \begin{bmatrix} 1\\x_1 & x_2\\x_3 & x_4\\ & 1 \end{bmatrix} \mapsto \begin{bmatrix} 1\\\frac{x_1x_4-x_2x_3}{x_4}\\x_3 & x_4\\ & 1 \end{bmatrix}.$$

Similarly, factorizing $\dot{g}y_{-}\dot{g}^{-1}$ as an element of $\mathcal{U}_2(g) \cdot \mathcal{U}_1(g)$, we find that

$$\tilde{\nu}_{g,2}(y \cdot B(\mathcal{A}_+)) = y_2 y \cdot B(\mathcal{A}_+) = \begin{bmatrix} 1 & \frac{x_2}{x_4} \\ & 1 & 1 \end{bmatrix} \cdot \dot{g} \cdot B(\mathcal{A}_+)$$

We have $N_g = \ell(g) = 1$, and the map $\bar{\nu}_{g,2} : \mathcal{O}_g \to Z_g = \mathbb{R}$ sends $\begin{bmatrix} 1 \\ x_1 & x_2 \\ x_3 & x_4 \\ 1 \end{bmatrix} \mapsto \frac{x_2}{x_4}$.

Torus action

We compute the maps from Section 3.8.2. Let $\tilde{\rho} \in Y(\mathcal{T})$ denote the group homomorphism $\tilde{\rho} : \mathbb{C}^* \to \mathbb{C}^* \times T$ sending $t \mapsto \tilde{\rho}(t) := (t^n, \operatorname{diag}(t^{n-1}, \ldots, t, 1))$. If $x \in \operatorname{GL}_n(\mathcal{A})$ is represented by a $\mathbb{Z} \times \mathbb{Z}$ matrix $(\tilde{x}_{i,j})$ then the element $y := \tilde{\rho}(t)x\tilde{\rho}(t)^{-1} \in \operatorname{GL}_n(\mathcal{A})$ satisfies $\tilde{y}_{i,j} = t^{j-i}\tilde{x}_{i,j}$ for all $i, j \in \mathbb{Z}$.

Example 3.9.19. Continuing the above example, we find that

$$\tilde{\rho}(t) \cdot y_2 y \cdot \tilde{\rho}(t)^{-1} \cdot B(\mathcal{A}_+) = \begin{bmatrix} 1 & \frac{tx_2}{x_4} \\ & 1 & 1 \end{bmatrix} \cdot \dot{g} \cdot B(\mathcal{A}_+), \quad \text{and} \quad \|y_2 y \cdot B(\mathcal{A}_+)\| = \frac{|x_2|}{|x_4|}$$

Thus the action of ϑ_g on Z_g is given by $\vartheta_g\left(t, \frac{x_2}{x_4}\right) = \frac{tx_2}{x_4}$. The pullback of this action to $\mathcal{O}_g \subset C_u^{(J)}$ via $\bar{\nu}_g^{-1}$ preserves x_3 , x_4 , and $x_1x_4 - x_2x_3$ (since it preserves $\bar{\nu}_{g,1}(x)$), but multiplies $\frac{x_2}{x_4}$ by t. Therefore it is given by

$$\bar{\nu}_g^{-1} \circ (\mathrm{id} \times \vartheta_g(t, \cdot)) \circ \bar{\nu}_g : \mathcal{O}_g \to \mathcal{O}_g, \quad \begin{bmatrix} 1 \\ x_1 & x_2 \\ x_3 & x_4 \\ 1 \end{bmatrix} \mapsto \begin{bmatrix} 1 \\ x_1 + (t-1) \frac{x_2 x_3}{x_4} & tx_2 \\ x_3 & x_4 \\ 1 \end{bmatrix}.$$

3.9.10 The maps κ and $\zeta_{u,v}^{(J)}$

The subset $\dot{u}G_0^{(J)}$ consists of matrices $x \in G$ such that $\Delta_{u[k]}^{\text{flag}}(x) \neq 0$. Suppose that $x = g^{(J)}\dot{u} \in \dot{u}U_{-}^{(J)}$. Then the elements $g_1^{(J)}\dot{u}$ and $g_2^{(J)}\dot{u}$ are obtained from x by setting some entries to zero, see Section 3.9.4. The map $x \mapsto \kappa_x x$ from Definition 3.4.23 sends $x = g^{(J)}\dot{u}$

to $g_1^{(J)}\dot{u}$, e.g., if $[x] = \begin{bmatrix} 1 \\ x_1 & x_2 \\ x_3 & x_4 \\ 1 \end{bmatrix}$ then $[\kappa_x x] = \begin{bmatrix} 1 \\ x_2 \\ x_4 \\ 1 \end{bmatrix}$ as in Example 3.9.1. Comparing this to Section 3.9.8, we see that if M = [x] is in u[k]-echelon form then $[\kappa_x x]$ is the *u*-truncation $M^{\operatorname{tr}_u^1}$.

Let now $(v, w) \in Q_J^{\succeq(u,u)}$, thus $\tau_{u\lambda} \leq^{\text{op}} g := f_{v,w}$, and denote $\mathcal{I}_g := (I_a)_{a \in \mathbb{Z}}$. The set $G_{u,v}^{(J)}$ from (3.6.1) consists of $x \in G$ such that $\Delta_{u[k]}^{\text{flag}}(x) \neq 0$ and $\Delta_{v[k]}^{\text{flag}}(\kappa_x x) \neq 0$. But recall from Example 3.9.5 that $v[k] = I_1$. Thus

$$G_{u,v}^{(J)} = \left\{ x \in G \mid \Delta_{u[k]}^{\text{flag}}(x) \neq 0 \text{ and } \Delta_{I_1}^{\text{tr}_u^1}(M) \neq 0 \right\}, \text{ where } M := \left[g^{(J)} \dot{u} \right].$$
(3.9.12)

Example 3.9.20. We compute the maps κ and $\zeta_{u,v}^{(J)}$ for our running example. Suppose that $x = g^{(J)}\dot{u}$ is given in Example 3.9.1, and let g = [2,4,5,7] as in Example 3.9.18. Then $g = s_2\tau_k$, so under the correspondence (3.9.4), we have $g = f_{v,w}$ for $v = s_2$ and $w = w^J = s_2s_1s_3s_2$, see also Example 3.9.4. Since $v[k] = I_1 = \{1,3\}$, we see that $x \in G_{u,v}^{(J)}$ whenever $x_4 \neq 0$. We have just computed that $[\kappa_x x] = \begin{bmatrix} 1 & x_2 \\ x_4 \\ 1 \end{bmatrix}$, thus $\dot{v}^{-1}\kappa_x x = \begin{bmatrix} 1 & x_4 \\ -x_2 & 1 \end{bmatrix}^{-1}$. Factorizing it as an element of $U_-^{(J)} \cdot L_J \cdot U^{(J)}$ via (3.9.1), we get

$$\dot{v}^{-1}\kappa_x x = \begin{bmatrix} 1 & & \\ x_4 & -1 \\ -x_2 & 1 \\ 1 \end{bmatrix} = \begin{bmatrix} 1 & & \\ -\frac{x_2}{x_4} & 1 \\ \frac{1}{x_4} & 1 \end{bmatrix} \cdot \begin{bmatrix} 1 & & \\ x_4 & & \\ 1 & -\frac{x_2}{x_4} \\ \frac{1}{x_4} \end{bmatrix} \cdot \begin{bmatrix} 1 & & \\ 1 & -\frac{1}{x_4} \\ 1 & 1 \end{bmatrix}, \quad [\dot{v}^{-1}\kappa_x x]_J = \begin{bmatrix} 1 & & \\ x_4 & & \\ 1 & -\frac{x_2}{x_4} \\ \frac{1}{x_4} \end{bmatrix}.$$

Thus we have computed $\eta(x) = [\dot{v}^{-1}\kappa_x x]_J$ from Definition 3.6.1. Since $x \in \dot{u}U_-^{(J)}$, we use Lemma 3.6.3(ii) to find

$$\zeta_{u,v}^{(J)}(x) = x\eta(x)^{-1} = \begin{bmatrix} 1 \\ x_1 & \frac{x_2}{x_4} & -1 & -x_2 \\ x_3 & 1 & -x_4 \\ \frac{1}{x_4} \end{bmatrix}, \quad \text{thus} \quad \zeta_{u,v}^{(J)}(x)\dot{w}^{-1} = \begin{bmatrix} 1 & 1 \\ -1 & -x_2 & x_1 & \frac{x_2}{x_4} \\ -x_4 & x_3 & 1 \\ \frac{1}{x_4} \end{bmatrix}$$

Therefore the bottom-right principal minors of $\zeta_{u,v}^{(J)}(x)\dot{w}^{-1}$ are

$$\Delta_1^{\pm}(\zeta_{u,v}^{(J)}(x)\dot{w}^{-1}) = \frac{1}{x_4}, \quad \Delta_2^{\pm}(\zeta_{u,v}^{(J)}(x)\dot{w}^{-1}) = \frac{x_3}{x_4}, \quad \Delta_3^{\pm}(\zeta_{u,v}^{(J)}(x)\dot{w}^{-1}) = \frac{x_1x_4 - x_2x_3}{x_4}. \quad (3.9.13)$$

By Proposition 3.9.17, the preimage of C_g under $\bar{\varphi}_u$ is described by $\Delta_{I_a}^{\operatorname{tr}_u^*}(M) \neq 0$ for all $a \in [n]$. Alternatively, as we showed in Section 3.7.7, the preimage of C_g under $\bar{\varphi}_u$ is described

by $\Delta_i^{\pm}(\zeta_{u,v}^{(J)}(x)\dot{w}^{-1}) \neq 0$ for all $i \in [n-1]$. The following result has been computationally checked for all $n \leq 5, k \in [n]$, and $(u, u) \preceq (v, w) \in Q_J$:

Conjecture 3.9.21. Let $(u, u) \preceq (v, w) \in Q_J$. Denote $g := f_{v,w}$, and let $\mathcal{I}_g := (I_a)_{a \in \mathbb{Z}}$ be the Grassmann necklace of g. Suppose that $x = g^{(J)}\dot{u} \in G_{u,v}^{(J)}$ and let M := [x]. Then

$$\Delta_{n+1-i}^{\pm}(\zeta_{u,v}^{(J)}(x)\dot{w}^{-1}) = \frac{\Delta_{I_i}^{\operatorname{tr}_u^i}(M)}{\Delta_{I_1}^{\operatorname{tr}_u^1}(M)} \quad \text{for all } i \in [n].$$
(3.9.14)

For example, compare (3.9.13) with (3.9.10). Recall also that when i = 1, $\Delta_n^{\pm}(\zeta_{u,v}^{(J)}(x)\dot{w}^{-1}) :=$ 1, so in this case (3.9.14) holds trivially.

3.9.11 Total positivity

We recall the background on the totally nonnegative Grassmannian $\operatorname{Gr}_{\geq 0}(k, n)$ of [Pos07]. By a result of Whitney [Whi52], $G_{\geq 0}$ is the set of matrices in $\operatorname{SL}_n(\mathbb{R})$ all of whose minors (of arbitrary sizes) are nonnegative. The following characterizations are well known:

$$(G/B)_{\geq 0} = \left\{ xB \in (G/B)_{\mathbb{R}} \mid \Delta_S^{\text{flag}}(x) \geq 0 \text{ for all } S \subset [n] \right\},$$
(3.9.15)

$$\operatorname{Gr}_{\geq 0}(k,n) = (G/P)_{\geq 0} = \left\{ xP \in (G/P)_{\mathbb{R}} \mid \Delta_S^{\operatorname{flag}}(x) \ge 0 \text{ for all } S \in \binom{[n]}{k} \right\}.$$
(3.9.16)

Eq. (3.9.16) is due to Rietsch, see [Lam16, Rmk. 3.8] for a proof. Eq. (3.9.15) can be easily deduced from the proof of Lemma 3.4.17 combined with the results of [Whi52]. We warn the reader that the analogous statement is false for other choices of J. For instance, when $G = \text{SL}_4$ and $J = \{2\}$, $(G/P)_{\geq 0}$ does not contain all $xP \in (G/P)_{\mathbb{R}}$ such that $\Delta_S^{\text{flag}}(x) \geq 0$ for all $S \in {[n] \choose 1} \cup {[n] \choose 3}$, see [Che11, §10.1].

For $f \in \text{Bound}(k, n)$, we let $\Pi_f^{>0} := \mathring{\Pi}_f \cap \text{Gr}_{\geq 0}(k, n)$ and $\Pi_f^{\geq 0} := \Pi_f \cap \text{Gr}_{\geq 0}(k, n)$, thus for $(v, w) \in Q_J$, we have $\Pi_{f_{v,w}}^{>0} = \Pi_{v,w}^{>0}$ and $\Pi_{f_{v,w}}^{\geq 0} = \Pi_{v,w}^{\geq 0}$ by Theorem 3.9.3.

Recall that if M = [x] is in u[k]-echelon form then $M' := [\kappa_x x]$ equals the *u*-truncation $M^{\operatorname{tr}^1_u}$ of M.

Proposition 3.9.22. Let $\tau_{u\lambda} \leq^{\text{op}} g \leq^{\text{op}} h \in \text{Bound}(k, n)$, and denote by $\mathcal{I}_g := (I_a)_{a \in \mathbb{Z}}$ the Grassmann necklace of g. Suppose that a matrix M in u[k]-echelon form belongs to $\operatorname{Gr}_{\geq 0}(k,n)$. Then

$$M^{\operatorname{tr}_{u}^{a}} \in \operatorname{Gr}_{\geq 0}(k, n) \quad and \quad \Delta_{I_{a}}^{\operatorname{tr}_{u}^{a}}(M) > 0 \quad for \ all \ a \in \mathbb{Z}.$$

$$(3.9.17)$$

Proof. Applying Theorem 3.9.3, we have $(u, u) \preceq (v, w) \preceq (v', w') \in Q_J$, where $g = f_{v,w}$ and $h = f_{v',w'}$. By (3.4.22), we get $v' \leq vr' \leq ur \leq wr' \leq w'$ for some $r, r' \in W_J$.

Suppose first that a = 1. Let $x \in G$ be such that $M = \left[g^{(J)}\dot{u}\right|$ and $xP \in \Pi_h^{>0}$, and denote $M' := M^{\operatorname{tr}_u^1}$. We may assume that $xB \in R_{v',w'}^{>0}$. By Corollary 3.6.10, we find that $\kappa_x xP \in \Pi_{\overline{v}',u}^{>0}$, where $\overline{v}' := v' \triangleleft r_w^{-1}$ for some $r_w \in W_J$ satisfying $r_w \ge r$, see Lemma 3.6.9(ii). This shows that $M' \in \operatorname{Gr}_{\ge 0}(k, n)$. Since $ur \le ur_w$, we find that $ur \triangleleft r_w^{-1} \le u$ by Lemma 3.4.6, therefore $ur \triangleleft r_w^{-1} = u$. Applying $\triangleleft r_w^{-1}$ to $v' \le vr' \le ur$ via Lemma 3.4.6(iii), we see that $\overline{v}' \le (vr' \triangleleft r_w^{-1}) \le u$. Let $v = v_1v_2$ for $v_1 \in W^J$ and $v_2 \in W_J$ be the parabolic factorization of v. Then $vr' \triangleleft r_w^{-1} \in v_1W_J$, thus $(v_1, v_1) \preceq (\overline{v}', u) \in Q_J$, which is equivalent to $\Delta_{v_1[k]}^{\operatorname{fag}}(\kappa_x x) > 0$. From Example 3.9.5 we have that $v[k] = I_1$, and $v_1[k] = v[k]$ since $v \in v_1W_J$, so $\Delta_{I_1}^{\operatorname{tr}_u}(M) = \Delta_{I_1}^{\operatorname{fag}}(\kappa_x x) > 0$. We have shown (3.9.17) for a = 1. Applying the cyclic shift $\chi : \operatorname{Gr}_{\ge 0}(k, n) \to \operatorname{Gr}_{\ge 0}(k, n)$ (which takes M to the matrix with rows $(M_{a+1})_{a\in[n]}$), we obtain (3.9.17) for all $a \in \mathbb{Z}$.

Note that our proof of Proposition 3.9.22 involves a lifting from G/P to G/B, so it does not stay completely inside Gr(k, n).

Problem 3.9.23. Give a self-contained proof of Proposition 3.9.22.

Example 3.9.24. We now consider an example for the case G/P = Gr(2,5). Let $u := s_2 \in W^J$, so $u[k] = \{1,3\}$. Consider $(v', w') \in Q_J$ given by $v' = s_1$, $w' = s_2s_1s_4s_3s_2$ as in Figure 3-2, thus $h := f_{v',w'} = [3,4,7,5,6]$. We use Marsh–Rietsch parametrizations¹ from Section 3.4.9 to compute $x \in G$ such that $xB \in R_{v',w'}^{>0}$ and $xP \in \Pi_h^{>0}$:

$$x := y_2(t_1)\dot{s}_1y_4(t_3)y_3(t_4)y_2(t_5) = \begin{bmatrix} -1 \\ t_1 & t_5 & 1 \\ t_4t_5 & t_4 & 1 \\ t_3t_4t_5 & t_3t_4 & t_3 & 1 \end{bmatrix}, \quad M := \begin{bmatrix} g^{(J)}\dot{u} \end{bmatrix} = \begin{bmatrix} \frac{1}{t_5} & \frac{1}{t_1} \\ \frac{1}{t_1} & \frac{1}{t_1} \\ -t_4t_5 \\ -t_3t_4t_5 \end{bmatrix}.$$

¹For the Grassmannian case, Marsh–Rietsch parametrizations are closely related to *BCFW bridge* parametrizations, see [BCFW05, AHBC⁺16, Kar16].

Observe that $xB \in (G/B)_{\geq 0}$ since all flag minors of x are nonnegative. (For instance, the first column of x consists of nonnegative entries.) In fact, flag minors of x are subtraction-free rational expressions in $\mathbf{t} = (t_1, t_3, t_4, t_5)$, cf. (3.5.19). The $n \times k$ matrix [x] is not in u[k]-echelon form, but the matrix $M := [g^{(J)}\dot{u}]$ is. Up to a common scalar, the 2×2 flag minors of M are the same as the corresponding flag minors of x, however, other (i.e., 1×1) flag minors of M are not necessarily nonnegative. The Grassmann necklace of h is $\mathcal{I}_h = [\{1, 2\}, \{2, 3\}, \{3, 4\}, \{4, 7\}, \{5, 7\}]$. Using Proposition 3.9.2(i), we check that indeed $xP \in \Pi_h^{>0}$.

Let us choose $(v, w) \in Q_J$ for $v = s_2 s_1$, $w = s_2 s_1 s_4 s_3 s_2$, so that $g := f_{v,w} = [2, 4, 8, 5, 6]$. The corresponding Le diagram is obtained from the one in Figure 3-2 (bottom left) by removing the dot in the bottom row. We have $(u, u) \preceq (v, w) \preceq (v', w')$ and $\tau_{u\lambda} \leq^{\text{op}} g \leq^{\text{op}} h$. We compute the elements $\kappa_x = h_2^{(J)} \in U_2^{(J)}$, $\pi_{\dot{u}P_-}(x)$, $\eta(x)$, and $\zeta_{u,v}^{(J)}(x) = \pi_{\dot{u}P_-}(x) \cdot \eta(x)^{-1}$ from Definition 3.6.1:

$$g^{(J)}\dot{u} = \begin{bmatrix} \frac{1}{t_{5}} & \frac{1}{t_{1}} - 1\\ -t_{4}t_{5} & 1\\ -t_{3}t_{4}t_{5} & 1 \end{bmatrix}, \quad \kappa_{x} = \begin{bmatrix} \frac{1}{-\frac{t_{5}}{t_{1}}} & 1\\ 1\\ t_{4}t_{5} & 1\\ t_{3}t_{4}t_{5} & 1 \end{bmatrix}, \quad \kappa_{x}x = \begin{bmatrix} -1\\ 1\\ \frac{t_{5}}{t_{1}} & t_{1}\\ t_{1} & t_{5} & 1\\ t_{3}t_{4}t_{3} & 1 \end{bmatrix}, \quad \pi_{u}x = \begin{bmatrix} 1\\ 1\\ t_{1} & t_{5} & 1\\ t_{1} & t_{5} & 1\\ t_{1} & t_{5} & t_{1}\\ t_{1} & t_{5} & t_{1}\\ t_{3}t_{4}t_{5} & t_{3}t_{4} & t_{3} \end{bmatrix}, \quad \eta(x) = \begin{bmatrix} t_{1} & t_{5} & 1\\ 1\\ t_{1} & t_{5} & t_{1}\\ t_{1} & t_{3}t_{4}t_{5} & t_{3}t_{4} & t_{3} \end{bmatrix}, \quad \eta(x) = \begin{bmatrix} t_{1} & t_{5} & 1\\ 1\\ t_{1} & t_{1} & t_{1}\\ t_{3}t_{4}t_{5} & t_{3}t_{4} & t_{3} \end{bmatrix}, \quad \eta(x) = \begin{bmatrix} t_{1} & t_{5} & 1\\ 1\\ t_{1} & t_{1} & t_{1}\\ t_{3}t_{4}t_{5} & t_{3}t_{4} & t_{3} \end{bmatrix},$$

We see that all flag minors of $\kappa_x x$ are nonnegative, cf. Lemma 3.6.9(ii). Observe that $\kappa_{g^{(J)}\dot{u}} = \kappa_x$ by Lemma 3.6.2(iii), so by Lemma 3.6.3(ii), we could alternatively compute $\zeta_{u,v}^{(J)}(x)$ as the product $g^{(J)}\dot{u} \cdot \eta(g^{(J)}\dot{u})^{-1}$:

$$\eta(g^{(J)}\dot{u}) = \begin{bmatrix} -1 & 1 \\ & 1 \\ & & 1 \end{bmatrix}, \quad \zeta_{u,v}^{(J)}(x) = g^{(J)}\dot{u} \cdot \eta(g^{(J)}\dot{u})^{-1} = \begin{bmatrix} 1 & 1 \\ \frac{t_5}{t_1} & \frac{1}{t_1} - 1 \\ & 1 \\ -t_4t_5 & 1 \\ -t_3t_4t_5 & 1 \end{bmatrix} \cdot \begin{bmatrix} 1 & -1 \\ & 1 \\ & 1 \\ & 1 \end{bmatrix}.$$

Finally, we compute the bottom-right $i \times i$ principal minors of $\zeta_{u,v}^{(J)}(x)\dot{w}^{-1}$ and observe that they are all nonzero subtraction-free expressions in **t**, agreeing with Theorems 3.6.4 and 3.6.14:

$$\zeta_{u,v}^{(J)}(x)\dot{w}^{-1} = \begin{bmatrix} -1 & -1 \\ -1 & \frac{1}{t_1} & -\frac{t_5}{t_1} \\ 1 & 1 & t_4t_5 \\ 1 & -1 & t_3t_4t_5 \end{bmatrix}, \qquad \Delta_1^{\pm}(\zeta_{u,v}^{(J)}(x)\dot{w}^{-1}) = t_3t_4t_5, \qquad \Delta_2^{\pm}(\zeta_{u,v}^{(J)}(x)\dot{w}^{-1}) = t_4t_5, \qquad \Delta_4^{\pm}(\zeta_{u,v}^{(J)}(x)\dot{w}^{-1}) = \frac{t_5}{t_1}.$$

Let us check that this agrees with Conjecture 3.9.21. The Grassmann necklace of g is $\mathcal{I}_g = [\{1,3\},\{2,3\},\{3,4\},\{4,8\},\{5,8\}]$ in window notation. We see that the corresponding u-truncated minors of $M = [g^{(J)}\dot{u}]$ are indeed given by

$$\Delta_{13}^{\operatorname{tr}_{u}^{1}}(M) = 1, \quad \Delta_{23}^{\operatorname{tr}_{u}^{2}}(M) = \frac{t_{5}}{t_{1}}, \quad \Delta_{34}^{\operatorname{tr}_{u}^{3}}(M) = t_{4}t_{5}, \quad \Delta_{48}^{\operatorname{tr}_{u}^{4}}(M) = t_{4}t_{5}, \quad \Delta_{58}^{\operatorname{tr}_{u}^{5}}(M) = t_{3}t_{4}t_{5}.$$

3.10 Further directions

In addition to Theorem 1.2.1 and [Her14], we expect the regularity theorem to hold for many other spaces occurring in total positivity. The most natural immediate direction is total positivity for Kac–Moody flag varieties.

Let \mathcal{G}^{\min} be a minimal Kac–Moody group, $\mathcal{U}^{\min}, \mathcal{U}^{\min}_{-}, \mathcal{B}^{\min}_{-}, \mathcal{B}^{\min}_{-}$ be unipotent and Borel subgroups, and \tilde{W} be the Weyl group as in Section 3.A. Furthermore, let $\mathcal{P}^{\min} \supset \mathcal{B}^{\min}$ denote a standard parabolic subgroup of \mathcal{G}^{\min} (a group of the form $\mathcal{G}^{\min} \cap \mathcal{P}_{Y}$ in the notation of [Kum02]).

Definition 3.10.1. Define the totally nonnegative part $\mathcal{U}_{\geq 0}^-$ of \mathcal{U}_{-}^{\min} to be the subsemigroup generated by $\{x_{\alpha_i}(t) \mid t \in \mathbb{R}_{>0}, \ 1 \leq i \leq r\}$. Define the totally nonnegative part of the flag variety $\mathcal{G}^{\min}/\mathcal{P}^{\min}$ to be the closure $(\mathcal{G}^{\min}/\mathcal{P}^{\min})_{\geq 0} := \overline{\mathcal{U}_{\geq 0}^- \mathcal{P}^{\min}/\mathcal{P}^{\min}}$.

When \mathcal{G}^{\min} is an affine Kac–Moody group of type A, Definition 3.10.1 agrees with the definition of Lam and Pylyavskyy (cf. [LP12, Theorem 2.6]) for the polynomial loop group.

Conjecture 3.10.2 (Regularity conjecture for Kac–Moody groups and flag varieties).

1. The intersection of $\mathcal{U}_{\geq 0}^-$ with the Bruhat stratification $\{\mathcal{B}^{\min}\dot{w}\mathcal{B}^{\min} \mid w \in \tilde{W}\}$ of \mathcal{G}^{\min} endows $\mathcal{U}_{\geq 0}^-$ with an (infinite) cell decomposition with closure partial order equal to the Bruhat order of \tilde{W} . Furthermore, the link of the identity in any (closed) cell is a regular CW complex homeomorphic to a closed ball.

- 2. The intersection of $(\mathcal{G}^{\min}/\mathcal{B}^{\min})_{\geq 0}$ with the open Richardson stratification \mathcal{R}_u^v of $\mathcal{G}^{\min}/\mathcal{B}^{\min}$ endows $(\mathcal{G}^{\min}/\mathcal{B}^{\min})_{\geq 0}$ with the structure of a regular CW complex. The closure partial order is the interval order of the Bruhat order of \tilde{W} , and after adding a minimum, every interval of the closure partial order is thin and shellable.
- 3. The intersection of $(\mathcal{G}^{\min}/\mathcal{P}^{\min})_{\geq 0}$ with the open projected Richardson stratification $\Pi_{v,w}^{\circ}$ of $\mathcal{G}^{\min}/\mathcal{P}^{\min}$ endows $(\mathcal{G}^{\min}/\mathcal{P}^{\min})_{\geq 0}$ with the structure of a regular CW complex. The closure partial order is the natural partial order on \mathcal{P} -Bruhat intervals of \tilde{W} , and after adding a minimum, every interval of the closure partial order is thin and shellable.

Note that every interval in the Bruhat order of \tilde{W} is known to be thin and shellable [BW82]. The stratification $\Pi_{v,w}^{\circ}$ and the \mathcal{P} -Bruhat order can be defined analogously to [KLS14].

We include a list of some other spaces occurring in total positivity which we expect to have a natural regular CW complex structure.

- 1. The totally nonnegative part of double Bruhat cells [FZ99]. It has been expected that a link of a double Bruhat cell inside another double Bruhat cell is a regular CW complex homeomorphic to a closed ball. Our Theorem 3.3.11 confirms this in type A, since double Bruhat cells for GL_n embed in the Grassmannian Gr(n, 2n), see [Pos07, Remark 3.11].
- 2. The space of planar electrical networks from Section 1.5 and the space of planar Ising models from Section 1.6. These spaces are homeomorphic to closed balls by Theorems 1.5.1 and 4.1.3, and have cell decompositions whose face poset is graded, thin, and shellable [HK18].
- 3. Amplituhedra [AHT14] and, more generally, Grassmann polytopes [Lam16]. Grassmann polytopes generalize convex polytopes into the Grassmannian Gr(k, n). The former are well known to be regular CW complexes homeomorphic to closed balls. We caution that not all Grassmann polytopes are balls.
- 4. The totally nonnegative part of the wonderful compactification [He07]. A cell decomposition of this space was constructed in [He07].

We expect that most spaces in this list are TNN spaces that admit a Fomin–Shapiro atlas.

3.A Appendix: Kac–Moody flag varieties

We recall some background on Kac–Moody groups, and refer to [Kum02] for all missing definitions. We start by introducing the minimal Kac–Moody group \mathcal{G}^{\min} and its flag variety $\mathcal{G}^{\min}/\mathcal{B}^{\min}$, and then explain how they relate to the polynomial loop group \mathcal{G} and its flag variety variety \mathcal{G}/\mathcal{B} from Section 3.7.

3.A.1 Kac–Moody Lie algebras

Suppose that \tilde{A} is a generalized Cartan matrix [Kum02, Dfn. 1.1.1]. Thus \tilde{A} is an $r \times r$ integer matrix for some $r \geq 1$. We assume \tilde{A} is symmetrizable, that is, there exists a diagonal matrix $D \in \operatorname{GL}_r(\mathbb{Q})$ such that $D\tilde{A}$ is a symmetric matrix. As in [Kum02, §1.1], denote by \mathfrak{g} the Kac-Moody Lie algebra associated with \tilde{A} , and let $\mathfrak{h} \subset \mathfrak{g}$ be its Cartan subalgebra, whose dual is denoted by \mathfrak{h}^* . Thus \mathfrak{h} and \mathfrak{h}^* are vector spaces over \mathbb{C} of dimension $\tilde{r} := 2r - \operatorname{rank}(\tilde{A})$, and we let $\langle \cdot, \cdot \rangle : \mathfrak{h} \times \mathfrak{h}^* \to \mathbb{C}$ denote the natural pairing.

We let $\Delta \subset \mathfrak{h}^*$ denote the root system of \mathfrak{g} , as defined in [Kum02, §1.2]. Let $\{\alpha_i\}_{i=1}^r \subset \mathfrak{h}^*$ be the simple roots and $\{\alpha_i^{\vee}\}_{i=1}^r \in \mathfrak{h}$ be the simple coroots.

Let $\Delta_{\rm re} \subset \Delta$ denote the set of real roots and $\Delta_{\rm im} \subset \Delta$ denote the set of imaginary roots, so $\Delta = \Delta_{\rm re} \sqcup \Delta_{\rm im}$. Also let $\Delta = \Delta^+ \sqcup \Delta^-$ denote the decomposition of Δ into positive and negative roots, and denote $\Delta_{\rm re}^+ := \Delta^+ \cap \Delta_{\rm re}$ and $\Delta_{\rm re}^- := \Delta_{\rm re} \cap \Delta^-$. Denote by \tilde{W} the Weyl group associated with \tilde{A} as in [Kum02, §1.3]. Thus \tilde{W} acts on Δ , and preserves the subset $\Delta_{\rm re}$. Moreover, \tilde{W} is generated by simple reflections $s_1, \ldots, s_r \in \tilde{W}$, and $(\tilde{W}, \{s_i\}_{i=1}^r)$ is a Coxeter group by [Kum02, Prop. 1.3.21]. We let (\tilde{W}, \leq) denote the Bruhat order on \tilde{W} and $\ell : \tilde{W} \to \mathbb{Z}_{\geq 0}$ denote the length function.

3.A.2 Kac–Moody groups

Let \mathcal{G}^{\min} be the minimal Kac-Moody group associated to \tilde{A} by Kac and Peterson [PK83, KP83], see [Kum02, §7.4]. For each real root $\alpha \in \Delta_{\mathrm{re}}$, there is a one-parameter subgroup $\mathcal{U}_{\alpha} \subset \mathcal{G}^{\min}$ by [Kum02, Dfn. 6.2.7].² For each $\alpha \in \Delta_{\mathrm{re}}$, we fix an isomorphism $x_{\alpha} : \mathbb{C} \xrightarrow{\sim} \mathcal{U}_{\alpha}$

²The results in [Kum02] are usually stated for the maximal Kac–Moody group which he denotes by \mathcal{G} . However, these results apply to \mathcal{G}^{\min} as well, see Remark 3.A.3.

of algebraic groups. Similarly to the subgroups U, U_-, T, B, B_- of G, we have subgroups $\mathcal{U}^{\min}, \mathcal{U}^{\min}_-, \mathcal{T}^{\min}_-, \mathcal{B}^{\min}_-, \mathcal{B}^{\max}_-, \mathcal{B}^{\max}$

We define a bracket closed subset $\Theta \subset \Delta_{\rm re}$ in the same way as in Section 3.4.2, and for a bracket closed subset $\Theta \subset \Delta_{\rm re}^+$ (resp., $\Theta \subset \Delta_{\rm re}^-$), we have a subgroup $\mathcal{U}(\Theta) \subset \mathcal{U}^{\rm min}$ (resp., $\mathcal{U}_{-}(\Theta) \subset \mathcal{U}_{-}^{\rm min}$), generated by \mathcal{U}_{α} for $\alpha \in \Theta$, see [Kum02, Eq. 6.1.1(6)] (resp., [Kum02, §6.2.7]). For $w \in \tilde{W}$, Inv(w) := $\Delta^+ \cap w^{-1}\Delta^- \subset \Delta_{\rm re}^+$ is a bracket closed subset of size $\ell(w)$, cf. [Kum02, Ex. 6.1.5(b)]. We state the Kac–Moody analog of Lemma 3.4.1(i).

Lemma 3.A.1 ([Kum02, Lemma 6.1.4]). Suppose that $\Theta = \bigsqcup_{i=1}^{n} \Theta_i$ and $\Theta, \Theta_1, \ldots, \Theta_n \subset \Delta_{\text{re}}^+$ are finite bracket closed subsets. Then $\mathcal{U}(\Theta), \mathcal{U}(\Theta_1), \ldots, \mathcal{U}(\Theta_n)$ are finite-dimensional unipotent algebraic groups, and the multiplication map gives a biregular isomorphism

$$\mathcal{U}(\Theta_1) \times \cdots \times \mathcal{U}(\Theta_n) \xrightarrow{\sim} \mathcal{U}(\Theta).$$
 (3.A.1)

3.A.3 Kac–Moody flag varieties

The Weyl group \tilde{W} equals $N_{\mathcal{G}^{\min}}(\mathcal{T}^{\min})/\mathcal{T}^{\min}$, where $N_{\mathcal{G}^{\min}}(\mathcal{T}^{\min})$ is the normalizer of \mathcal{T}^{\min} in \mathcal{G}^{\min} , cf. [Kum02, Lemma 7.4.2]. For $f \in \tilde{W}$, we denote by $\dot{f} \in \mathcal{G}^{\min}$ an arbitrary representative of f in $N_{\mathcal{G}^{\min}}(\mathcal{T}^{\min})$.

By [Kum02, Lemma 7.4.2, Ex. 7.4.E(9), and Thm. 5.2.3(g)], we have Bruhat and Birkhoff decompositions of \mathcal{G}^{\min} :

$$\mathcal{G}^{\min} = \bigsqcup_{f \in \tilde{W}} \mathcal{B}^{\min} \dot{f} \mathcal{B}^{\min}, \quad \mathcal{G}^{\min} = \bigsqcup_{h \in \tilde{W}} \mathcal{B}^{\min}_{-} \dot{h} \mathcal{B}^{\min}.$$
(3.A.2)

We let $\mathcal{G}^{\min}/\mathcal{B}^{\min}$ denote the *Kac–Moody flag variety* of \mathcal{G}^{\min} . For each $h, f \in \tilde{W}$, we have Schubert cells $\mathring{\mathcal{X}}^{f} := \mathcal{B}^{\min}\dot{f}\mathcal{B}^{\min}/\mathcal{B}^{\min}$ and opposite Schubert cells $\mathring{\mathcal{X}}_{h} := \mathcal{B}^{\min}_{-}\dot{h}\mathcal{B}^{\min}/\mathcal{B}^{\min}$. If $h \not\leq f \in \tilde{W}$ then by [Kum02, Lemma 7.1.22(b)], $\mathring{\mathcal{X}}^{f} \cap \mathring{\mathcal{X}}_{h} = \emptyset$. For $h \leq f$, we denote by $\mathring{\mathcal{R}}_{h}^{f} := \mathring{\mathcal{X}}_{h} \cap \mathring{\mathcal{X}}^{f}$. Therefore (3.7.3) follows from (3.A.2). The flag variety $\mathcal{G}^{\min}/\mathcal{B}^{\min}$ is a projective ind-variety by [Kum02, §7.1], the Schubert cell $\mathring{\mathcal{X}}^f$ and Schubert variety \mathcal{X}^f are finite-dimensional subvarieties, while the opposite Schubert cell $\mathring{\mathcal{X}}_h$ and opposite Schubert variety \mathcal{X}_h are ind-subvarieties.

Proposition 3.A.2. For $h \leq f \in \tilde{W}$, $\mathcal{X}_h \cap \mathcal{X}^f$ is a closed irreducible $(\ell(f) - \ell(h))$ dimensional subvariety of \mathcal{X}^f , and $\mathring{\mathcal{R}}_h^f$ is an open dense subset of $\mathcal{X}_h \cap \mathcal{X}^f$.

Proof. By (3.7.5), $\mathring{\mathcal{X}}^f$ is $\ell(f)$ -dimensional, and by [Kum02, Lemma 7.3.10], $\mathring{\mathcal{X}}_h \cap \mathcal{X}^f$ has codimension $\ell(h)$ in \mathcal{X}^f . The rest follows by [Kum17, Prop. 6.6].

For $g \in \tilde{W}$, let $\mathcal{C}_g := \dot{g} \mathcal{B}_{-}^{\min} \mathcal{B}^{\min} / \mathcal{B}^{\min}$. We have

$$\mathcal{G}^{\min}/\mathcal{B}^{\min} = \bigsqcup_{h \le f} \mathring{\mathcal{R}}_h^f \quad \text{and} \quad \mathcal{C}_g = \bigsqcup_{h \le g \le f} (\mathcal{C}_g \cap \mathring{\mathcal{R}}_h^f),$$
 (3.A.3)

where the unions are taken over $h, f \in \tilde{W}$. The first part of (3.A.3) follows from (3.A.2), and for the second part, see the proof of Proposition 3.8.2(iii).

Remark 3.A.3. Let $\hat{\mathcal{G}} \supset \mathcal{G}^{\min}$ be the "maximal" Kac–Moody group (denoted \mathcal{G} in [Kum02]) associated to \tilde{A} , and let $\hat{\mathcal{B}} \supset \mathcal{B}^{\min}$ be its standard positive Borel subgroup. Then the standard negative Borel subgroup of $\hat{\mathcal{G}}$ is still \mathcal{B}^{\min}_{-} . By [Kum02, Eq. 7.4.5(2)], we may identify $\mathcal{G}^{\min}/\mathcal{B}^{\min} \xrightarrow{\sim} \hat{\mathcal{G}}/\hat{\mathcal{B}}$. By [Kum02, Eq. 7.4.2(3)], $\hat{\mathcal{X}}^f$ coincides with the variety $\hat{\mathcal{B}}f\hat{\mathcal{B}}/\hat{\mathcal{B}}$ in [Kum02, Dfn. 7.1.13] for $f \in \tilde{W}$. Similarly, for $h \in \tilde{W}$, $\hat{\mathcal{X}}_h = \mathcal{B}^{\min}_{-} \cdot \dot{h}\mathcal{B}^{\min}/\mathcal{B}^{\min}$ coincides with the variety $\mathcal{B}^h_{\emptyset} := \mathcal{B}^{\min}_{-}h\hat{\mathcal{B}}/\hat{\mathcal{B}}$ defined in the last paragraph of [Kum02, §7.1.20].

3.A.4 Affine Kac–Moody groups and polynomial loop groups

Suppose that \tilde{A} is the affine Cartan matrix associated to a simple and simply-connected algebraic group G. Thus we have r = |I| + 1, $\tilde{r} = |I| + 2$, and \tilde{A} is defined by [Kum02, Eq. 13.1.1(7)]. Let \mathcal{G} denote the polynomial loop group from Section 3.7. Our goal is to explain that the flag varieties \mathcal{G}/\mathcal{B} and $\mathcal{G}^{\min}/\mathcal{B}^{\min}$ are isomorphic.

Let $C \subset T \subset G$ be the center of G, and let $\tilde{C} \subset \mathcal{T}^{\min} \subset \mathcal{G}^{\min}$ be the center of \mathcal{G}^{\min} , see [Kum02, Lemma 6.2.9(c)]. By [Kum02, Cor. 13.2.9], there exists a surjective group homomorphism $\psi : \mathcal{G}^{\min} \to (\mathbb{C}^* \ltimes \mathcal{G})/C$ with kernel \tilde{C} , where \mathbb{C}^* acts on \mathcal{G} as in Section 3.8.2, see also [Kum02, Dfn. 13.2.1]. The groups $\mathcal{U}, \mathcal{U}_- \subset \mathcal{G}$ are identified with the groups $\mathcal{U}^{\min}, \mathcal{U}_-^{\min} \subset \mathcal{G}^{\min}$, and we have $\mathcal{T}/C \cong \mathcal{T}^{\min}/\tilde{C}$. Thus ψ induces an isomorphism $\mathcal{G}^{\min}/\mathcal{B}^{\min} \xrightarrow{\sim} \mathcal{G}/\mathcal{B}$ between the affine Kac–Moody flag variety and the affine flag variety. The Weyl groups \tilde{W} of \mathcal{G} and \mathcal{G}^{\min} are isomorphic by [Kum02, Prop. 13.1.7], and the root systems Δ coincide by [Kum02, Cor. 13.1.4]. Therefore the subsets $\mathcal{X}^f, \mathcal{X}_h, \mathcal{R}_h^f, \mathcal{C}_g$ of \mathcal{G}/\mathcal{B} get sent by ψ to the corresponding subsets of $\mathcal{G}^{\min}/\mathcal{B}^{\min}$. As explained in the last paragraph of [Kum02, §13.2.8], G can be viewed as a subset of \mathcal{G}^{\min} as well, and the restriction of ψ to G is the quotient map $G \to G/C$.

We justify some other statements that we used in Sections 3.7.1 and 3.8.2. For (3.7.2), see [Kum02, §13.1]. For (3.7.6), see [Kum02, §6.1.13]. For a description of $Y(\mathcal{T})$ from Section 3.8.2, see [Kum02, §13.2.2]. For a description of the pairing $\langle \cdot, \cdot \rangle : Y(\mathcal{T}) \times X(\mathcal{T}) \to \mathbb{Z}$ in the same section, see [Kum02, §13.1.1].

3.A.5 Gaussian decomposition and affine charts

By [Kum02, Thm. 7.4.14], \mathcal{G}^{\min} is an *affine ind-group*. Similarly, $\mathcal{U}_{-}^{\min}, \mathcal{U}_{-}^{\min}, \mathcal{T}, \mathcal{B}^{\min}, \mathcal{B}_{-}^{\min}$ are affine ind-groups, see e.g. [Kum02, §7.4] and [Kum02, Cor. 7.3.8].

Let $\mathcal{G}_0^{\min} := \mathcal{B}_-^{\min} \mathcal{B}^{\min}$ and $g \in \tilde{W}$. Recall the subgroups $\mathcal{U}_1(g)$ and $\mathcal{U}_2(g)$ from (3.7.4). Then $\mathcal{U}_1(g)$ is a closed $\ell(g)$ -dimensional subgroup of $\mathcal{U}^{\min} \cong \mathcal{U}$ and $\mathcal{U}_2(g)$ is a closed indsubgroup of $\mathcal{U}_-^{\min} \cong \mathcal{U}_-$.

Proof of Lemma 3.8.1. For (i), see [Kum02, Prop. 7.4.11]. For (ii), we use an argument given in [Wil13, Prop. 2.5]: both maps are bijective morphisms by [Kum02, Lemma 6.1.3]. In particular, it follows that $\dot{g}\mathcal{U}_{-}^{\min}\dot{g}^{-1} \subset \mathcal{G}_{0}^{\min}$ and for $x \in \dot{g}\mathcal{U}_{-}^{\min}\dot{g}^{-1}$, we have $[x]_{0} = 1$. The inverse maps are given by $\mu_{21}^{-1}(x) = ([x]_{-}, [x]_{+}), \ \mu_{12}^{-1}(x) = ([x^{-1}]_{+}^{-1}, [x^{-1}]_{-}^{-1})$. They are regular morphisms by (i), which proves (ii).

Proof of (3.7.5). The map $\dot{g}\mathcal{U}_{-}^{\min}\dot{g}^{-1} \xrightarrow{\sim} \mathcal{C}_g$ is a biregular isomorphism for $g = \mathrm{id}$ by [Kum02, Lemma 7.4.10]. Since \tilde{W} acts on $\mathcal{G}^{\min}/\mathcal{B}^{\min}$ by left multiplication, the case of general $g \in \tilde{W}$ follows as well. Since $\mathcal{U}_1(g)$, $\mathcal{U}_2(g)$ are closed ind-subvarieties of $\dot{g}\mathcal{U}_{-}^{\min}\dot{g}^{-1}$ and $\mathring{\mathcal{X}}^g$, $\mathring{\mathcal{X}}_g$ are closed ind-subvarieties of \mathcal{C}_g , it suffices to show that the image of $\mathcal{U}_1(g)$ equals $\mathring{\mathcal{X}}^g$ while the image of $\mathcal{U}_2(g)$ equals $\overset{\circ}{\mathcal{X}}_g$. By [Kum02, Ex. 7.4.E(9) and Eq. 5.2.3(11)], we have

$$\mathcal{U}^{\min} = (\mathcal{U}^{\min} \cap \dot{g}\mathcal{U}^{\min}_{-}\dot{g}^{-1}) \cdot (\mathcal{U}^{\min} \cap \dot{g}\mathcal{U}^{\min}\dot{g}^{-1}) = \mathcal{U}_1(g) \cdot (\mathcal{U}^{\min} \cap \dot{g}\mathcal{U}^{\min}\dot{g}^{-1}),$$
$$\mathcal{U}^{\min}_{-} = (\mathcal{U}^{\min}_{-} \cap \dot{g}\mathcal{U}^{\min}_{-}\dot{g}^{-1}) \cdot (\mathcal{U}^{\min}_{-} \cap \dot{g}\mathcal{U}^{\min}\dot{g}^{-1}) = \mathcal{U}_2(g) \cdot (\mathcal{U}^{\min}_{-} \cap \dot{g}\mathcal{U}^{\min}\dot{g}^{-1}).$$

Thus

$$\mathcal{B}^{\min} \dot{g} \mathcal{B}^{\min} = \mathcal{U}_1(g) \cdot (\mathcal{U}^{\min} \cap \dot{g} \mathcal{U}^{\min} \dot{g}^{-1}) \cdot \dot{g} \mathcal{B}^{\min} = \mathcal{U}_1(g) \cdot \dot{g} \cdot \mathcal{B}^{\min},$$
$$\mathcal{B}^{\min}_{-} \dot{g} \mathcal{B}^{\min} = \mathcal{U}_2(g) \cdot (\mathcal{U}^{\min}_{-} \cap \dot{g} \mathcal{U}^{\min} \dot{g}^{-1}) \cdot \dot{g} \mathcal{B}^{\min} = \mathcal{U}_2(g) \cdot \dot{g} \cdot \mathcal{B}^{\min}.$$

4

Ising model

The goal of this chapter is to describe a cell decomposition of the space $\overline{\mathcal{X}}_n$ of planar Ising networks from Section 1.6, describe this space by inequalities, and show that it is homeomorphic to a ball. As we mentioned in Section 1.6, we also recognize Kramers–Wannier's duality [KW41] as the cyclic shift on $\operatorname{Gr}_{\geq 0}(n, 2n)$ in Theorem 4.2.4. We then explain (Remark 4.2.6) the connection between the planar Ising model at critical temperature and the unique cyclically symmetric point $X_0 \in \operatorname{Gr}_{\geq 0}(n, 2n)$ from Section 2.2.1. We also express (Theorem 4.2.13) generalized Griffiths' inequalities of [Gri67, KS68] as manifestly positive linear combinations of the Plücker coordinates of our embedding. We explain in Corollary 4.2.9 how the known formula for Plücker coordinates in terms of the dimer model gives a new expression for boundary correlation functions, which is related to Dubédat's formula for squared correlation functions [Dub11]. Finally, we solve the *inverse problem* in Section 4.2.5: given a boundary correlation matrix $M \in \operatorname{Mat}_n(\mathbb{R})$ of the Ising model on a planar graph G embedded in a disk, we show that if G is *reduced* then the edge weights of the Ising model are uniquely and explicitly determined by M.

This chapter is organized as follows. We state our main result (Theorem 4.1.3) in Section 4.1, and then list several applications of our construction in Section 4.2. We give some background on the totally nonnegative Grassmannian in Section 4.3, and study the totally nonnegative orthogonal Grassmannian in Section 4.4. After that, we prove our main results. In Section 4.5, we show that the formula for boundary correlations in terms of the dimer model indeed yields the same result as the embedding ϕ from Section 4.1. In Section 4.6, we prove that $\overline{\mathcal{X}}_n$ is homeomorphic to a ball and discuss the cyclic symmetry of this space. We explain how to express generalized Griffiths' inequalities as positive sums of Plücker coordinates in Section 4.7, and list several conjectures in Section 4.8.

4.1 Main results

We give some background necessary to formulate the main results of this chapter.

4.1.1 The Ising model

A planar Ising network is a pair N = (G, J) where G = (V, E) is a planar graph embedded in a disk and $J : E \to \mathbb{R}_{>0}$ is a function assigning positive real numbers to the edges of G. We always label the vertices of G on the boundary of the disk by $b_1, \ldots, b_n \in V$ in counterclockwise order. Given a planar Ising network N = (G, J), the Ising model on N (with no external field and free boundary conditions) is a probability measure on the space $\{-1, 1\}^V$ of spin configurations on the vertices of G. Given a spin configuration $\sigma: V \to \{-1, 1\}$, its probability is given by

$$\mathbf{P}(\sigma) := \frac{1}{Z} \prod_{\{u,v\} \in E} \exp\left(J_{\{u,v\}}\sigma_u\sigma_v\right), \qquad (4.1.1)$$

where Z is the partition function:

$$Z := \sum_{\sigma \in \{-1,1\}^V} \prod_{\{u,v\} \in E} \exp\left(J_{\{u,v\}}\sigma_u\sigma_v\right).$$

$$(4.1.2)$$

Our main focus will be boundary two-point correlation functions. Let $[n] := \{1, 2, ..., n\}$. Given $i, j \in [n]$, we define the corresponding correlation function by

$$\langle \sigma_i \sigma_j \rangle := \sum_{\sigma \in \{-1,1\}^V} \mathbf{P}(\sigma) \sigma_{b_i} \sigma_{b_j}.$$
 (4.1.3)

Clearly, we have $\langle \sigma_i \sigma_j \rangle = \langle \sigma_j \sigma_i \rangle$, and if i = j then the correlation function $\langle \sigma_i \sigma_i \rangle$ is equal to 1. We denote by $\operatorname{Mat}_n^{\operatorname{sym}}(\mathbb{R}, 1) \subset \operatorname{Mat}_n(\mathbb{R})$ the space of all $n \times n$ symmetric real matrices

with ones on the diagonal. Thus we obtain a matrix $M = M(G, J) := (m_{i,j}) \in \operatorname{Mat}_n^{\operatorname{sym}}(\mathbb{R}, 1)$ given by $m_{i,j} := \langle \sigma_i \sigma_j \rangle$. Let us denote

 $\mathcal{X}_n := \{ M(G,J) \mid (G,J) \text{ is a planar Ising network with } n \text{ boundary vertices} \}.$

Denote by $\overline{\mathcal{X}}_n$ the closure of \mathcal{X}_n in the space $\operatorname{Mat}_n(\mathbb{R})$ of $n \times n$ real matrices. (In other words, $\overline{\mathcal{X}}_n$ can be defined as the space of all boundary correlation matrices M(G, J) where J is allowed to take values in $[0, \infty]$, or equivalently where G is obtained from a planar graph embedded in a disk by contracting some edges that may connect boundary vertices, as we discuss in Section 4.5.) We will see later (Proposition 4.5.4) that $\overline{\mathcal{X}}_n$ admits a natural stratification into cells indexed by *matchings on* [2n], that is, by perfect matchings of the complete graph K_{2n} (also called *medial pairings*). For n = 3, all matchings on [2n] are shown in Figure 4-7.

4.1.2 The orthogonal Grassmannian

Recall the definition of $\operatorname{Gr}_{\geq 0}(k, n)$ from Section 2.2.

Definition 4.1.1. The orthogonal Grassmannian $OG(n, 2n) \subset Gr(n, 2n)$ is defined by

$$OG(n,2n) := \left\{ X \in Gr(n,2n) \mid \Delta_I(X) = \Delta_{[2n] \setminus I}(X) \text{ for all } I \in \binom{[2n]}{n} \right\}.$$

Its totally nonnegative part $OG_{\geq 0}(n, 2n) \subset Gr_{\geq 0}(n, 2n)$ is the intersection

$$OG_{\geq 0}(n,2n) := OG(n,2n) \cap Gr_{\geq 0}(n,2n).$$

The space $OG_{\geq 0}(n, 2n)$ has been first considered in [HW13] in the context of the scattering amplitudes of ABJM theory. Postnikov defined a stratification of $Gr_{\geq 0}(k, n)$ into *positroid cells*, which induces a stratification of $OG_{\geq 0}(n, 2n)$. As it was observed in [HW13, HWX14], the strata of $OG_{\geq 0}(n, 2n)$ are also naturally labeled by matchings on [2n]. We prove this in Section 4.4.

$$M = \begin{pmatrix} 1 & m_{12} & m_{13} & m_{14} \\ m_{12} & 1 & m_{23} & m_{24} \\ m_{13} & m_{23} & 1 & m_{34} \\ m_{14} & m_{24} & m_{34} & 1 \end{pmatrix} \quad \mapsto \quad \widetilde{M} = \begin{pmatrix} 1 & 1 & m_{12} & -m_{12} & -m_{13} & m_{13} & m_{14} & -m_{14} \\ -m_{12} & m_{12} & 1 & 1 & m_{23} & -m_{23} & -m_{24} & m_{24} \\ m_{13} & -m_{13} & -m_{23} & m_{23} & 1 & 1 & m_{34} & -m_{34} \\ -m_{14} & m_{14} & m_{24} & -m_{24} & -m_{34} & m_{34} & 1 & 1 \end{pmatrix}$$

Figure 4-1: An example of the map $M \mapsto \widetilde{M}$ for n = 4.

4.1.3 An embedding

Given a matrix $M = (m_{i,j}) \in \operatorname{Mat}_{n}^{\operatorname{sym}}(\mathbb{R}, 1)$, one can construct an element $\phi(M) \in \operatorname{OG}(n, 2n)$ using the following rules. We will describe an $n \times 2n$ matrix $\widetilde{M} = (\widetilde{m}_{i,j})$, so that for all $i, j \in [n]$, each of $\widetilde{m}_{i,2j-1}$ and $\widetilde{m}_{i,2j}$ is equal to either $m_{i,j}$ or $-m_{i,j}$, as in Figure 4-1. Explicitly, for i = j we put $\widetilde{m}_{i,2i-1} = \widetilde{m}_{i,2i} = m_{i,i} = 1$, and for $i \neq j$ we set

$$\widetilde{m}_{i,2j-1} = -\widetilde{m}_{i,2j} = (-1)^{i+j+\mathbb{1}(i$$

where $\mathbb{1}(i < j)$ denotes 1 if i < j and 0 otherwise.

Remark 4.1.2. For each $i \in [n]$, the sum of columns 2i - 1 and 2i of \widetilde{M} is equal to $2e_i$, where e_i is the *i*-th standard basis vector in \mathbb{R}^n . Thus the matrix \widetilde{M} has full rank, and we denote by $\phi(M) \in \operatorname{Gr}(n, 2n)$ its row span.

One can check that in fact $\phi(M)$ belongs to OG(n, 2n), see Corollary 4.4.6. We have thus constructed a map $\phi : \operatorname{Mat}_{n}^{\operatorname{sym}}(\mathbb{R}, 1) \to OG(n, 2n)$. Since boundary correlation matrices of planar Ising networks belong to the space $\operatorname{Mat}_{n}^{\operatorname{sym}}(\mathbb{R}, 1)$, ϕ restricts to a map $\phi : \overline{\mathcal{X}}_{n} \to OG(n, 2n)$. We are ready to state our main result.

Theorem 4.1.3. The restriction $\phi : \overline{\mathcal{X}}_n \to OG(n, 2n)$ is a stratification-preserving homeomorphism between $\overline{\mathcal{X}}_n$ and $OG_{\geq 0}(n, 2n)$. Moreover, both spaces are homeomorphic to an $\binom{n}{2}$ -dimensional closed ball.

We prove the second part of Theorem 4.1.3 in Section 4.6, where we also deduce its first part from Theorems 4.4.17 and 4.5.5.

Remark 4.1.4. The second sentence of Theorem 4.1.3 is an application of the machinery developed in Chapter 2. The fact that the image $\phi(\overline{\mathcal{X}}_n)$ is a subset of $\operatorname{Gr}_{\geq 0}(n, 2n)$ can be deduced in a straightforward fashion from the work of Lis [Lis17], see Section 4.5.2.

Example 4.1.5. We illustrate Theorem 4.1.3 in the case n = 2. Let $\sigma_{12} := \langle \sigma_1 \sigma_2 \rangle$, then the boundary correlation matrix M has the form

$$M = \begin{pmatrix} 1 & \sigma_{12} \\ \sigma_{12} & 1 \end{pmatrix}.$$
 (4.1.5)

By definition, $\sigma_{12} \leq 1$, and we also have $\sigma_{12} \geq 0$ by one of the *Griffiths' inequalities* [Gri67, Theorem 1]. In fact, if *G* has a single edge connecting the vertices b_1 and b_2 then it is easy to check that depending on $J_{\{b_1,b_2\}}$, σ_{12} can be any number strictly between 0 and 1. If we remove the edge $\{b_1, b_2\}$ from *G*, we get $\sigma_{12} = 0$. Thus \mathcal{X}_n consists of all matrices *M* of the form (4.1.5) for $0 \leq \sigma_{12} < 1$. If we contract the edge $\{b_1, b_2\}$, we get $\sigma_{12} = 1$. The resulting graph will no longer be embedded in a disk, because the boundary vertices b_1 and b_2 will get identified. This is an example of a *generalized planar Ising network* that we introduce in Section 4.5. We see that the closure $\overline{\mathcal{X}}_n$ of \mathcal{X}_n consists of all matrices *M* of the form (4.1.5) for $0 \leq \sigma_{12} \leq 1$, and is stratified into three cells $\{\sigma_{12} = 0\}$, $\{0 < \sigma_{12} < 1\}$, and $\{\sigma_{12} = 1\}$. These three cells correspond to three possible matchings on $\{1, 2, 3, 4\}$, namely, $\{\{1, 2\}, \{3, 4\}\}, \{\{1, 3\}, \{2, 4\}\}, \text{ and }\{\{1, 4\}, \{2, 3\}\}$, respectively.

We have

$$\widetilde{M} = \begin{pmatrix} 1 & 1 & \sigma_{12} & -\sigma_{12} \\ -\sigma_{12} & \sigma_{12} & 1 & 1 \end{pmatrix},$$
(4.1.6)

and $\phi(M) \in \operatorname{Gr}(n, 2n)$ is the row span of \widetilde{M} . The maximal minors of \widetilde{M} are

$$\Delta_{12}(\widetilde{M}) = \Delta_{34}(\widetilde{M}) = 2\sigma_{12}, \quad \Delta_{14}(\widetilde{M}) = \Delta_{23}(\widetilde{M}) = 1 - \sigma_{12}^2,$$
$$\Delta_{13}(\widetilde{M}) = \Delta_{24}(\widetilde{M}) = 1 + \sigma_{12}^2.$$

It follows that $\phi(M)$ belongs to OG(n, 2n) for all $\sigma_{12} \in \mathbb{R}$, and moreover, we get $\phi(M) \in OG_{\geq 0}(n, 2n)$ precisely when $0 \leq \sigma_{12} \leq 1$. Note that \widetilde{M} is a matrix but $\phi(M)$ is an element of the Grassmannian, and thus the Plucker coordinates of $\phi(M)$ are only defined up to

rescaling. Nevertheless, we can recover σ_{12} from these minors as follows:

$$\sigma_{12} = \frac{\Delta_{12}(\phi(M))}{\Delta_{13}(\phi(M)) + \Delta_{14}(\phi(M))}.$$
(4.1.7)

Thus we see that for n = 2, the map ϕ is indeed a homeomorphism, and both spaces $\overline{\mathcal{X}}_n$ and $OG_{\geq 0}(n, 2n)$ are homeomorphic to [0, 1], which is an $\binom{n}{2} = 1$ -dimensional closed ball.

4.2 Consequences of the main construction

In this section, we give further results on the relationship between the Ising model and the orthogonal Grassmannian.

4.2.1 Reconstructing correlations from minors

Our first goal is, given an element $X \in \text{OG}_{\geq 0}(n, 2n)$, to find explicitly a matrix $M = (m_{i,j}) \in \text{Mat}_n^{\text{sym}}(\mathbb{R}, 1)$ such that X is the row span of \widetilde{M} . For the case n = 2, this was done in (4.1.7). In order to deal with the general case, we give the following important definition.

Definition 4.2.1. Given a subset $S \subset [n]$, we denote by $\mathcal{E}_n(S) \subset {\binom{[2n]}{n}}$ the collection of *n*-element subsets I of [2n] such that for each $i \in [n]$, the intersection $I \cap \{2i - 1, 2i\}$ has even size if and only if $i \in S$.

The following result, proved in Section 4.5, is a simple consequence of Remark 4.1.2.

Lemma 4.2.2. Let $M = (m_{i,j}) \in \operatorname{Mat}_n^{\operatorname{sym}}(\mathbb{R}, 1)$ be a matrix. Then for each $i, j \in [n]$, we have

$$m_{i,j} = \frac{\sum_{I \in \mathcal{E}_n(\{i,j\})} \Delta_I(\phi(M))}{\sum_{I \in \mathcal{E}_n(\emptyset)} \Delta_I(\phi(M))} = 2^{-n} \sum_{I \in \mathcal{E}_n(\{i,j\})} \Delta_I(\widetilde{M}).$$
(4.2.1)

We stress again that unlike \widetilde{M} , the maximal minors of $\phi(M)$ are defined up to a common scalar, so it only makes sense to talk about their ratios. However, for the specific matrix \widetilde{M} , we have

$$\sum_{I \in \mathcal{E}_n(\emptyset)} \Delta_I(\widetilde{M}) = 2^n, \tag{4.2.2}$$

by the multilinearity of the determinant, see Remark 4.1.2. Thus (4.2.2) explains why the two expressions for $m_{i,j}$ given in (4.2.1) are actually equal.

For example, for n = 2, (4.2.2) becomes

$$\Delta_{13}(\widetilde{M}) + \Delta_{14}(\widetilde{M}) + \Delta_{23}(\widetilde{M}) + \Delta_{24}(\widetilde{M}) = 4$$

and for i = 1 and j = 2, Lemma 4.2.2 gives another expression for σ_{12} :

$$\sigma_{12} = \frac{\Delta_{12}(\phi(M)) + \Delta_{34}(\phi(M))}{\Delta_{13}(\phi(M)) + \Delta_{14}(\phi(M)) + \Delta_{23}(\phi(M)) + \Delta_{24}(\phi(M))},$$

which is easily seen to be equivalent to (4.1.7).

4.2.2 Cyclic symmetry and Kramers–Wannier's duality

A nice application of Theorem 4.1.3 is a cyclic symmetry of the space $\overline{\mathcal{X}}_n$, which comes from the cyclic symmetry of $OG_{\geq 0}(n, 2n)$. It turns out that the cyclic shift operation on $OG_{\geq 0}(n, 2n)$ corresponds to a generalization of the Kramers-Wannier duality [KW41] that switches between the *high and low temperature expansions* for the Ising model.

Let $k \leq N$, and consider a linear operator $S : \mathbb{R}^N \to \mathbb{R}^N$ mapping a row vector $v = (v_1, \ldots, v_N) \in \mathbb{R}^N$ to $v \cdot S = (v_2, v_3, \ldots, v_N, (-1)^{k-1}v_1)$, see Section 2.2.1. As a matrix, S is given by $S_{i+1,i} = 1$ for $i \in [N-1]$, and $S_{1,N} = (-1)^{k-1}$. A simple observation is that multiplying a $k \times N$ matrix A with nonnegative maximal minors by S on the right yields another $k \times N$ matrix with nonnegative maximal minors. Since multiplication on the right commutes with the left $\operatorname{GL}_k(\mathbb{R})$ -action, we get a cyclic shift operator on $\operatorname{Gr}_{\geq 0}(k, N)$ mapping $X \in \operatorname{Gr}_{\geq 0}(k, N)$ to $X \cdot S \in \operatorname{Gr}_{\geq 0}(k, N)$. It is clear from the definitions that for $\operatorname{Gr}_{\geq 0}(n, 2n)$, this action restricts to a cyclic shift action on $\operatorname{OG}_{\geq 0}(n, 2n)$. For example, if $X \in \operatorname{OG}_{\geq 0}(2, 4)$ is the row span of the matrix \widetilde{M} given in (4.1.6) then $X \cdot S$ is represented by

$$\widetilde{M} \cdot S = \begin{pmatrix} 1 & \sigma_{12} & -\sigma_{12} & -1 \\ \sigma_{12} & 1 & 1 & \sigma_{12} \end{pmatrix}.$$
(4.2.3)

One can check that the row span $X \cdot S$ of this matrix again belongs to $OG_{\geq 0}(2,4)$. By

Theorem 4.1.3, there must exist a matrix $M' = \begin{pmatrix} 1 & \sigma'_{12} \\ \sigma'_{12} & 1 \end{pmatrix}$ such that $\phi(M') = X \cdot S$ in $OG_{\geq 0}(2,4)$ (i.e., such that $\widetilde{M'}$ is obtained from the matrix in (4.2.3) by row operations). The value of σ'_{12} can be found from the minors of $X \cdot S$ using (4.1.7):

$$\sigma_{12}' = \frac{\Delta_{12}(X \cdot S)}{\Delta_{13}(X \cdot S) + \Delta_{14}(X \cdot S)} = \frac{1 - \sigma_{12}^2}{1 + \sigma_{12}^2 + 2\sigma_{12}} = \frac{1 - \sigma_{12}}{1 + \sigma_{12}}$$

Thus the cyclic shift operation on $OG_{\geq 0}(n, 2n)$ yields an automorphism of $\overline{\mathcal{X}}_n$ which has order 2n for n > 2 and order n for n = 1, 2. For n = 2, it sends σ_{12} to $\frac{1-\sigma_{12}}{1+\sigma_{12}}$.

Let us now formulate a generalization of the duality of [KW41].

Definition 4.2.3. Let N = (G, J) be a connected¹ planar Ising network. The *dual planar* Ising network $N^* := (G^*, J^*)$ is defined as follows. The graph $G^* = (V^*, E^*)$ is the planar dual graph of G, with boundary vertices b_1^*, \ldots, b_n^* placed counterclockwise on the boundary of the disk so that b_i^* is between b_i and b_{i+1} . For $e \in E$, we denote by e^* the edge of G^* that crosses $e \in E$, and thus we have $E^* = \{e^* \mid e \in E\}$. The edge parameters $J_{e^*}^* \in \mathbb{R}_{>0}$ are defined uniquely by the condition that

$$\sinh(2J_{e^*}^*) = \frac{1}{\sinh(2J_e)}$$
(4.2.4)

for all $e \in E$.

For example, if G is the graph in Figure 4-4 (left) then its dual G^* is shown in Figure 4-4 (middle). Note that we have $\sinh(2t) = 1/\sinh(2t)$ if and only if $t = \frac{1}{2}\log(\sqrt{2}+1)$ is the *critical temperature* of the Ising model. We also remark that applying the duality twice yields the same planar Ising network except that its boundary vertex labels are cyclically shifted: $(b_i^*)^* = b_{i+1}$. We prove the following result in Section 4.6.

Theorem 4.2.4. Let N = (G, J) be a connected planar Ising network with dual planar Ising network $N^* := (G^*, J^*)$. Then the correlation matrices M := M(G, J) and $M^* := M(G^*, J^*)$

¹This definition can be easily extended to all (not necessarily connected) generalized planar Ising networks.

are related by the cyclic shift on $OG_{\geq 0}(n, 2n)$:

$$\phi(M) \cdot S = \phi(M^*).$$

According to Theorem 1.1.1, the space $\operatorname{Gr}_{\geq 0}(k, N)$ is homeomorphic to a closed ball. The main ingredient of the proof of this result is the cyclic symmetry of $\operatorname{Gr}_{\geq 0}(k, N)$. In Section 4.6, we use the above cyclic symmetry of $\operatorname{OG}_{\geq 0}(n, 2n)$ in a similar way to show that it is a closed ball, which by the first part of Theorem 4.1.3 implies that the space $\overline{\mathcal{X}}_n$ of boundary correlation matrices is homeomorphic to a closed ball as well.

Recall from Section 2.2.1 that we have a unique cyclically symmetric point $X_0 \in \text{Gr}_{\geq 0}(n, 2n)$.

It will follow from our proof of Theorem 4.1.3 that this point X_0 actually belongs to $OG_{\geq 0}(n, 2n)$, and thus corresponds to some special planar Ising network with n boundary vertices. For instance, for n = 2 this is the Ising network N with one edge e such that $J_e = \frac{1}{2} \log(\sqrt{2} + 1)$. This planar Ising network is *self-dual*, i.e., satisfies $N = N^*$. However, it is easy to see that for n = 3 there are no self-dual planar Ising networks. Nevertheless, as our next result shows, for each n, there exists a (usually not unique) planar Ising network N = (G, J) with n boundary vertices such that the boundary correlation matrices of N and N^* coincide: $M(G, J) = M(G^*, J^*)$.

Proposition 4.2.5. For each $n \ge 1$, there exists a unique boundary correlation matrix $M_0 \in \overline{\mathcal{X}}$ of some planar Ising network such that the element $\phi(M_0) \in \mathrm{OG}_{\ge 0}(n, 2n)$ is cyclically symmetric, i.e., satisfies $\phi(M_0) \cdot S = \phi(M_0)$. For any planar Ising network N = (G, J) satisfying M(G, J) = M, we have $M(G, J) = M(G^*, J^*)$.

See Section 4.6 for the proof.

Remark 4.2.6. Consider a planar Ising network N = (G, J) such that G is the intersection of the square lattice of small side length δ with a disk, and let $J_e = \frac{1}{2} \log(\sqrt{2} + 1)$ be critical for all $e \in E$. The dual network $N^* = (G^*, J^*)$ is "very close" to N in the sense that it is obtained by shifting N by $(\delta/2, \delta/2)$ and making some adjustments near the boundary of the disk. Thus one could argue that the boundary correlations of N are "very close" to being cyclically symmetric, in which case we can find them explicitly from (4.2.2) and (2.2.1). It



Figure 4-2: Transforming a graph G (left) with one edge e connecting two boundary vertices b_1 and b_2 into a bipartite graph G^{\Box} (middle) with eight edges. Four of those edges are incident to the boundary and have weight 1, and the rest have weights s_e , c_e , s_e , c_e , as shown in the figure. The corresponding medial graph G^{\times} from Section 4.2.5 is shown on the right.

seems plausible to us that this approach may be applied to studying the universality of the scaling limit as $\delta \to 0$. The notion of being "very close" is asymptotic and thus is left beyond the scope of this paper.

4.2.3 Reduction to the dimer model

Lemma 4.2.2 shows that each two-point correlation function is a ratio of two sums of minors of an element of $OG_{\geq 0}(n, 2n)$. In the next section, we apply a well known result that each minor of an element of $Gr_{\geq 0}(k, N)$ is equal to a weighted sum of matchings in a certain planar bipartite graph.

Suppose that we are given a planar Ising network N = (G, J). We introduce two functions $s, c: E \to (0, 1)$ satisfying $s_e^2 + c_e^2 = 1$ for each $e \in E$, as follows. Given $e \in E$, we set

$$s_e := \operatorname{sech}(2J_e) = \frac{2}{\exp(2J_e) + \exp(-2J_e)}; \quad c_e := \tanh(2J_e) = \frac{\exp(2J_e) - \exp(-2J_e)}{\exp(2J_e) + \exp(-2J_e)}.$$
(4.2.5)

Next, we transform G into a weighted planar bipartite graph (a *plabic graph* in the sense of [Pos07]) G^{\Box} embedded in a disk, as in Figures 4-2 and 4-4: we replace each edge $e \in E$ of G by a bipartite square as in Figure 4-2 (middle), and connect two such squares if the corresponding edges of G share both a vertex and a face of G. Additionally, we connect each of the 2n boundary vertices of G^{\Box} , which we label d_1, \ldots, d_{2n} in counterclockwise order, to a unique vertex of G^{\Box} in the interior of the disk in an obvious way (as in Figure 4-4). Thus d_i is a white (resp., black) vertex if i is odd (resp., even). See Definitions 4.4.12 and 4.5.2 for a precise description of the rules for constructing G^{\Box} from G. We call G^{\Box} the plabic graph associated with G.

Let us now describe the *boundary measurement map* of [Pos07, Tal08], as explained in [Lam16].

Definition 4.2.7. An almost perfect matching of G^{\Box} is a collection \mathcal{A} of edges of G^{\Box} such that every vertex of G^{\Box} is incident to at most one edge in \mathcal{A} , and every non-boundary vertex of G^{\Box} is incident to exactly one edge in \mathcal{A} . The boundary of \mathcal{A} is a subset $\partial(\mathcal{A}) \subset [2n]$ which consists of all odd indices i such that d_i is not incident to an edge of \mathcal{A} together with all even indices i such that d_i is incident to an edge of \mathcal{A} . We define the weight wt(\mathcal{A}) of \mathcal{A} to be the product of weights of all edges in \mathcal{A} .

It is not hard to see that $\partial(\mathcal{A})$ has size *n* for any almost perfect matching \mathcal{A} of G^{\Box} . We are prepared to give a formula for the boundary correlation functions which is very similar to Kenyon and Wilson's *grove measurement* formula [KW11]. See Section 4.6 for the proof.

Theorem 4.2.8. Let N = (G, J) be a planar Ising network and M = M(G, J) be its boundary correlation matrix. Consider the element $\phi(M) \in OG_{\geq 0}(n, 2n)$, and let G^{\Box} be the weighted planar bipartite graph described above. Then up to a common rescaling, for every $I \in {\binom{[2n]}{n}}$ we have

$$\Delta_I(\phi(M)) = \sum_{\mathcal{A}:\partial(\mathcal{A})=I} \operatorname{wt}(\mathcal{A}), \qquad (4.2.6)$$

where the sum is over almost perfect matchings \mathcal{A} of G^{\Box} with boundary I.

For example, consider the graph G^{\Box} in Figure 4-2 (middle). There is a single almost perfect matching of G^{\Box} with boundary $\{1, 2\}$, shown in Figure 4-3 (left). Similarly, there are two almost perfect matchings of G^{\Box} with boundary $\{1, 3\}$ and one almost perfect matching with boundary $\{1, 4\}$, also shown in Figure 4-3. Therefore by Theorem 4.2.8 we get

$$\Delta_{12}(\phi(M)) = c_e, \quad \Delta_{13}(\phi(M)) = c_e^2 + s_e^2 = 1, \quad \Delta_{14}(\phi(M)) = s_e,$$



Figure 4-3: Some almost perfect matchings of G^{\Box} , together with their boundaries and weights.

up to a common rescaling. By (4.1.7), we should have

$$\sigma_{12} = \frac{\Delta_{12}(\phi(M))}{\Delta_{13}(\phi(M)) + \Delta_{14}(\phi(M))} = \frac{c_e}{1 + s_e},\tag{4.2.7}$$

where $s_e = \operatorname{sech}(2J_e)$ and $c_e = \tanh(2J_e)$ are expressed in terms of J_e as in (4.2.5). Simplifying the expressions, we get

$$\sigma_{12} = \frac{\exp(J_e) - \exp(-J_e)}{\exp(J_e) + \exp(-J_e)}.$$
(4.2.8)

On the other hand, by the definition of the Ising model, the partition function is equal to $Z = 2(\exp(J_e) + \exp(-J_e))$ and thus the correlation $\langle \sigma_1 \sigma_2 \rangle$ is

$$\langle \sigma_1 \sigma_2 \rangle = \frac{2}{Z} \left(\exp(J_e) - \exp(-J_e) \right) = \frac{\exp(J_e) - \exp(-J_e)}{\exp(J_e) + \exp(-J_e)},$$

in agreement with Theorem 4.2.8.

Lemma 4.2.2 and Theorem 4.2.8 together give a new simple way to express boundary correlations in terms of almost perfect matchings which we summarize in the following corollary.

Corollary 4.2.9. Let N = (G, J) be a planar Ising network. Then for all $i, j \in [n]$, the corresponding boundary correlation function is given by

$$\langle \sigma_i \sigma_j \rangle = \frac{\sum_{\mathcal{A}:\partial(\mathcal{A})\in\mathcal{E}_n(\{i,j\})} \operatorname{wt}(\mathcal{A})}{\sum_{\mathcal{A}:\partial(\mathcal{A})\in\mathcal{E}_n(\emptyset)} \operatorname{wt}(\mathcal{A})},$$
(4.2.9)



Figure 4-4: A planar graph G embedded in a disk (left), its dual G^* (middle), and the corresponding plabic graph G^{\Box} (right).

where the sums in the numerator and the denominator are over almost perfect matchings \mathcal{A} of G^{\Box} .

Remark 4.2.10. Corollary 4.2.9 is related to *Dubédat's bosonization identity* [Dub11]. Let us introduce a planar bipartite graph \widehat{G}^{\Box} which is obtained from G^{\Box} by simply adding an extra edge connecting d_{2i-1} to d_{2i} for all $i \in [n]$. Then there is an elegant formula (see Proposition 4.5.8) expressing the *squared* boundary correlation $\langle \sigma_i \sigma_j \rangle^2$ as a ratio of sums of perfect matchings in \widehat{G}^{\Box} . We explain how to relate this formula to (4.2.9) in Section 4.5.1. We thank Marcin Lis for bringing the paper [Dub11] to our attention.

4.2.4 Generalized Griffiths' inequalities

As we have already noted, the fact that we have $J_e > 0$ for all edges e implies that all two-point correlation functions $\langle \sigma_u \sigma_v \rangle$ are nonnegative [Gri67]. Equation (4.2.1) shows that $\langle \sigma_u \sigma_v \rangle$ is a positive linear combination of the minors of \widetilde{M} , and thus its nonnegativity follows from Theorem 4.1.3. More generally, for every subset $A \subset [n]$, define

$$\langle \sigma_A \rangle := \langle \prod_{i \in A} \sigma_{b_i} \rangle = \sum_{\sigma \in \{-1,1\}^V} \mathbf{P}(\sigma) \prod_{i \in A} \sigma_{b_i}$$

to be the expectation of the product of the spins in A. The following generalized Griffiths' inequalities were proved in [KS68].

Proposition 4.2.11 ([KS68]). For every $A \subset [n]$, we have

 $\langle \sigma_A \rangle \ge 0.$

For every $A, B \subset [n]$, we have

$$\langle \sigma_A \sigma_B \rangle - \langle \sigma_A \rangle \langle \sigma_B \rangle \ge 0.$$

Here $\langle \sigma_A \sigma_B \rangle = \langle \prod_{i \in A \oplus B} \sigma_{b_i} \rangle$, where $A \oplus B = (A \setminus B) \cup (B \setminus A)$ denotes the symmetric difference of A and B.

The goal of our next result is to explain how both inequalities in Proposition 4.2.11 also arise as positive linear combinations of the minors of the matrix \widetilde{M} , where M = M(G, J) is the boundary correlation matrix.

Definition 4.2.12. For $A \subset [n]$, we define $\widetilde{A} := \{2i - 1 \mid i \in A\} \cup \{2i \mid i \in A\}$, and for $\epsilon \in \{0, 1\}$, we let $\mathcal{D}^{\epsilon}(A) \subset {\binom{[2n]}{n}}$ be the set of all $I \in {\binom{[2n]}{n}}$ such that the sum of elements of $I \cap \widetilde{A}$ is equal to ϵ modulo 2.

Recall also the notation $\mathcal{E}_n(S)$ from Definition 4.2.1. We prove the following result in Section 4.7.

Theorem 4.2.13. For every $A \subset [n]$, we have

$$\langle \sigma_A \rangle = 2^{-n} \sum_{I \in \mathcal{E}_n(A)} \Delta_I(\widetilde{M}).$$
 (4.2.10)

For every $A, B \subset [n]$, there exists $\epsilon \in \{0, 1\}$, given explicitly in (4.7.1), such that

$$\langle \sigma_A \sigma_B \rangle - \langle \sigma_A \rangle \langle \sigma_B \rangle = 2^{-n+1} \sum_{I \in \mathcal{E}_n(A \oplus B) \cap \mathcal{D}^{\epsilon}(B)} \Delta_I(\widetilde{M}).$$
 (4.2.11)

Thus the inequalities of Proposition 4.2.11 become manifestly true when expressed in terms of minors of \widetilde{M} , which are nonnegative by Theorem 4.1.3.

For example, when n = 2 and $A = B = \{1, 2\}$, we have $\epsilon = 1$ by (4.7.1), and thus (4.2.11)



Figure 4-5: A planar graph G embedded in a disk (left), the corresponding medial graph G^{\times} (middle) and its medial strands (right).

becomes

$$1 - \sigma_{12}^2 = \frac{\Delta_{14}(\bar{M}) + \Delta_{23}(\bar{M})}{2}$$

4.2.5 Inverse problem

In this section, we concentrate on answering the following question.

Question 4.2.14. Given a planar Ising network N = (G, J), is it possible to reconstruct J from the matrix M(G, J)? In other words, is it true that the function $J : E \to \mathbb{R}_{>0}$ is uniquely determined by G and M(G, J)?

Of course, the answer to this question is negative if, for example, G has more than $\binom{n}{2}$ edges. In order to fix this, we introduce *medial graphs*. Namely, given a planar graph G embedded in a disk, the *medial graph* G^{\times} associated with G is a planar graph obtained from G as in Figure 4-2 (right) and Figure 4-5 (middle). It has 2n boundary vertices d_1, \ldots, d_{2n} , each of degree 1, and |E| interior vertices, each of degree 4. See Section 4.5 for a precise description.

Since each interior vertex of G^{\times} has degree 4, we define a *medial strand* in G^{\times} to be a path that starts at a boundary vertex d_i of G^{\times} , follows the only edge of G^{\times} incident to it, and then goes "straight" at each interior vertex of degree 4, until it reaches another boundary vertex d_j of G^{\times} . (More precisely, a medial strand is determined by the condition that whenever two of its edges share a vertex, they do not share a face.) Clearly there are *n* medial strands in G^{\times} , and each of them connects d_i to d_j for some $i, j \in [2n]$, giving rise to a matching τ_G on [2n] called the *medial pairing associated with G*:

$$\tau_G := \{\{i, j\} \subset [2n] \mid \text{a medial strand in } G^{\times} \text{ starts at } d_i \text{ and ends at } d_j \}.$$

Thus τ_G is a partition of [2n] into n sets, each of size 2. For example, for the graph G in Figure 4-5 (left), there is a medial strand that starts at vertex d_3 , then follows the midpoints of edges e_2 , e_3 , e_8 , e_7 , and then terminates at vertex d_8 . Thus the medial pairing τ_G of Gcontains a pair $\{3, 8\}$. Following the other five medial strands, we find

$$\tau_G = \{\{1, 4\}, \{2, 11\}, \{3, 8\}, \{5, 9\}, \{6, 10\}, \{7, 12\}\}, \qquad (4.2.12)$$

see Figure 4-5 (right).

Definition 4.2.15. For $i < j, i' < j' \in [2n]$, we say that pairs $\{i, j\}$ and $\{i', j'\}$ form a crossing if either i < i' < j < j' or i' < i < j' < j. For a matching τ of [2n], we let $\operatorname{xing}(\tau)$ denote the number of pairs in τ that form a crossing.

For example, if τ_G is given in (4.2.12) then $\operatorname{xing}(\tau_G) = 9$. We are now ready to state an important definition, introduced in [CIM98].

Definition 4.2.16. We say that G is *reduced* if its number |E| of edges equals $xing(\tau_G)$.

For example, the graph G in Figure 4-5 (left) is reduced since it has $|E| = 9 = \operatorname{xing}(\tau_G)$ edges.

We will see later in Proposition 4.5.4 that if we fix G and let J vary, the space of matrices M(G, J) obtained in such a way is an open ball of dimension $\operatorname{xing}(\tau_G)$. Since J varies over $\mathbb{R}^{E}_{>0}$, we see that if G is not reduced (in which case clearly $|E| > \operatorname{xing}(\tau_G)$) then the answer to Question 4.2.14 is negative. On the other hand, if G is reduced then the answer turns out to be always positive, as we show in Section 4.6.

Theorem 4.2.17. Let N = (G, J) be a planar Ising network such that G is reduced. Then the map $J \mapsto M(G, J)$ is injective, i.e., for each matrix $M \in \operatorname{Mat}_n^{\operatorname{sym}}(\mathbb{R}, 1)$, there is at most one function $J : E \to \mathbb{R}_{>0}$ such that M = M(G, J). Theorem 4.1.3 combined with our results in Section 4.4 gives a simple explicit way to test whether for a given reduced graph G and a matrix $M \in \operatorname{Mat}_n^{\operatorname{sym}}(\mathbb{R}, 1)$ there exists a function $J: E \to \mathbb{R}_{>0}$ such that M = M(G, J).

In order to describe a recursive way of reconstructing J from M(G, J), we introduce operations of *adjoining a boundary spike* and *adjoining a boundary edge* to G. An identical construction in the case of electrical networks has been considered in [CIM98].

Definition 4.2.18. Let N' = (G', J') be a planar Ising network, where G' = (V', E') has n boundary vertices. Given $k \in [n]$, we say that another planar Ising network N = (G, J) with n boundary vertices is obtained from N' by adjoining a boundary spike at k if the vertex b_k in G is incident to a single edge e and contracting this edge in G yields G'. Similarly, we say that N is obtained from N' by adjoining a boundary edge between k and k+1 if G contains an edge e connecting b_k and b_{k+1} , and removing this edge from G yields G'. In both cases, we additionally require that the restriction of $J : E \to \mathbb{R}_{>0}$ to $E' = E \setminus \{e\}$ coincides with $J' : E' \to \mathbb{R}_{>0}$.

When adjoining boundary edges, we allow for k = n, in which case we set k + 1 := 1. We denote $t := J_e$, and our first goal will be to reconstruct t from the matrix M(G, J).

Definition 4.2.19. Let $i \in [2n]$. Consider a total order \prec_i on [2n] given by

$$i \prec_i i+1 \prec_i \cdots \prec_i 2n \prec_i 1 \prec_i \cdots \prec_i i-1,$$

where the indices are taken modulo 2n. For a planar Ising network N = (G, J) and $i \in [2n]$, define subsets $I_i^{\min}(G), I_i^{\max}(G) \in {\binom{[2n]}{n}}$ whose disjoint union is [2n] as follows. For each unordered pair $\{a, b\}$ of τ_G , we may assume that $a \prec_i b$, and then we let $a \in I_i^{\min}(G)$ and $b \in I_i^{\max}(G)$. In particular, we always have $i \in I_i^{\min}(G)$ and $i - 1 \in I_i^{\max}(G)$.

For example, recall that if G is the graph from Figure 4-5 (left) then τ_G is given by (4.2.12). For i = 7 and i = 12, we have

$$I_7^{\min}(G) = \{7, 8, 9, 10, 11, 1\}, \quad I_7^{\max}(G) = \{6, 5, 4, 3, 2, 12\},$$

$$I_{12}^{\min}(G) = \{12, 1, 2, 3, 5, 6\}, \quad I_{12}^{\max}(G) = \{11, 10, 9, 8, 7, 5\}.$$

$$(4.2.13)$$

We prove our next result in Section 4.6.

Theorem 4.2.20. Let G be a reduced planar graph embedded in a disk.

Suppose that N = (G, J) is obtained from N' = (G', J') by adjoining a boundary spike e at k ∈ [n]. Let M := M(G, J), k̃ := 2k − 1, and t := J_e. Then for I := I^{min}_{k+1}(G), we have

$$s_e = \operatorname{sech}(2t) = \frac{\Delta_I(\phi(M))}{\Delta_{I \cup \{\tilde{k}\} \setminus \{\tilde{k}+1\}}(\phi(M))}$$

Suppose that N = (G, J) is obtained from N' = (G', J') by adjoining a boundary edge e between k ∈ [n] and k + 1. Let M := M(G, J), k̃ := 2k, and t := J_e. Then for I := I^{max}_{k+1}(G), we have

$$c_e = \tanh(2t) = \frac{\Delta_I(\phi(M))}{\Delta_{I \cup \{\tilde{k}+1\} \setminus \{\tilde{k}\}}(\phi(M))}$$

For example, the graph G in Figure 4-5 (left) can be obtained from another reduced graph by adjoining a boundary spike e_4 at k = 6, so we have $\tilde{k} = 11$ in the first part of Theorem 4.2.20. Since $I = I_{12}^{\min}(G) = \{1, 2, 3, 5, 6, 12\}$ by (4.2.13), we have

$$s_{e_4} = \frac{\Delta_{\{1,2,3,5,6,12\}}(\phi(M))}{\Delta_{\{1,2,3,5,6,11\}}(\phi(M))}$$

Similarly, G can be obtained from another reduced graph by adjoining a boundary edge e_9 between k = 3 and k + 1 = 4, so we have $\tilde{k} = 6$ in the second part of Theorem 4.2.20. Since $I = I_7^{\max}(G) = \{2, 3, 4, 5, 6, 12\}$ by (4.2.13), we have

$$c_{e_9} = \frac{\Delta_{\{2,3,4,5,6,12\}}(\phi(M))}{\Delta_{\{2,3,4,5,7,12\}}(\phi(M))}.$$

Since both functions sech, $\tanh : (0, \infty) \to (0, 1)$ are strictly monotone, it follows from Theorem 4.2.20 that we can reconstruct $t = J_e$ uniquely from M(G, J) whenever e is either a boundary spike or a boundary edge of G. This constitutes the first step of our reconstruction algorithm, in view of the following result.
Proposition 4.2.21. Suppose that G is a connected reduced planar graph embedded in a disk, having at least one edge. Then G is obtained from another reduced graph G' by adjoining either a boundary spike or a boundary edge.

We note that the graph G' above need not be connected. Also, if G itself is not connected then it is clearly enough to solve the inverse problem for each connected component of G separately, and thus we may assume that G is connected. See Lemma 4.4.18 for a generalization of Proposition 4.2.21.

Proposition 4.2.21 says that given a reduced graph G and a matrix M(G, J), we can reconstruct J_e for at least one edge e of G. A natural thing to do now would be to contract eif it is a boundary spike and remove e if it is a boundary edge, obtaining the reduced graph G'. Our next goal is to explain the relationship between the matrices M(G, J) and M(G', J')in the case when N = (G, J) is obtained from N' = (G', J') by adjoining either a boundary spike or a boundary edge.

We note that these two operations look like they have a very different effect on the boundary correlation matrix. For example, adjoining a boundary spike at k only changes the correlation $\langle \sigma_i \sigma_j \rangle$ when either i or j is equal to k, but adjoining a boundary edge between k and k + 1 in general changes all entries of the boundary correlation matrix. Surprisingly, these two operations have exactly the same form when written in terms of the matrix \widetilde{M} , as we now explain. (In fact, it is clear that applying the duality from Section 4.2.2 switches the roles of boundary spikes and boundary edges.)

Suppose that N = (G, J) is obtained from N' = (G', J') by adjoining a boundary spike e at $k \in [n]$ (resp., a boundary edge e between k and k+1). Define $\tilde{k} := 2k-1$ (resp., $\tilde{k} := 2k$), $t := J_e$, $s_e := \operatorname{sech}(2t)$, $c_e := \tanh(2t)$, as in Theorem 4.2.20, and let $g = g_{\tilde{k}}(t)$ be a $2n \times 2n$ matrix which coincides with the identity matrix except that it contains a 2×2 block $R_{\tilde{k}}$ in rows and columns indexed by \tilde{k} and $\tilde{k} + 1$. When \tilde{k} is odd, we set $R_{\tilde{k}} := \begin{pmatrix} 1/c_e & s_e/c_e \\ s_e/c_e & 1/c_e \end{pmatrix}$

and when \tilde{k} is even, we set $R_{\tilde{k}} := \begin{pmatrix} 1/s_e & c_e/s_e \\ c_e/s_e & 1/s_e \end{pmatrix}$. In the case where we have $\tilde{k} = 2n$, i.e., when we are adding a boundary edge between k = n and k + 1 = 1, the relevant entries of g are $g_{2n,2n} = g_{1,1} = 1/s_e$ and $g_{1,2n} = g_{2n,1} = (-1)^{n-1}c_e/s_e$. This sign twist is related to the

cyclic symmetry of $\operatorname{Gr}_{\geq 0}(n, 2n)$ as we explained in Section 4.2.2. For example, for n = 2 and $\tilde{k} = 1, 4$, we have

$$g_1(t) = \begin{pmatrix} 1/c_e & s_e/c_e & 0 & 0\\ s_e/c_e & 1/c_e & 0 & 0\\ 0 & 0 & 1 & 0\\ 0 & 0 & 0 & 1 \end{pmatrix}, \quad g_4(t) = \begin{pmatrix} 1/s_e & 0 & 0 & -c_e/s_e\\ 0 & 1 & 0 & 0\\ 0 & 0 & 1 & 0\\ -c_e/s_e & 0 & 0 & 1/s_e \end{pmatrix}$$

Recall that given an element $X \in Gr(n, 2n)$ and a $2n \times 2n$ invertible real matrix g, an element $X \cdot g \in Gr(n, 2n)$ is well defined as the row span of $A \cdot g$ where A is any $n \times 2n$ matrix whose row span is X.

Theorem 4.2.22. Suppose that N = (G, J) is obtained from N' = (G', J') by adjoining a boundary spike e at $k \in [n]$ (resp., a boundary edge e between k and k+1). Let M = M(G, J), M' = M(G', J'), and $g_{\tilde{k}}(t)$ be as above. Then we have

$$\phi(M) = \phi(M') \cdot g_{\tilde{k}}(t).$$

Theorems 4.2.20 and 4.2.22 give the following inductive algorithm for reconstructing the function $J: E \to \mathbb{R}_{>0}$ for a given reduced graph G = (V, E) from the matrix M = M(G, J). The problem is trivial when G has no edges. Otherwise by Proposition 4.2.21, there is either a boundary spike or a boundary edge e in G. The matrix M gives an element $\phi(M) \in OG_{\geq 0}(n, 2n)$, from which we compute either s_e or c_e using Theorem 4.2.20 and thus find $t = J_e$. After that, we contract e in G if it is a boundary spike and remove it if it is a boundary edge, and also modify the matrix M accordingly: we let

$$X' := \phi(M) \cdot (g_{\tilde{k}}(t))^{-1} \in \mathrm{OG}_{\geq 0}(n, 2n),$$

where $(g_{\tilde{k}}(t))^{-1}$ can be found using

$$\begin{pmatrix} 1/c_e & s_e/c_e \\ s_e/c_e & 1/c_e \end{pmatrix}^{-1} = \begin{pmatrix} 1/c_e & -s_e/c_e \\ -s_e/c_e & 1/c_e \end{pmatrix}, \quad \begin{pmatrix} 1/s_e & c_e/s_e \\ c_e/s_e & 1/s_e \end{pmatrix}^{-1} = \begin{pmatrix} 1/s_e & -c_e/s_e \\ -c_e/s_e & 1/s_e \end{pmatrix}.$$

By Lemma 4.2.2, we have $X' = \phi(M') \in \text{OG}_{\geq 0}(n, 2n)$ for a unique matrix $M' \in \text{Mat}_n^{\text{sym}}(\mathbb{R}, 1)$. We then express the entries of M' in terms of the Plücker coordinates of X' using (4.2.1). It follows that this $n \times n$ matrix M' is equal to M(G', J'), so we set G := G' and proceed recursively until G has no edges left, splitting G' into connected components if necessary. This finishes a constructive proof of Theorem 4.2.20. Alternatively, we deduce Theorem 4.2.20 from Theorem 4.1.3 at the end of Section 4.6.

Another question similar to Question 4.2.14 is the following.

Question 4.2.23. Given an $n \times n$ matrix $M \in \operatorname{Mat}_{n}^{\operatorname{sym}}(\mathbb{R}, 1)$, does there exist a planar Ising network N = (G, J) such that M = M(G, J)?

The answer to this question is provided by Theorem 4.1.3: the answer is "yes" if and only if all minors of the matrix \widetilde{M} are nonnegative. There are exponentially many minors to check, that is, $\binom{2n}{n}$, and in general one needs to check all of them to ensure that \widetilde{M} is totally nonnegative. However, checking whether $\phi(M) \in \text{OG}(n, 2n)$ belongs to $\text{OG}_{>0}(n, 2n) :=$ $\text{OG}(n, 2n) \cap \text{Gr}_{>0}(n, 2n)$ (defined in (4.3.1)), as opposed to $\text{OG}_{\geq 0}(n, 2n)$, can be done in polynomial time. More precisely, one needs to check only $n^2 + 1$ minors of \widetilde{M} , as it follows from the results of [Pos07]. These minors are algebraically independent as functions on Gr(n, 2n), but when restricted to OG(n, 2n), this is no longer the case. Thus if all of them are positive then it follows that $\phi(M) \in \text{OG}_{>0}(n, 2n)$, but in general one could check less minors and arrive at the same conclusion. See Section 4.8 for further discussion.

4.3 Background on the combinatorics of the totally nonnegative Grassmannian

In this section, we recall some combinatorial objects related to $\operatorname{Gr}_{\geq 0}(k, n)$ that were introduced by Postnikov [Pos07]. Most of the results in this section can be found in either [Pos07] or [Lam16].

Recall that the totally nonnegative Grassmannian $\operatorname{Gr}_{\geq 0}(k, N)$ is the subset of the real Grassmannian $\operatorname{Gr}(k, N)$ where all Plücker coordinates are nonnegative. Given a point $X \in$

 $\operatorname{Gr}_{\geq 0}(k, N)$, define the matroid $\mathcal{M}_X \subset {\binom{[N]}{k}}$ of X by

$$\mathcal{M}_X := \left\{ J \in \binom{[N]}{k} \mid \Delta_J(X) > 0 \right\}.$$

Given a collection $\mathcal{M} \subset {\binom{[N]}{k}}$, define the *positroid cell* $\Pi_{\mathcal{M}}^{>0} \subset \operatorname{Gr}_{\geq 0}(k, N)$ by

$$\Pi_{\mathcal{M}}^{>0} := \{ X \in \operatorname{Gr}_{\geq 0}(k, N) \mid \mathcal{M}_X = \mathcal{M} \}$$

For example, one can take $\mathcal{M} = {\binom{[N]}{k}}$, in which case the positroid cell $\Pi_{\mathcal{M}}^{>0}$ coincides with the totally positive Grassmannian $\operatorname{Gr}_{>0}(k, n)$:

$$\operatorname{Gr}_{>0}(k,N) := \left\{ X \in \operatorname{Gr}(k,N) \mid \Delta_I(X) > 0 \text{ for all } I \in \binom{[N]}{k} \right\}.$$
(4.3.1)

A collection $\mathcal{M} \subset {\binom{[N]}{k}}$ is called a *positroid* if $\Pi_{\mathcal{M}}^{>0}$ is nonempty. Positroids are special kinds of matroids, and have a very nice structure which we now explain.

Recall that for $i \in [N]$, the total order \prec_i on [N] is given by $i \prec_i i + 1 \prec_i \cdots \prec_i N \prec_i 1 \prec_i \cdots \prec_i i - 1$.

Definition 4.3.1. For two sets $I, J \in {\binom{[N]}{k}}$, we write $I \prec_i J$ if $I = \{i_1 \prec_i \cdots \prec_i i_k\}$, $J = \{j_1 \prec_i \cdots \prec_i j_k\}$, and $i_s \prec_i j_s$ for $1 \leq s \leq k$. It turns out that if \mathcal{M} is a positroid then for each i it has a unique \prec_i -minimal element which we denote $I_i^{\min}(\mathcal{M})$. Thus $I_i^{\min}(\mathcal{M})$ satisfies $I_i^{\min}(\mathcal{M}) \prec_i J$ for all $J \in \mathcal{M}$. Similarly, we let $I_i^{\max}(\mathcal{M})$ be the unique \prec_i -maximal element of \mathcal{M} .

Definition 4.3.2. A sequence $\mathcal{I} := (I_1, \ldots, I_N)$ of k-element subsets of [N] is called a Grassmann necklace if for each $i \in [N]$ there exists $j_i \in [N]$ such that $I_{i+1} = I_i \setminus \{i\} \cup \{j_i\}$.

Here (and everywhere in this section) the index i + 1 is taken modulo N.

There is a simple bijection between positroids and Grassmann necklaces, which sends a positroid \mathcal{M} to the sequence $\mathcal{I}(\mathcal{M}) := (I_1^{\min}(\mathcal{M}), I_2^{\min}(\mathcal{M}), \dots, I_N^{\min}(\mathcal{M}))$, which is a Grassmann necklace for each positroid \mathcal{M} . Each Grassmann necklace \mathcal{I} is encoded by an associated *decorated permutation* $\pi_{\mathcal{I}} : [N] \to [N]$ which sends $i \in [N]$ to the index j_i from Definition 4.3.2. (When *i* is a fixed point of $\pi_{\mathcal{I}}$, i.e., $\pi_{\mathcal{I}}(i) = i$, there is an extra bit of data in $\pi_{\mathcal{I}}$ recording whether $i \in I_i$ or $i \notin I_i$, but this will not be important for our exposition.) The map $\mathcal{I} \mapsto \pi_{\mathcal{I}}$ is a bijection between Grassmann necklaces and decorated permutations.

Remark 4.3.3. Under the above correspondence, a positroid \mathcal{M} gives rise to a decorated permutation $\pi_{\mathcal{M}}$ such that for $i \in [N]$, $\pi_{\mathcal{M}}(i)$ is equal to the unique element of the set $I_{i+1}^{\min}(\mathcal{M}) \setminus I_i^{\min}(\mathcal{M})$, if it is nonempty, and is equal to i otherwise. It is not hard to see that $\pi_{\mathcal{M}}^{-1}(i)$ is the unique element of the set $I_i^{\max}(\mathcal{M}) \setminus I_{i+1}^{\max}(\mathcal{M})$.

See [Pos07, Section 16] for a detailed description of these objects and bijections between them.

A plabic graph is a planar bipartite graph $G^{\Box} = (V^{\Box}, E^{\Box})$ embedded in a disk such that it has N boundary vertices d_1, \ldots, d_N , each of degree 1. (Postnikov considers more general plabic graphs where vertices of the same color are allowed to be connected by an edge, but for our purposes it is sufficient to work with bipartite graphs.) Recall that the notion of an almost perfect matching is given in Definition 4.2.7. Given an almost perfect matching \mathcal{A} of G, we define its boundary $\partial(\mathcal{A}) \subset [N]$ to be the set

 $\partial(\mathcal{A}) = \{i \in [N] \mid d_i \text{ is black and is not incident to an edge of } \mathcal{A}\} \cup \{i \in [N] \mid d_i \text{ is white and is incident to an edge of } \mathcal{A}\}.$

It turns out that for every G^{\Box} there exists an integer $0 \le k \le N$ such that every almost perfect matching \mathcal{A} of G^{\Box} satisfies $|\partial(\mathcal{A})| = k$. The number k is given explicitly in terms of the number of black and white vertices of G^{\Box} , see [Lam16, Eq. (9)].

Definition 4.3.4. Each plabic graph G^{\Box} gives rise to a decorated permutation $\pi_{G^{\Box}}$, as follows. A strand in G^{\Box} is a path that turns maximally right (resp., maximally left) at each black (resp., white) vertex. If a strand that starts at b_i ends at b_j for some $i, j \in [N]$ then we put $\pi_{G^{\Box}}(i) := j$, which defines a decorated permutation $\pi_{G^{\Box}} : [n] \to [n]$. (For each *i* such that $\pi_{G^{\Box}}(i) = i, \pi_{G^{\Box}}$ also contains the information whether *i* was black or white in G^{\Box} .)

Since decorated permutations are in bijection with Grassmann necklaces and positroids, each plabic graph G^{\Box} gives rise to a Grassmann necklace $\mathcal{I}_{G^{\Box}}$ and a positroid $\mathcal{M}_{G^{\Box}}$.

A weighted plabic graph is a pair (G^{\Box}, wt) where G^{\Box} is a plabic graph and $\mathrm{wt} : E^{\Box} \to \mathbb{R}_{>0}$ is a weight function assigning positive real numbers to the edges of G^{\Box} . For an almost perfect matching \mathcal{A} of G^{\Box} , recall that wt(\mathcal{A}) is the product of weights of edges of \mathcal{A} . We can consider a collection Meas $(G^{\Box}, \text{wt}) := (\Delta_I(G^{\Box}, \text{wt}))_{I \in \binom{[N]}{k}} \in \mathbb{RP}^{\binom{N}{k}-1}$ of polynomials given for $I \in \binom{[N]}{k}$ by

$$\Delta_I(G^{\Box}, \mathrm{wt}) := \sum_{\mathcal{A}:\partial(\mathcal{A})=I} \mathrm{wt}(\mathcal{A}), \qquad (4.3.2)$$

where the sum is over all almost perfect matchings \mathcal{A} of G^{\Box} with boundary I. It turns out that $(\Delta_I(G^{\Box}, \operatorname{wt}))_{I \in \binom{[N]}{k}}$ is the collection of Plücker coordinates of some point $X \in \operatorname{Gr}_{\geq 0}(k, N)$. The following result is implicit in [PSW09].

Theorem 4.3.5 ([Lam16, Corollary 7.14]). Given a weighted plabic graph (G^{\Box}, wt) , there exists a unique point $X \in Gr(k, N)$ such that

$$\operatorname{Meas}(G^{\Box}, \operatorname{wt}) = (\Delta_I(X))_{I \in \binom{[N]}{k}}$$

as elements of the projective space $\mathbb{RP}^{\binom{N}{k}-1}$. The point X belongs to $\operatorname{Gr}_{\geq 0}(k, N)$ and in fact to the positroid cell $\Pi^{>0}_{\mathcal{M}_{G^{\square}}}$, where $\mathcal{M}_{G^{\square}}$ is the positroid whose decorated permutation is $\pi_{G^{\square}}$. Every point $X \in \Pi^{>0}_{\mathcal{M}_{C^{\square}}}$ arises in this way from some weight function $\operatorname{wt} : E^{\square} \to \mathbb{R}_{>0}$.

The map $\operatorname{Meas}(G^{\Box}, \cdot) : \mathbb{R}_{>0}^{E^{\Box}} \to \operatorname{Gr}_{\geq 0}(k, N)$ sending wt $\mapsto X$ is not usually injective. To see this, observe that each interior vertex of G^{\Box} is incident to precisely one edge of each almost perfect matching \mathcal{A} . Thus rescaling the weights of all edges incident to a single interior vertex (i.e. applying a *gauge transformation*) does not change the value of Meas. We denote by $\mathbb{R}_{>0}^{E^{\Box}}/$ Gauge the space of gauge-equivalence classes of functions wt : $E^{\Box} \to \mathbb{R}_{>0}$, so that wt and wt' are the same in $\mathbb{R}_{>0}^{E^{\Box}}/$ Gauge if and only if wt' can be obtained from wt by a sequence of gauge transformations. It is not hard to see that $\mathbb{R}_{>0}^{E^{\Box}}/$ Gauge is homeomorphic to an open ball of dimension $F(G^{\Box}) - 1$, where $F(G^{\Box})$ denotes the number of faces of G^{\Box} .

Thus by Theorem 4.3.5, Meas gives rise to a map

$$\overline{\mathrm{Meas}}: \mathbb{R}^{E^{\square}}_{>0} / \operatorname{Gauge} \to \Pi^{>0}_{\mathcal{M}_{G^{\square}}} \subset \mathrm{Gr}_{\geq 0}(k, N)$$

which turns out to be injective for some plabic graphs G^{\Box} . More precisely, let us say that G^{\Box} is *reduced* if all of the following conditions are satisfied:

- no strand in G^{\Box} intersects itself;
- there are no closed strands in G^{\Box} ;
- no two strands in G^{\Box} have a bad double crossing.

Here two strands are said to form a *bad double crossing* if there are two vertices $u, v \in V^{\Box}$ such that both strands first pass through u and then through v. The following result can be found in [Pos07, Lam16].

Theorem 4.3.6. For each positroid \mathcal{M} , there exists a reduced plabic graph G^{\Box} such that $\mathcal{M} = \mathcal{M}_{G^{\Box}}$. Given a reduced plabic graph G^{\Box} , the map $\overline{\text{Meas}} : \mathbb{R}_{>0}^{E^{\Box}} / \text{Gauge} \to \Pi_{\mathcal{M}_{G^{\Box}}}^{>0}$ is a homeomorphism. Thus the positroid cell $\Pi_{\mathcal{M}_{G^{\Box}}}^{>0}$ is homeomorphic to $\mathbb{R}^{F(G^{\Box})-1}$. In addition, we have

$$\operatorname{Gr}_{\geq 0}(k, N) = \bigsqcup_{\mathcal{M}} \Pi_{\mathcal{M}}^{>0}, \qquad (4.3.3)$$

where the union is over all positroids $\mathcal{M} \subset {[N] \choose k}$.

The last ingredient from the theory of plabic graphs that we will need is *BCFW bridges*, introduced in [AHBC⁺16, BCFW05]. Our exposition will follow [Lam16, Section 7].

Recall that each boundary vertex of a plabic graph is incident to a unique edge.

Definition 4.3.7. Given $i \in [N]$, we say that a plabic graph G^{\Box} has a *removable bridge* between i and i+1 if there exists a path of length 3 between d_i and d_{i+1} in G^{\Box} . (In particular, these vertices have to be of different color).

Here we again allow i = N and i + 1 = 1. We refer to the middle edge of this path of length 3 as a *bridge between* i and i + 1. There are two types of bridges, since i can be incident either to a white or to a black interior vertex. It turns out that the weight of the bridge can always be recovered from the minors of the corresponding element of the Grassmannian. The following result can be found in [Lam16, Proposition 7.10] and [Lam18, Proposition 3.10], and is the main ingredient of the proof of Theorem 4.2.20.

Theorem 4.3.8. Let (G^{\Box}, wt) be a weighted reduced plabic graph, and suppose that it has a removable bridge between i and i + 1. Assume that the weights of the edges incident to d_i and d_{i+1} are both equal to 1 (which can always be achieved using gauge transformations). Let $e \in E^{\square}$ be the bridge between i and i+1, and denote $X := \overline{\text{Meas}}(G^{\square}, \text{wt}) \in \text{Gr}_{\geq 0}(k, N)$.

• If i is white then for $I := I_{i+1}^{\min}(\mathcal{M}_{G^{\square}})$, we have

$$\operatorname{wt}(e) = \frac{\Delta_I(X)}{\Delta_{I \cup \{i\} \setminus \{i+1\}}(X)}$$

• If i is black then for $I := I_{i+1}^{\max}(\mathcal{M}_{G^{\square}})$, we have

$$\operatorname{wt}(e) = \frac{\Delta_I(X)}{\Delta_{I \cup \{i+1\} \setminus \{i\}}(X)}.$$

We will also need to explain how removing a bridge changes the corresponding element of the Grassmannian. For $i \in [N-1]$ and $t \in \mathbb{R}$, define $x_i(t) \in \operatorname{Mat}_N(\mathbb{R})$ to be a $N \times N$ matrix with ones on the diagonal and a single nonzero off-diagonal entry in row i and column i + 1equal to t. We also define $x_N(t)$ to be the matrix with ones on the diagonal and the entry in row N, column 1 equal to $(-1)^{k-1}t$. We define $y_i(t)$ to be the matrix transpose of $x_i(t)$ for $i \in [N]$.

Lemma 4.3.9 ([Lam16, Lemma 7.6]). Let $(G^{\Box}, \operatorname{wt})$ be a weighted plabic graph, and suppose that it has a removable bridge between i and i + 1. Assume that the weights of the edges incident to d_i and d_{i+1} are both equal to 1. Let $e \in E^{\Box}$ be the bridge between i and i + 1 with weight $\operatorname{wt}(e) = t$, and denote $X := \overline{\operatorname{Meas}}(G^{\Box}, \operatorname{wt}) \in \operatorname{Gr}_{\geq 0}(k, N)$. Let $(G^{\Box'}, \operatorname{wt'})$ be obtained from $(G^{\Box}, \operatorname{wt})$ by removing e, and define $X' := \overline{\operatorname{Meas}}(G^{\Box'}, \operatorname{wt})$. Then for all $I \in {[N] \choose k}$ we have the following.

• If i is white then $X' = X \cdot x_i(-t)$, and

$$\Delta_I(X') = \begin{cases} \Delta_I(X) - t\Delta_{I \setminus \{i+1\} \cup \{i\}}(X), & \text{if } i+1 \in I \text{ but } i \notin I; \\ \Delta_I(X), & \text{otherwise.} \end{cases}$$

• If i is black then $X' = X \cdot y_i(-t)$, and

$$\Delta_{I}(X') = \begin{cases} \Delta_{I}(X) - t\Delta_{I \setminus \{i\} \cup \{i+1\}}(X), & \text{if } i \in I \text{ but } i+1 \notin I; \\ \Delta_{I}(X), & \text{otherwise.} \end{cases}$$

4.4 The totally nonnegative orthogonal Grassmannian

In this section, we discuss how the stratification of Gr(n, 2n) induces a stratification of the totally nonnegative orthogonal Grassmannian $OG_{\geq 0}(n, 2n)$. We remark that some of the statements below have appeared in [HW13, HWX14], but mostly without proofs.

Recall from Definition 4.1.1 that the orthogonal Grassmannian $OG(n, 2n) \subset Gr(n, 2n)$ is the set of $X \in Gr(n, 2n)$ such that $\Delta_I(X) = \Delta_{[2n]\setminus I}(X)$ for all *n*-element sets $I \subset [2n]$. In the literature, the term "orthogonal Grassmannian" usually refers to the set of subspaces where a certain non-degenerate symmetric bilinear form vanishes. Over the complex numbers, there is only one such bilinear form up to isomorphism, but over the real numbers, one needs to choose a signature. Following [HW13], define a non-degenerate symmetric bilinear form $\eta : \mathbb{R}^{2n} \times \mathbb{R}^{2n} \to \mathbb{R}$ by $\eta(u, v) := u_1v_1 - u_2v_2 + \cdots + u_{2n-1}v_{2n-1} - u_{2n}v_{2n}$. Let us also introduce another subset $OG_{-}(n, 2n) \subset Gr(n, 2n)$ consisting of all $X \in Gr(n, 2n)$ such that $\Delta_I(X) = -\Delta_{[2n]\setminus I}(X)$ for all *n*-element sets $I \subset [2n]$.² We justify our terminology as follows.

Proposition 4.4.1. For a subspace $X \in Gr(n, 2n)$, the following are equivalent:

- $X \in OG(n, 2n) \sqcup OG_{-}(n, 2n);$
- for any two vectors $u, v \in X \subset \mathbb{R}^{2n}$, we have $\eta(u, v) = 0$.

Proof. Given an $k \times N$ matrix $A = (a_{i,j})$, define another $k \times N$ matrix $\operatorname{alt}(A) := ((-1)^j a_{i,j})$. Taking row spans and setting k := n, N := 2n, we get a map alt : $\operatorname{Gr}(n, 2n) \to \operatorname{Gr}(n, 2n)$. It is a classical result (see e.g. [Hoc75, Section 7] or [Kar17, Lemma 1.11]) that for $X \in \operatorname{Gr}(n, 2n)$ and $I \in {[2n] \choose n}$, we have $\Delta_{[2n] \setminus I}(\operatorname{alt}(X^{\perp})) = c\Delta_I(X)$, where \perp denotes the orthogonal complement of $X \subset \mathbb{R}^{2n}$ with respect to the standard scalar product $\langle \cdot, \cdot \rangle$ on \mathbb{R}^{2n} and $c \in \mathbb{R}$

²We thank David Speyer for suggesting to consider both OG(n, 2n) and $OG_{-}(n, 2n)$.

is some nonzero constant. Note that $\eta(u, v) = \langle \operatorname{alt}(u), v \rangle$ for $u, v \in \mathbb{R}^{2n}$, which shows that η vanishes on X if and only if $\Delta_I(X) = c\Delta_{[2n]\setminus I}(X)$ for all $I \in \binom{[2n]}{n}$. Applying this equality twice, we get $\Delta_I(X) = c\Delta_{[2n]\setminus I}(X) = c^2\Delta_I(X)$, and thus $c = \pm 1$. We are done with the proof.

Remark 4.4.2. Lusztig [Lus94] has defined the totally nonnegative part $(G/P)_{\geq 0}$ of any partial flag variety G/P inside a split reductive algebraic group G over \mathbb{R} . Rietsch showed that the space $\operatorname{Gr}_{>0}(k,n)$ is a special case of $(G/P)_{>0}$, see [Lam16, Remark 3.8]. For a specific choice of G = O(n, n) (i.e. the split orthogonal group, which corresponds to the Dynkin diagram of type D_n) and a maximal parabolic subgroup $P \cong SL_n(\mathbb{R})$ (corresponding to the Dynkin diagram of type A_{n-1} , obtained from D_n by removing a leaf adjacent to a degree 3 vertex), G/P becomes equal to OG(n, 2n). If we had $(G/P)_{\geq 0} = OG_{\geq 0}(n, 2n)$ then the fact that $OG_{\geq 0}(n, 2n)$ is a closed ball would follow from the results of [GKL18]. However, the relationship between Lusztig's $(G/P)_{\geq 0}$ and $OG_{\geq 0}(n, 2n)$ remains unclear to us. For instance, the cell decomposition of $(G/P)_{\geq 0}$ conjectured by Lusztig and proved by Rietsch [Rie98, Rie99] appears to have a different number of cells than the cell decomposition of $OG_{\geq 0}(n, 2n)$ indexed by matchings on [2n] that we consider in this paper. David Speyer Spe18 has also informed us that the space E_n of electrical networks can be realized as a subset of the Lagrangian Grassmannian LG(n-1, 2n-2), which is also equal to G/P when G is the symplectic group $\operatorname{Sp}_{2n-2}(\mathbb{C})$ (corresponding to the Dynkin diagram of type C_{n-1}). The relationship between this subset and $(G/P)_{\geq 0}$ is again unclear in this case.

Remark 4.4.3. A different relation between the Ising model and spin representations of the orthogonal group can be found in [Kau49, SMJ78, Pal07].

Remark 4.4.4. The generators $g_i(t)$ from Section 4.2.5 belong to O(n, n), and moreover, they are hyperbolic rotation matrices, since for $t \in \mathbb{R}_{>0}$ and $c := \tanh(2t)$, $s := \operatorname{sech}(2t)$, there exists a unique $r(t) \in \mathbb{R}$ such that $\begin{pmatrix} 1/c & s/c \\ s/c & 1/c \end{pmatrix} = \begin{pmatrix} \cosh(r(t)) & \sinh(r(t)) \\ \sinh(r(t)) & \cosh(r(t)) \end{pmatrix}$. It would thus be interesting to find an analog of the theory of [LP15] for the orthogonal group rather than the symplectic group. **Remark 4.4.5.** A more standard choice of coordinates for OG(n, 2n) is to consider the set OG'(n, 2n) of all $X \in Gr(n, 2n)$ where another symmetric bilinear form, $\eta'(u, v) :=$ $u_1v_{n+1} + u_2v_{n+2} + \cdots + u_nv_{2n}$, vanishes. Consider a $2n \times 2n$ matrix J with the only nonzero entries given by $J_{2j-1,j} = J_{2j,j} = J_{2j-1,j+n} = 1/2$, $J_{2j,j+n} = -1/2$, for all $j \in [n]$. Then the map $X \mapsto X \cdot J$ gives a bijection between $OG(n, 2n) \sqcup OG_{-}(n, 2n)$ and OG'(n, 2n). Moreover, for $M \in Mat_n^{sym}(\mathbb{R}, 1)$, the matrix $M \cdot J$ has the form $[I_n|M']$ for a skew-symmetric matrix M'given by $m'_{i,j} = (-1)^{i+j+1}m_{i,j}$ for $i \neq j$ and $m'_{i,j} = 0$ for i = j. A standard way to work with OG'(n, 2n) is to consider *spinor coordinates*, which are essentially *Pfaffians* of the matrix M'above, see e.g. [HS10, Section 5]. It was shown in [GBK78] that these Pfaffians are multipoint boundary correlation functions for the Ising model, as we explain in Proposition 4.7.7. We thank David Speyer for this remark.

Proposition 4.4.1 allows one to deduce that the image of the map ϕ is contained inside the orthogonal Grassmannian.

Corollary 4.4.6. We have $\phi(\operatorname{Mat}_n^{\operatorname{sym}}(\mathbb{R},1)) \subset \operatorname{OG}(n,2n)$.

Proof. Let $M \in \operatorname{Mat}_{n}^{\operatorname{sym}}(\mathbb{R}, 1)$. It is obvious from the definition of \widetilde{M} that if $u, v \in \mathbb{R}^{2n}$ are any two rows of \widetilde{M} then we have $\eta(u, v) = 0$, and thus by Proposition 4.4.1 we get $\phi(\operatorname{Mat}_{n}^{\operatorname{sym}}(\mathbb{R}, 1)) \subset \operatorname{OG}(n, 2n) \sqcup \operatorname{OG}_{-}(n, 2n)$. But note that $\operatorname{OG}(n, 2n)$ and $\operatorname{OG}_{-}(n, 2n)$ are not connected to each other inside $\operatorname{OG}(n, 2n) \sqcup \operatorname{OG}_{-}(n, 2n)$, however, $\operatorname{Mat}_{n}^{\operatorname{sym}}(\mathbb{R}, 1) \cong \mathbb{R}^{\binom{n}{2}}$ is connected. Thus $\phi(\operatorname{Mat}_{n}^{\operatorname{sym}}(\mathbb{R}, 1))$ is connected, and clearly the image of the identity matrix $I_{n} \in \operatorname{Mat}_{n}^{\operatorname{sym}}(\mathbb{R}, 1)$ belongs to $\operatorname{OG}(n, 2n)$ and not to $\operatorname{OG}_{-}(n, 2n)$. The result follows. \Box

Proposition 4.4.7. Let $X \in OG_{\geq 0}(n, 2n)$, and let $\mathcal{M} := \mathcal{M}_X$ be the positroid of X with decorated permutation $\pi_{\mathcal{M}}$. Then $\pi_{\mathcal{M}}$ is a fixed-point free involution: if $\pi_{\mathcal{M}}(i) = j$ then $i \neq j$ and $\pi_{\mathcal{M}}(j) = i$.

Proof. It is clear from Definition 4.3.1 that we have $I_i^{\min}(\mathcal{M}) = [2n] \setminus I_i^{\max}(\mathcal{M})$, because $X \in OG(n, 2n)$. Suppose now that $\pi_{\mathcal{M}}(i) = j$ and that $i \neq j$. By Remark 4.3.3, $\pi_{\mathcal{M}}^{-1}(i)$ is the unique element of the set

$$I_i^{\max}(\mathcal{M}) \setminus I_{i+1}^{\max}(\mathcal{M}) = ([2n] \setminus I_i^{\min}(\mathcal{M})) \setminus ([2n] \setminus I_{i+1}^{\min}(\mathcal{M})) = I_{i+1}^{\min}(\mathcal{M}) \setminus I_i^{\min}(\mathcal{M}),$$

which is equal to $\{\pi_{\mathcal{M}}(i)\} = \{j\}$. Thus $\pi_{\mathcal{M}}^{-1}(i) = j$, equivalently, $\pi_{\mathcal{M}}(j) = i$, so $\pi_{\mathcal{M}}$ is an involution. It remains to show that it is *fixed-point free*, i.e., that $\pi_{\mathcal{M}}(i) \neq i$ for all $i \in [2n]$. We can have $\pi_{\mathcal{M}}(i) = i$ if either i is a *loop* or a *coloop* of \mathcal{M} , that is, if either $i \in I$ for all $I \in \mathcal{M}$ or $i \notin I$ for all $I \in \mathcal{M}$, respectively. Choose some $I \in \mathcal{M}$. Then $[2n] \setminus I$ also belongs to \mathcal{M} , which shows that i is neither a loop nor a coloop of \mathcal{M} . We are done with the proof.

Remark 4.4.8. Recall that given a matching τ on [2n], Definition 4.2.19 gives two disjoint sets $I_i^{\min}(\tau)$ and $I_i^{\max}(\tau)$ for each $i \in [2n]$. It is easy to check that if π is the fixed-point free involution corresponding to τ then $I_i^{\min}(\mathcal{M}_{\pi}) = I_i^{\min}(\tau)$ and $I_i^{\max}(\mathcal{M}_{\pi}) = I_i^{\max}(\tau)$.

In Section 4.2.5, we described how to transform a planar graph G embedded in a disk into a medial graph G^{\times} , and then how to obtain a medial pairing τ_G from G^{\times} . Not all matchings can be obtained in this way, for example, when n = 2, the matching $\{\{1, 4\}, \{2, 3\}\}$ is not a medial pairing of any graph G. It will thus be more convenient for us to work with medial graphs rather than matchings. In Section 4.5, we introduce generalized planar Ising networks which correspond to all matchings on [2n].

Definition 4.4.9. A medial graph is a planar graph $G^{\times} = (V^{\times}, E^{\times})$ embedded in a disk, such that it has 2n boundary vertices $d_1, d_2, \ldots, d_{2n} \in V^{\times}$ in counterclockwise order, each of degree 1, and such that every other vertex of G^{\times} has degree 4.

The non-boundary vertices (the ones that have degree 4) are called *interior vertices* of G^{\times} , and we let $V_{\text{int}}^{\times} := V^{\times} \setminus \{d_1, \ldots, d_{2n}\}$ denote the set of such vertices. Each medial graph G^{\times} gives rise to a *medial pairing* $\tau_{G^{\times}}$, as in Section 4.2.5. We say that a medial graph G^{\times} is *reduced* if the number of its interior vertices equals $\operatorname{xing}(\tau_{G^{\times}})$. Equivalently, G^{\times} is reduced if every edge of G^{\times} belongs to some medial strand connecting two boundary vertices, no medial strand intersects itself, and no two medial strands intersect more than once.

Lemma 4.4.10. For every matching τ on [2n], there exists a reduced medial graph G^{\times} satisfying $\tau_{G^{\times}} = \tau$.

Proof. For each pair $\{i, j\} \in \tau$, connect d_i with d_j by a straight line segment. Then perturb each line segment slightly so that every point inside the disk would belong to at most two

segments, obtaining a *pseudoline arrangement*. Let G^{\times} be obtained from this pseudoline arrangement by putting an interior vertex at each intersection point. It is clear that G^{\times} is a reduced medial graph whose medial pairing is τ . Alternatively, G^{\times} can be constructed by induction on $\operatorname{xing}(\tau)$ in an obvious way using the poset P_n from Definition 4.4.15.

Let us say that a medial network $N^{\times} = (G^{\times}, J^{\times})$ is a medial graph G^{\times} together with a function $J^{\times} : V_{int}^{\times} \to \mathbb{R}_{>0}$. Thus if N = (G, J) is a planar Ising network then the edges of Gcorrespond to the interior vertices of the corresponding medial graph G^{\times} and thus the Ising network N gives rise to a medial network $N^{\times} = (G^{\times}, J^{\times})$, as described in Sections 4.2.5 and 4.5. In the remainder of this section, we will work with medial networks rather than with planar Ising networks.

Every medial graph gives rise to a plabic graph. In order to describe this correspondence, we first introduce a canonical way to orient each medial graph, as described in [HWX14].

Proposition 4.4.11. Let G^{\times} be a medial graph. Then there exists a unique orientation of the edges of G^{\times} such that:

- 1. for $i \in [2n]$, d_i is a source if and only if i is odd;
- 2. each interior vertex $v \in V_{int}^{\times}$ of G^{\times} is incident to two incoming and two outgoing arrows so that their directions alternate around v.

Proof. If G^{\times} is connected then it is easy to see that there are just two orientations satisfying the second condition, since we can color the faces of G^{\times} in a bipartite way and then orient all black faces clockwise and all white faces counterclockwise, or vice versa. One of these two orientations will satisfy the first condition. If G^{\times} has C connected components then there are 2^{C} orientations of G satisfying the second condition, but there will still be one of them that satisfies the first condition, because the number of vertices of each connected component is even.

Definition 4.4.12. Given a medial network $N^{\times} = (G^{\times}, J^{\times})$, the associated weighted plabic graph (G^{\Box}, wt) is constructed as follows. First, orient the edges of G^{\times} as in Proposition 4.4.11, and then for each oriented edge e of G^{\times} , put a white vertex e° of G^{\Box} close to the source of e and a black vertex e° of G^{\Box} close to the target of e. If the source (resp.,



Figure 4-6: Uncrossing the pairs $\{i, j\}$ and $\{i', j'\}$.

the target) of e is a boundary vertex d_i then we set $e^{\circ} := d_i$ (resp., $e^{\bullet} := d_i$). Now, for each edge $e \in E^{\Box}$ of G^{\Box} , connect e^{\bullet} and e° by an edge of G^{\Box} , and set its weight to 1: wt($\{e^{\bullet}, e^{\circ}\}$) := 1. Additionally for every interior vertex $v \in V_{int}^{\times}$ of G^{\Box} incident to edges $e_1, e_2, e_3, e_4 \in E^{\Box}$ in counterclockwise order so that v is the target of e_1 and e_3 and the source of e_2 and e_4 , add four edges $\{e_1^{\bullet}, e_2^{\circ}\}, \{e_2^{\circ}, e_3^{\bullet}\}, \{e_3^{\bullet}, e_4^{\circ}\}, \{e_4^{\circ}, e_1^{\bullet}\}$ to G^{\Box} . The weights of these edges are given by

$$wt(\{e_1^{\bullet}, e_2^{\circ}\}) = wt(\{e_3^{\bullet}, e_4^{\circ}\}) := s_v, \quad wt(\{e_2^{\circ}, e_3^{\bullet}\}) = wt(\{e_4^{\circ}, e_1^{\bullet}\}) := c_v, \tag{4.4.1}$$

where s_v and c_v are given by (4.2.5), that is,

$$s_v := \operatorname{sech}(2J_v^{\times}) = \frac{2}{\exp(2J_v^{\times}) + \exp(-2J_v^{\times})}; \quad c_v := \tanh(2J_v^{\times}) = \frac{\exp(2J_v^{\times}) - \exp(-2J_v^{\times})}{\exp(2J_v^{\times}) + \exp(-2J_v^{\times})}.$$

This defines a weighted plabic graph (G^{\Box}, wt) associated to the medial network $N^{\times} = (G^{\times}, J^{\times})$.

Recall that for a medial graph G^{\times} , the corresponding medial pairing $\tau_{G^{\times}} = \{\{i_1, j_1\}, \ldots, \{i_n, j_n\}\}$ is a matching on [2n]. We define a permutation $\pi_{G^{\times}} : [2n] \to [2n]$ by setting $\pi_{G^{\times}}(i_k) := j_k$ and $\pi_{G^{\times}}(j_k) := i_k$ for all $k \in [2n]$. Thus $\pi_{G^{\times}}$ is a fixed-point free involution.

Lemma 4.4.13. A medial graph G^{\times} is reduced if and only if the corresponding plabic graph G^{\Box} is reduced. We have $\pi_{G^{\times}} = \pi_{G^{\Box}}$.

Proof. This is straightforward to check from the definitions, since the medial strands correspond to the strands in G^{\Box} from Definition 4.3.4.

Proposition 4.4.14. Given a medial network $N^{\times} = (G^{\times}, J^{\times})$, let (G^{\Box}, wt) be the corresponding weighted plabic graph. Then $\overline{\text{Meas}}(G^{\Box}, \text{wt})$ yields an element of $\Pi^{>0}_{\mathcal{M}} \cap \text{OG}_{\geq 0}(n, 2n)$, where $\mathcal{M} = \mathcal{M}_{\pi_{G^{\times}}}$ is the positroid corresponding to the fixed-point free involution $\pi_{G^{\times}}$.

Proof. By Lemma 4.4.13, we have $\pi_{G^{\times}} = \pi_{G^{\square}}$. Let $X \in \operatorname{Gr}_{\geq 0}(n, 2n)$ be the point given by $\overline{\operatorname{Meas}}(G^{\square}, \operatorname{wt})$. By Theorem 4.3.5, X belongs to $\Pi^{>0}_{\mathcal{M}}$, and it remains to show that X belongs to $\operatorname{OG}_{\geq 0}(n, 2n)$. Given an almost perfect matching \mathcal{A} of G^{\square} , let us define $S^{\times}(\mathcal{A}) \subset E^{\times}$ to be the set of edges e of G^{\times} such that the edge $\{e^{\bullet}, e^{\circ}\}$ of G^{\square} belongs to \mathcal{A} . We claim that for all sets $R \subset E^{\times}$ of edges of G^{\times} , we have

$$\sum_{S^{\times}(\mathcal{A})=R} \operatorname{wt}(\mathcal{A}) = \sum_{S^{\times}(\mathcal{A})=E^{\times}\setminus R} \operatorname{wt}(\mathcal{A}), \qquad (4.4.2)$$

where the sums are over almost perfect matchings of G^{\Box} . The left hand side of (4.4.2) is equal to the product over all interior vertices $v \in V_{int}^{\times}$ of G^{\times} of q(v), where q(v) is equal to either c_e , s_e , $c_e^2 + s_e^2$, 1, or 0, depending on which of the four edges of G^{\times} adjacent to v belong to R. It is clear from Figure 4-3 that replacing R with its complement does not affect this product. (The only non-trivial change is replacing $c_e^2 + s_e^2$ with 1, but recall that we have $c_e^2 + s_e^2 = 1$ by construction.) This proves (4.4.2), and clearly if two almost perfect matchings $\mathcal{A}, \mathcal{A}'$ of G^{\Box} satisfy $S^{\times}(\mathcal{A}') = E^{\times} \setminus S^{\times}(\mathcal{A})$ then they also satisfy $\partial(\mathcal{A}') = [2n] \setminus \partial(\mathcal{A})$, finishing the proof.

For a matching τ on [2n], let \mathcal{M}_{τ} be the positroid corresponding to the fixed-point free involution $\pi : [2n] \to [2n]$ associated with τ , and denote by $\Pi_{\tau}^{>0} := \Pi_{\mathcal{M}_{\tau}}^{>0}$. Following [Lam18], denote by P_n the partially ordered set (*poset*) of all matchings τ on [2n]. (It is easy to see that P_n has $\frac{(2n)!}{n!2^n}$ elements.) The covering relations of P_n are described as follows. Given a matching τ on [2n], suppose that the pairs $\{i, j\}, \{i', j'\} \in \tau$ form a crossing, as in Definition 4.2.15. Introduce two matchings

$$\tau' := \tau \setminus \{\{i, j\}, \{i', j'\}\} \cup \{\{i, j'\}, \{i', j\}\};$$

$$\tau'' := \tau \setminus \{\{i, j\}, \{i', j'\}\} \cup \{\{i, i'\}, \{j, j'\}\}.$$

$$(4.4.3)$$



Figure 4-7: The Hasse diagram of the poset P_3 .

Definition 4.4.15. We say that τ' and τ'' are obtained from τ by uncrossing the pairs $\{i, j\}$ and $\{i', j'\}$ (see Figure 4-6). In addition, if $\operatorname{xing}(\tau') + 1 = \operatorname{xing}(\tau)$ (resp., $\operatorname{xing}(\tau'') + 1 = \operatorname{xing}(\tau)$), we write $\tau' \leq \tau$ (resp., $\tau'' \leq \tau$), and let P_n be the poset whose order relation \leq is the transitive closure of \leq .

Remark 4.4.16. Equivalently, as explained in [Lam18, Section 4.5], given a medial graph G^{\times} with medial pairing τ , we have $\tau' \ll \tau$ if and only if "uncrossing" the unique vertex $v \in V_{\text{int}}^{\times}$ of G^{\times} that belongs to the intersection of medial strands connecting d_i to d_j and $d_{i'}$ to $d_{j'}$ yields a *reduced* medial graph with medial pairing τ' . Here *uncrossing* an interior vertex of a medial graph means replacing its neighborhood in one of the two ways shown in Figure 4-6.

By [Lam18, Lemma 4.13], the poset P_n is graded with grading given by $xing(\tau)$, and by [HK18, Lam15], P_n is a shellable Eulerian poset. See Figure 4-7 for the case n = 3.

We are now ready to state the main result of this section.

Theorem 4.4.17.

(i) Given a reduced medial graph G^{\times} , let G^{\Box} be the corresponding plabic graph with positroid $\mathcal{M} := \mathcal{M}_{G^{\Box}}$. Then the map $J^{\times} \mapsto \overline{\mathrm{Meas}}(G^{\Box}, \mathrm{wt})$ is a homeomorphism between $\mathbb{R}_{>0}^{E^{\times}}$ and $\Pi_{\mathcal{M}}^{>0} \cap \mathrm{OG}_{\geq 0}(n, 2n)$. (ii) The set $OG_{\geq 0}(n, 2n)$ is a disjoint union of cells

$$OG_{\geq 0}(n,2n) = \bigsqcup_{\tau \in P_n} \left(\Pi_{\tau}^{>0} \cap OG_{\geq 0}(n,2n) \right),$$
(4.4.4)

and each cell $\Pi^{>0}_{\tau} \cap \mathrm{OG}_{>0}(n, 2n)$ is homeomorphic to $\mathbb{R}^{\mathrm{xing}(\tau)}$.

(iii) For $\tau \in P_n$, the closure of the cell $\Pi_{\tau}^{>0} \cap \mathrm{OG}_{\geq 0}(n, 2n)$ in $\mathrm{Gr}(n, 2n)$ equals

$$\overline{(\Pi_{\tau}^{>0} \cap \operatorname{OG}_{\geq 0}(n,2n))} = \bigsqcup_{\sigma \in P_n: \sigma \leq \tau} \left(\Pi_{\sigma}^{>0} \cap \operatorname{OG}_{\geq 0}(n,2n) \right).$$
(4.4.5)

Proof. As we have shown in Proposition 4.4.7, for every $X \in OG_{\geq 0}(n, 2n)$, the decorated permutation $\pi = \pi_{\mathcal{M}_X}$ associated with the positroid \mathcal{M}_X of X is a fixed-point free involution, and thus (4.4.4) follows from (4.3.3). The remainder of part (ii) (that each cell is an open ball) follows from part (i), which we prove now. Thus we fix a reduced medial graph G^{\times} and the corresponding plabic graph G^{\Box} , reduced by Lemma 4.4.13. Let $\psi : \mathbb{R}_{\geq 0}^{E^{\times}} \to \Pi_{\mathcal{M}}^{>0} \cap$ $OG_{\geq 0}(n, 2n)$ be the map that sends $J^{\times} \mapsto \overline{\mathrm{Meas}}(G^{\Box}, \mathrm{wt})$. We first show that ψ is injective. By Theorem 4.3.6, it suffices to show that the map $J^{\times} : V_{\mathrm{int}}^{\times} \to \mathbb{R}_{>0}$ can be reconstructed from the corresponding weight function $\mathrm{wt} \in \mathbb{R}_{>0}^{E^{\Box}}/\mathrm{Gauge}$. Fix an interior vertex $v \in V_{\mathrm{int}}^{\times}$ of G^{\times} and consider the corresponding four vertices $v_1, v_2, v_3, v_4 \in V^{\Box}$ of G^{\Box} on the four edges of G^{\times} incident to v. Let $e_{12}, e_{23}, e_{34}, e_{14}$ be the four edges of G^{\Box} forming a square around v. We have $\mathrm{wt}(e_{12}) = \mathrm{wt}(e_{34}) = s_v$ and $\mathrm{wt}(e_{23}) = \mathrm{wt}(e_{14}) = c_v$, as in (4.4.1). Suppose that we have applied a gauge transformation to wt obtaining another weight function wt' . Thus we have rescaled all edges adjacent to the vertex v_k by some number $t_k \in \mathbb{R}_{>0}$ for $1 \leq k \leq 4$. Therefore

$$\operatorname{wt}'(e_{12}) = t_1 t_2 s_v, \quad \operatorname{wt}'(e_{34}) = t_3 t_4 s_v, \quad \operatorname{wt}'(e_{23}) = t_2 t_3 c_v, \quad \operatorname{wt}'(e_{14}) = t_1 t_4 c_v.$$

In order for wt' to come from some other map $(J^{\times})': V_{\text{int}}^{\times} \to \mathbb{R}_{>0}$, we must have

$$t_1 t_2 s_v = t_3 t_4 s_e = s'_v, \quad t_2 t_3 c_v = t_1 t_4 c_e = c'_v, \quad (s'_v)^2 + (c'_v)^2 = 1,$$

where $s'_v = \operatorname{sech}(2(J^{\times})'_v)$ and $c'_v = \tanh(2(J^{\times})'_v)$. But the above equations imply that $t_1 = t_2 = t_3 = t_4 = 1$, and it follows that ψ is injective.

Clearly ψ is continuous, and we now prove that it is surjective, and that its inverse is also continuous. We need the following simple observation, whose proof we leave as an exercise to the reader.

Lemma 4.4.18. Suppose that G^{\times} is a connected medial graph having at least one interior vertex. Then there exists an interior vertex $v \in V_{int}^{\times}$ and an index $i \in [2n]$ such that v is connected in G^{\times} to both d_i and d_{i+1} (modulo 2n).

Note that Proposition 4.2.21 follows from Lemma 4.4.18 as an immediate corollary.

We now return to the proof of Theorem 4.4.17, part (i). Let τ be a matching on [2n], π be the corresponding fixed-point free involution, $\mathcal{M} = \mathcal{M}_{\pi}$ be the corresponding positroid. Choose a reduced medial graph G^{\times} with medial pairing $\tau_{G^{\times}} = \tau$ (which exists by Lemma 4.4.10), and let G^{\Box} be the associated plabic graph. If G^{\times} is not connected then each of its connected components contains an even number of vertices. Moreover, in this case G^{\Box} induces the same partition of boundary vertices into connected components, and it is clear from the definition of the map Meas and Theorem 4.3.5 that each minor of $\overline{\text{Meas}}(G^{\Box}, \text{wt})$ is a product of the individual minors for each of the connected components. Thus the problem naturally separates into several independent problems, one for each connected component of G^{\times} , and in what follows, we assume that G^{\times} is connected.

Let $X \in \Pi_{\mathcal{M}}^{>0} \cap \operatorname{OG}_{\geq 0}(n, 2n)$. By Theorem 4.3.6, there exists a weight function wt : $E^{\Box} \to \mathbb{R}_{>0}$ such that $\overline{\operatorname{Meas}}(G^{\Box}, \operatorname{wt}) = X$, and our goal is to show that there exists a unique function $J^{\times} : V_{\operatorname{int}}^{\times} \to \mathbb{R}_{>0}$ such that $\psi(J^{\times}) : E^{\Box} \to \mathbb{R}_{>0}$ is obtained from wt using gauge transformations. Choose $v \in V_{\operatorname{int}}^{\times}$ and $i \in [2n]$ as in Lemma 4.4.18. Thus G^{\Box} contains a removable bridge between i and i+1 (modulo 2n). Denote by v_1 and v_2 the vertices of G^{\Box} adjacent to d_i and d_{i+1} respectively, and denote by v_3 and v_4 the other two vertices of G^{\Box} so that v_1, v_2, v_3, v_4 surround v in counterclockwise order. Applying gauge transformations to v_1 and v_2 , we may assume that wt($\{v_1, d_i\}$) = wt($\{v_2, d_{i+1}\}$) = 1. Let $s := \operatorname{wt}(\{v_1, v_2\}) > 0$. Applying gauge transformations to v_3 and v_4 , we may assume that wt($\{v_3, v_4\}$) = s, and wt($\{v_1, v_4\}$) = wt($\{v_2, v_3\}$) = c for some $c \in \mathbb{R}_{>0}$. Now, let $I := I_{i+1}^{\min}(\mathcal{M})$ and $J := [2n] \setminus I =$ $I_{i+1}^{\max}(\mathcal{M})$. Thus $i + 1 \in I$ and $i \in J$. Choose some $n \times 2n$ matrix representing X and denote by $X_k \in \mathbb{R}^n$ its k-th column. For $u \in \mathbb{R}^n$ and $k \in [2n]$, let $X(k \to u)$ denote the matrix obtained from X by replacing its k-th column with u. We introduce linear functions $h_I, h_J : \mathbb{R}^n \to \mathbb{R}$ as follows:

$$h_I(u) := \Delta_I(X(i+1 \to u)), \quad h_J(u) := \Delta_J(X(i \to u)).$$

Denote $u := X_i$ and $w := X_{i+1}$. Since $X \in OG(n, 2n)$, we get $h_I(w) = h_J(u)$ and $h_I(u) = h_J(w)$. Let $(G^{\Box'}, wt')$ be obtained from (G^{\Box}, wt) by removing the bridge $\{v_1, v_2\}$, and let $X' \in \operatorname{Gr}_{\geq 0}(n, 2n) := \overline{\operatorname{Meas}}(G^{\Box'}, wt')$. By the first part of Theorem 4.3.8, we have $s = h_I(w)/h_I(u)$. By Lemma 4.3.9, we have $(X')_i = u$ while $(X')_{i+1} = w - su$. Now, after removing degree 2 vertices³ v_1 and v_2 from $G^{\Box'}$ and denoting the resulting graph $G^{\Box''}$, each of the vertices d_i and d_{i+1} changes color and becomes adjacent to an edge of weight $\operatorname{wt'}(\{d_i, v_4\}) = \operatorname{wt'}(\{d_{i+1}, v_3\}) = c$. Let us define $\operatorname{wt''}$ to be the same as $\operatorname{wt'}$ except that $\operatorname{wt''}(\{d_i, v_4\}) = \operatorname{wt''}(\{d_{i+1}, v_3\}) := 1$, and let $X'' := \overline{\operatorname{Meas}}(G^{\Box''}, \operatorname{wt''})$. It is clear from the definition of $\overline{\operatorname{Meas}}$ that $(X'')_i = cu$ and $(X'')_{i+1} = \frac{1}{c}(w - su)$. Finally, $G^{\Box''}$ has a removable bridge between i and i + 1 so by the second part of Theorem 4.3.8, the weight $\operatorname{wt''}(\{v_3, v_4\})$ of this bridge must be equal to

wt''({
$$v_3, v_4$$
}) = $\frac{\Delta_J(X'')}{\Delta_{J\cup\{i+1\}\setminus\{i\}}(X'')} = \frac{h_J(cu)}{h_J\left(\frac{1}{c}(w-su)\right)},$

which after substituting $s := h_I(w)/h_I(u)$, $h_J(w) := h_I(u)$, $h_J(u) := h_I(w)$, and using the linearity of h_I , transforms into

wt''({
$$v_3, v_4$$
}) = $\frac{c^2 h_I(u) h_I(w)}{h_I(u)^2 - h_I(w)^2}$.

By construction, wt["]($\{v_3, v_4\}$) is equal to s, and thus after substituting $h_I(w) := sh_I(u)$ the above equation becomes

$$s = \frac{c^2 s h_I(u)^2}{(1-s^2)h_I(u)^2} = \frac{c^2 s}{1-s^2}$$

³It is well known that adding/removing vertices of degree 2 does not affect the result of $\overline{\text{Meas}}$, see [Lam16, Section 4.5 (M2)].

which is equivalent to $c^2 + s^2 = 1$. Since we have s, c > 0, it follows that 0 < s, c < 1 and thus there exists a unique $t \in \mathbb{R}_{>0}$ satisfying $s = \operatorname{sech}(2t)$ and $c = \tanh(2t)$. Moreover, it is clear that t depends continuously on the minors of X, since the denominators $\Delta_I(X) = \Delta_J(X)$ must be positive. Setting $J_v^{\times} := t$, we uncross (in the sense of Remark 4.4.16) the interior vertex v in G^{\times} so that the corresponding graph G^{\Box} would be obtained by removing the bridges $\{v_1, v_2\}$ and $\{v_3, v_4\}$, and proceed by induction, finishing the proof of part (i) (and therefore of part (ii) as well).

It remains to prove (4.4.5). There is a certain partial order (called the *affine Bruhat* order) on the set of decorated permutations such that two decorated permutations π, σ satisfy $\sigma \leq \pi$ if and only if the closure of the positroid cell $\Pi_{\mathcal{M}_{\pi}}^{>0}$ contains $\Pi_{\mathcal{M}_{\sigma}}^{>0}$. We have

$$\Pi_{\mathcal{M}_{\pi}} = \bigsqcup_{\sigma \leq \pi} \Pi_{\mathcal{M}_{\sigma}}^{>0},$$

see e.g. [Lam16, Theorem 8.1]. Moreover, the restriction of the affine Bruhat order to the set of fixed-point free involutions coincides with the poset P_n from Definition 4.4.15. Thus we have

$$\overline{(\Pi_{\tau}^{>0} \cap \mathrm{OG}_{\geq 0}(n,2n))} \subset \bigsqcup_{\sigma \in P_n: \sigma \leq \tau} \left(\Pi_{\sigma}^{>0} \cap \mathrm{OG}_{\geq 0}(n,2n) \right),$$

and it remains to prove that the left hand side of (4.4.5) contains the right hand side, i.e., that for all pairs $\sigma \leq \tau$ in P_n , the cell $\Pi_{\sigma}^{>0} \cap \operatorname{OG}_{\geq 0}(n, 2n)$ is contained inside the closure of $\Pi_{\tau}^{>0} \cap \operatorname{OG}_{\geq 0}(n, 2n)$. Clearly it is enough to consider the case $\sigma < \tau$. Then σ is obtained from τ by uncrossing some pairs $\{i, j\}$ and $\{i', j'\}$. Moreover, since G^{\times} is reduced, it contains a unique vertex $v \in V_{\text{int}}^{\times}$ which belongs to the medial strands connecting i to j and i' to j', and one of the two ways of uncrossing v yields a reduced medial graph with medial pairing σ , see Remark 4.4.16. But the two ways of uncrossing v correspond to sending J_v^{\times} to either 0 or ∞ , or equivalently, sending either $s_v \to 1, c_v \to 0$ or $s_v \to 0, c_v \to 1$. By part (i), we indeed see that $\Pi_{\sigma}^{>0} \cap \operatorname{OG}_{\geq 0}(n, 2n)$ is a subset of the closure of $\Pi_{\tau}^{>0} \cap \operatorname{OG}_{\geq 0}(n, 2n)$, finishing the proof of Theorem 4.4.17.

4.5 From the Ising model to the orthogonal Grassmannian

In this section, we study the relationship between the space $\overline{\mathcal{X}}_n$ and the space $OG_{>0}(n, 2n)$.

We start by slightly extending the notion of a planar Ising network so that contracting an edge in such a network would yield another such network. Throughout, we assume that a planar graph embedded in a disk has no *loops* (i.e. edges connecting a vertex to itself) or interior vertices of degree 1.

Definition 4.5.1. A generalized planar Ising network is a pair N = (G, J) where G = (V, E)is a planar graph embedded in a disk and $J : E \to \mathbb{R}_{>0} \cup \{\infty\}$. We denote $E = E_{\text{fin}} \sqcup E_{\infty}$, where $E_{\infty} := \{e \in E \mid J_e = \infty\}$. The Ising model associated to N is a probability measure on the space

$$\{-1,1\}^{V/E_{\infty}} := \{\sigma : V \to \{-1,1\} \mid \sigma_u = \sigma_v \text{ for all } \{u,v\} \in E_{\infty}\}.$$

The definitions of the probability $\mathbf{P}(\sigma)$ of a spin configuration $\sigma \in \{-1, 1\}^{V/E_{\infty}}$, the partition function Z, and a two-point boundary correlation $\langle \sigma_i \sigma_j \rangle$ are obtained from the corresponding definitions (4.1.1), (4.1.2), and (4.1.3) by replacing $\{-1, 1\}^V$ with $\{-1, 1\}^{V/E_{\infty}}$ and E with E_{fin} . As before, we let $M(G, J) = (\langle \sigma_i \sigma_j \rangle) \in \text{Mat}_n^{\text{sym}}(\mathbb{R}, 1)$ denote the boundary correlation matrix. Thus we have $m_{i,j} = 1$ whenever there exists a path connecting b_i to b_j by edges in E_{∞} .

Definition 4.5.2. To each generalized planar Ising network N = (G, J) we associate a medial network $N^{\times} = (G^{\times}, J^{\times})$. First suppose that $E = E_{\text{fin}}$, i.e., that J only takes values in $\mathbb{R}_{>0}$. Then the medial graph $G^{\times} = (V^{\times}, E^{\times})$ is obtained from G as in Figures 4-2 and 4-4. More precisely, the vertex set V^{\times} is given by

$$V^{\times} = \{d_1, \dots, d_{2n}\} \sqcup \{v_e \mid e \in E\},\$$

where the d_1, \ldots, d_{2n} are boundary vertices placed counterclockwise on the boundary of the disk so that b_i is between d_{2i-1} and d_{2i} , while v_e is the midpoint of the edge $e \in E$ of G.

The edges of G^{\times} are described as follows. If $e, e' \in E$ share both a vertex and a face then we connect v_e to $v_{e'}$ in G^{\times} . In addition, for each $i \in [n]$, we connect d_{2i-1} (resp., d_{2i}) with v_e where $e \in E$ is the first in the clockwise (resp., counterclockwise) order edge of G incident to b_i . If b_i is isolated in G then we connect d_{2i-1} to d_{2i} in G^{\times} . Thus each vertex $v_e \in V^{\times}$ has degree 4, and each boundary vertex d_i , $i \in [2n]$, has degree 1 in G^{\times} . Finally, we set $J_{v_e}^{\times} := J_e$.

Suppose now that $E \neq E_{\text{fin}}$, and thus J takes the value of ∞ on some edges of G. Let $N_{\text{fin}} = (G, J_{\text{fin}})$ be obtained from N by setting $(J_{\text{fin}})_e := 1$ for all $e \in E_{\infty}$ and $(J_{\text{fin}})_e := J_e$ for $e \in E_{\text{fin}}$. Let $N_{\text{fin}}^{\times} := (G_{\text{fin}}^{\times}, J_{\text{fin}}^{\times})$ be the medial network associated to N_{fin} . Then the medial network $N^{\times} = (G^{\times}, J^{\times})$ associated to N is obtained from N_{fin}^{\times} by "uncrossing" (see Remark 4.4.16) the vertices v_e of G_{fin}^{\times} for all $e \in E_{\infty}$. There are two ways to uncross the vertex v_e as in Figure 4-6, and we choose the one where no edge of the resulting graph G^{\times} intersects the corresponding edge e of G. This uniquely defines the medial graph G^{\times} , and we set $J_{v_e}^{\times} := J_e$ for all $e \in E_{\text{fin}}$.

The notion of a generalized planar Ising network is equivalent to the notion of a *cactus network* introduced in [Lam18, Section 4.1], where he also assigns a medial graph to it in the same way as in Definition 4.5.2.

Remark 4.5.3. Given a planar Ising network N = (G, J), the above procedure assigns a medial network $N^{\times} = (G^{\times}, J^{\times})$ to it. In Section 4.4, we assign a weighted plabic graph (G^{\Box}, wt) to N^{\times} . It is trivial to check that the same weighted plabic graph (G^{\Box}, wt) gets assigned to N = (G, J) in the construction described in Section 4.2.3.

To each medial graph G^{\times} (and thus to each generalized planar Ising network) we have associated a medial pairing τ in Section 4.2.5. Let us denote

 $\mathcal{X}_{\tau} := \{ M(G, J) \mid N = (G, J) \text{ is a generalized planar Ising network with medial pairing } \tau \}.$

The following stratification of $\overline{\mathcal{X}}_n$ will be deduced from Theorem 4.1.3 at the end of Section 4.6.

Proposition 4.5.4. The space $\overline{\mathcal{X}}_n$ decomposes as

$$\overline{\mathcal{X}}_n = \bigsqcup_{\tau \in P_n} \mathcal{X}_{\tau},\tag{4.5.1}$$

and for each $\tau \in P_n$, \mathcal{X}_{τ} is homeomorphic to $\mathbb{R}^{\operatorname{xing}(\tau)}$, with closure relations given by the poset P_n .

Given a (generalized) planar Ising network N = (G, J), we have described two ways to assign an element of $X \in OG_{\geq 0}(n, 2n)$ to N. First, one can take the boundary correlation matrix M = M(G, J), and let $X := \phi(M)$, as we did in Section 4.1.3. Second, one can construct a medial network $N^{\times} = (G^{\times}, J^{\times})$ as above, transform it into a weighted plabic graph (G^{\Box}, wt) , and then put $X' := \overline{\text{Meas}}(G^{\Box}, \text{wt})$, as we did in Section 4.2.3. Theorem 4.4.17, part (i) shows that the second map $J \mapsto \overline{\text{Meas}}(G^{\Box}, \text{wt})$ gives a homeomorphism between $\mathbb{R}_{>0}^E$ and $\Pi_{\tau}^{>0} \cap OG_{\geq 0}(n, 2n)$, where τ is the medial pairing of G^{\times} . The goal of the rest of this section is to show that the outputs $X = \phi(M)$ and $X' = \overline{\text{Meas}}(G^{\Box}, \text{wt})$ of these two maps coincide.

Theorem 4.5.5. Let N = (G, J) be a (generalized) planar Ising network with boundary correlation matrix M = M(G, J). Define $X := \phi(M)$. Let $N^{\times} = (G^{\times}, J^{\times})$ and (G^{\Box}, wt) be the medial network and the weighted plabic graph corresponding to N, and put $X' := \overline{\text{Meas}}(G^{\Box}, wt)$. Then X = X' in Gr(n, 2n).

We give two proofs of Theorem 4.5.5, one using Dubédat's results [Dub11], and one using a formula of Lis [Lis17] for boundary correlations in terms of the random alternating flow model. Note that it suffices to prove Theorem 4.5.5 only for planar Ising networks, since the corresponding statement for generalized planar Ising networks is obtained by taking the limit $J_e \to \infty$ for all $e \in E_{\infty}$.

Before we proceed with the proofs, we need several preliminary results.

Proof of Lemma 4.2.2. By (4.2.2), it is enough to show that $m_{i,j} = 2^{-n} \sum_{I \in \mathcal{E}_n(\{i,j\})} \Delta_I(\widetilde{M})$. For $k \in [n]$, denote by $e_k \in \mathbb{R}^n$ the k-th standard basis vector, and for $k \in [2n]$, denote by $(\widetilde{M})_k$ the k-th column of \widetilde{M} . Consider an $n \times n$ matrix A with columns

$$e_1, e_2, \ldots, e_{i-1}, (\widetilde{M})_{2i-1}, (\widetilde{M})_{2i}, e_{i+1}, \ldots, e_{j-1}, e_{j+1}, \ldots, e_n.$$

Since by Remark 4.1.2, $2e_k = (\widetilde{M})_{2k-1} + (\widetilde{M})_{2k}$ for all $k \in [n]$, we can expand det A in terms of minors of \widetilde{M} :

$$\det A = 2^{-n+2} \sum_{I \in \mathcal{E}_n(\{i,j\}): 2i-1, 2i \in I} \Delta_I(\widetilde{M}).$$

On the other hand, since most of the columns of A are basis vectors, we can compute its determinant directly: det $A = 2m_{i,j}$, where the sign in (4.1.4) is chosen so that we would have det $A = 2m_{i,j}$ and not det $A = -2m_{i,j}$. Similarly, we can define an $n \times n$ matrix B with columns

$$e_1, e_2, \ldots, e_{i-1}, e_{i+1}, \ldots, e_{j-1}, (\widetilde{M})_{2j-1}, (\widetilde{M})_{2j}, e_{j+1}, \ldots, e_n$$

We have $\det B = 2m_{i,j}$ as well, and

$$\det B = 2^{-n+2} \sum_{I \in \mathcal{E}_n(\{i,j\}): 2j-1, 2j \in I} \Delta_I(\widetilde{M}).$$

It remains to note that for all $I \in \mathcal{E}_n(\{i, j\})$, we have either $2i - 1, 2i \in I$ or $2j - 1, 2j \in I$, but not both. Thus

$$4m_{i,j} = \det A + \det B = 2^{-n+2} \sum_{I \in \mathcal{E}_n(\{i,j\})} \Delta_I(\widetilde{M}),$$

finishing the proof.

Lemma 4.5.6. Let $J := \{1, 3, ..., 2n - 1\}$ and $X' \in OG_{\geq 0}(n, 2n)$. Then for all $I \in {\binom{[2n]}{n}}$, we have

$$\Delta_I(X') \le \Delta_J(X').$$

Proof. This follows from Skandera's inequalities [Ska04] for $\operatorname{Gr}_{\geq 0}(n, 2n)$. Namely, by [FP16, Theorem 6.1], we have $\Delta_I(X')\Delta_{[2n]\setminus I}(X') \leq \Delta_J(X')\Delta_{[2n]\setminus J}(X')$ for all $X' \in \operatorname{Gr}_{\geq 0}(n, 2n)$. In particular, if $X' \in \operatorname{OG}_{\geq 0}(n, 2n)$, this becomes $(\Delta_I(X'))^2 \leq (\Delta_J(X'))^2$, which finishes the proof.

An important consequence of the above lemma is that for $J := \{1, 3, ..., 2n - 1\}$ and all $X' \in OG_{\geq 0}(n, 2n)$, we have $\Delta_J(X') > 0$, since we must have $\Delta_I(X') > 0$ for some $I \in {\binom{[2n]}{n}}$.

Lemma 4.5.7. The image $\phi(\operatorname{Mat}_{n}^{\operatorname{sym}}(\mathbb{R}, 1))$ contains $\operatorname{OG}_{\geq 0}(n, 2n)$. Equivalently, for any $X' \in \operatorname{OG}_{\geq 0}(n, 2n)$, there exists a matrix $M' \in \operatorname{Mat}_{n}^{\operatorname{sym}}(\mathbb{R}, 1)$ such that $X' = \phi(M')$ as elements of $\operatorname{Gr}(n, 2n)$.

Proof. We are going to use Lemma 4.2.2. Choose some $n \times 2n$ matrix A representing X' in Gr(n, 2n), and let $\widetilde{I_n}$ be the $n \times 2n$ matrix given by $(\widetilde{I_n})_{i,2i-1} = (\widetilde{I_n})_{i,2i} = 1$ and the remaining entries being zero. Remark 4.1.2 says that for a matrix $M' \in \operatorname{Mat}_n^{\operatorname{sym}}(\mathbb{R}, 1)$, we have $\widetilde{M'} \cdot (\widetilde{I_n})^T = 2I_n$, where I_n is the $n \times n$ identity matrix, and T denotes matrix transpose. Let $B := A \cdot (\widetilde{I_n})^T$. We claim that if $X' \in OG_{\geq 0}(n, 2n)$ then B is an invertible matrix. Indeed, by the multilinearity of the determinant, we have $\det B = \sum_{I \in \mathcal{E}_n(\emptyset)} \Delta_I(A)$. This sum contains only nonnegative terms, and the term $\Delta_{\{1,3,\dots,2n-1\}}(A)$ is positive by Lemma 4.5.6. Thus $B \in \operatorname{GL}_n(\mathbb{R})$ is invertible, and we can consider the matrix $C := 2 \cdot B^{-1} \cdot A$, which represents the same element X' in Gr(n, 2n). The matrix C satisfies $C \cdot (\widetilde{I_n})^T = 2I_n$, in particular, $C_{i,2j-1} = -C_{i,2j}$ for $i \neq j \in [n]$. We define the $n \times n$ matrix $M' = (m'_{i,j})$ by $m'_{i,i} := 1$ and $m'_{i,j} := (-1)^{i+j+1} (i < j) C_{i,2j-1}$ for $i \neq j \in [n]$, in agreement with (4.1.4). It turns out that M' is a symmetric matrix, since its entries can be recovered from the minors of C as follows. As we have mentioned in the proof of Lemma 4.2.2, for each $I \in \mathcal{E}_n(\{i, j\})$, we have either $2i-1, 2i \in I$ or $2j-1, 2j \in I$, but not both. Thus we can write $m'_{i,j} = 2^{-n+2} \sum_{I} \Delta_{I}(C)$, where the sum is over all $I \in \mathcal{E}_n(\{i, j\})$ such that $2i - 1, 2i \in I$. Similarly, we have $m'_{j,i} =$ $2^{-n+2}\sum_{I}\Delta_{I}(C)$, where the sum is over all $I \in \mathcal{E}_{n}(\{i, j\})$ such that $2j - 1, 2j \in I$. Since $\Delta_I(C) = \Delta_{[2n]\setminus I}(C)$ (because C represents $X' \in OG_{\geq 0}(n, 2n)$), we see that $m'_{i,j} = m'_{j,i}$, and thus M' belongs to $\operatorname{Mat}_n^{\operatorname{sym}}(\mathbb{R}, 1)$. Similarly, using

$$2^{n-1}C_{i,2i-1} = \sum_{I \in \mathcal{E}_n(\emptyset): 2i-1 \in I} \Delta_I(C) = \sum_{I \in \mathcal{E}_n(\emptyset): 2i \in I} \Delta_I(C) = 2^{n-1}C_{i,2i},$$

and $C_{i,2i-1} + C_{2i} = 2$, we get $C_{i,2i-1} = C_{i,2i} = 1$, and thus $\phi(M') = X'$. We are done with the proof of Lemma 4.5.7.

In order to prove Theorem 4.5.5, we need to show that $X := \phi(M)$ equals to X' :=

 $\overline{\text{Meas}}(G^{\Box}, \text{wt})$ as an element of Gr(n, 2n). By Theorem 4.4.17, we know that $X' \in OG_{\geq 0}(n, 2n)$.

By Lemma 4.5.7, we get a matrix $M' \in \operatorname{Mat}_n^{\operatorname{sym}}(\mathbb{R}, 1)$ such that $\phi(M') = X'$. Since $X = \phi(M)$, we have X = X' if and only if M = M'. By Lemma 4.2.2, the entries $m'_{i,j}$ of M' can be written as ratios of sums of minors of X'. By Theorem 4.3.5, each such minor is a sum over almost perfect matchings of G^{\Box} with prescribed boundary. Putting it all together, we get the following: for $i \neq j \in [n]$,

$$m_{i,j}' = \frac{\sum_{\mathcal{A}:\partial(\mathcal{A})\in\mathcal{E}_n(\{i,j\})} \operatorname{wt}(\mathcal{A})}{\sum_{\mathcal{A}:\partial(\mathcal{A})\in\mathcal{E}_n(\emptyset)} \operatorname{wt}(\mathcal{A})},$$
(4.5.2)

where the sums are over almost perfect matchings in G^{\Box} . Our goal is to show that $m'_{i,j}$ equals to $m_{i,j} := \langle \sigma_i \sigma_j \rangle$.

4.5.1 Dubédat's bosonization identity

Recall from Remark 4.2.10 that the planar bipartite graph $\widehat{G}^{\square} = (\widehat{V}^{\square}, \widehat{E}^{\square})$ is obtained from G^{\square} by adding an edge of weight 1 connecting d_{2i-1} to d_{2i} for each $i \in [n]$. We view \widehat{G}^{\square} as a weighted graph embedded in the sphere, and for each $i \in [n]$, we view the vertex b_i of G as a point inside of the square face of \widehat{G}^{\square} which contains d_{2i-1} and d_{2i} . We let $\operatorname{Match}(\widehat{G}^{\square})$, denote the set of *perfect matchings* of \widehat{G}^{\square} . Given such a perfect matching $\mathcal{A} \in \operatorname{Match}(\widehat{G}^{\square})$, its *weight* is the product of weights of its edges. We say that a *generic path* is a continuous path P in the sphere whose endpoints belong to the interiors of faces of \widehat{G}^{\square} , and which intersects every edge of \widehat{G}^{\square} at most once. Clearly every path in G is a generic path. We denote by $\widehat{E}_{-}^{\square}(P) \subset \widehat{E}^{\square}$ the set of edges of \widehat{G}^{\square} that intersect P. Given a perfect matching $\mathcal{A} \in \operatorname{Match}(\widehat{G}^{\square})$ and a generic path P, we define $\operatorname{wt}_{-}(\mathcal{A}, P) := (-1)^{|\widehat{E}_{-}^{\square}(P) \cap \mathcal{A}|} \operatorname{wt}(\mathcal{A})$. In other words, $\operatorname{wt}_{-}(\mathcal{A}, P)$ is the product of weights of edges in \mathcal{A} , where the weight of every edge that intersects P is negated. We are ready to state Dubédat's formula, as explained in [dT14, Corollary 1]. See also [BdT12] and [DCL17, Remark 4] for related results.

Proposition 4.5.8 ([Dub11]). Let N = (G, J) be a planar Ising network with n boundary vertices. Let $i, j \in [n]$, and choose some path P connecting b_i to b_j in G. Then the squared

boundary correlation function is given by

$$\langle \sigma_i \sigma_j \rangle^2 = \frac{\sum_{\mathcal{A} \in \text{Match}(\widehat{G}^{\square})} \text{wt}_{-}(\mathcal{A}, P)}{\sum_{\mathcal{A} \in \text{Match}(\widehat{G}^{\square})} \text{wt}(\mathcal{A})}.$$
(4.5.3)

Example 4.5.9. If G has two vertices and one edge as in Figure 4-2, then \widehat{G}^{\Box} has the following form:



There is exactly one path P in G that connects b_1 to b_2 , and $\widehat{E}_{-}^{\square}(P)$ consists of two edges of weight s_e . There are five perfect matchings of \widehat{G}^{\square} with weights s_e^2 , s_e , s_e , 1, and c_e^2 , respectively. Thus (4.5.3) in this case becomes

$$\langle \sigma_1 \sigma_2 \rangle^2 = \frac{1 - 2s_e + s_e^2 + c_e^2}{1 + 2s_e + s_e^2 + c_e^2} = \frac{1 - s_e}{1 + s_e}.$$

Recall from (4.2.7) that (4.2.9) in this case yields $\langle \sigma_1 \sigma_2 \rangle = \frac{c_e}{1+s_e}$. These two formulas agree:

$$\left(\frac{c_e}{1+s_e}\right)^2 = \frac{1-s_e}{1+s_e} \iff c_e^2 = (1+s_e)(1-s_e) \iff c_e^2 + s_e^2 = 1$$

First proof of Theorem 4.5.5. Recall that our goal is to show that $m'_{i,j} = \langle \sigma_i \sigma_j \rangle$, where $m'_{i,j}$ is given by (4.5.2). Since the right hand side of (4.5.2) is manifestly positive, it is enough to show $(m'_{i,j})^2 = \langle \sigma_i \sigma_j \rangle^2$, which is given by (4.5.3). Observe that every perfect matching \mathcal{A}' of \widehat{G}^{\Box} restricts to an almost perfect matching \mathcal{A} of G^{\Box} such that $\partial(\mathcal{A}) \in \mathcal{E}_n(\emptyset)$. Moreover, this gives a weight-preserving bijection between $\operatorname{Match}(\widehat{G}^{\Box})$ and such almost perfect matchings of G^{\Box} . Thus the denominators of (4.5.2) and (4.5.3) are equal, and therefore it suffices to show

$$\left(\sum_{\mathcal{A}:\partial(\mathcal{A})\in\mathcal{E}_n(\{i,j\})}\operatorname{wt}(\mathcal{A})\right)^2 = \left(\sum_{\mathcal{A}\in\operatorname{Match}(\widehat{G}^{\square})}\operatorname{wt}_-(\mathcal{A},P)\right)\left(\sum_{\mathcal{A}\in\operatorname{Match}(\widehat{G}^{\square})}\operatorname{wt}(\mathcal{A})\right),\qquad(4.5.4)$$

where the sum on the left is over almost perfect matchings of G^{\Box} .

Denote $\partial(P)$ to be the set of $i \in [n]$ such that $\{d_{2i-1}, d_{2i}\}$ intersects P. The following simple observation shows that the right hand side of (4.5.3) does not really depend on the choice of a generic path P.

Lemma 4.5.10. Let P, P' be two generic paths with the same endpoints. Then we have

$$(-1)^{|\widehat{E}_{-}^{\square}(P)|+|\partial(P)|} \left(\sum_{\mathcal{A}\in \operatorname{Match}(\widehat{G}^{\square})} \operatorname{wt}_{-}(\mathcal{A}, P) \right) = (-1)^{|\widehat{E}_{-}^{\square}(P')|+|\partial(P')|} \left(\sum_{\mathcal{A}\in \operatorname{Match}(\widehat{G}^{\square})} \operatorname{wt}_{-}(\mathcal{A}, P') \right).$$

Proof. Since we can transform P continuously into P' in a generic way, it is enough to show the above equality when there is a single vertex v of \widehat{G}^{\Box} that lies inside the region bounded by P and P'. If v belongs to the interior of the disk then it has degree 3 in \widehat{G}^{\Box} , so $|\partial(P)| = |\partial(P')|$ but $|\widehat{E}^{\Box}_{-}(P)|$ and $|\widehat{E}^{\Box}_{-}(P')|$ have different parity. Similarly, if $v = d_j$ for some $j \in [2n]$ then it has degree 2 in \widehat{G}^{\Box} and thus $|\widehat{E}^{\Box}_{-}(P)| + |\partial(P)|$ and $|\widehat{E}^{\Box}_{-}(P')| + |\partial(P')|$ again have different parity. On the other hand, each perfect matching \mathcal{A} of \widehat{G}^{\Box} contains exactly one edge incident to v, and thus we also have $\operatorname{wt}_{-}(\mathcal{A}, P) = -\operatorname{wt}_{-}(\mathcal{A}, P')$. The result follows.

Observe that when P is a path in G, it necessarily intersects an even number of edges of \widehat{G}^{\Box} , and $\partial(P) = \emptyset$. Suppose now that P connects b_i to b_j . Consider another generic path P' which connects b_i to b_j but only intersects two edges of \widehat{G}^{\Box} , namely, $\{d_{2i-1}, d_{2i}\}$ and $\{d_{2j-1}, d_{2j}\}$. By Lemma 4.5.10, we have $\sum_{\mathcal{A} \in \text{Match}(\widehat{G}^{\Box})} \text{wt}_{-}(\mathcal{A}, P) = \sum_{\mathcal{A} \in \text{Match}(\widehat{G}^{\Box})} \text{wt}_{-}(\mathcal{A}, P')$, and thus it suffices to show (4.5.4) for P replaced with P'.

For simplicity, we denote $a := d_{2i-1}$, $b := d_{2i}$, $c := d_{2j-1}$, $d := d_{2j}$. Define $\widehat{G}_{i,j}^{\Box}$ to be the subgraph obtained from \widehat{G}^{\Box} by removing the edges $\{a, b\}$ and $\{c, d\}$. Given a set $S \subset \{a, b, c, d\}$, denote by $\widehat{G}_{i,j}^{\Box}(S)$ the graph obtained from $\widehat{G}_{i,j}^{\Box}$ by removing the vertices of $\{a, b, c, d\} \setminus S$ (each such vertex is removed together with the unique edge incident to it). Finally, define

$$\kappa_S := \sum_{\mathcal{A} \in \operatorname{Match}(\widehat{G}_{i,j}^{\square}(S))} \operatorname{wt}(\mathcal{A})$$

for each $S \subset \{a, b, c, d\}$. Thus we have $\kappa_S = \kappa_{\{a, b, c, d\} \setminus S}$ for all S by Proposition 4.4.14. After

replacing P with P', (4.5.4) becomes

$$\left(\kappa_{a,d} + \kappa_{b,c}\right)^2 = \left(\kappa_{a,b,c,d} - \kappa_{a,b} - \kappa_{c,d} + \kappa_{\emptyset}\right) \left(\kappa_{a,b,c,d} + \kappa_{a,b} + \kappa_{c,d} + \kappa_{\emptyset}\right),\tag{4.5.5}$$

where we denote $\kappa_{\{a,d\}} = \kappa_{a,d}$, etc. Since $\kappa_{a,d} = \kappa_{b,c}$, the left hand side equals $4\kappa_{a,d}^2 = 4\kappa_{a,d}\kappa_{b,c}$. Similarly, the right hand side equals $4(\kappa_{\emptyset} - \kappa_{a,b})(\kappa_{\emptyset} + \kappa_{a,b}) = 4(\kappa_{\emptyset}^2 - \kappa_{a,b}^2) = 4(\kappa_{a,b,c,d}\kappa_{\emptyset} - \kappa_{a,b}\kappa_{c,d})$. Thus we need to show $\kappa_{a,d}\kappa_{b,c} = \kappa_{a,b,c,d}\kappa_{\emptyset} - \kappa_{a,b}\kappa_{c,d}$, or equivalently,

$$\kappa_{a,d}\kappa_{b,c} + \kappa_{a,b}\kappa_{c,d} = \kappa_{a,b,c,d}\kappa_{\emptyset}.$$
(4.5.6)

This is easy to prove bijectively using standard double-dimer arguments. For instance, superimposing a pair of matchings $(\mathcal{A}, \mathcal{A}') \in \operatorname{Match}(\widehat{G}_{i,j}^{\Box}(\{a,d\})) \times \operatorname{Match}(\widehat{G}_{i,j}^{\Box}(\{b,c\}))$ gives a union of cycles in $\widehat{G}_{i,j}^{\Box}$ together with a path Q connecting a to d and a path R connecting b to c. Thus $\mathcal{A} \oplus R$ (the symmetric difference of sets of edges) is a perfect matching of $\widehat{G}_{i,j}^{\Box}(\{a,b,c,d\})$, while $\mathcal{A}' \oplus R$ is a perfect matching of $\widehat{G}_{i,j}^{\Box}(\{a,b\})$. The case $(\mathcal{A}, \mathcal{A}') \in$ $\operatorname{Match}(\widehat{G}_{i,j}^{\Box}(\{a,b\})) \times \operatorname{Match}(\widehat{G}_{i,j}^{\Box}(\{c,d\}))$ is completely similar, and together they give a bijection between the left and the right hand sides of (4.5.6). We are done with the first proof of Theorem 4.5.5.

4.5.2 Random alternating flows of Lis

For our second proof, we use a formula due to Lis [Lis17], which he proved using the *random* currents model of [GHS70], see also [DC16, LW16]. Let us say that a clockwise bidirected edge (resp., a counterclockwise bidirected edge) is a directed cycle of length two in the plane which is oriented clockwise (resp., counterclockwise).

Suppose we are given a planar Ising network N = (G, J) with n boundary vertices and two disjoint subsets $A, B \subset [n]$ of the same size. We define $G^{A \sqcup B} = (V^{A \sqcup B}, E^{A \sqcup B})$ to be the graph obtained from G by adding a boundary spike at b_i for all $i \in A \sqcup B$.

An (A, B)-alternating flow F on G is a graph obtained from $G^{A \sqcup B}$ by replacing each edge $\{u, v\} \in E^{A \sqcup B}$ of $G^{A \sqcup B}$ by either

(a) an undirected edge, or

- (b) a directed edge $u \to v$ or $v \to u$, or
- (c) a clockwise or a counterclockwise bidirected edge,

so that the vertex b_i is incident to an outgoing (resp., incoming) edge if $i \in A$ (resp., if $i \in B$), and so that every other vertex $v \in V$ of G is incident to an even number of directed edges of F, and their directions alternate around v. The set of all (A, B)-alternating flows on G is denoted $\mathcal{F}_{A,B}(G)$.

For $e \in E$, we put

$$x_e := \tanh(J_e) = \frac{\exp(J_e) - \exp(-J_e)}{\exp(J_e) + \exp(-J_e)}, \quad y_e := \operatorname{sech}(J_e) = \frac{2}{\exp(J_e) + \exp(-J_e)}.$$

(Recall that x_e and y_e are not the same as $c_e = \tanh(2J_e)$ and $s_e = \operatorname{sech}(2J_e)$.) Given an edge $e \in E^{A \sqcup B}$ and an (A, B)-alternating flow $F \in \mathcal{F}_{A,B}(G)$, we set

$$\mathbf{w}(F,e) = \begin{cases} 2x_e/y_e^2, & \text{if } e \text{ is a directed edge in } F;\\ 2x_e^2/y_e^2, & \text{if } e \text{ is a bidirected edge in } F;\\ 1, & \text{otherwise.} \end{cases}$$

Following [Lis17, Eq. (4.2)], the weight of an (A, B)-alternating flow F is given by

$$\mathbf{w}(F) := 2^{|A| - |V(F)|} \prod_{e \in E^{A \sqcup B}} \mathbf{w}(F, e), \qquad (4.5.7)$$

where V(F) denotes the set of vertices $v \in V^{A \sqcup B} \setminus \{b_i \mid i \in A \sqcup B\}$ incident to a directed or a bidirected edge in F (note that b_i is always incident to a directed edge in F when $i \in A \sqcup B$).

Remark 4.5.11. The equivalence of (4.5.7) and [Lis17, Eq. (4.2)] is explained in the proof of [Lis17, Lemma 4.2].

We will be interested in the two special cases $A = B = \emptyset$ and $A = \{a\}, B = \{b\}$ for $a \neq b \in [n]$. We denote the corresponding graphs by G^{\emptyset} and $G^{a,b}$, respectively. Denote also $\mathcal{F}_{\emptyset}(G) := \mathcal{F}_{\emptyset,\emptyset}(G)$ and $\mathcal{F}_{a,b}(G) := \mathcal{F}_{\{a\},\{b\}}(G)$.

Lemma 4.5.12 ([Lis17, Lemma 5.2]). Let N = (G, J) be a planar Ising network with n boundary vertices, and let $i \neq j \in [n]$. Then the boundary correlation $\langle \sigma_i \sigma_j \rangle$ equals

$$\langle \sigma_i \sigma_j \rangle = \frac{\sum_{F \in \mathcal{F}_{a,b}(G)} w(F)}{\sum_{F \in \mathcal{F}_{\emptyset}(G)} w(F)}.$$
(4.5.8)

Second proof of Theorem 4.5.5. For a flow $F \in \mathcal{F}_{A,B}(G)$, let U(F) denote the set of vertices $v \in V^{A \sqcup B}$ that are not incident to a directed or a bidirected edge of F. Thus |U(F)| = |V| - |V(F)|, and we set

$$\tilde{\mathbf{w}}(F) := 2^{|V|} \mathbf{w}(F) = 2^{|A| + |U(F)|} \prod_{e \in E^{A \sqcup B}} \mathbf{w}(F, e)$$

Suppose that we are given a flow $F \in \mathcal{F}_{A,B}(G)$ together with a map $\alpha : U(F) \to \{-1, 1\}$. We say that the pair (F, α) is a *spinned flow*. The weight of a spinned flow is defined to be $\tilde{w}(F, \alpha) = 2^{|A|} \prod_{e \in E^{A \sqcup B}} w(F, e)$, so that $\tilde{w}(F) = \sum_{\alpha \in \{-1,1\}^{U(F)}} \tilde{w}(F, \alpha)$. We then define an order relation \leq on spinned flows by writing $(F, \alpha) \leq (F', \alpha')$ if all of the following conditions are satisfied:

- F' is obtained from F by making some undirected edges bidirected (thus $U(F') \subset U(F)$),
- the restriction of α to U(F') equals α' , and
- for every vertex $v \in U(F) \setminus U(F')$ such that $\alpha(v) = 1$ (resp., $\alpha(v) = -1$), all bidirected edges of F' incident to v are clockwise (resp., counterclockwise) bidirected edges.

Even though α' can be obtained from α by restricting it to $U(F') \subset U(F)$, we can also reconstruct α from (F', α') , since every vertex $v \in U(F) \setminus U(F')$ is incident to at least one bidirected edge of F', and either all such edges are clockwise (in which case we must have $\alpha(v) = 1$) or counterclockwise (in which case we must have $\alpha(v) = -1$).

Given a spinned flow (F, α) , we say that an undirected edge e of F is *active* if there exists a spinned flow $(F', \alpha') > (F, \alpha)$ such that e is bidirected in F'. Thus any $(F', \alpha') > (F, \alpha)$ is obtained from (F, α) by making some active edges bidirected. (An active edge can become either a clockwise or a counterclockwise bidirected edge but not both.) Equivalently, for



Figure 4-8: The correspondence θ between almost perfect matchings of G^{\Box} (top) and minimal spinned flows (bottom), where + (resp., -) next to a vertex v denotes $\alpha(v, e) = 1$ (resp., $\alpha(v, e) = -1$).

every undirected edge e of F and a vertex v incident to e, we set $\alpha(v, e) := \alpha(v)$ if $v \in U(F)$, and $\alpha(v, e) := 1$ (resp., $\alpha(v, e) := -1$) if $v \in V \setminus U(F)$ and after replacing e by a clockwise (resp., counterclockwise) bidirected edge, the directions of edges still alternate around v. Then an undirected edge $e = \{u, v\}$ of F is active if and only if we have $\alpha(v, e) = \alpha(u, e)$.

We say that a spinned flow (F, α) is *minimal* if it is minimal with respect to our order relation \leq . Equivalently, (F, α) is minimal if F has no bidirected edges. We denote $\mathcal{F}_{A,B}^{\min}(G)$ the set of all minimal spinned flows (F, α) where $F \in \mathcal{F}_{A,B}(G)$. For $(F, \alpha) \in \mathcal{F}_{A,B}^{\min}(G)$, we define its weight

$$\bar{\mathbf{w}}(F,\alpha) := \sum_{(F',\alpha') \ge (F,\alpha)} \tilde{\mathbf{w}}(F',\alpha') = 2^{|A|} \prod_{e \in E^{A \sqcup B}} \bar{\mathbf{w}}(F,\alpha,e),$$

where

$$\bar{\mathbf{w}}(F,\alpha,e) = \begin{cases} 2x_e/y_e^2, & \text{if } e \text{ is a directed edge in } F; \\ 1+2x_e^2/y_e^2 = (1+x_e^2)/y_e^2, & \text{if } e \text{ is an active edge of } (F,\alpha); \\ 1, & \text{otherwise.} \end{cases}$$
(4.5.9)

Here we used the fact that $x_e^2 + y_e^2 = 1$.

It thus follows that

$$\sum_{F \in \mathcal{F}_{A,B}(G)} \tilde{\mathbf{w}}(F) = \sum_{(F,\alpha) \in \mathcal{F}_{A,B}^{\min}(G)} \bar{\mathbf{w}}(F,\alpha).$$

Our goal is to give a map θ from almost perfect matchings of G^{\Box} to minimal spinned flows, which locally is defined in Figure 4-8. Namely, each edge $e = \{u, v\}$ of G corresponds to four interior vertices of G^{\Box} , as in Figure 4-2. Every almost perfect matching \mathcal{A} of G^{\Box} assigns a single edge to each of those four vertices, and there are seven ways to do so, as in Figure 4-8 (top). The product of the weights of edges of \mathcal{A} incident to one of the four vertices of G^{\Box} equals, respectively, to $c_e, c_e, s_e, s_e, 1, s_e^2, c_e^2$, see Figure 4-8 (top).

Similarly, for every minimal spinned flow (F, α) , e may be directed from u to v, or directed from v to u, or undirected, in which case the functions $\alpha(u, e), \alpha(v, e) \in \{-1, 1\}$ are well defined. As shown in Figure 4-8, the two matchings of weight c_e correspond to the case where e is directed in F, and the remaining five matchings correspond to e being undirected in F. Specifically, the two matchings of weight s_e correspond to the two cases where $\alpha(u, e) \neq$ $\alpha(v, e)$, the matching of weight 1 corresponds to the case $\alpha(u, e) = \alpha(v, e) = 1$, and the two matchings of weights s_e^2, c_e^2 correspond to a single case $\alpha(u, e) = \alpha(v, e) = -1$.

It is straightforward to check that these rules give a well defined map θ from the set of almost perfect matchings of G^{\Box} to the set of minimal spinned flows on G. Moreover, it is easy to check that the set $J := \partial(\mathcal{A}) \subset [2n]$ determines uniquely two disjoint sets $A, B \subset [n]$ such that $\theta(\mathcal{A}) \in \mathcal{F}_{A,B}^{\min}(G)$. Namely, we have $A = \{i \in [n] \mid 2i - 1, 2i \notin J\}$ and $B = \{i \in [n] \mid 2i - 1, 2i \in J\}$. Finally, let $(F, \alpha) \in \mathcal{F}_{A,B}^{\min}(G)$ be a minimal spinned flow, then we claim that

$$\bar{\mathbf{w}}(F,\alpha) = \frac{1}{\prod_{e \in E} s_e} \sum_{\mathcal{A}: \theta(\mathcal{A}) = (F,\alpha)} \mathrm{wt}(\mathcal{A}), \qquad (4.5.10)$$

where the sum is over almost perfect matchings \mathcal{A} of G^{\Box} . To see why this is the case, note that the multiplicative contribution of an edge $e \in E$ to $\bar{w}(F, \alpha)$ is given by (4.5.9). On the other hand, it is clear from Figure 4-8 that for any two almost perfect matchings $\mathcal{A}, \mathcal{A}'$ such that $\theta(\mathcal{A}) = \theta(\mathcal{A}')$, we have $S^{\times}(\mathcal{A}) = S^{\times}(\mathcal{A}')$, where $S^{\times}(\mathcal{A})$ is defined in the proof of Proposition 4.4.14. Thus the total weight of almost perfect matchings in the preimage of (F, α) under θ equals to the product over all edges $e \in E$ of q(e), defined in the proof of Proposition 4.4.14 as

$$q(e) = \begin{cases} c_e, & \text{if } e \text{ is a directed edge in } F; \\ 1 = s_e^2 + c_e^2, & \text{if } e \text{ is an active edge of } (F, \alpha); \\ s_e, & \text{otherwise.} \end{cases}$$
(4.5.11)

Indeed, if $e = \{u, v\}$ is an active edge of (F, α) then we either have $\alpha(u, e) = \alpha(v, e) = 1$ in which case *e* corresponds locally to a single matching of weight 1, or we have $\alpha(u, e) = \alpha(v, e) = -1$ in which case *e* corresponds locally to two matchings of weights s_e^2 and c_e^2 , which can be interchanged in every almost perfect matching in the preimage of (F, α) under θ . It remains to note that the right hand side of (4.5.11) can be obtained from the right hand side of (4.5.9) by multiplying by s_e :

$$s_e \bar{\mathbf{w}}(F, \alpha, e) = \begin{cases} 2x_e s_e / y_e^2 = c_e, & \text{if } e \text{ is a directed edge in } F;\\ (1 + x_e^2) s_e / y_e^2 = 1, & \text{if } e \text{ is an active edge of } (F, \alpha);\\ s_e, & \text{otherwise.} \end{cases}$$

Thus $s_e \bar{w}(F, \alpha, e) = q(e)$, which proves (4.5.10). This implies that the right hand sides of (4.5.8) and (4.5.2) are equal, finishing the second proof of Theorem 4.5.5.

4.6 Cyclic symmetry and a homeomorphism with a ball

By Theorems 4.5.5 and 4.4.17, the map ϕ is a stratification-preserving homeomorphism from $\overline{\mathcal{X}}_n$ to $OG_{\geq 0}(n, 2n)$, which is the first part of Theorem 4.1.3. In this section, we follow the strategy of Chapter 2 to prove the second part of Theorem 4.1.3, which states that $\overline{\mathcal{X}}_n$ is homeomorphic to a closed ball of dimension $\binom{n}{2}$.

Recall that the cyclic shift $2n \times 2n$ matrix S was defined in Section 4.2.2. We let S^T denote the matrix transpose of S. Also, recall from Corollary 2.2.4 that $\exp(t(S+S^T))$ sends $\operatorname{Gr}_{\geq 0}(k, N)$ to $\operatorname{Gr}_{>0}(k, N)$ for all t > 0. Here the totally positive Grassmannian $\operatorname{Gr}_{>0}(k, N)$ is defined in (4.3.1). Let us define the *totally positive orthogonal Grassmannian* to be the

intersection

$$\mathrm{OG}_{>0}(n,2n) := \mathrm{Gr}_{>0}(n,2n) \cap \mathrm{OG}(n,2n).$$

Lemma 4.6.1. For all $X \in OG_{\geq 0}(n, 2n)$ and all t > 0, we have $X \cdot \exp(t(S + S^T)) \in OG_{>0}(n, 2n)$.

Proof. In view of Corollary 2.2.4, it suffices to show that $X \cdot \exp(t(S + S^T)) \in OG(n, 2n)$. By Proposition 4.4.1, it is enough to prove that $\exp(t(S + S^T))$ belongs to the Lie group O(n, n) consisting of all $2n \times 2n$ matrices g preserving the bilinear form η , i.e., satisfying $\eta(gu, gv) = \eta(u, v)$ for all $u, v \in \mathbb{R}^{2n}$. It is a standard fact from Lie theory that $\exp(t(S + S^T))$ is such a matrix if and only if $S + S^T$ belongs to the Lie algebra of O(n, n). Let $D := \operatorname{diag}(1, -1, 1, -1, \ldots, 1, -1)$ be a $2n \times 2n$ diagonal matrix with $D_{i,i} = (-1)^{i-1}$ for $1 \leq i \leq 2n$. Then the Lie algebra of O(n, n) consists of all $2n \times 2n$ matrices B such that $B \cdot D = -D \cdot B^T$. It is easy to check that $S + S^T$ belongs to this Lie algebra. We are done with the proof. \Box

Example 4.6.2. For n = 2, the computation we need to check that $S + S^T$ belongs to the Lie algebra of O(n, n) goes as follows.

$$(S+S^{T}) \cdot D = \begin{pmatrix} 0 & 1 & 0 & -1 \\ 1 & 0 & 1 & 0 \\ 0 & 1 & 0 & 1 \\ -1 & 0 & 1 & 0 \end{pmatrix} \cdot \begin{pmatrix} 1 & 0 & 0 & 0 \\ 0 & -1 & 0 & 0 \\ 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & -1 \end{pmatrix} = \begin{pmatrix} 0 & -1 & 0 & 1 \\ 1 & 0 & 1 & 0 \\ 0 & -1 & 0 & -1 \\ -1 & 0 & 1 & 0 \end{pmatrix};$$
$$D \cdot (S+S^{T})^{T} = \begin{pmatrix} 1 & 0 & 0 & 0 \\ 0 & -1 & 0 & 0 \\ 0 & -1 & 0 & 0 \\ 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & -1 \end{pmatrix} \cdot \begin{pmatrix} 0 & 1 & 0 & -1 \\ 1 & 0 & 1 & 0 \\ 0 & 1 & 0 & 1 \\ -1 & 0 & 1 & 0 \end{pmatrix} = \begin{pmatrix} 0 & 1 & 0 & -1 \\ -1 & 0 & -1 & 0 \\ 0 & 1 & 0 & 1 \\ 1 & 0 & -1 & 0 \end{pmatrix}.$$

This shows $(S + S^T) \cdot D = -D \cdot (S + S^T)^T$ for n = 2.

Remark 4.6.3. For all $X \in \operatorname{Gr}_{\geq 0}(n, 2n)$, it was shown in Chapter 2 that the limit of $X \cdot \exp(t(S + S^T))$ as $t \to \infty$ is the unique cyclically symmetric element $X_0 \in \operatorname{Gr}_{\geq 0}(n, 2n)$ from Section 2.2.1. It follows from Lemma 4.6.1 that this point X_0 belongs to $\operatorname{OG}_{\geq 0}(n, 2n)$.

Proof of Theorem 4.1.3. As we have already discussed, the first part is a direct consequence of Theorems 4.5.5 and 4.4.17. The second part follows from Lemma 4.6.1 together with an argument completely identical to the one in Chapter 2, which we briefly outline here.

It was shown in Section 2.2 that the space $\operatorname{Gr}_{\geq 0}(n, 2n)$ can be explicitly realized as a subset of \mathbb{R}^N so that the image of $\operatorname{Gr}_{>0}(n, 2n)$ would be an embedded submanifold of \mathbb{R}^N , and that the action of $\exp(t(S + S^T))$ on $\operatorname{Gr}_{\geq 0}(n, 2n)$ extends to a contractive flow on \mathbb{R}^N . Since $\operatorname{OG}(n, 2n)$ is an embedded submanifold of $\operatorname{Gr}(n, 2n)$, we see that $Q := \operatorname{OG}_{>0}(n, 2n)$ becomes an embedded submanifold of \mathbb{R}^N whose closure is $\overline{Q} := \operatorname{OG}_{\geq 0}(n, 2n)$ in \mathbb{R}^N . By Lemma 4.6.1, the contractive flow $\exp(t(S + S^T))$ restricts to $\operatorname{OG}_{\geq 0}(n, 2n)$ and satisfies (2.1.1). The result follows.

Theorem 4.1.3 establishes the correspondence between the planar Ising model and the totally nonnegative orthogonal Grassmannian. Having finished its proof, we are in a position to deduce several other results stated in Section 4.2.

Proof of Theorem 4.2.4. This follows easily from studying the relationship of the map

$$(G,J)\mapsto (G^{\square},\mathrm{wt})\mapsto \overline{\mathrm{Meas}}(G^{\square},\mathrm{wt})\in \mathrm{OG}_{\geq 0}(n,2n)$$

with the duality map $(G, J) \mapsto (G^*, J^*)$. Namely, a planar Ising network N = (G, J)corresponds to a weighted plabic graph (G^{\Box}, wt) then the dual planar Ising network $N^* = (G^*, J^*)$ corresponds to a weighted plabic graph $((G^*)^{\Box}, \text{wt}^*)$ so that $(G^*)^{\Box}$ is obtained from G^{\Box} by switching the colors of all vertices and cyclically relabeling boundary vertices (i.e., $d_i^* := d_{i+1}$), and wt* is obtained from wt by swapping s_e and c_e for all $e \in E$. More precisely, for each $e \in E$ we have $\sinh(2J_e) \sinh(2J_{e^*}) = 1$ by (4.2.4). On the other hand, by (4.2.5), we have $\sinh(2J_e) = \frac{c_e}{s_e}$ and $\sinh(2J_{e^*}) = \frac{c_{e^*}}{s_{e^*}}$, so $s_{e^*} = c_e$ and $c_{e^*} = s_e$. It thus follows from the definition of Meas given in (4.3.2) that the minor Δ_I of Meas (G^{\Box}, wt) equals the minor $\Delta_{I'}$ of Meas $((G^*)^{\Box}, \text{wt}^*)$ where $I' = \{i+1 \mid i \in I\}$ (modulo 2n) for all $I \in {[2n] \choose n}$. This is equivalent to having Meas $(G^{\Box}, \text{wt}) \cdot S = Meas((G^*)^{\Box}, \text{wt}^*)$, which finishes the proof.

Proof of Proposition 4.2.5. We know from Section 2.2.1 that there exists a unique cyclically symmetric element $X_0 \in \text{Gr}_{\geq 0}(n, 2n)$, and by Remark 4.6.3, we have $X_0 \in \text{OG}_{\geq 0}(n, 2n)$. By
Theorem 4.1.3, X_0 corresponds to a unique boundary correlation matrix $M_0 \in \overline{\mathcal{X}}$ of a planar Ising network (i.e., $\phi(M_0) = X_0$). Since the operation $N = (G, J) \mapsto N^* = (G^*, J^*)$ amounts to applying the cyclic shift on $OG_{\geq 0}(n, 2n)$ by Theorem 4.2.4, we see that $M_0 = M(G, J)$ if and only if $M(G^*, J^*) = M(G, J)$.

Proof of Theorem 4.2.8. This also follows easily from Theorem 4.1.3 combined with (4.3.2).

Proof of Theorem 4.2.17. Follows from Theorem 4.1.3 and part (i) of Theorem 4.4.17. \Box

Proof of Theorem 4.2.20. Follows from Theorems 4.3.8 and 4.1.3. \Box

Proof of Theorem 4.2.22. Indeed, adjoining a boundary spike e to G' amounts to adding a pair of bridges to $(G')^{\Box}$. Adding bridges to $(G')^{\Box}$ translates into acting by $x_{\tilde{k}}(s_e)$ and $y_{\tilde{k}+1}(s_e)$ on $\overline{\text{Meas}}((G')^{\Box}, \text{wt'})$ by Lemma 4.3.9. However, we also rescale the edges incident to \tilde{k} and $\tilde{k} + 1$ by c_e between adding the two bridges, which amounts to multiplying $\overline{\text{Meas}}((G')^{\Box}, \text{wt})$ by a diagonal matrix $D_{\tilde{k}}(c_e)$ whose (\tilde{k}, \tilde{k}) -th and $(\tilde{k} + 1, \tilde{k} + 1)$ -th entries are equal to c_e and $1/c_e$, respectively. Thus if N = (G, J) is obtained from N' = (G', J') by adjoining a boundary spike, then the matrices M = M(G, J) and M' = M(G', J') are related by

$$\phi(M) = \overline{\text{Meas}}(G^{\Box}, \text{wt}) = \overline{\text{Meas}}((G')^{\Box}, \text{wt}') \cdot x_{\tilde{k}}(s_e) \cdot D_{\tilde{k}}(c_e) \cdot y_{\tilde{k}+1}(s_e) = \overline{\text{Meas}}((G')^{\Box}, \text{wt}') \cdot g_{\tilde{k}}(t),$$

which is equal to $\phi(M') \cdot g_{\tilde{k}}(t)$. Here $x_{\tilde{k}}(s_e) \cdot D_{\tilde{k}}(c_e) \cdot y_{\tilde{k}+1}(s_e) = g_{\tilde{k}}(t)$ reduces to the following 2×2 matrix computation, which relies on $s_e^2 + c_e^2 = 1$:

$$\begin{pmatrix} 1 & s_e \\ 0 & 1 \end{pmatrix} \cdot \begin{pmatrix} c_e & 0 \\ 0 & 1/c_e \end{pmatrix} \cdot \begin{pmatrix} 1 & 0 \\ s_e & 1 \end{pmatrix} = \begin{pmatrix} 1/c_e & s_e/c_e \\ s_e/c_e & 1/c_e \end{pmatrix}.$$

We are done with the case of adjoining a boundary spike. The case of adjoining a boundary edge is completely similar, and also follows by applying the duality from Section 4.2.2, which switches between s_e and c_e due to (4.2.4). We are done with the proof.

Proof of Proposition 4.5.4. It follows from Theorem 4.5.5 that ϕ sends \mathcal{X}_{τ} homeomorphically onto the cell $\Pi_{\tau}^{>0} \cap \mathrm{OG}_{\geq 0}(n, 2n)$, and thus the result follows from Theorem 4.1.3 combined with Theorem 4.4.17 (ii).

4.7 Generalized Griffiths' inequalities

In this section, our goal is to prove Theorem 4.2.13. Note that (4.2.10) follows from (4.2.11)by taking disjoint A and B such that |B| = 1. We thus focus on proving (4.2.11). Let us fix two subsets $A, B \subset [n]$, and let $C := A \oplus B$ be their symmetric difference. If C has odd size then both sides of (4.2.11) become zero. Thus we assume that the size of C is even. Recall that $\mathcal{E}_n(C) \subset {\binom{[2n]}{n}}$ consists of all *n*-element subsets I of [2n] such that $I \cap \{2i - 1, 2i\}$ has even size if and only if $i \in C$. (In particular, $\mathcal{E}_n(C)$ is empty when C has odd size.)

Throughout, we also fix a matrix $M = (m_{i,j}) \in \operatorname{Mat}_n^{\operatorname{sym}}(\mathbb{R}, 1)$, and we treat the entries $m_{i,j} = m_{j,i}$ as indeterminates for $i \neq j$.

Our first goal is to give a formula for the minors $\Delta_I(\widetilde{M})$ for $I \in \mathcal{E}_n(C)$.

Definition 4.7.1. Denote n' := n - |C|/2. Let $\alpha : [2n] \to [2n']$ be the unique orderpreserving map such that $\alpha(2i - 1) = \alpha(2i)$ if and only if $i \in C$. Let $\beta : [2n'] \to [n]$ be the unique order-preserving map such that the composition $\beta \circ \alpha : [2n] \to [n]$ sends both 2i - 1and 2i to i for all $i \in [n]$.

Example 4.7.2. Suppose that n = 4 and $C = \{1,3\}$. Then n' = 3, and the map $\alpha : [8] \rightarrow [6]$ sends the top row entries of the 2-line array $\begin{vmatrix} 1 & 2 & 3 & 4 & 5 & 6 & 7 & 8 \\ 1 & 1 & 2 & 3 & 4 & 4 & 5 & 6 \end{vmatrix}$ to the corresponding bottom row entries (i.e., $\alpha(1) = \alpha(2) = 1, \alpha(3) = 2$, etc.). Similarly, $\beta : [6] \rightarrow [4]$ sends the top row entries of $\begin{vmatrix} 1 & 2 & 3 & 4 & 5 & 6 \\ 1 & 2 & 2 & 3 & 4 & 4 \end{vmatrix}$ to its bottom row entries, giving rise to a composite map $\beta \circ \alpha$ represented by a 3-line array $\begin{vmatrix} 1 & 2 & 3 & 4 & 5 & 6 \\ 1 & 1 & 2 & 3 & 4 & 5 & 6 \\ 1 & 1 & 2 & 2 & 3 & 4 & 4 \end{vmatrix}$

For disjoint subsets $I, J \subset [2N]$ of the same size, we say that π is a matching between Iand J if π contains |I| = |J| pairs, and for each pair $\{i, j\} \in \pi$, we have either $i \in I, j \in J$ or $i \in J, j \in I$. The set of matchings between I and J is denoted by Match(I, J). For a subset $K \subset [2N]$ of even size, a matching on K is a partition of K into |K|/2 disjoint subsets of size 2, and we let Match(K) denote the set of matchings on K. Thus $Match(I, J) \subset Match(I \sqcup J)$,

and Match([2n]) is as a set equal to P_n . The function xing naturally extends to Match(I, J)and Match(K).

For each $I \in \mathcal{E}_n(C)$, we denote $I' := \alpha(I)$, and it is easy to check that we have $I' \in {\binom{[2n']}{n'}}$ for $I \in \mathcal{E}_n(C)$. Given a matching π on [2n'], we define a monomial $m_{\beta,\pi} := \prod_{\{i,j\}\in\pi} m_{\beta(i),\beta(j)}$. Similarly, given a subset $K \subset [n]$ of even size and a matching $\pi \in \text{Match}(K)$, we set $m_{\pi} := \prod_{\{i,j\}\in\pi} m_{i,j}$.

Proposition 4.7.3. For $I \in \mathcal{E}_n(C)$, we have

$$\Delta_I(\widetilde{M}) = 2^{|C|/2} \sum_{\pi \in \operatorname{Match}(I', [2n'] \setminus I')} (-1)^{\operatorname{xing}(\pi)} m_{\beta, \pi}.$$

Proof. This is essentially [Pos07, Proposition 5.2], see also [Lis17, Eq. (2.2)].

Remark 4.7.4. For any $I \in {\binom{[2n]}{n}}$, there exists a unique $C \subset [n]$ such that $I \in \mathcal{E}_n(C)$. Thus Proposition 4.7.3 actually gives a formula for all maximal minors of \widetilde{M} in terms of the entries of M.

Example 4.7.5. Let n = 4 and $C = \{1,3\}$ as in Example 4.7.2, so n' = 3. The matrices M and \widetilde{M} are given in Figure 4-1. Let $I := \{1, 2, 4, 7\}$. We have $I \in \mathcal{E}_n(C)$ since $|I \cap \{1, 2\}| = 2$ and $|I \cap \{5, 6\}| = 0$ are both even, while $|I \cap \{3, 4\}| = |I \cap \{7, 8\}| = 1$ are both odd. Next, $I' = \alpha(I) = \{1, 3, 5\} \in \binom{[2n']}{n'}$. Computing the maximal minor $\Delta_I(\widetilde{M})$, we find

$$\Delta_I(\widetilde{M}) = 2(m_{14}m_{23}m_{24} - m_{13}m_{24}^2 + m_{12}m_{24}m_{34} + m_{12}m_{23} + m_{14}m_{34} + m_{13}).$$

These six terms correspond to the six elements of $Match(I', [2n'] \setminus I') = Match(\{1, 3, 5\}, \{2, 4, 6\})$. For instance, the term $-m_{13}m_{24}^2$ comes from the matching $\pi = \{\{1, 4\}, \{3, 6\}, \{5, 2\}\}$ with $xing(\pi) = 3$, while the term m_{13} comes from the matching $\pi = \{\{1, 4\}, \{3, 2\}, \{5, 6\}\}$ with $xing(\pi) = 0$.

Definition 4.7.6. We introduce two disjoint subsets $A', B' \subset [2n']$ by:

$$A' := \{ i \in [2n'] \mid \beta(i) \in A \setminus B \} \cup \{ i \in [2n'] \mid \beta(i) = \beta(i+1) \in A \cap B \};$$
$$B' := \{ i \in [2n'] \mid \beta(i) \in B \setminus A \} \cup \{ i+1 \in [2n'] \mid \beta(i) = \beta(i+1) \in A \cap B \}.$$

Define the number $\epsilon \in \{0, 1\}$ mentioned in Theorem 4.2.13 by

$$\epsilon \equiv 1 + \sum_{i \in B'} i \pmod{2}. \tag{4.7.1}$$

Next, we state a classical result expressing correlations of the Ising model in terms of *Pfaffians*. Given a set $K \subset [n]$ of even size, we define

$$\operatorname{Pf}_{K}(M) := \sum_{\pi \in \operatorname{Match}(K)} (-1)^{\operatorname{xing}(\pi)} m_{\pi}.$$

If the size of K is odd, we set $Pf_K(M) := 0$. The following classical result expresses multipoint correlations in terms of two-point correlations.

Proposition 4.7.7 ([GBK78, Theorem A]). Given a planar Ising network N = (G, J), let M = M(G, J) be its boundary correlation matrix. Then for every set $K \subset [n]$, we have

$$\langle \sigma_K \rangle = \operatorname{Pf}_K(M) = \sum_{\pi \in \operatorname{Match}(K)} (-1)^{\operatorname{xing}(\pi)} \prod_{\{i,j\} \in \pi} \langle \sigma_i \sigma_j \rangle.$$

Thus Theorem 4.2.13 becomes a consequence of the following result.

Theorem 4.7.8. We have

$$\operatorname{Pf}_{C}(M) - \operatorname{Pf}_{A}(M) \operatorname{Pf}_{B}(M) = \frac{1}{2^{n-1}} \sum_{I \in \mathcal{E}_{n}(A \oplus B) \cap \mathcal{D}^{\epsilon}(B)} \Delta_{I}(\widetilde{M}).$$
(4.7.2)

Both sides of (4.7.2) are polynomials in the entries of M by Propositions 4.7.3 and 4.7.7.

Remark 4.7.9. It may look like the right hand side of (4.7.2) is not symmetric with respect to A and B, but in fact it is easy to see that

$$\mathcal{E}_n(A \oplus B) \cap \mathcal{D}^{\epsilon}(B) = \mathcal{E}_n(A \oplus B) \cap \mathcal{D}^{\epsilon'}(A),$$

where $\epsilon' \equiv 1 + n + \sum_{i \in A'} i \pmod{2}$.

Before we prove Theorem 4.7.8, we state a lemma which will be used repeatedly later.

Lemma 4.7.10. Let $[2N] = K_1 \sqcup K_2$ for two sets K_1, K_2 of even size. Let $\pi_1 \in Match(K_1)$, $\pi_2 \in Match(K_2)$, and let $\pi_1 \sqcup \pi_2 \in Match([2N])$ be obtained by superimposing π_1 and π_2 . Then

$$\operatorname{xing}(\pi_1 \sqcup \pi_2) - \operatorname{xing}(\pi_1) - \operatorname{xing}(\pi_2) \equiv |K_1|/2 + \sum_{i \in K_1} i \equiv |K_2|/2 + \sum_{i \in K_2} i \pmod{2}.$$
 (4.7.3)

Proof. Suppose that there is $i \in K_1$ such that i > 1 and $i - 1 \notin K_1$. Then replacing K_1 with $K_1 \setminus \{i\} \cup \{i - 1\}$ and modifying π_1, π_2 accordingly changes the parity of each side of (4.7.3). Applying this operation repeatedly until $K_1 = [|K_1|]$, the result follows.

Proof of Theorem 4.7.8. First, it is straightforward to check that if $i \in [n] \setminus (A \cup B)$ then removing *i* from [n] does not affect the left and right hand sides of (4.7.2). Thus from now on we assume that $A \cup B = [n]$.

Assume first that $A \cap B = \emptyset$. This implies $C = A \sqcup B = [n]$, *n* is even, n' = n/2, A' = A, and B' = B. For a matching $\pi \in \text{Match}([n])$, we are going to compare the coefficients of m_{π} on both sides of (4.7.2), and show that in all cases they are equal.

Here for two disjoint subsets I and J, we say that a matching $\pi \in \operatorname{Match}(I \sqcup J)$ restricts to I and J and write $\pi \in \operatorname{Match}|_{I,J}^{\operatorname{res}}$ if for all $\{i, j\} \in \pi$ we have either $\{i, j\} \subset I$ or $\{i, j\} \subset J$. We denote by $\pi|_I \in \operatorname{Match}(I)$ and $\pi|_J \in \operatorname{Match}(J)$ the corresponding restricted matchings. Thus the set $\operatorname{Match}|_{I,J}^{\operatorname{res}} \subset \operatorname{Match}(I \sqcup J)$ is in bijection with $\operatorname{Match}(I) \times \operatorname{Match}(J)$.

For $\pi \in Match([n])$, the coefficient of m_{π} in $Pf_C(M) - Pf_A(M) Pf_B(M)$ is equal to

$$c_{\text{left}}(\pi) = \begin{cases} (-1)^{\text{xing}(\pi)} - (-1)^{\text{xing}(\pi|_A)} (-1)^{\text{xing}(\pi|_B)} & \text{if } \pi \in \text{Match} \mid_{A,B}^{\text{res}}; \\ (-1)^{\text{xing}(\pi)} & \text{otherwise.} \end{cases}$$

For the right hand side of (4.7.2), observe that by Definition 4.2.12, a given set $I \in \mathcal{E}_n(C)$ belongs to $\mathcal{D}^{\epsilon}(B)$ if and only if

$$\sum_{i \in I \cap \widetilde{B}} i \equiv 1 + \sum_{i \in B} i \pmod{2},$$

because B = B'. Since C = [n], we have $I \in \mathcal{E}_n(C)$ if and only if $I \cap \{2i-1, 2i\}$ has even size

for all $i \in [n]$. Let us say that a set $I \in \mathcal{E}_n(C)$ is compatible with π if $\pi \in Match(I', [2n'] \setminus I')$. It is clear that the coefficient of m_{π} in the right hand side of (4.7.2) is equal to

$$c_{\text{right}}(\pi) := \frac{2^{n/2}}{2^{n-1}} (-1)^{\operatorname{xing}(\pi)} N(\pi),$$

where $N(\pi)$ is the number of $I \in \mathcal{E}_n(C) \cap \mathcal{D}^{\epsilon}(B)$ compatible with π . We claim that $N(\pi)$ is given by

$$N(\pi) = \begin{cases} 2^{n/2} & \text{if } \pi \in \text{Match} \mid_{A,B}^{\text{res}} \text{ and } |B|/2 \equiv 1 + \sum_{i \in B} i \pmod{2}; \\ 0 & \text{if } \pi \in \text{Match} \mid_{A,B}^{\text{res}} \text{ and } |B|/2 \not\equiv 1 + \sum_{i \in B} i \pmod{2}; \\ 2^{n/2-1} & \text{if } \pi \notin \text{Match} \mid_{A,B}^{\text{res}}. \end{cases}$$
(4.7.4)

Indeed, assume first $\pi \notin \text{Match}|_{A,B}^{\text{res}}$. Then there exists a pair $\{i, j\} \in \pi$ such that $i \in A$ and $j \in B$. Note that there are a total of $2^{n/2}$ sets $I \in \mathcal{E}_n(C)$ compatible with π . Each such set satisfies either $2i - 1, 2i \in I, 2j - 1, 2j \notin I$ or $2j - 1, 2j \in I, 2i - 1, 2i \notin I$, so they naturally split into pairs $\{I, I \oplus \{2i-1, 2i, 2j-1, 2j\}\}$. Exactly one set I in each pair satisfies $\sum_{i \in I \cap \widetilde{B}} i \equiv \epsilon \pmod{2}$. Thus the total number $N(\pi)$ of sets $I \in \mathcal{E}_n(C) \cap \mathcal{D}^{\epsilon}(B)$ compatible with π equals $2^{n/2-1}$ in this case.

Assume now that $\pi \in \text{Match}|_{A,B}^{\text{res}}$. Then for any $I \in \mathcal{E}_n(C)$ compatible with π , we have $\sum_{i \in I \cap \tilde{B}} i \equiv |B|/2 \pmod{2}$. Thus, either all I compatible with π belong to $\mathcal{E}_n(C) \cap \mathcal{D}^{\epsilon}(B)$, in which case we get $2^{n/2}$ of them, or they all belong to $\mathcal{E}_n(C) \cap \mathcal{D}^{1-\epsilon}(B)$, in which case we get $N(\pi) = 0$. It is easy to check that the former case happens exactly when $|B|/2 \equiv 1 + \sum_{i \in B} i \pmod{2}$. This shows (4.7.4), which, combined with (4.7.3), clearly implies $c_{\text{left}}(\pi) = c_{\text{right}}(\pi)$. We are done with the case $A \cap B = \emptyset$.

Assume now that $A \cap B \neq \emptyset$. Since we are assuming $A \cup B = [n]$, we have $2n' = n + |A \cap B|$, and $[2n'] = A' \sqcup B'$.

For $k, k + 1 \in [2n']$ such that $\beta(k) = \beta(k + 1) = j$, let the *flipping* of a matching $\pi \in \text{Match}([2n'])$ at j be a matching π' obtained from π' by "swapping" the elements k, k + 1, i.e., $\pi' = \pi \setminus \{\{a, k\}, \{b, k + 1\}\} \cup \{\{a, k + 1\}, \{b, k\}\}$ for some $a, b \in [2n']$. (If $\{k, k + 1\} \in \pi$ then we set $\pi' := \pi$.)

Note that two different matchings $\pi, \pi' \in \operatorname{Match}([2n'])$ can yield the same monomial $m_{\beta,\pi}$ if they differ by a flipping at some $j \in A \cap B$. We write in this case $\pi \sim \pi'$, and denote $\Pi = [\pi]$ the equivalence class of matchings $\pi \in \operatorname{Match}([2n'])$ with respect to this equivalence relation. Thus we have $m_{\beta,\pi} = m_{\beta,\pi'}$ if and only if $\pi \sim \pi'$, and we denote $m_{\beta,[\pi]} = m_{\beta,\pi}$.

We say that π is trivial on $j \in A \cap B$, denoted $\pi \perp j$, if the pair $\{k, k+1\} = \beta^{-1}(j)$ belongs to π . We say that π is trivial on $A \cap B$, denoted $\pi \perp A \cap B$, if π is trivial on all elements of $A \cap B$. It is easy to see that triviality depends only on the equivalence class of π , justifying the notation $\Pi \perp j$ and $\Pi \perp A \cap B$. In the case $\Pi \perp A \cap B$, Π consists of just a single element π , so we define $\operatorname{xing}(\Pi) := \operatorname{xing}(\pi)$ in this case.

Let $\pi \in Match([2n'])$ be a matching. Consider a graph $\Gamma_{\pi} = ([2n'], E(\pi))$ with vertex set [2n'] and edge set

$$E(\pi) = \pi \cup \{\{k, k+1\} \mid k \in [2n'] \text{ is such that } \beta(k) = \beta(k+1)\} = \pi \cup \{\beta^{-1}(j) \mid j \in A \cap B\}.$$

(Here if π is trivial on j then the corresponding pair $\{k, k+1\} = \beta^{-1}(j)$ belongs to both π and $\{\beta^{-1}(j) \mid j \in A \cap B\}$, so Γ_{π} contains *two* edges connecting k to k+1.)

Each connected component of Γ_{π} contains an even number of vertices and is either a cycle or a path. We denote by $\operatorname{Conn}(\Gamma_{\pi})$ the set of connected components of Γ_{π} and by $\operatorname{Cyc}(\Gamma_{\pi}) \subset$ $\operatorname{Conn}(\Gamma_{\pi})$ the set of cycles of Γ_{π} . Clearly, flipping π at $j \in A \cap B$ preserves the set of vertices of each connected component of Γ_{π} . In particular, we have $\operatorname{cyc}(\pi) := |\operatorname{Cyc}(\pi)| = |\operatorname{Cyc}(\pi')|$ for all $\pi \sim \pi'$, and thus we set $\operatorname{cyc}([\pi]) := \operatorname{cyc}(\pi)$.

For each equivalence class Π of matchings we are going to compare the coefficients of $m_{\beta,\Pi}$ on both sides of (4.7.2), and show that they are equal. (Recall that we have $m_{\beta,\pi} = m_{\beta,\pi'}$ if and only if $\pi \sim \pi'$, and in particular we have $m_{\beta,\Pi} \neq m_{\beta,\Pi'}$ for $\Pi \neq \Pi'$.)

The coefficient of $m_{\beta,\Pi}$ in the left hand side of (4.7.2) equals

$$c_{\text{left}}(\Pi) = \begin{cases} (-1)^{\text{xing}(\Pi)} & \text{if } \Pi \cap \text{Match} \mid_{A',B'}^{\text{res}} = \emptyset \text{ and } \Pi \perp A \cap B; \\ -(-1)^{\text{xing}(\pi|_{A'}) + \text{xing}(\pi|_{B'})} 2^{\text{cyc}(\Pi)} & \text{if } \pi \in \Pi \cap \text{Match} \mid_{A',B'}^{\text{res}} \text{ and } \Pi \not\perp A \cap B; \\ 0 & \text{if } \Pi \cap \text{Match} \mid_{A',B'}^{\text{res}} = \emptyset \text{ and } \Pi \not\perp A \cap B. \end{cases}$$

Note that the case $\Pi \cap \text{Match}|_{A',B'}^{\text{res}} \neq \emptyset$, $\Pi \perp A \cap B$ is impossible because $A \cap B \neq \emptyset$. For the second case $\pi \in \Pi \cap \text{Match}|_{A',B'}^{\text{res}}$, $\Pi \not\perp A \cap B$, the parity of $\operatorname{xing}(\pi|_{A'}) + \operatorname{xing}(\pi|_{B'})$ is uniquely determined, even if π itself may not be uniquely determined. Indeed, any two $\pi, \pi' \in \Pi \cap \text{Match}|_{A',B'}^{\text{res}}$ can be obtained from each other by flipping all $j \in S$ for some $S \subset A \cap B$ such that $\beta^{-1}(S)$ is a union of cycles of Γ_{π} (and thus a union of cycles of $\Gamma_{\pi'}$). Clearly in this case we have $\operatorname{xing}(\pi|_{A'}) + \operatorname{xing}(\pi|_{B'}) = \operatorname{xing}(\pi'|_{A'}) + \operatorname{xing}(\pi'|_{B'})$.

Recall that $I \in \mathcal{E}_n(C)$ is compatible with π if $\pi \in \operatorname{Match}(I', [2n'] \setminus I')$. In this case we also say that I' is compatible with π . Note that the map $I \mapsto I' = \alpha(I)$ is injective on $\mathcal{E}_n(C)$, and we denote by $\mathcal{E}'_n(C) := \{I' \mid I \in \mathcal{E}_n(C)\} \subset {\binom{[2n']}{n'}}$ the image of this map. Thus $I' \in \mathcal{E}'_n(C)$ if and only if |I'| = n' and $|I' \cap \{k, k+1\}| = 1$ for all $k \in [2n']$ such that $\beta(k) = \beta(k+1)$.

It is clear that the coefficient of $m_{\beta,\Pi}$ in the right hand side of (4.7.3) is equal to

$$c_{\text{right}}(\Pi) = \frac{2^{n-n'}}{2^{n-1}} \sum_{(\pi,J)} (-1)^{\operatorname{xing}(\pi)},$$

where the sum is over all pairs (π, J) such that $\pi \in \Pi$ and $J \in \mathcal{E}_n(C)$ is compatible with π . We claim that this sum equals

$$\sum_{(\pi,J)} (-1)^{\operatorname{xing}(\pi)} = \begin{cases} (-1)^{\operatorname{xing}(\Pi)} 2^{n'-1} & \text{if } \Pi \cap \operatorname{Match}|_{A',B'}^{\operatorname{res}} = \emptyset \text{ and } \Pi \perp A \cap B; \\ -(-1)^{\operatorname{xing}(\pi|_{A'}) + \operatorname{xing}(\pi|_{B'})} 2^{n'-1 + \operatorname{cyc}(\Pi)} & \text{if } \pi \in \Pi \cap \operatorname{Match}|_{A',B'}^{\operatorname{res}} \text{ and } \Pi \not\perp A \cap B; \\ 0 & \text{if } \Pi \cap \operatorname{Match}|_{A',B'}^{\operatorname{res}} = \emptyset \text{ and } \Pi \not\perp A \cap B. \end{cases}$$

Consider the first case $\Pi \cap \text{Match}|_{A',B'}^{\text{res}} = \emptyset$, $\Pi \perp A \cap B$. Let π be the unique element of Π . Pick some $j \in A \cap B$, and let $\{k, k+1\} := \beta^{-1}(j)$. For each pair $\{i, i'\} \in \pi$ except for $\{k, k+1\}$, choose arbitrarily which of i and i' belongs to J' and which does not. There are total $2^{n'-1}$ ways to do this. For each of the $2^{n'-1}$ ways, the condition $\sum_{i \in J \cap \widetilde{B}} i \equiv \epsilon$ (mod 2) uniquely determines whether k or k+1 must belong to J' in order for J to belong to $\mathcal{E}_n(C) \cap \mathcal{D}^{\epsilon}(B)$. We are done with the first case.

Consider now the third case $\Pi \cap \text{Match}|_{A',B'}^{\text{res}} = \emptyset$, $\Pi \not\perp A \cap B$. It follows that there is a pair $\{i, i'\}$ common to all $\pi \in \Pi$ such that $\beta(i) \in A \setminus B$ and $\beta(i') \in B \setminus A$. There is also a pair $\{k, k+1\} = \beta^{-1}(j)$ for some $j \in A \cap B$ such that k and k+1 are not connected to each other in any $\pi \in \Pi$. Consider a map $\gamma : \mathcal{E}_n(C) \cap \mathcal{D}^{\epsilon}(B) \to \mathcal{E}_n(C) \cap \mathcal{D}^{\epsilon}(B)$ defined as follows. We put $\gamma(I) = J$ for $I, J \in \mathcal{E}_n(C) \cap \mathcal{D}^{\epsilon}(B)$ if $J' = I' \oplus \{i, i', k, k+1\}$. Let π' be obtained from π by flipping at j. We claim that $I \in \mathcal{E}_n(C) \cap \mathcal{D}^{\epsilon}(B)$ is compatible with π if and only if $\gamma(I) \in \mathcal{E}_n(C) \cap \mathcal{D}^{\epsilon}(B)$ is compatible with π' . Moreover, $\operatorname{xing}(\pi')$ differs from $\operatorname{xing}(\pi)$ by 1. Thus, we have a sign-reversing involution that cancels all the terms in $\sum_{(\pi,J)} (-1)^{\operatorname{xing}(\pi)}$, proving that it is equal to 0 in the third case.

Finally, consider the second case $\pi \in \Pi \cap \text{Match}|_{A',B'}^{\text{res}}$, $\Pi \not\perp A \cap B$. We are going to show that

$$\sum_{(\pi,J)} (-1)^{\operatorname{xing}(\pi)} = -(-1)^{\operatorname{xing}(\pi|_{A'}) + \operatorname{xing}(\pi|_{B'})} 2^{n' - 1 + \operatorname{cyc}(\Pi)}$$

Fix a matching $\pi \in \Pi \cap \text{Match}|_{A',B'}^{\text{res}}$. We claim that for any $\pi' \in \Pi$, there exists $\epsilon_{\pi'} \in \{0,1\}$ such that for all $I \in \mathcal{E}_n(C)$ compatible with π' , we have $I \in \mathcal{D}^{\epsilon_{\pi'}}(B)$, that is,

$$\sum_{i\in I\cap\widetilde{B}}i\equiv\epsilon_{\pi'}\pmod{2}.$$

Indeed, each component of $\Gamma_{\pi'}$ is a bipartite graph (a path or a cycle with an even number of vertices) so let us color its vertices black and white in a bipartite way. It is easy to check that $I \in \mathcal{E}_n(C)$ is compatible with π' if and only if for each connected component of $\Gamma_{\pi'}$, I'contains either all white vertices or all black vertices of this component. Let $S \subset [2n']$ be the set of vertices of a connected component of $\Gamma_{\pi'}$, and let $J \in \mathcal{E}_n(C)$ be such that $J' = I' \oplus S$ (thus J' is obtained from I' by switching from white to black inside the component S). It is straightforward to check that because π' is equivalent to $\pi \in \text{Match}|_{A',B'}^{\text{res}}$, we have

$$\sum_{i \in I \cap \widetilde{B}} i \equiv \sum_{i \in J \cap \widetilde{B}} i \pmod{2}.$$

We thus define $\epsilon_{\pi'} := \sum_{i \in I \cap \widetilde{B}} i$ for some $I \in \mathcal{E}_n(C)$ compatible with π' , and we have shown that $\epsilon_{\pi'}$ does not depend on the choice of I.

Next, flipping π' at some $j \in A \cap B$ changes $\epsilon_{\pi'}$ into $1 - \epsilon_{\pi'}$. Thus we have $\epsilon_{\pi'} = \epsilon$ for precisely half of the matchings $\pi' \in \Pi$, and for each such matching π' , there are $2^{|\operatorname{Conn}(\Gamma_{\pi'})|} = 2^{n'-|A\cap B|}$ sets $J \in \mathcal{E}_n(C)$ compatible with π' . Since $\Pi \cap \operatorname{Match}|_{A',B'}^{\operatorname{res}} \neq \emptyset$, we have $\Pi \not\perp j$ for each $j \in A \cap B$, and thus $|\Pi| = 2^{|A \cap B|}$. Therefore the total number of pairs (π', J) such that $\pi' \in \Pi$ and $J \in \mathcal{E}_n(C) \cap \mathcal{D}^{\epsilon}(B)$ equals $2^{n'-1}$, and for each of them, the parity of $\operatorname{xing}(\pi')$ is the same, because it satisfies

$$\epsilon_{\pi} - \epsilon \equiv \operatorname{xing}(\pi) - \operatorname{xing}(\pi').$$

Thus in order to finish the proof, it suffices to show that

$$\operatorname{xing}(\pi) - \operatorname{xing}(\pi|_{A'}) - \operatorname{xing}(\pi|_{B'}) \not\equiv \epsilon_{\pi} - \epsilon \pmod{2}. \tag{4.7.5}$$

Let $J \in \mathcal{E}_n(C)$ be compatible with π . Then by the definition of ϵ_{π} and ϵ , we have

$$\epsilon_{\pi} - \epsilon \equiv \sum_{i \in J \cap \widetilde{B}} i + \sum_{i \in B'} i + 1 \pmod{2}.$$

Combining this with Lemma 4.7.10, Equation (4.7.5) transforms into

$$|B'|/2 + \sum_{i \in B'} i \equiv \sum_{i \in J \cap \widetilde{B}} i + \sum_{i \in B'} i \pmod{2},$$

equivalently, $|B'|/2 \equiv \sum_{i \in J \cap \tilde{B}} i \pmod{2}$, which follows in a straightforward way since $\pi \in$ Match $|_{A',B'}^{\text{res}}$, J contains either all white or all black vertices in each connected component of Γ_{π} , and hence the contribution of each connected component to the left and right hand side is the same. We are done with the proof of Theorem 4.7.8, which implies Theorem 4.2.13 as discussed previously.

4.8 Open problems and future directions

In this section, we briefly list several questions that in our opinion would be worth exploring further.

According to (4.4.4), $OG_{\geq 0}(n, 2n)$ is a union of cells labeled by matchings τ on [2n], and each such cell $\Pi_{\tau}^{>0} \cap OG_{\geq 0}(n, 2n)$ is homeomorphic to $\mathbb{R}^{\operatorname{xing}(\tau)}$. It would be nice to understand the topology closures of these cells. In fact, we have a conjecture, analogous to [Pos07, Conjecture 3.6].

Conjecture 4.8.1. The cell decomposition (4.4.4) gives a regular CW complex structure on $OG_{\geq 0}(n, 2n)$. In other words, the closure of each cell $\Pi_{\tau}^{>0} \cap OG_{\geq 0}(n, 2n)$, given by (4.4.5), is homeomorphic to a closed xing(τ)-dimensional ball.

As we have already mentioned, the poset P_n of cells in $OG_{\geq 0}(n, 2n)$ has been studied in the context of electrical networks. In particular, it has been shown to be shellable and Eulerian by [Lam15, HK18], which shows that P_n is the face poset of *some* regular CW complex by a result of [Bjö84]. This leads to our next question.

Question 4.8.2. Does there exist a natural stratification-preserving homeomorphism between the compactification E_n of the space of response matrices of planar electrical networks (as studied in [Lam18]) and the space $\overline{\mathcal{X}}_n$ of boundary correlation matrices of planar Ising networks?

Recall that both spaces have cell decompositions into cells indexed by matchings on [2n], and both spaces are homeomorphic to a closed $\binom{n}{2}$ -dimensional ball by Theorems 1.5.1 and 4.1.3. Similarly to Conjecture 4.8.1, the space E_n is believed to be a regular CW complex with face poset P_n . There are many more surprising analogies between the two spaces:

- In both cases, a planar graph yields a point in the cell corresponding to its medial pairing.
- Two reduced planar graphs yield the same point if and only if they are connected by the corresponding $Y \Delta$ (or *star-triangle*) moves.⁴
- In both cases, there is an embedding of the space of boundary measurements into the totally nonnegative Grassmannian, as in Theorem 4.1.3 and [Lam18, Theorem 5.8].⁵

⁴In fact, under our map $G \mapsto G^{\square}$, applying a $Y - \Delta$ move to G corresponds to applying the superurban renewal of [KP16] to G^{\square} .

⁵The corresponding decorated permutations differ by a "shift by 1", i.e., if $\pi : [n] \to [n]$ is a fixed-point free involution then Lam embeds the electrical response matrix into the cell $\Pi_{\pi'}^{>0}$ of $\operatorname{Gr}_{\geq 0}(n-1,2n)$, where $\pi'(i) := \pi(i) - 1$ modulo *n* for all $i \in [n]$. An analogous construction in the context of the *amplituhedron* of [AHT14] is related to going from the *momentum space* to the *momentum-twistor space*, where one performs a "shift by 2". It remains an open problem to define the amplituhedron and related objects in the context of ABJM amplitudes. We thank Thomas Lam for pointing this out to us.

- Both spaces can be realized as subsets of the partial flag variety G/P for a suitable choice of G and P, see Remark 4.4.2.
- The cyclic shift inside the corresponding Grassmannian amounts to the duality operation for Ising networks as in Section 4.2.2, and for electrical networks it corresponds to taking the dual graph and replacing each conductance by its reciprocal, as easily follows from the results of [Lam18, Section 5].
- Adding boundary spikes and boundary edges translates into adding pairs of bridges to the corresponding plabic graph, see Theorem 4.2.20 and [Lam18, Proposition 5.12].

Our next question is related to Remark 4.2.6.

Problem 4.8.3. Explain rigorously the relationship between the scaling limit of planar Ising networks at critical temperature and the unique cyclically symmetric point $X_0 \in$ $OG_{\geq 0}(n, 2n)$ from Section 2.2.1.

Our main result establishes a correspondence between total positivity and planar Ising networks, and thus potentially allows to apply results and intuition from one area to another. For example, asymptotic properties of plabic graphs have not yet been studied, while asymptotic properties of planar Ising networks have rich and important well-studied structure. Similarly, the space $\operatorname{Gr}_{\geq 0}(k, n)$ is usually studied in the context of *cluster algebras* and *canonical bases* of Lusztig, see e.g. [FZ02, Lus97]. For instance, Theorem 4.2.13 expresses Griffiths' inequalities as positive linear sums of minors of $\phi(M)$. But the theory of cluster algebras gives a much larger family of rational functions of the minors that all take positive values on $\operatorname{Gr}_{\geq 0}(k, n)$.

Problem 4.8.4. Give an interpretation of the values of other cluster variables in the cluster algebra of the Grassmannian in terms of the planar Ising model.

Another direction is related to Question 4.2.23 and the discussion after it: what is the minimal number of minors one needs to check in order to test whether a given element $X \in OG(n, 2n)$ belongs to $OG_{\geq 0}(n, 2n)$? A similar question for electrical networks has been discussed in [Ken12, Section 4.5.3]. This question also makes sense when X belongs to a

lower-dimensional cell inside $OG_{\geq 0}(n, 2n)$. Note also that in the case of the Grassmannian, collections of such minors have a very nice structure [OPS15] as they form *clusters* in the associated cluster algebra. It is not clear to us whether there exists a similar "cluster structure" on $OG_{\geq 0}(n, 2n)$.

Finally, there has been a rich interplay between the areas of scattering amplitudes and total positivity, giving rise to canonical differential forms on positroid cells inside $\operatorname{Gr}_{\geq 0}(k, n)$, see [AHT14, AHBC⁺16, AHBL17, GL18]. A similar result for electrical networks can be found in [Ken12, Theorem 4.13], which gives an explicit expression for the Jacobian of a certain natural map. In [HWX14, Section 2.4.2], an expression for another Jacobian was given for $\operatorname{OG}_{\geq 0}(n, 2n)$ in the context of ABJM scattering amplitudes. It would thus be interesting to understand their Jacobian in the language of planar Ising networks, as well as develop an analog of the amplituhedron for which $\operatorname{OG}_{\geq 0}(n, 2n)$ plays the role of $\operatorname{Gr}_{\geq 0}(k, n)$.

Bibliography

- [AHBC⁺16] Nima Arkani-Hamed, Jacob Bourjaily, Freddy Cachazo, Alexander Goncharov, Alexander Postnikov, and Jaroslav Trnka. Grassmannian Geometry of Scattering Amplitudes. Cambridge University Press, Cambridge, 2016.
- [AHBL17] Nima Arkani-Hamed, Yuntao Bai, and Thomas Lam. Positive geometries and canonical forms. arXiv:1703.04541, 2017.
- [AHT14] Nima Arkani-Hamed and Jaroslav Trnka. The amplituhedron. J. High Energy Phys., (10):33, 2014.
- [AKSM04] Víctor Ayala, Wolfgang Kliemann, and Luiz A. B. San Martin. Control sets and total positivity. *Semigroup Forum*, 69(1):113–140, 2004.
- [BB05] Anders Björner and Francesco Brenti. Combinatorics of Coxeter groups, volume 231 of Graduate Texts in Mathematics. Springer, New York, 2005.
- [BCFW05] Ruth Britto, Freddy Cachazo, Bo Feng, and Edward Witten. Direct proof of the tree-level scattering amplitude recursion relation in Yang-Mills theory. *Phys. Rev. Lett.*, 94(18):181602, 4, 2005.
- [BD94] Yuly Billig and Matthew Dyer. Decompositions of Bruhat type for the Kac-Moody groups. Nova J. Algebra Geom., 3(1):11–39, 1994.
- [BdT12] Cédric Boutillier and Béatrice de Tilière. Height representation of XOR-Ising loops via bipartite dimers. arXiv:1211.4825v2, 2012.
- [BGPZ19] Pavle V. M. Blagojević, Pavel Galashin, Nevena Palić, and Günter M. Ziegler. Some more amplituhedra are contractible. *Selecta Math.* (N.S.), 25(1):25:8, 2019.
- [Bjö80] Anders Björner. Shellable and Cohen-Macaulay partially ordered sets. *Trans. Amer. Math. Soc.*, 260(1):159–183, 1980.
- [Bjö84] A. Björner. Posets, regular CW complexes and Bruhat order. *European J. Combin.*, 5(1):7–16, 1984.
- [Bor91] Armand Borel. *Linear algebraic groups*, volume 126 of *Graduate Texts in Mathematics*. Springer-Verlag, New York, second edition, 1991.

- [BPZ84a] A. A. Belavin, A. M. Polyakov, and A. B. Zamolodchikov. Infinite conformal symmetry in two-dimensional quantum field theory. *Nuclear Phys. B*, 241(2):333–380, 1984.
- [BPZ84b] A. A. Belavin, A. M. Polyakov, and A. B. Zamolodchikov. Infinite conformal symmetry of critical fluctuations in two dimensions. J. Statist. Phys., 34(5-6):763-774, 1984.
- [Bro60] Morton Brown. A proof of the generalized Schoenflies theorem. Bull. Amer. Math. Soc., 66:74–76, 1960.
- [Bro62] Morton Brown. Locally flat imbeddings of topological manifolds. Ann. of Math. (2), 75:331–341, 1962.
- [BW82] Anders Björner and Michelle Wachs. Bruhat order of Coxeter groups and shellability. Adv. in Math., 43(1):87–100, 1982.
- [BZ97] Arkady Berenstein and Andrei Zelevinsky. Total positivity in Schubert varieties. Comment. Math. Helv., 72(1):128–166, 1997.
- [CDCH⁺14] Dmitry Chelkak, Hugo Duminil-Copin, Clément Hongler, Antti Kemppainen, and Stanislav Smirnov. Convergence of Ising interfaces to Schramm's SLE curves. C. R. Math. Acad. Sci. Paris, 352(2):157–161, 2014.
- [CdVGV96] Yves Colin de Verdière, Isidoro Gitler, and Dirk Vertigan. Réseaux électriques planaires. II. Comment. Math. Helv., 71(1):144–167, 1996.
- [Che11] Nicolas Chevalier. Total positivity criteria for partial flag varieties. J. Algebra, 348:402–415, 2011.
- [CHI15] Dmitry Chelkak, Clément Hongler, and Konstantin Izyurov. Conformal invariance of spin correlations in the planar Ising model. Ann. of Math. (2), 181(3):1087–1138, 2015.
- [CIM98] E. B. Curtis, D. Ingerman, and J. A. Morrow. Circular planar graphs and resistor networks. *Linear Algebra Appl.*, 283(1-3):115–150, 1998.
- [CS12] Dmitry Chelkak and Stanislav Smirnov. Universality in the 2D Ising model and conformal invariance of fermionic observables. *Invent. Math.*, 189(3):515–580, 2012.
- [Dav08] Michael W. Davis. The geometry and topology of Coxeter groups, volume 32 of London Mathematical Society Monographs Series. Princeton University Press, Princeton, NJ, 2008.
- [DC16] Hugo Duminil-Copin. Random currents expansion of the Ising model. arXiv:1607.06933v2, 2016.
- [DCL17] Hugo Duminil-Copin and Marcin Lis. On the double random current nesting field. arXiv:1712.02305v1, 2017.

- [DK74] Gopal Danaraj and Victor Klee. Shellings of spheres and polytopes. *Duke Math.* J., 41:443–451, 1974.
- [dT14] Béatrice de Tilière. Bipartite dimer representation of squared 2d-Ising correlations. arXiv:1407.6274v2, 2014.
- [Dub11] Julien Dubédat. Exact bosonization of the Ising model. arXiv:1112.4399v1, 2011.
- [Ele16] Balazs Elek. Toric surfaces with equivariant Kazhdan-Lusztig atlases. arXiv:1610.04667v1, 2016.
- [FH91] William Fulton and Joe Harris. Representation theory, volume 129 of Graduate Texts in Mathematics. Springer-Verlag, New York, 1991. A first course, Readings in Mathematics.
- [FP16] Miriam Farber and Alexander Postnikov. Arrangements of equal minors in the positive Grassmannian. *Adv. Math.*, 300:788–834, 2016.
- [Fre82] Michael Hartley Freedman. The topology of four-dimensional manifolds. J. Differential Geom., 17(3):357–453, 1982.
- [FS00] Sergey Fomin and Michael Shapiro. Stratified spaces formed by totally positive varieties. *Michigan Math. J.*, 48:253–270, 2000. Dedicated to William Fulton on the occasion of his 60th birthday.
- [FZ99] Sergey Fomin and Andrei Zelevinsky. Double Bruhat cells and total positivity. J. Amer. Math. Soc., 12(2):335–380, 1999.
- [FZ02] Sergey Fomin and Andrei Zelevinsky. Cluster algebras. I. Foundations. J. Amer. Math. Soc., 15(2):497–529, 2002.
- [GBK78] J. Groeneveld, R.J. Boel, and P.W. Kasteleyn. Correlation-function identities for general planar ising systems. *Physica A: Statistical Mechanics and its Applications*, 93(1):138 – 154, 1978.
- [GHS70] Robert B. Griffiths, C. A. Hurst, and S. Sherman. Concavity of magnetization of an Ising ferromagnet in a positive external field. *J. Mathematical Phys.*, 11:790–795, 1970.
- [GK50] F. R. Gantmaher and M. G. Krein. Oscillyacionye matricy i yadra i malye kolebaniya mehaničeskih sistem. Gosudarstv. Isdat. Tehn.-Teor. Lit., Moscow-Leningrad, 1950. 2d ed. Translated into English by A. Eremenko [GK02].
- [GK02] F. P. Gantmacher and M. G. Krein. Oscillation Matrices and Kernels and Small Vibrations of Mechanical Systems. AMS Chelsea Publishing, Providence, RI, revised edition, 2002. Translation based on the 1950 Russian original. Edited and with a preface by Alex Eremenko.

- [GKL17] Pavel Galashin, Steven N. Karp, and Thomas Lam. The totally nonnegative Grassmannian is a ball. arXiv:1707.02010, 2017.
- [GKL18] Pavel Galashin, Steven N. Karp, and Thomas Lam. The totally nonnegative part of G/P is a ball. arXiv:1801.08953v2, 2018.
- [GKL19] Pavel Galashin, Steven N. Karp, and Thomas Lam. Regularity theorem for totally nonnegative flag varieties. arXiv:1904.00527, 2019.
- [GL18] Pavel Galashin and Thomas Lam. Parity duality for the amplituhedron. arXiv:1805.00600, 2018.
- [GP18] Pavel Galashin and Pavlo Pylyavskyy. Ising model and the positive orthogonal Grassmannian. arXiv:1807.03282v3, 2018.
- [Gri67] Robert B. Griffiths. Correlations in Ising Ferromagnets. I. Journal of Mathematical Physics, 8(3):478–483, 1967.
- [GY09] K. R. Goodearl and M. Yakimov. Poisson structures on affine spaces and flag varieties. II. Trans. Amer. Math. Soc., 361(11):5753–5780, 2009.
- [Had84] Ziad Haddad. Infinite-dimensional flag varieties. ProQuest LLC, Ann Arbor, MI, 1984. Thesis (Ph.D.)–Massachusetts Institute of Technology.
- [Had85] Ziad Haddad. A Coxeter group approach to Schubert varieties. In Infinitedimensional groups with applications (Berkeley, Calif., 1984), volume 4 of Math. Sci. Res. Inst. Publ., pages 157–165. Springer, New York, 1985.
- [Hat02] Allen Hatcher. *Algebraic topology*. Cambridge University Press, Cambridge, 2002.
- [He07] Xuhua He. Minimal length elements in some double cosets of Coxeter groups. Adv. Math., 215(2):469–503, 2007.
- [He09] Xuhua He. A subalgebra of 0-Hecke algebra. J. Algebra, 322(11):4030–4039, 2009.
- [Her14] Patricia Hersh. Regular cell complexes in total positivity. Invent. Math., 197(1):57–114, 2014.
- [HK18] Patricia Hersh and Richard Kenyon. Shellability of face posets of electrical networks and the CW poset property. arXiv:1803.06217v2, 2018.
- [HKL19] Xuhua He, Allen Knutson, and Jiang-Hua Lu. In preparation, 2019.
- [HL15] Xuhua He and Thomas Lam. Projected Richardson varieties and affine Schubert varieties. Ann. Inst. Fourier (Grenoble), 65(6):2385–2412, 2015.
- [Hoc75] Melvin Hochster. Topics in the homological theory of modules over commutative rings. Published for the Conference Board of the Mathematical Sciences by the American Mathematical Society, Providence, R.I., 1975.

- [HS10] Andre Henriques and David E. Speyer. The multidimensional cube recurrence. Adv. Math., 223(3):1107–1136, 2010.
- [HS13] Clément Hongler and Stanislav Smirnov. The energy density in the planar Ising model. Acta Math., 211(2):191–225, 2013.
- [Hum75] James E. Humphreys. *Linear algebraic groups*. Springer-Verlag, New York-Heidelberg, 1975. Graduate Texts in Mathematics, No. 21.
- [HW13] Yu-Tin Huang and Congkao Wen. ABJM amplitudes and the positive orthogonal Grassmannian. arXiv:1309.3252v2, 2013.
- [HWX14] Yu-Tin Huang, Congkao Wen, and Dan Xie. The positive orthogonal Grassmannian and loop amplitudes of ABJM. J. Phys. A, 47(47):474008, 48, 2014.
- [Isi25] E. Ising. Beitrag zur Theorie des Ferromagnetismus. Zeitschrift fur Physik, 31:253–258, February 1925.
- [Kar16] Rachel Karpman. Bridge graphs and Deodhar parametrizations for positroid varieties. J. Combin. Theory Ser. A, 142:113–146, 2016.
- [Kar17] Steven N. Karp. Sign variation, the Grassmannian, and total positivity. J. Combin. Theory Ser. A, 145:308–339, 2017.
- [Kar18] Steven N. Karp. Moment curves and cyclic symmetry for positive Grassmannians. arXiv:1805.06004v1, 2018.
- [Kau49] Bruria Kaufman. Crystal Statistics. II. Partition Function Evaluated by Spinor Analysis. *Phys. Rev.*, 76:1232–1243, Oct 1949.
- [Ken12] R. Kenyon. The Laplacian on planar graphs and graphs on surfaces. In Current developments in mathematics, 2011, pages 1–55. Int. Press, Somerville, MA, 2012.
- [KL79] David Kazhdan and George Lusztig. Representations of Coxeter groups and Hecke algebras. *Invent. Math.*, 53(2):165–184, 1979.
- [KLS13] Allen Knutson, Thomas Lam, and David E. Speyer. Positroid varieties: juggling and geometry. *Compos. Math.*, 149(10):1710–1752, 2013.
- [KLS14] Allen Knutson, Thomas Lam, and David E. Speyer. Projections of Richardson varieties. J. Reine Angew. Math., 687:133–157, 2014.
- [Kos93] Antoni A. Kosinski. Differential manifolds, volume 138 of Pure and Applied Mathematics. Academic Press, Inc., Boston, MA, 1993.
- [KP83] Victor G. Kac and Dale H. Peterson. Regular functions on certain infinitedimensional groups. In Arithmetic and geometry, Vol. II, volume 36 of Progr. Math., pages 141–166. Birkhäuser Boston, Boston, MA, 1983.

- [KP16] Richard Kenyon and Robin Pemantle. Double-dimers, the Ising model and the hexahedron recurrence. J. Combin. Theory Ser. A, 137:27–63, 2016.
- [KS68] D. G. Kelly and S. Sherman. General Griffiths' Inequalities on Correlations in Ising Ferromagnets. *Journal of Mathematical Physics*, 9(3):466–484, 1968.
- [KS77] Robion C. Kirby and Laurence C. Siebenmann. Foundational essays on topological manifolds, smoothings, and triangulations. Princeton University Press, Princeton, N.J.; University of Tokyo Press, Tokyo, 1977. With notes by John Milnor and Michael Atiyah, Annals of Mathematics Studies, No. 88.
- [Kum02] Shrawan Kumar. Kac-Moody groups, their flag varieties and representation theory, volume 204 of Progress in Mathematics. Birkhäuser Boston, Inc., Boston, MA, 2002.
- [Kum17] Shrawan Kumar. Positivity in *T*-equivariant *K*-theory of flag varieties associated to Kac-Moody groups. *J. Eur. Math. Soc. (JEMS)*, 19(8):2469–2519, 2017. With an appendix by Masaki Kashiwara.
- [KW41] H. A. Kramers and G. H. Wannier. Statistics of the two-dimensional ferromagnet. I. Phys. Rev. (2), 60:252–262, 1941.
- [KW11] Richard W. Kenyon and David B. Wilson. Boundary partitions in trees and dimers. Trans. Amer. Math. Soc., 363(3):1325–1364, 2011.
- [KW19] Steven N. Karp and Lauren K. Williams. The m = 1 amplituhedron and cyclic hyperplane arrangements. Int. Math. Res. Not. IMRN, (5):1401–1462, 2019.
- [KWY13] Allen Knutson, Alexander Woo, and Alexander Yong. Singularities of Richardson varieties. *Math. Res. Lett.*, 20(2):391–400, 2013.
- [Lam15] Thomas Lam. The uncrossing partial order on matchings is Eulerian. J. Combin. Theory Ser. A, 135:105–111, 2015.
- [Lam16] Thomas Lam. Totally nonnegative Grassmannian and Grassmann polytopes. In Current developments in mathematics 2014, pages 51–152. Int. Press, Somerville, MA, 2016.
- [Lam18] Thomas Lam. Electroid varieties and a compactification of the space of electrical networks. *Adv. Math.*, 338:549–600, 2018.
- [Lee13] John M. Lee. Introduction to smooth manifolds, volume 218 of Graduate Texts in Mathematics. Springer, New York, second edition, 2013.
- [Lis17] Marcin Lis. The planar Ising model and total positivity. J. Stat. Phys., 166(1):72–89, 2017.
- [LP12] Thomas Lam and Pavlo Pylyavskyy. Total positivity in loop groups, I: Whirls and curls. *Adv. Math.*, 230(3):1222–1271, 2012.

[LP15] Thomas Lam and Pavlo Pylyavskyy. Electrical networks and Lie theory. Algebra Number Theory, 9(6):1401–1418, 2015. [Lus90] G. Lusztig. Canonical bases arising from quantized enveloping algebras. J. Amer. Math. Soc., 3(2):447–498, 1990. [Lus94] G. Lusztig. Total positivity in reductive groups. In Lie theory and geometry, volume 123 of Progr. Math., pages 531–568. Birkhäuser Boston, Boston, MA, 1994. [Lus97] George Lusztig. Total positivity and canonical bases. In Algebraic groups and Lie groups, volume 9 of Austral. Math. Soc. Lect. Ser., pages 281–295. Cambridge Univ. Press, Cambridge, 1997. [Lus98a] G. Lusztig. Total positivity in partial flag manifolds. Represent. Theory, 2:70– 78, 1998. [Lus98b] George Lusztig. Introduction to total positivity. In *Positivity in Lie theory:* open problems, volume 26 of De Gruyter Exp. Math., pages 133–145. de Gruyter, Berlin, 1998. A. T. Lundell and S. Weingram. The Topology of CW Complexes. The university [LW69] series in higher mathematics. Springer-Verlag, New York, 1969. [LW16] Titus Lupu and Wendelin Werner. A note on Ising random currents, Ising-FK, loop-soups and the Gaussian free field. *Electron. Commun. Probab.*, 21:Paper No. 13, 7, 2016. [Mil65] John Milnor. Lectures on the h-cobordism theorem. Notes by L. Siebenmann and J. Sondow. Princeton University Press, Princeton, N.J., 1965. [MR04] R. J. Marsh and K. Rietsch. Parametrizations of flag varieties. *Represent.* Theory, 8:212–242, 2004. [MR15] Robert Marsh and Konstanze Rietsch. The *B*-model connection and mirror symmetry for Grassmannians. arXiv:1307.1085, 2015. |Ons44| Lars Onsager. Crystal statistics. I. A two-dimensional model with an orderdisorder transition. Phys. Rev. (2), 65:117–149, 1944. [OPS15]Suho Oh, Alexander Postnikov, and David E. Speyer. Weak separation and plabic graphs. Proc. Lond. Math. Soc. (3), 110(3):721-754, 2015. [Pal07] John Palmer. Planar Ising correlations, volume 49 of Progress in Mathematical Physics. Birkhäuser Boston, Inc., Boston, MA, 2007. [Per02] Grisha Perelman. The entropy formula for the Ricci flow and its geometric applications. arXiv:math/0211159, 2002.

[Per03a] Grisha Perelman. Finite extinction time for the solutions to the Ricci flow on certain three-manifolds. arXiv:math/0307245, 2003. [Per03b] Grisha Perelman. Ricci flow with surgery on three-manifolds. arXiv:math/0303109, 2003. [PK83] Dale H. Peterson and Victor G. and Kac. Infinite flag varieties and conjugacy theorems. Proc. Nat. Acad. Sci. U.S.A., 80(6 i.):1778–1782, 1983. [Pos07] Alexander Postnikov. Total positivity, Grassmannians, and networks. Preprint, http:// math.mit.edu/~apost/papers/tpgrass.pdf, 2007. [PSW09] Alexander Postnikov, David Speyer, and Lauren Williams. Matching polytopes, toric geometry, and the totally non-negative Grassmannian. J. Algebraic Combin., 30(2):173-191, 2009.[Rie98] Konstanze Christina Rietsch. Total Positivity and Real Flag Varieties. Ph.D. thesis, Massachusetts Institute of Technology, 1998. [Rie99] Konstanze Rietsch. An algebraic cell decomposition of the nonnegative part of a flag variety. J. Algebra, 213(1):144–154, 1999. [Rie06] K. Rietsch. Closure relations for totally nonnegative cells in G/P. Math. Res. Lett., 13(5-6):775-786, 2006. [Rie08] Konstanze Rietsch. A mirror symmetric construction of $qH^*_T(G/P)_{(q)}$. Adv. Math., 217(6):2401–2442, 2008. [RW08] Konstanze Rietsch and Lauren Williams. The totally nonnegative part of G/Pis a CW complex. *Transform. Groups*, 13(3-4):839–853, 2008. [RW10] Konstanze Rietsch and Lauren Williams. Discrete Morse theory for totally non-negative flag varieties. Adv. Math., 223(6):1855–1884, 2010. [Sag16] Collaborative 2016.SageMath, Inc. CoCalc Computation Online, https://cocalc.com/. [Sch30] Isac Schoenberg. Uber variationsvermindernde lineare Transformationen. Math. Z., 32(1):321–328, 1930. [Sco79] R. F. Scott. Note on a theorem of Prof. Cayley's. Messeng. Math., 8:155–157, 1879. [Ska04] Mark Skandera. Inequalities in products of minors of totally nonnegative matrices. J. Algebraic Combin., 20(2):195–211, 2004. [Sma61] Stephen Smale. Generalized Poincaré's conjecture in dimensions greater than four. Ann. of Math. (2), 74:391-406, 1961.

- [Smi10] Stanislav Smirnov. Conformal invariance in random cluster models. I. Holomorphic fermions in the Ising model. Ann. of Math. (2), 172(2):1435–1467, 2010.
- [SMJ78] Mikio Sato, Tetsuji Miwa, and Michio Jimbo. Holonomic quantum fields. I. Publ. Res. Inst. Math. Sci., 14(1):223–267, 1978.
- [Sni10] Michelle Bernadette Snider. Affine patches on positroid varieties and affine pipe dreams. ProQuest LLC, Ann Arbor, MI, 2010. Thesis (Ph.D.)–Cornell University.
- [Spe18] David Speyer. Private communication, 2018.
- [Spr98] T. A. Springer. *Linear algebraic groups*, volume 9 of *Progress in Mathematics*. Birkhäuser Boston, Inc., Boston, MA, second edition, 1998.

[Stu88] Bernd Sturmfels. Totally positive matrices and cyclic polytopes. In *Proceedings* of the Victoria Conference on Combinatorial Matrix Analysis (Victoria, BC, 1987), volume 107, pages 275–281, 1988.

- [Tal08] Kelli Talaska. A formula for Plücker coordinates associated with a planar network. *Int. Math. Res. Not. IMRN*, pages Art. ID rnn 081, 19, 2008.
- [Tal11] Kelli Talaska. Combinatorial formulas for Le-coordinates in a totally nonnegative Grassmannian. J. Combin. Theory Ser. A, 118(1):58–66, 2011.
- [Whi52] A. M. Whitney. A reduction theorem for totally positive matrices. J. Analyse Math., 2:88–92, 1952.
- [Wil07] Lauren K. Williams. Shelling totally nonnegative flag varieties. J. Reine Angew. Math., 609:1–21, 2007.
- [Wil13] Harold Williams. Double Bruhat cells in Kac-Moody groups and integrable systems. *Lett. Math. Phys.*, 103(4):389–419, 2013.
- [Yan52] C. N. Yang. The spontaneous magnetization of a two-dimensional Ising model. *Physical Rev.* (2), 85:808–816, 1952.